Regulation of Macrophage Inflammatory Protein-1α Expression by Haemopoietic Growth Factors

By

David Ian Jarmin

Submission for the Degree of Doctor of Philosophy

in

The Faculty of Medicine at the University of Glasgow

January 1998

©David Jarmin 1998

ProQuest Number: 13818693

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 13818693

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346



11141 (copy 1)

,

GLASGOW UNIVERSITY Lidrary

Declaration

I declare that all work presented in this thesis was performed by me personally unless acknowledged otherwise.

Dedication

This thesis is dedicated to my family. In memory of Beatrice and Stanley Jarmin, and Hilda Skinner.

Acknowledgements

I would firstly like to gratefully acknowledge the support, guidance, encouragement, supervision, inspiration and boundless enthusiasm of my supervisor, Dr. Gerry Graham. In particular I would like to thank him for his willingness to always find time for me to harass him and for putting up with an enormous amount of cheek. Most of all I would like to thank him for showing me the secret to becoming a great scientist.....namely the ability to shamelessly scrounge all your solutions! Thanks must go to all members of R2, both past and present, for their assistance in the laboratory and for making my time in R2 such great fun. In particular I would like to thank Dr. Robert Nibbs, for his help and advice over the last 3 years and especially for his unending cheerfulness! Likewise thanks must go to Julie Connelly, for her constant state of happiness and for always sparing time to share her experiences concerning the many joys of thesis writing! Thanks must also go to my advisor, Dr. Paul Harrison, for his advice and to Dr. Ian Pragnell, for his wonderful couscous evenings! I would like to acknowledge the assistance of Professors Donald Metcalf and Ashley Dunn, in kindly providing the GM-CSF transgenic mouse tissue samples. I must also thank Dr. Thomas Baumruker for his gift of reagents and for his continued support over the last 5 years. I would like to acknowledge the help provided during my PhD by all the staff at the Beatson Institute, but particularly that of Liz Gordon, Bill Simpson and Ian White.

Thanks must also go to all my friends for making my time in Glasgow so enjoyable. I would especially like to thank Lorraine Blower and Chris Denning for always being there. Also, I must thank Dr. Vicky Heath, Dewi Owens, Stacey Hoare, Dr. Gillian Hirst and 'Shazz Bob' for their morale support, generally in the form of drinking or playing squash! Finally, special thanks must go to my parents for all their generous financial support and encouragement over the last 3 years (and more!)

I acknowledge that this work was supported by a grant from the Medical Research Council.

ABSTRACT

Macrophage inflammatory protein-1 alpha (MIP-1 α) is a pro-inflammatory cytokine and a regulator of haemopoiesis. Little is currently known regarding the network of regulatory factors controlling the expression of MIP-1 α , within the bone marrow microenvironment. This study was therefore initiated to further understand how expression of this gene is regulated by haemopoietic growth factors. Macrophages are likely to be an important source of MIP-1 α within the bone marrow and this study has therefore used primary murine bone marrow-derived macrophages and Northern blot analysis to examine the expression of MIP-1 α following treatment with various growth factors.

The results presented in this thesis show that treatment of bone marrowderived macrophages with granulocyte-macrophage colony-stimulating factor (GM-CSF) results in a striking increase in MIP-1 α mRNA levels, relative to control cells. This increase in MIP-1 α mRNA expression was also followed by an increase in MIP-1 α protein levels in these cells, suggesting that the potent stimulatory effect of GM-CSF on macrophage-derived MIP-1 α expression may be of physiological or pathological significance. Since GM-CSF, interleukin- (IL-) 3 and IL-5 all share a common β -chain subunit in their receptors and can functionally substitute for each other in certain cell types, primary macrophages were also treated with IL-3 or IL-5. A similar increase in mRNA levels was found when macrophages were treated with IL-3, but not IL-5. An increase in the expression of mRNAs for three other CC chemokines: MIP-1 β , JE and MARC, was also found following treatment with GM-CSF or IL-3, implying that the potent stimulatory effects of GM-CSF and IL-3 are not restricted to MIP-1 α and may therefore also have an impact on the expression of other CC chemokines.

Previous work from our group has provided evidence of an endogenous reciprocal relationship in macrophages, between MIP-1 α and transforming growth factor-beta (TGF- β), in which TGF- β acts to suppress MIP-1 α expression. To investigate the potential effect of GM-CSF and IL-3 on the relationship between TGF- β and MIP-1 α , the effect of TGF- β on the GM-CSF-or IL-3-induced expression of MIP-1 α mRNA was assessed. TGF- β was able to suppress the GM-CSF-or IL-3-

v

induced expression of MIP-1 α mRNA, but this suppression was not complete. These results suggest that GM-CSF and IL-3 may therefore be two potential stimulatory signals, that may be sufficiently potent to overcome the TGF- β mediated block on MIP-1 α expression. Similar effects on the expression of the GM-CSF or IL-3 induced expression of MIP-1 β , JE and MARC were also observed. The significance of this is unclear, but may reflect the widely recognised function of TGF- β as an immunosuppressive molecule.

A previous study in our laboratory suggested that TGF- β reversibly downregulated the expression of MIP-1 α receptors on the murine FDCPmix cell line. Therefore, the effect of GM-CSF and IL-3 on the expression of various receptors for MIP-1 α on bone marrow-derived macrophages was also examined and an increase in the expression of CCR1 mRNA levels was observed, but no change was seen in the level of mRNA for CCR5. This suggests that CCR1 and CCR5 expression may be differently regulated in bone marrow-derived macrophages. TGF- β was observed to have only a minimal suppressive effect on the expression of CCR1 and CCR5 mRNA. Binding studies revealed that treatment of bone marrow macrophages with GM-CSF also resulted in an increase in the overall levels of MIP-1 α receptors on these cells. Scatchard analysis showed that GM-CSF increased the number of MIP-1a receptors with no change in the affinity of the remaining receptors for MIP-1 α ligand. The K_d's for the receptors detected in this study are similar to those previously reported for murine CCR1, but different to those for CCR5. This observation, together with the lack of effect of GM-CSF on CCR5 expression suggests, that this increase in MIP-1a receptor numbers probably reflects an increase in the cell surface expression of CCR1. To investigate the functional significance of this increase in MIP-1 α receptors, the effect of GM-CSF on the ability of macrophages to mobilise calcium in response to MIP-1 α was examined. In one experiment, treatment of bone marrow-derived macrophages with GM-CSF appeared to result in a slight increase in the ability of these cells to respond to MIP-1 α . In most cases however, their ability to flux was highly variable, suggesting they were not an ideal choice for calcium flux analysis and therefore the pro-monocytic human THP-1 cell line was used instead. GM-CSF stimulation of THP-1 cells resulted in an increase in the expression of CCR1 mRNA, similar to that observed in bone marrow-derived macrophages, but no increase in calcium mobilisation was observed in GM-CSF treated THP-1 cells in response to MIP-1 α . The functional significance of the observed GM-CSF-induced increase in CCR1 expression therefore remains to be determined.

Using RT-PCR analysis, a preliminary investigation of MIP-1 α expression in tissues derived from a GM-CSF transgenic mouse, revealed that an increased expression of MIP-1 α occurred in samples from the eye (and perhaps also the bone marrow) of a GM-CSF transgenic mouse. Although preliminary, these results suggest that increased expression of MIP-1 α may occur under circumstances in which elevated GM-CSF expression occurs.

The data presented in this thesis suggest that expression of GM-CSF or IL-3 may be one mechanism through which the levels of MIP-1 α and other β chemokines produced within the bone marrow by macrophages is increased. Indeed, the observation that TGF- β was only partially able to inhibit the induction of MIP-1 α by GM-CSF and IL-3, implies that GM-CSF and IL-3 may be sufficiently potent to overcome the proposed TGF- β mediated block on MIP-1 α expression. Although it is impossible to say with certainty, whether these observations reflect an involvement in stem cell proliferative control or in inflammatory responses, these results clearly have implications for both. In summary, the results presented in this thesis suggest that the expression of MIP-1 α and other chemokines in bone marrow macrophages, is regulated through the combined interaction of the positive stimuli exerted by GM-CSF and IL-3 and the negative stimulus of TGF- β .

CONTENTS

SECTION

Title Page i Declaration ii Dedication ii Acknowledgements iii Abstract iv vii Contents List of Figures and Tables xvi Abbreviations xviii

PART I: INTRODUCTION

CHA	PTER 1: 7	The Haemopoietic System	1
1.1	The Haem	opoietic System: An Overview	1
1.2	The Stem	Cell Compartment	1
1.3	The In Viv	o CFU-S Assay	3
1.4	In vitro Cu	lture of Haemopoietic Stem Cells	5
	1.4.1	In vitro stem cell assays	5
	1.4.2	The stromal microenvironment	6
1.5	Progenitor	Cells and Lineages	7
1.6	Inhibition	of Haemopoiesis	8
1.7	The Strom	al Microenvironment and the Production of Soluble Mediators	9
	1.7.1	Bone marrow and stromal production of cytokines	
		and growth factors	10
	1.7.2	Localisation of growth factors	12
1.8	Haemopoi	etic Regulators and the Cytokine Network:	13
	Redundand	cy or Eloquent Control ?	

PAGE

CHA	PTER 2: 0	Growth Factors and Cytokines	17
2.1	Interleukin	1-1 (IL-1)	17
2.2	Interleukin	n-3 (IL-3)	18
2.3	Interleukin	n-4 (IL-4)	21
2.4	Interleukin	n-5 (IL-5)	21
2.5	Interleukin	1-6 (IL-6)	23
2.6	Interleukin	1-7 (IL-7)	23
2.7	Interleukin	n-8 (IL-8)	24
2.8	Interleukin	-11 (IL-11)	24
2.9	Macrophag	ge Colony-Stimulating Factor (M-CSF)	25
2.10	Granulocy	te Colony-Stimulating Factor (G-CSF)	26
2.11	Granulocy	te-Macrophage Colony-Stimulating Factor (GM-CSF)	27
2.12	Stem Cell	Factor (SCF)	30
2.13	Leukemia	Inhibitory Factor (LIF)	32
2.14	The Transf	forming Growth Factor Beta (TGF-β) Superfamily	34
	2.14.1	The TGF-β family	35
	2.14.2	Structure of TGF- β : TGF- β exists in a latent form	35
	2.14.3	Expression of TGF-β	37
	2.14.4	The role of TGF- β as an immunoregulator	37
	2.14.5	TGF- β as a regulator of cell proliferation	- 39
	2.14.6	The role of TGF- β in haemopoiesis	40
2.15	The Role of	of Growth Factors in the Regulatory Control of Haemopoiesis	41
	2.15.1	The stochastic model of differentiation	41
	2.15.2	The instructive or deterministic model of differentiation	42
2.16	Summary		43
CHA	PTER 3: N	/IIP-1 $lpha$ and the Chemokine Superfamily	44
3.1	The Chem	okine Superfamily: Structure and Family Subdivision	44
3.2	The CXC	Chemokines	46
	3.2.1	Interleukin-8 (IL-8)	46
	3.2.2	Stromal Cell Derived Factor-1 (SDF-1)	47

	3.2.3	IP-10 and Mig	48;
	3.2.4	Other CXC Chemokines	49)
3.3	The CC C	hemokines	49)
	3.3.1	The Monocyte Chemotactic Proteins	50)
		3.3.1.1 Monocyte Chemotactic Protein-1 (MCP-1) / JE	50)
		3.3.1.2 MCP-2 and MCP-3	51
		3.3.1.3 MCP-4 and MCP-5	52)
	3.3.2	RANTES	522
	3.3.3	Other CC Chemokines	533
3.4	The C Sub	family: Lymphotactin	533
3.5	The CX ₃ C	Subfamily: Fractalkine / Neurotactin	533
3.6	MIP-1a		54
	3.6.1	Characterisation of MIP-1a	54
	3.6.2	Normal Expression of MIP-1a	56
	3.6.3	Expression in Disease and Inflammation	57/
	3.6.4	Pro-inflammatory Functions of MIP-1α	58
	3.6.5	The Role of MIP-1 α as a Haemopoietic Regulator	61
3.7	Chemokin	e Receptors	64
	3.7.1	Early characterisation of receptors	64
	3.7.2	The 7TMsGPCR superfamily	66
	3.7.3	Receptors for the CC chemokines	68
		3.7.3.1 CCR1	68
		3.7.3.2 CCR2	69
		3.7.3.3 CCR3	69
		3.7.3.4 CCR4	70
		3.7.3.5 CCR5	70
		3.7.3.6 Other CC Receptors	70
	3.7.4	Receptors for the CXC Chemokines	71
3.8	The Duffy	Antigen for Chemokines (DARC)	71
3.9	Chemokin	es, their Receptors and Significance to HIV Pathogenesis	72
3.10	Virally En	coded Chemokine and Chemokine Receptor Homologues	74

3.11 In Summary	75
AIMS OF THESIS	77
PART II: MATERIALS AND METHODS	

CHAPTER 4: MATERIALS

4.1	Plasmids for Generation of Probe Fragments	78
4.2	Plasmids for Use in RT-PCR Analysis	79
4.3	PCR Primers: Identity	80
4.4	PCR Primers: Sequences	81
4.5	Tissue Culture Supplies	82
4.6	Cytokines and Antibodies	83
4.7	Kits	83
4.8	Membranes, Paper and X-ray film	84
4.9	Nucleotides	84
4.10	Protein and DNA Markers	84
4.11	Enzymes, Enzyme Buffers and Enzyme Inhibitors	85
4.12	Columns	85
4.13	Chemicals	85
4.14	Bacterial Culture	86
	4.14.1 Solutions	86
	4.14.2 Host strains	87
4.15	Solutions	87
	4.15.1 Agarose gel electrophoresis	87
	4.15.2 DNase I digestion	87
	4.15.3 Northern blotting and hybridisation	87
	4.15.4 SDS-Polyacrylamide gel electrophoresis	88
	and Western blotting	
	4.15.5 Binding studies	89
	4.15.6 Calcium mobilisation studies	89

CHAPTER 5: METHODS

5.1	Mice		90
5.2	Cell Lines		90
	5.2.1	J774.2	90
	5.2.2	THP-1	90
	5.2.3	HEK 293-hCCR5 stable transfectants	90
5.3	Bone Mar	row-Derived Macrophages	92
	5.3.1	Harvesting of bone marrow	92
	5.3.2	Derivation of bone marrow derived macrophages	92
5.4	Stimulatio	n of Cell Lines and Bone Marrow-Derived Macrophages	92
5.5	DNA Met	hodology	93
	5.5.1	Transformation of bacterial host cells with plasmid DNA	93
	5.5.2	Growth of plasmids	93
	5.5.3	Plasmid Preparation	94
	5.5.4	Digestion of plasmid DNA with restriction enzymes	95
	5.5.5	Agarose gel electrophoresis of DNA	95
	5.5.6	Isolation of probe fragments for generating	96
		radiolabelled probes	
5.6	RNA Metl	hodology	97
	5.6.1	RNase free environment	. 97
	5.6.2	Isolation of total RNA	97
	5.6.3	Isolation of polyA ⁺ mRNA	98
	5.6.4	Agarose gel electrophoresis of RNA	99
5.7	Southern a	and Northern blotting of DNA and RNA	99
5.8	Probing of	Blots with Radiolabelled Probes	100
	5.8.1	Preparation of radiolabelled probes	100
	5.8.2	Pre-hybridisation and hybridisation conditions for	100
		Northern and Southern blots	
	5.8.3	Washing of blots and exposure to X-ray film	101
	5.8.4	Stripping of blots	101
	5.8.5	Scanning densitometric analysis of blots	101

5.9	Protein De	tection	102
	5.9.1	Preparation of cell conditioned medium	102
	5.9.2	Purification of MIP-1 α from bone marrow-derived	102
		macrophage cell conditioned medium	
	5.9.3	SDS Polyacrylamide gel electrophoresis	102
	5.9.4	Western blotting	103
5.10	Reverse Tr	anscriptase Polymerase Chain Reaction	104
	(RT-PCR)	Methodology	
	5.10.1	DNase I digestion of total RNA	104
	5.10.2	RT-PCR methodology I (amplification from total RNA)	105
	5.10.3	RT-PCR methodology II (amplification from polyA ⁺ mRNA)	105
5.11	Receptor B	inding Studies	106
	5.11.1	¹²⁵ I labelling of MIP-1α protein	106
	5.11.2	MIP-1 α receptor binding studies on bone marrow	106
		derived-macrophages	
5.12	Calcium Fl	ux Studies	107
	5.12.1	Calcium flux studies in bone marrow-derived macrophages	107
	5.12.2	Calcium flux studies in THP-1 cells	108

PART III: RESULTS AND DISCUSSION

CHAPTER 6: RESULTS

6.1	Overview		109
6.2	Effect of C	β -CSF, LIF and IL-6 on the Expression of MIP-1 α in BMM	109
6.3.	Effect of C	$GM-CSF$ on Expression of MIP-1 α mRNA in BMM	111
	6.3.1	GM-CSF induces MIP- α expression	111
	6.3.2	Induction of MIP- α expression by GM-CSF is rapid,	111
		but transient	
	6.3.3	Induction of MIP- α mRNA expression in response to a	114
		titration of GM-CSF concentration	

	6.3.4	Induction of MIP- α mRNA expression is not the result	114
		of factors present in serum or L929 conditioned medium	
	6.3.5	GM-CSF directly induces the expression of MIP-1 α mRNA	116
	6.3.6	GM-CSF induces MIP-1 α protein expression in BMM	119
6.4	Effect of I	L-3 on MIP-1α mRNA Expression in BMM	121
	6.4.1	Titration of the inductive effect of IL-3 on MIP-1 α	121
		mRNA expression	
6.5	Effect of I	L-5 on MIP-1α mRNA Expression	123
	6.5.1	IL-5 does not induce expression of MIP-1 α mRNA in BMM	123
	6.5.2	BMM do not express the IL-5R α chain	123
	6.5.3	IL-5 does not induce expression of MIP-1 α mRNA in	126
		bone marrow cells	
6.6	Effect of C	GM-CSF on the Expression of mRNA for Other CC chemokines	126
	6.6.1	GM-CSF induces expression of MIP-1 β mRNA	128
	6.6.2	GM-CSF and IL-3 induce expression of JE and	128
		MARC mRNA	
6.7	Effect of T	GF- β 1 on the GM-CSF and IL-3 Induced Expression	128
	of CC Che	mokines	
	6.7.1	TGF- β 1 can suppress the induction of MIP-1 α mRNA	131
		expression induced by GM-CSF and IL-3	
	6.7.2	The suppression of GM-CSF induced MIP-1 α expression	131
		by TGF-β1 is dose-dependent	
	6.7.3	TGF- β 1 can also suppress the induction of JE and	134
		MARC mRNA expression induced by GM-CSF and IL-3	
6.8	Effect of C	β M-CSF, IL-3 and TGF- β 1 on the Expression of MIP-1 α	134
	Receptors		
	6.8.1	GM-CSF and IL-3 differentially regulate the expression	136
		of CCR1 and CCR5 in BMM	
	6.8.2	GM-CSF and IL-3 also differentially regulate the expression	136
		of CCR1 and CCR5 in splenic cells	

	6.8.3	GM-CSF and IL-3 do not affect the expression of	139
		other MIP-1α receptors in BMM	
	6.8.4	TGF- β 1 does not markedly alter the expression of CCR1	139
		and CCR5 in BMM	
6.9	Effect of C	GM-CSF on MIP-1α Receptor Levels on BMM	141
6.10	Effect of C	M -CSF on the Ability of Cells Expressing MIP-1 α Receptors	145
	to Mobilise	e Calcium in Response to MIP-1a	
	6.10.1	Effect of GM-CSF on the ability of BMM to mobilise	145
		calcium in response to MIP-1 α	
	6.10.2	GM-CSF induces the expression of CCR1 mRNA in	146
		THP-1 cells	
	6.10.3	GM-CSF does not alter the ability of THP-1 cells to mobilise	149
		calcium in response to MIP-1a	
6.11	Analysis of	MIP-1α Expression in GM-CSF Transgenic Mouse Tissues	149
	6.11.1 Ove	erview: A potential link between MIP-1 α and the	149
	path	nologies observed in GM-CSF transgenic mice	
	6.11.2 RT-	PCR analysis of MIP-1α expression in GM-CSF transgenic	151
	tiss	ues	
	6.11.3 MII	$P-1\alpha$ expression appears elevated in the eye of GM-CSF	152
	tran	sgenic mice	
	6.11.4 MII	P-1 α expression may be elevated in the bone marrow	155
	of C	GM-CSF transgenic mice	
	6.11.5 Exp	pression of MIP-1 α does not appear elevated in striated	159
	mus	scle or peritoneal cells of GM-CSF transgenic mice	

xiv

CHAPTER 7: DISCUSSION

7.1	Introduction	165
7.2	Induction of MIP-1 α Expression by GM-CSF and IL-3	166
7.3	Effect of IL-5 on MIP-1α expression	169
7.4	Induction of Other CC Chemokines by GM-CSF and IL-3	169

7.5	TGF- β 1 Suppression of Members of the MIP-1 α Superfamily	172
7.6	Expression of CCR1 in BMM is Regulated Differently	173
	to that of CCR5, in Response to GM-CSF and IL-3	
7.7	Analysis of MIP-1 α Expression in GM-CSF Transgenic Mouse Tissues	175
7.8	The Role of GM-CSF, IL-3 and TGF- β in Regulating the Expression	178
	of MIP-1a in BMM?	
7.9	The Importance of MIP-1 α in Disease Pathology	181
7.10	Conclusions and Aspects for Further Investigation	182

REFERENCES

187

LIST OF FIGURES AND TABLES

Figure	Figure Title	Page
Figure 1.1	Overview of Haemopoiesis	2
Figure 1.2	Haemopoiesis and the Stromal Microenvironment	11
Figure 3.1	Structure of CCR1	6 7
Figure 6.1	Northern blot analysis of the effect of G-CSF, IL-6 and LIF on the	
	expression of MIP-1a mRNA in BMM	110
Figure 6.2	Northern blot analysis of the effect of GM-CSF on the expression of	
	MIP-1a mRNA in BMM	112
Figure 6.3	Northern blot analysis of the effect of GM-CSF on a time course	
	of MIP-1 α expression in BMM	113
Figure 6.4	Northern blot analysis of the effect of a titration of GM-CSF on	
	MIP-1 α mRNA expression in BMM	115
Figure 6.5	Northern blot analysis of the effect of GM-CSF on the expression	
	of MIP-1 α mRNA in BMM, in the presence or absence of either	
	serum or L-929 CM	117
Figure 6.6	Northern blot analysis of the effect of a neutralising GM-CSF antibody	
	on the expression of MIP-1 α mRNA in BMM, under LPS free conditions	118
Figure 6.7	Western blot analysis of the effect of GM-CSF on MIP-1 α protein expression	120
Figure 6.8	Northern blot analysis of the effect of IL-3 on the expression of MIP-1 α	
	mRNA in BMM	122
Figure 6.9	Northern blot analysis of the effect of a titration of IL-5 on MIP-1 α	
	mRNA expression in BMM	124
Figure 6.10	RT-PCR analysis of the expression of the IL-5R α chain in BMM	125
Figure 6.11	RT-PCR analysis of the effect of IL-5 on the expression of MIP-1 α mRNA	
	in murine bone marrow cells	127
Figure 6.12	Northern blot analysis of the effect of GM-CSF and IL-3 on the expression	
	of MIP-1 β mRNA in BMM	129
Figure 6.13	Northern blot analysis of the effect of GM-CSF and IL-3 on the expression	
	of JE and MARC mRNA in BMM	130
Figure 6.14	Northern blot analysis of the effect of TGF- $\beta 1$ on the induction of	
	MIP-1 α expression mRNA by GM-CSF and IL-3	132
Figure 6.15	Northern blot analysis of the effect of a titration of TGF- $\beta 1$ on the induction	
	of MIP-1 α expression mRNA by GM-CSF	133
Figure 6.16	Northern blot analysis of the effect of TGF- $\beta 1$ on the induction of MARC	
	and JE mRNA expression by GM-CSF and IL-3	135

Figure	Figure Title	Page
Figure 6.17	Northern blot analysis of the effect of GM-CSF and IL-3 on the expression	
	of CCR1 and CCR5 mRNA in BMM	137
Figure 6.18	Northern blot analysis of the effect of GM-CSF and IL-3 on the expression	
	of CCR1 and CCR5 mRNA in splenic cells	138
Figure 6.19	RT-PCR analysis of the expression of CCR3, CCR4, D6 and CCR2	
	following stimulation with GM-CSF or IL-3	140
Figure 6.20	Northern blot analysis of the effect of TGF- β 1 on the expression	
	of CCR1 and CCR5 mRNA, in the presence of GM-CSF and IL-3	142
Figure 6.21	Scatchard analysis of the effect of GM-CSF on the binding of ^{125}I MIP-1 α	
	to MIP-1 α receptors on BMM	143
Figure 6.22	Plot of the effect of GM-CSF on BMM MIP-1 α receptor numbers	144
Figure 6.23	Effect of GM-CSF stimulation on the ability of BMM to mobilise calcium	
	in response to MIP-1a	147
Figure 6.24	Northern blot analysis of the effect of human GM-CSF on the expression	
	of human CCR1 mRNA in THP-1 cells	148 ,
Figure 6.25	Effect of GM-CSF stimulation on the ability of THP-1 cells to mobilise	
	calcium in response to MIP-1 α	150
Figure 6.26	RT-PCR analysis of MIP-1 α expression in wild type and GM-CSF	
	transgenic mouse eye	153
Figure 6.27	Southern blot analysis of MIP-1 α and β -Actin expression detected by	
	RT-PCR analysis, in wild type and GM-CSF transgenic mouse eye	154
Figure 6.28	RT-PCR analysis of MIP-1 α expression in wild type and GM-CSF	
	transgenic mouse bone marrow	157
Figure 6.29	Southern blot analysis of MIP-1 α and β -Actin expression detected by RT-	
	PCR analysis, in wild type and GM-CSF transgenic mouse bone marrow	158
Figure 6.30	RT-PCR analysis of MIP-1 α expression in wild type and GM-CSF	
	transgenic mouse leg muscle	160
Figure 6.31	Southern blot analysis of MIP-1 α and β -Actin expression detected by RT-	
	PCR analysis, in wild type and GM-CSF transgenic mouse leg muscle	161
Figure 6.32	RT-PCR analysis of MIP-1 α expression in wild type and GM-CSF	
	transgenic mouse peritoneal cells	162
Figure 6.33	Southern blot analysis of MIP-1 α and β -Actin expression detected by RT-PCR	
	analysis, in wild type and GM-CSF transgenic mouse peritoneal cells	163
Figure 7.1	Summary of proposed interactions between MIP-1 α , GM-CSF, IL-3	
	and TGF-β	184

Table	Table Title	Page
Table 1.1	Summary of Cytokines and Growth Factors	14
Table 3.1	The chemokine superfamilyso far	45
Table 3.2	The human chemokine receptor family and their ligands	65
Table 4.1	Plasmids for generation of probe fragments	78
Table 4.2	Plasmids for use in RT-PCR analysis	79
Table 4.3	PCR primers- identity	80
Table 4.4	PCR primers- sequences	81
Table 5.1	Tissue culture cell lines	91
Table 5.2	Analysis of bone marrow-derived macrophage populations	91

1

xvii

ABBREVIATIONS

³² P	³² Phosphorous
5-FU	5-fluorouracil
7TMsGPCR	7 transmembrane spanning G protein-coupled
	receptor
Ab	antibody
Ag	antigen
AIDS	Acquired immune deficiency syndrome
AML	acute myelogenous leukaemia
Ara-C	cytosine arabinoside
ATP	adenosine triphosphate
βc	common β chain receptor subunit
βπ_3	IL-3 receptor β chain
BFU	burst-forming unit
BM	bone marrow
BMM	bone marrow-derived macrophages
BSA	bovine serum albumin
С	cysteine
cDNA	complementary deoxyribonucleic acid
CFC	colony-forming cell
CFU-A	colony-forming unit arbitrary
CFU-E	colony-forming unit erythroid
CFU-GM	colony-forming unit granulocyte-macrophage
CFU-S	colony-forming unit spleen
СМ	conditioned medium
CML	chronic myeloid leukaemia
CNS	central nervous system
CNTF	ciliary neurotrophic factor
CSF	colony-stimulating factor
CTP	cytidine triphosphate
d	day (e.g. day 8)
DARC	Duffy antigen receptor for chemokines
dATP	deoxyadenosine triphosphate
dCTP	deoxycytidine triphosphate
DEAE	diethylaminoethyl
dGTP	deoxyguanosine triphosphate
DHS	donor horse serum
DMEM	Dulbecco's modified Eagle's medium
DMSO	dimethylsulphoxide
DNA	deoxyribonucleic acid
DNase	deoxyribonuclease
dNTP	deoxynucleoside triphosphates
dTTP	deoxythymidine triphosphate
DTT	dithiothreitol
EAE	experimental autoimmune encephalomyelitis
ECM	extracellular matrix
EDTA	ethylenediaminetetraacetic acid
Epo	erythropoietin

ES	embryonic stem
FCS	fetal calf serum
GAPDH	glyceraldehyde-3-phosphate dehydrogenase
G-CSF	granulocyte colony-stimulating factor
GM	granulocyte-macrophage
GM-CSF	granulocyte-macrophage colony-stimulating
	factor
GTP	guanosine triphosphate
HCl	hydrochloric acid
HEK	human embryonic kidney
HEPES	N-2-hydroxyethylpiperazine-N'-2-
	ethanesulphonic acid
HIV	human immunodeficiency virus
HPC	haemopoietic progenitor cell
HPP-CFC	high proliferative potential colony-forming cell
HSC	haemopoietic stem cell
HU	hydroxyurea
IFN	interferon
Ig	immunoglobulin (e.g. IgE)
-s IL-	interleukin (e.g. IL-3)
IP10	interferon-inducible protein-10
IP3	inositol triphosphate
KSHV	kaposi's sarcoma-associated herpesvirus
LB	Luria broth
LIF	leukemia inhibitory factor
LPS	lipopolysaccharide
LTRC	long-term repopulating cell
LTC-IC	long-term culture intitiating cell
LTBMC	long-term bone marrow culture
M-CSF	macrophage colony-stimulating factor
MCP	monocyte chemotactic protein
 MEM	minimal essential medium
Mig	monokine induced by interferon- γ
MIP	macrophage inflammatory protein
MOPS	3-(N-morpholino) propanesulphonic acid
MPIF	myeloid progenitor inhibitory factor
mRNA	messenger ribonucleic acid
MRP	macrophage inflammatory protein related protein
MS	multiple sclerosis
m.w.	molecular weight
NK	natural killer cell
OD	optical density
ор	osteopetrotic
ŌSM	oncostatin M
PAGE	polyacrylamide gel electrophoresis
PBS	phosphate buffered saline
PCR	polymerase chain reaction
PDGF	platelet-derived growth factor
PF-4	platelet factor four

РКС	protein kinase C
PLC	phospholipase C
PTX	perussis toxin
r	recombinant
R	receptor
RA	rheumatoid arthritis
RANTES	regulated on activation normal T-cell expressed
	and secreted
Rh	receptor homologue
RNA	ribonucleic acid
RNase	ribonuclease
RT	reverse transcriptase
SCF	stem cell factor
SCI	stem cell inhibitor
SDF	stromal derived factor
SDS	sodium dodecyl sulphate
Sl	steel
SSC	standard saline citrate
TEMED	tetramethylenediamine
tg	transgenic
TGF	transforming growth factor
T _h	T helper
TNF	tumour necrosis factor
TPO	thrombopoietin
Tris	tris(hydroxymethyl)aminomethane
U	units
UTP	uridine triphosphate
UV	ultraviolet
v/v	volume for volume
w/v	weight for volume
wt	wild type

.

PART I: INTRODUCTION

CHAPTER 1: The Haemopoietic System

1.1 The Haemopoietic System: An Overview

The haemopoietic system is essential for the maintained viability of an individual. It can be considered as a highly complex cellular system that enables the derivation of mature immuno-haemopoietic cells, through the processes of proliferation, commitment and differentiation. There are at least eight mature blood cell types, functionally distinct from each other, but which are ultimately derived from a single self-renewing pluripotent stem cell that differentiates via various different committed progenitor cells (some of which are summarised in figure 1.1). This complex system has evolved to cope with the on-going production of the vast numbers of cells needed on a daily basis, simply to maintain the stasis of the mature immuno-haemopoietic system. In fact it is estimated that as many as 3.7×10^{11} haemopoietic cells are needed per day to maintain human homeostasis. In addition to this homeostatic role, the haemopoietic system must also be able to respond rapidly with a dramatic increase in cell numbers, in order to be able to combat infection and disease (inducible haemopoiesis). The many processes involved in regulating the haemopoietic system are likely to be complex and elaborate and yet they must also be remarkably efficient because despite the enormous scope for abnormalities, diseases such as leukaemia and anaemia are relatively rare.

1.2 The Stem Cell Compartment

One can consider the stem cell compartment in its simplest form, as "a heterogeneous compartment consisting of cells displaying varying degrees of self-renewal capacity and/or differentiation potential" (Graham and Pragnell 1992a). This is perhaps best illustrated by considering the haemopoietic system as a three-tiered hierarchical system, analagous to a pyramid, which forms an ordered continuum. At the apex of this pyramid are the most primitive cells with a high capacity for self-renewal, the greatest repopulating ability and which are most resistant to proliferation and differentiation stimuli. As one moves further down the hierachy, the cells exhibit a reduced capacity for self-renewal and increased tendency for differentiation. At the base of this pyramid are the mature cells. It is this remarkable ability to undergo self-

Figure 1.1: Overview of the Haemopoietic System

This figure summarises the ability of the haemopoietic system to generate all the necessary different mature blood cell types. Shown is a single pluripotent stem cell, as well as the various different progenitor cells and their lineages. Progenitors are defined by *in vitro* assays, according to the cell type to which they give rise and are known as colony-forming units (CFU) or burst-forming units (BFU). Abbreviations: E=Erythroid, Meg=Megakaryocyte, GM=Granulocyte/Macrophage. For example, CFU-GM represents a progenitor giving rise to granulocytes and macrophages.





renewal and generate differentiated progeny which defines a stem cell. Stem cells constitute only a tiny proportion of the total haemopoietic system, probably between 0.01 and 0.1 % of the total bone marrow of an adult mouse, and it is believed that the majority of these are held in a quiescent or G_0 state (Hodgson *et al* 1982, Harrison and Lerner 1991). Stem cells are triggered into an actively cycling state in order to replenish progenitor and mature cell compartments, as a result of stress to the system (e.g. following chemotherapy, or during infection). Following replenishment, the stem cells re-enter quiescence.

Significant progress in understanding the nature of haemopoietic stem cells came from a series of experiments in which it was discovered that transplanted murine bone marrow contained progenitor cells that were able to give rise to individual macroscopic colonies of myeloid, erythroid and megakaryocytic cells on the surface of spleens of lethally irradiated recipient mice (Till and McCulloch 1961). This seminal experiment, defined the first *in-vivo* stem cell assay (the CFU-S assay) and produced the first evidence for primitive quiescent stem cells with a capacity for self-renewal and differentiation. Due to the difficulty in distinguishing stem cells on the basis of their morphology, much of what we now know about stem cells has been derived through the development and usage of such *in vivo* and other, *in vitro*, assays.

1.3 The In Vivo CFU-S Assay

The aforementioned colony forming unit-spleen (CFU-S) assay (Till and McCulloch 1961) was also important in demonstrating that stem cells are in fact a heterogeneous population of cells. In this assay, transplantation of donor bone marrow cells to a lethally irradiated mouse resulted in the reconstitution of the ablated haemopoietic system and enabled the mouse to survive an otherwise potentially lethal dose of radiation. During reconstitution, colonies formed on the spleen that were donor-derived and which could be detected at various stages of maturity, depending upon when the CFU-S assay was read (Siminovich *et al* 1964). Colonies were first visible on the spleen at day 7 or 8 and other colonies were visible at a later time, around day 12. Subsequent studies, in which individual day 7/8 spleen colonies were transplanted into a similarly lethally irradiated mouse, demonstrated very little secondary CFU-S colony formation and indicated that the early appearing (day 7/8) colonies were derived from cells with a limited self-renewal/differentiation potential.

Cells derived from individual day 12 colonies were however able to generate multiple secondary CFU-S colonies on serial transplantation, indicating that cells within these later-derived colonies had a greater capacity for self-renewal (Wolf and Priestly, 1986). Put more simply, colonies appearing early at day 7/8 were derived from more mature cells, whilst the later appearing day 12 colonies were derived from more primitive cells. Originally it was assumed that the day 12 colonies, since they appeared later, were derived directly from the earlier appearing day 7/8 colonies by differentiation and self-renewal. However studies have shown that 50 % of the day 7/8 colonies (mostly erythroid in morphology) subsequently regressed, following which, colonies were observed at day 12, which had a multi-lineage constitution (Magli et al 1982). The precise origin of the day 12 colonies remains in controversy, with three possible explanations. They may have arisen as a result of re-circulation of CFU-S cells that originally seeded in the bone marrow, but which subsequently moved to the spleen (Van Zant 1984), or a second explanation is that some stem cells have migrated out of the day 7/8 colonies and formed *de novo* colonies, that appear later at day 12. The third and simplest explanation is that the day 12 colonies simply take longer to appear. Additional evidence for the heterogeneity of stem cells, at least within the CFU-S compartment, has arisen through studies using cell cycle specific toxic drugs, such as 5-fluorouracil (5-FU). These studies demonstrated that the day 7/8 CFU-S cells were extremely sensitive to the lethal effects of these drugs, whereas the later day 12 CFU-S cells were less sensitive (Hodgson et al 1982). This suggests that the day 7/8 cells are largely in cycle, whilst the day 12 cells are in a more quiescent state and accordingly represent a more primitive population of stem cells.

Following the early work on the CFU-S assay, it was believed that the day 12 CFU-S cell could be the 'ultimate' or 'totipotent' haemopoietic stem cell *in vivo*. Since then however, various lines of evidence have emerged suggesting that the day 12 CFU-S cell is not the ultimate haemopoietic stem cell. Perhaps most obviously, is the inability of the day 12 CFU-S cell to give rise to cells of the lymphopoietic lineages. Also, despite the importance of donor CFU-S cells in the reconstitution of short-term haemopoiesis in a lethally irradiated mouse (Jones *et al* 1989), these cells are unable to establish long-term *in vitro* or *in vivo* haemopoiesis (van der Sluijs *et al* 1990). This would suggest the existence of another more primitive or 'pre-CFU-S' cell, the existence of which was previously suggested (Hodgson and Bradley 1979).

This pre-CFU-S stem cell would have the ability to give rise to both lymphoid and myeloid lineages (from which the more mature CFU-S cell would be derived as a myeloid committed stem cell). To date this cell has not been purified to homogeneity, but it has been further characterised. Pre-CFU-S cells have been enriched for and separated from, CFU-S cells on the basis of its ability to retain the dye Rhodamine 123 (Ploemacher and Brons 1988, 1989). These cells have been termed long-term repopulating cells (LTRC), for their ability to maintain long-term haemopoiesis. These cells alone are not sufficient however, to effectively establish haemopoietic repopulation in a lethally irradiated animal, nor are they radioprotective. CFU-S cells alone will initiate reconstitution, but this will only persist for a short time and mice die within about 4 weeks. Full, long-term reconstitution only occurs with transplantation of both more mature CFU-S type stem cells, as well as the primitive pre-CFU-S stem cells (Jones et al 1990). Therefore, in simplistic terms one can subdivide the stem cell compartment into two subcompartments, comprised of the more primitive long-term engrafting stem cells and the short-term or transiently engrafting stem cells (reviewed by Graham and Wright 1997).

1.4 In vitro Culture of Haemopoietic Stem Cells

1.4.1 In vitro stem cell assays

As mentioned in previous sections, much of what is known about the stem cell compartment has been derived from the development of *in vitro* assays. The basis for these assays is the fact that clonogenic bone marrow cells cultured in the presence of appropriate growth factors give rise to colonies in semi-solid culture, without the addition of supporting stromal cells. The precise combination of cytokines and their concentrations vary from assay to assay. The size, morphology and composition of these colonies can then be assessed to give an indication of the level of primitiveness of the cell from which they were derived. In general, only differentiation along myeloid and erythroid lineages is observed.

A number of *in vitro* assays have been reported, that detect a range of cells within the haemopoietic stem cell compartment. The precise role that many of these cells play in normal steady-state haemopoiesis *in vivo* is still uncertain. These assays have however, enabled identification of several primitive stem cell types and haemopoietic stem cell regulatory factors. Some of these assays are outlined briefly below and were recently reviewed by Graham and Wright (1997).

The high proliferative potential colony-forming cell (HPP-CFC) was originally identified by Bradley and Hodgson (1979), using a clonal *in vitro* assay that detected macroscopic colonies in agar culture. These HPP-CFC differ from more mature haemopoietic cells, by virtue of their relative resistance to 5-fluorouracil, their requirement for a combination of regulatory growth factors and their ability to form large colonies (>0.5 mm) in agar culture. HPP-CFC have been further defined and now form three distinct populations of stem cells, known as HPP-CFC 1, 2 and 3. Studies with 5-fluorouracil have suggested that HPP-CFC 1 cells are the more primitive, whilst HPP-CFC 2 and 3 are more mature in phenotype. It is believed that the HPP-CFC population forms part of the stem cell compartment detected by the *in vivo* CFU-S assay (reviewed by Bertoncello 1992, Graham and Wright 1997).

The CFU-A assay is an *in vitro* assay that enables detection of primitive progenitor cells, that are HPP-CFC-like and which are believed to be equivalent to the population of cells detected by the *in vivo* day 12 CFU-S assay. The CFU-A assay detects a cell with an incidence of approximately 200 per 1 x 10^5 cells in normal bone marrow and which forms colonies >2 mm in diameter (Pragnell *et al* 1988, Lorimore *et al* 1990).

1.4.2 The stromal microenvironment

The most accurate *in vitro* culture model, of *in vivo* haemopoietic stem cell activity, is obtained through the use of stromal or long-term bone marrow cultures (LTBMC). The bone marrow microenvironment is a complex 3-dimensional structure consisting of a meshwork of heterogeneous cell types and extracellular matrix proteins. It provides an environment that allows the maintenance, self-renewal, proliferation and differentiation of haemopoietic stem cells and ultimately their migration to the circulation (Dexter 1982, Deryugina and Muller-Sieberg 1993). Bone marrow stromal cells are a crucial component in this micro-environment and include macrophages, fibroblasts, adipocytes, and endothelial cells (Gordon *et al* 1980, Mori *et al* 1990, Greenberger 1991). Stromal cells can be grown from freshly isolated bone marrow as *in vitro* cultures and will, under the appropriate conditions, form confluent adherent layers that will support the proliferation, differentiation and limited self-

renewal of haemopoietic progenitors in the absence of added haemopoietic growth factors. The first stromal cultures were developed using murine haemopoietic tissue (Dexter *et al* 1977a, 1977b) and were subsequently adapted to human cells (Gartner and Kaplan 1980). Whilst cells with long-term repopulating ability *in vivo* can be maintained for several weeks *in vitro* (Fraser *et al* 1991), both human and murine stromal-dependent LTBMCs have a limited life span. In these LTBMCs, committed progenitor cells rapidly undergo terminal differentiation and within 4-5 weeks these initial committed progenitor cells have disappeared and at this time, other clonogenic progenitors are detected. These cells are derived from primitive cells, termed long-term culture initiating cells (LTC-IC) (Sutherland *et al* 1989). These LTC-IC appear to be present in normal bone marrow at a frequency of approximately 1-5 per 1 x 10^4 cells and were recently shown to be capable of self-renewal and expansion *in vitro* (Petzer *et al* 1996). Despite their limitations, LTBMC have provided a useful *in vitro* culture system for modelling the structure of the stromal microenvironment and its interactions with haemopoietic stem cells.

1.5 Progenitor Cells and Lineages

Haemopoietic stem cells are able to undergo self-renewal, and this has been postulated to be of two types. 1) Symmetric self-renewal, which results in the generation of two identical daughter stem cells and 2) asymmetric self-renewal, that results in an identical daughter stem cell and a daughter cell that is committed to differentiation along a specific lineage(s). As shown previously in figure 1.1, differentiation occurs through various lineages and via a variety of morphologically identifiable intermediate progenitor cells, ultimately giving rise to terminally differentiated mature cells. These haemopoietic progenitor cells (HPCs) have only a limited capacity for self-renewal and are defined through the ability to give lineage specific progeny in *in vitro* assays. These assays, as those used for HSCs, are based on the ability of progenitor cells to form colonies upon culture in semi-solid agar or upon methylcellulose, in the presence of the appropriate haemopoietic growth factors. These "colony assays" led first to the growth and clonal differentiation of macrophages and granulocytes from their progenitor cells (Paran et al 1970, Pike and Robinson 1970) and subsequently to the in vitro generation of the majority of other blood cell lineages (Stephenson et al 1971, Metcalf et al 1975). The HPCs are defined according to the precursor cell type to which they give rise and are known as colony-forming units (CFU) or burst-forming units (BFU). So for example, cells from early and late progenitors giving rise to colonies containing erythroid cells are known as BFU-E or CFU-E, respectively, whilst those progenitors giving rise to granulocytes or macrophages are called CFU-GM.

These assays have enabled the identification, characterisation and cloning of a multitude of regulatory factors, that have been implicated in the process of stem cell function. The function of some of these regulatory factors and their overall importance to the regulation of haemopoiesis will be discussed in greater detail in chapter 2.

1.6 Inhibition of Haemopoiesis

The existence of a multitude of positively acting or stimulatory haemopoietic growth factors and cytokines, and their potential importance in regulating haemopoiesis, is well documented. What is less certain however, is the role inhibitory molecules play in haemopoiesis. One can argue that the action of positive regulators alone, is sufficient to control the proliferation of haemopoietic stem cells. There are however several problems with this argument. On a superficial level, one can question if this arrangement would provide a sufficiently rapid and tightly controlled regulation. Under normal steady state conditions, it is estimated that only 10 % or fewer of the haemopoietic stem cells are in active cell cycle, with the remainder in a quiescent or G₀ phase. One can envisage that stress to the system, such as that of inflammation, could result in inducible haemopoiesis (see section 1.7) and activation of previously quiescent stem cells. Under circumstances of extreme stress, such as bone marrow regeneration following chemotherapy or radiotherapy, even the most primitive stem cells are actively cycling (Harrison and Lerner 1991). Following activation, the majority of the cells within the haemopoietic stem cell compartment then return to their normal quiescent state. The existence of a variety of inhibitory molecules, that actively inhibit the cell cycle have been proposed to be responsible for the return to, or even maintenance of, stem cell quiescence. Whilst it is possible that the action of stimulatory regulators alone may be sufficient to control haemopoietic stem cell proliferation, the existence of both positive and negatively acting regulators would allow for a far more flexible and responsive system of control. In such a model, positive and negative regulators could both operate, and the relative balance between their opposing actions would ultimately control the state of haempoietic stem cell proliferation.

Further support for the existence of negative regulators comes from the identification of molecules that have the ability to inhibit stem cell proliferation (reviewed in Axelrad 1990, Graham and Pragnell 1990, Graham 1997). Both novel inhibitory molecules, such as the Tetrapeptide (AcSDKP), and Haemoregulatory peptide (pEEDCK) (both reviewed in Graham and Pragnell 1990, Guigon and Bonnet 1995, Graham 1997) have been identified, as well as previously described molecules such as TGF- β 1, MIP-1 α and TNF- α . Interestingly some of these 'cytokines' appear to be able to function bi-directionally, in that they can both stimulate as well as inhibit, the growth of some cell types (Keller *et al* 1994). This will be discussed further in sections 2.14.6 and 3.6.5, with regard to the known functions of TGF- β 1 and MIP-1 α respectively.

1.7 The Stromal Microenvironment and the Production of Soluble Mediators

As mentioned previously, the stromal microenvironment is crucial to the regulatory processes involved in haemopoiesis in the bone marrow. Much of what is known about the potential role of stromal regulation of haemopoiesis has been determined through the investigation of LTBMC and other in vitro culture systems designed to mimic the stromal microenvironment. A multitude of studies has been reported, particularly with regard to further understanding the relationship between the stromal cell network and the various cytokines and other soluble mediators implicated in regulating haemopoiesis. At this point, it is worth mentioning inducible haemopoiesis (as opposed to constitutive haemopoiesis), which can be defined as haemopoiesis induced as a result of an increased need for mature blood cells (e.g. during infection and inflammatory responses). Whilst these two forms of haemopoiesis may operate under different circumstances and via different mechanistic routes, a complex network of interacting cytokines and growth factors has been implicated in both. Regardless of which model(s) of haemopoiesis one subscribes to (stochastic or instructive; see sections 2.15.1 and 2.15.2), this network of cytokines is strongly implicated in a multitude of roles, such as the control of mobilisation, selfrenewal, survival, proliferation, commitment, differentiation and migration of cells

within the haemopoietic system. Further discussion of the potential role of this network in the control of haemopoiesis is detailed in chapter 2.

1.7.1 Bone marrow and stromal production of cytokines and growth factors

Steady state or constitutive haemopoiesis is regulated in the bone marrow by direct cell-cell contact between bone marrow stroma and haemopoietic stem and progenitor cells, and by locally produced soluble mediators (see figure 1.2).

Direct cell-cell contact may not be a pre-requisite in LTBMC, which mimic the stromal microenvironment, since stromal cells separated from haemopoietic cells by a thin porous mesh can still support haemopoiesis (Verfaillie 1992). This suggests that these stromal cells can produce soluble mediators that enable haemopoiesis to be maintained (Verfaillie 1993, Verfaillie *et al* 1994). Many studies have examined production of such soluble factors both *in vivo* and *in vitro*, as discussed below.

Attempts to detect mRNA expression and protein production has been complicated by the difficulty in detection of these factors and to date the results are still very confused and sometimes even contradictory. Constitutive expression of macrophage colony-stimulating factor (M-CSF) and granulocyte-macrophage colonystimulating factor (GM-CSF) by a variety of cell types, including fibroblasts and bone marrow stromal cells in long-term culture, have been widely reported (Piersma et al 1984, Sieff et al 1988, Schaafsma et al 1989, Kittler et al 1992). Detection of constitutive expression of granulocyte colony-stimulating factor (G-CSF), stem cell factor (SCF) and interleukin (IL)-6 mRNA has also been documented (Kittler et al 1992, Guba et al 1992). The role of IL-3 in the regulation of constitutive haemopoiesis is however, still very confused. Whilst the ability of IL-3 to function as a multilineage colony-stimulating factor during the earlier stages of haemopoietic stem cell differentiation in vitro are well documented (Sonoda et al 1988, Koike et al 1987) attempts to detect constitutive IL-3 expression, without the addition of inducing stimuli, have been largely unsucessful (Guba et al 1992, Cluitmans et al 1995). Expression of IL-3 has however, been detected in irradiated stromal cells using the sensitive technique of polymerase chain reaction (PCR) amplification and addition of IL-3 specific antibodies resulted in the abrogation of the ability of these stromal cell lines to support the growth of factor dependent cell lines (Kittler et al 1992). These results suggest that whilst the presence of growth factors in LTBMC stromal cells and



Figure 1.2 : Haemopoiesis and the Stromal Microenvironment The above figure shows a simple diagrammatic representation of the haemopoietic microenvironment. It summarises some of the processes that occur within the microenvironment, such as those of adhesion, stimulation, inhibition, self-renewal, commitment, differentiation and migration. The production of a variety of soluble mediators is believed to be an important part of the regulation of these processes and their production by stromal cells is also indicated in the above figure.
bone marrow may be undetectable by techniques such as Northern blotting and PCR, their actual presence (albeit at extremely low levels) can not be excluded.

Whilst many reports have documented the constitutive expression of GM-CSF and G-CSF by a variety of cell types crucially important to the bone marrow microenvironment, Cluitmans and his collegues were unable to detect their constitutive expression in normal human bone marrow (Cluitmans et al 1995). This study suggested that whilst constitutive expression of the cytokines M-CSF, SCF, IL-1β, IL-4, IL-6, IL-7, tumour necrosis factor-beta (TNF-β), transforming growth factorbeta (TGF- β) and macrophage inflammatory protein-1 alpha (MIP-1 α) was detectable in human bone marrow, expression of G-CSF and GM-CSF was not. This raises the question of whether or not GM-CSF and G-CSF play a role in uninduced/normal steady-state haemopoiesis. However, it should be noted that although Cluitmans et al were unable to detect constitutive GM-CSF expression in normal bone marrow, this does not preclude their expression at subliminal levels. Indeed, evidence from studies on mice deficient for GM-CSF, suggest that GM-CSF is critical for the homeostasis of normal lung physiology and resistance to local infection (Stanley et al 1994) and indicates that at least some subliminal GM-CSF expression must occur in vivo. Likewise, studies on mice deficient for G-CSF indicate at least some role for G-CSF in the in vivo production of granulocytes (Lieschke et al 1994).

1.7.2 Localisation of growth factors

Many soluble and insoluble mediators are produced by the stromal microenvironment, though in many cases, detection of these has proved extremely difficult. This suggests that if some of these factors are indeed produced within the bone marrow, they may be at extremely low levels (perhaps even below levels detectable with current techniques). One possible explanation for this may lie with the apparent existence of growth factor localisation. It is now evident that some cytokines and growth factors (such as SCF, M-CSF and IL-1) exist not just as soluble forms, but also as biologically active membrane or extra-cellular membrane (ECM) bound forms (Kurt-Jones *et al* 1985, Stein *et al* 1990). Other growth factors, e.g. GM-CSF, IL-3 and leukemia inhibitory factor (LIF) are also able to bind to components of the ECM (Gordon *et al* 1987, Roberts *et al* 1988). The ability to bind components of the ECM, as found within the stromal microenvironment, would enable these growth factors to

be held in an active, but highly localised fashion. There is some intriguing evidence that certain components of the ECM, such as heparan sulphate and other glycosaminoglycan (GAG) derivatives, can function directly on their own to regulate growth factor expression (Wrenshall *et al* 1991, Yang and Yang 1995). With the possible exception of fibroblast growth factor (Spivak-Kroizman *et al* 1994), these proteoglycan interactions are clearly not essential for growth factor function as exogenous soluble growth factors will function in the absence of such interactions. This does not however exclude their involvement in a localised niche within the stem cell microenvironment, perhaps in the sequestration or presentation of growth factors.

1.8 Haemopoietic Regulators and the Cytokine Network: Redundancy or Eloquent Control ?

There are a vast number of cytokines and growth factors and some of these, are summarised in table 1.1. Whilst not all play a direct role in regulating haemopoiesis, a large number of them clearly do. Given the known complexity and precision of haemopoiesis, it is not really a surprise that a large number of regulatory molecules have been identified to date and indeed further regulators continue to be identified. What is perhaps surprising, is the seemingly overlapping actions or redundancy of many regulators.

There does at first sight appear to be a great deal of evidence supporting the notion of redundancy (reviewed in Metcalf 1993). It is clear for instance, that more than one factor can stimulate the formation of the same type of colony, e.g. G-CSF, GM-CSF, IL-3, and IL-6 can all function to stimulate formation of granulocytes (Metcalf 1993). A similar number of factors can also stimulate the formation of eosinophil and mast cell colonies. Such an arrangement is seemingly wasteful and redundant, as well as offering a greatly increased risk of dysregulation. A second argument is that certain regulators, particularly LIF, IL-6 and IL-11 (Metcalf 1991, Kishimoto *et al* 1995), appear to have an unusually wide range of target cells. In many cases these cytokines also exhibit (at least partial) overlaps in their actions.

It is now evident that the receptors for these cytokines and growth factors can be subdivided into 3 major categories (reviewed in Park *et al* 1990a, Ullrich and Schlessinger 1990, Cosman *et al* 1990, Bazan 1990, Hara *et al* 1996, Bagley *et al* 1997), as follows: 1) Receptors belonging to the immunoglobulin superfamily, e.g.

Table 1.1: Summary of Cytokines and Growth Factors

Interleukins and Colony-S	timulating Factors	
IL-1α	IL-10	G-CSF
IL-1β	IL-11	GM-CSF
IL-2	IL-12	M-CSF
IL-3	IL-13	SCF
IL-4	IL-14	LIF
IL-5	IL-15	Epo
IL-6	IL-16	ТРО
IL-7	IL-17	TNF-α
IL-8	IL-18	TNF-β
IL-9		Ftl3L
		CNTF
		OSM
<u>TGF-β superfamily</u>	Interferons	Other Growth Factors
TGF-β1	IFN-α	EGF
TGF-β2	IFN-β	TGF-a
TGF-β3	IFN-γ	acidic FGF
Inhibin		basic FGF
Activin		PDGFα
Bone Morphogenetic Proteins (BMPs)		PDGFβ
		IGF-I
		IGF-II
		NGF
		HGF

Chemokines

See Table 3.1 for a full list of Chemotactic cytokines (Chemokines)

Abbreviations used:

IL-, interleukin; CNTF, ciliary neurotrophic factor; EGF, epidermal growth factor; Epo, erythropoietin; FGF, fibroblast growth factor; Ftl3L, Ftl3 ligand; G-CSF, granulocyte colony-stimulating factor; GM-CSF, granulocyte-macrophage colony-stimulating factor; HGF, hepatocyte growth factor; IGF, insulinlike growth factor; IFN, interferon; LIF, leukemia inhibitory factor; M-CSF, macrophage colony-stimulating factor; NGF, nerve growth factor; OSM, oncostatin M; PDGF, platelet-derived growth factor; SCF, stem cell factor; TGF, transforming growth factor; TNF, tumour necrosis factor; TPO, thrombopoietin.

IL-1R; 2) the tyrosine kinase receptors, which include receptors for M-CSF and SCF; and 3) receptors belonging to the "cytokine receptor superfamily", that are subdivided further on the basis of homology and which includes the haemopoietin receptors. Given their homology, it is almost certain that evolution of some of these receptors, has arisen through duplication of a common ancestral gene. This is most evident in the structure of the haemopoietin receptors. Many are heterodimers, with an α -chain specific for the cytokine cognate to that receptor and a β -chain that is shared among members of the same subfamily. For instance GM-CSF, IL-3 and IL-5 share a common β -chain (reviewed in Miyajima et al 1993). This β -chain is proposed to transduce the signal from the receptor complex to the intracellular compartment and may explain why GM-CSF, IL-3 and IL-5 all show the overlapping ability to stimulate eosinophil proliferation and functions. A similar situation exists for IL-6, IL-11, oncostatin M (OSM) and LIF, which share a common subunit chain (the gp130 subunit) and this is a probable explanation for the appearance of many of the overlapping pleiotropic functions of these cytokines (reviewed in Kishimoto et al 1995).

A multitude of studies has now been reported in which individual (and even combinations of) specific growth factors or cytokines have been deleted in mice. The various phenotypes of these mice, suggest that whilst some facets of their function may overlap, others are clearly unique. As indicated above, M-CSF, G-CSF, IL-3, SCF and IL-6 can all stimulate the formation of granulocytes. However, the gene for each of these has been separately deleted in mice and each shows a very distinct phenotype. Such deletion analysis has suggested functions for these molecules in both constitutive and inducible haemopoiesis, i.e. distinct and non-redundant roles. This suggests, that to say that some factors are redundant is over simplistic. Indeed, there is another alternative to redundancy as an explanation for the plethora of regulatory immuno-haemopoietic cytokines. A complex system such as haemopoiesis must be inherently flexible, in order to be able to respond rapidly to a multitude of sudden stresses to the system. A network of interacting cytokines allows such flexibility and enables rapid, yet controlled responses. From an evolutionary point of view this makes more sense too, as a lack of such flexibility could prove lethal. A network with overlapping functions also provides a fail-safe mechanism, by which one cytokine can substitute for another should that one not function properly.

There are clearly many other logical reasons why a such a complex interactive network makes sense. It can provide an in built amplification mechanism, whereby one cytokine has the ability to induce the expression of other cytokines and even autoregulate its expression (Lotem and Sachs 1991).

An additional level of control exists, with the potential to couple cytokine expression to that of cytokine receptor expression (Walker *et al* 1985). For instance IL-6 induces expression of receptors for M-CSF and IL-3 (Lotem *et al* 1989) and TGF- β downregulates expression of receptors for GM-CSF, G-CSF and IL-3 (Jacobsen *et al* 1991a).

Clearly, inducing or stimulating factors are only part of the cytokine network and the network is in fact likely to be a balance between these positive signals and those exerted by molecules with inhibitory functions, such as TGF- β , TNF- α and MIP-1 α . The role of inhibitory regulators in haemopoiesis was briefly discussed in section 1.6, whilst a discussion of the inhibitory roles of TGF- β and MIP-1 α can be found in sections 2.14.6 and 3.6.5 respectively. The following sections of chapter 2 summarise the roles and importance of some of the growth factors and cytokines (previously indicated in table 1.1), which together form an interacting network of regulatory molecules implicated in the survival, growth, differentiation and function of cells within the immuno-haemopoietic system.

CHAPTER 2: Growth Factors and Cytokines

This chapter will focus on some of the key regulators of the immunohaemopoietic system and discuss some of their known functions. This can not possibly be an exhaustive review, due to the overwhelming literature available. Instead, it is intended to familiarise the reader with some of the most important points concerning the role of specific growth factors and cytokines, in the network of cytokines that is implicated in the regulation of the immuno-haemopoietic system.

2.1 Interleukin-1 (IL-1)

IL-1 is a cytokine responsible for mediating a variety of processes in host defence, inflammation and haemopoiesis. IL-1 is most closely related to tumour necrosis factor (TNF), but is a member of a small cytokine subfamily that includes IL- 1α , IL-1 β and IL-1 receptor antagonist (IL-1Ra). IL-1 was originally described as a product of macrophages, but is now known to be produced by a variety of cell types, including fibroblasts, keratinocytes, T and B lymphocytes, astrocytes and microglial cells. IL-1 α and IL-1 β are two related, but distinct molecules, that are synthesised as 31 kD precursors (proIL-1 α and proIL-1 β). Unlike proIL-1 α , pro-IL-1 β requires cleavage for optimal secretion and activity and this occurs through the action of IL-1 β converting enzyme (ICE) (Thornberry et al 1992, Wilson et al 1994). Whilst both IL- 1α and IL-1 β exist as extracellular proteins, IL-1 α can also exist as a membrane bound form, that is also biologically active (Matsushima et al 1986, Kurt-Jones et al 1985). In most tissues, IL-1 β mRNA predominates over IL-1 α mRNA and in fact IL-1 α is rarely found in the circulation. The many inflammatory functions of IL-1 will not be discussed further and the reader is referred to reviews on this subject, such as that by Dinarello (1994).

IL-1 affects various levels of haemopoiesis and its direct haemopoietic effects were first identified as hemopoietin-1, an activity that modulated the response of very primitive progenitor cells to colony-stimulating factors (Stanley *et al* 1986, Mochizuki *et al* 1987). IL-1 induces the production of GM-CSF, G-CSF, M-CSF and IL-3 and other cytokines, particularly from bone marrow stromal cells (Zucali *et al* 1986). IL-1 has been shown to act synergistically with IL-3, IL-6, G-CSF and GM-CSF, to induce

single and multi-lineage colony formation (Moore *et al* 1987, Bradley *et al* 1989). *In vivo*, IL-1 has been shown to protect stem cells and promote granulopoiesis following cytotoxic drug treatment or irradiation (Fibbe *et al* 1989, Oppenheim *et al* 1989). IL-1 β also appears to stimulate platelet production *in vivo*, as shown in a murine model (Williams and Morrissey 1989). *In vivo* experiments have also suggested that IL-1 can function as an inhibitor of progenitor cell proliferation. IL-1 α reversibly inhibits late-stage erythropoiesis (Johnson *et al* 1989) and prolonged administration of IL-1 β has been shown to inhibit the formation of granulocyte-macrophage (GM) colonies, albeit indirectly through induction of TNF- α , a known inhibitor (Gasparetto *et al* 1989).

2.2 Interleukin-3 (IL-3)

During the search for colony-stimulating activities, several activities were found in spleen conditioned medium. These activities were designated by various names, including erythroid burst (colony)-promoting activity (BPA), mast cell growth factor (MCGF), Thy-1-inducing activity, hemopoietin-2, 20 α hydroxysteroid dehydrogenase (20 α SDH) inducing activity, and interleukin-3. These activities were suggested in fact, to be the result of a single colony-stimulating factor and were subsequently re-named multi-CSF, which is now more commonly known as IL-3. Unusually, IL-3 appears to be potentially involved in regulating the production of all the major myeloid blood cell types, including macrophages, megakaryocytes, mast cells, eosinophils, neutrophils and erythrocytes.

The major source of IL-3 appears to be antigen or mitogenically activated T lymphocytes. However, in addition to activated T cells, NK cells, (Cuturi *et al* 1989), mast cell lines (Wodner-Filipowicz *et al* 1989), megakaryocytic cell lines (Avaraham *et al* 1992) and thymic epithelial cells (Dalloul *et al* 1991) have been shown to express IL-3. Reports of IL-3 expression in cells within the bone marrow microenvironment, such as fibroblasts and endothelial cells have largely been negative (Cluitmans *et al* 1995) although one study, using a sensitive PCR-based detection method, did report the detection of IL-3 expression in irradiated bone marrow (Kittler *et al* 1992). The persistent failure to detect IL-3 in either bone marrow or normal serum has led to the suggestion, that IL-3 does not in fact play a role in constitutive haemopoiesis. Instead,

some researchers have suggested that it functions as a mediator of inflammation, due to its immune cell-derived sources.

Whilst the precise role of IL-3 in vivo remains controversial, the in vitro ability of IL-3 to function as a multilineage colony-stimulating factor is well documented. In vitro, recombinant IL-3 induces the formation of colonies in cultures of bone marrow cells in soft agar. It promotes the formation of mixed erythroid/myeloid, macrophage, neutrophil, eosinophil and basophil colonies (Hapel et al 1985, Rennick et al 1985). IL-3 has also been reported to support the growth of erythroid and megakaryocyte lineages, but the formation of mature cells does not occur in the absence of erythropoietin or thrombopoietin. A number of reports have provided evidence that IL-3, along with GM-CSF, is more specific for early stages of haemopoietic development. For instance, IL-3 is required for the proliferation and differentiation of multipotential haemopoietic progenitors in culture, though IL-3 does not appear to promote the entry of these quiescent progenitor cells into active cell cycle (Suda et al 1985). It has also been reported that multipotent progenitors become less sensitive to IL-3 as they mature into multilineage colonies (Koike et al 1986). Taken together these, and other studies, suggest that in vitro, IL-3 acts primarily on multipotent progenitors in the earlier stages of haemopoietic development.

IL-3 can clearly also function through synergy with other growth factors. For example, IL-3 synergises with M-CSF, to enhance M-CSF receptor levels and generate 'giant' macrophage colonies from HPP-CFCs, a progenitor cell with high proliferative potential (see section 1.4.2) (McNiece *et al* 1984, Bartelmez *et al* 1985). IL-3 has also been reported to act in synergy with IL-6 to enhance IL-3 dependent proliferation of multipotential haemopoietic progenitors (Ikebuchi *et al* 1987) and with IL-4 to enhance mast cell growth (Smith *et al* 1986). Interestingly, further insight into the role of IL-3 synergy with other growth factors has come from the study of particular mouse strains (mostly A/J, AKR and NZB), which appear to be less responsive or 'hyporesponsive' to IL-3 (Kincade *et al* 1979, Morris *et al* 1990). Progenitor cells from these mice strains are defective in their ability to form colonies or proliferate in response to IL-3, and yet they are still able to exhibit a synergistic response to IL-3 in combination with M-CSF, indicating that responses to IL-3 are not totally absent in these strains. (Breen *et al* 1990, Morris *et al* 1990). The genetic basis for this hyporesponsiveness has recently been identified, as being due to a defect in the

gene encoding the α chain for the IL-3 receptor (IL-3R α). Unusually however, this defect is not absolute and some functional receptors do appear to be expressed by haemopoietic bone marrow derived mast cells. (Leslie *et al* 1996). Further study of such hyporesponsive strains may well provide further insight into mechanisms by which IL-3 is able to synergise with other growth factors.

As discussed above, IL-3 is almost undetectable under most circumstances and many in vivo studies have proved confusing and difficult to interpret. Some have however, proved useful and suggest that there is at least some potential role for IL-3 in vivo. For instance, continuous infusion of IL-3 into mice was shown to result in increased numbers of progenitor and myeloid committed precursor cells in the spleen, but not in the bone marrow (Kindler et al 1986). Similar results were obtained by intraperitoneal injection of IL-3, which resulted in large increases in macrophage, eosinophil and neutrophil cell numbers, as well as enhanced peritoneal phagocytic activity (Metcalf et al 1986). It is possible however, that these changes could be attributable to chemotaxis and mobilisation of cells from the circulation in response to an IL-3 induced mediator, suggesting a role for IL-3 in inflammatory processes. Some evidence for a role for IL-3 to function in vivo in constitutive haemopoiesis came from studies using mice that had been pre-treated with purified lactoferrin, (Broxmeyer et al 1987a) which has been shown to suppress release of some haemopoietic growth factors in vitro and to inhibit myelopoiesis in vivo. In this way, the authors were able to suppress the background of normal haemopoiesis and demonstrate that a small infusion of IL-3 resulted in an change in cell cycle status and increased the number of progenitor cells in both the bone marrow and the spleen. In another study, the authors were also able to demonstrate that IL-3 could synergise with GM-CSF and M-CSF in vivo, to increase the number of progenitor cells and the absolute number of progenitor cells (Broxmeyer et al 1987b).

Whilst the effects of IL-3 appear to predominantly affect the early stages of haemopoiesis, effects on mature, fully differentiated cells are also evident. For instance, IL-3 can potentiate mature eosinophil functions, such as degranulation (Fujisawa *et al* 1990) and stimulate histamine release from basophils (Kurimoto *et al* 1989). IL-3 has also been reported to affect the functions of mature monocytic phagocytes, such as cytotoxicity (Cannistra *et al* 1988), as well as stimulating macrophage growth outside the bone marrow (Chen *et al* 1988).

The *in vivo* role of IL-3 may however not be essential, as mice deficient in the βc or β_{IL-3} components of the IL-3 receptor exhibit apparently normal haemopoiesis (Nicola *et al* 1996), as do mice completely lacking IL-3 functions (Nishinakamura *et al* 1996a). The role of IL-3 *in vivo* therefore remains controversial.

2.3 Interleukin-4 (IL-4)

IL-4 was initially characterised as B cell growth factor (BCGF), subsequently renamed B-cell stimulatory factor and is now known as IL-4. The main function of IL-4 appears to be in regulating the switch in immunoglobulin (Ig) synthesis to IgE (and IgG1 in mice), although it clearly has many other pleiotropic functions (reviewed in Paul 1991). Besides regulation of Ig class-switching, IL-4 can also function as a haemopoietic regulator. It has been shown to enhance the IL-3 mediated growth of mast cells (Mosmann *et al* 1986). IL-4 acts as a co-stimulator with other haemopoietic growth factors, such as G-CSF, erythropoietin (Epo) and IL-6, inducing the proliferation of CFU-GM, BFU-E and CFU-GEMM (Rennick *et al* 1987, Peschel *et al* 1987, Sonoda *et al* 1990).

Several inhibitory actions have been described for IL-4, such as macrophage colony formation (Jansen *et al* 1989) and IL-4 has also been shown to induce expression of a reversible inhibitory activity from bone marrow stromal cells (Peschel *et al* 1989). Taken together, these studies suggest that IL-4 has a potential role in the regulation of various stages of haemopoiesis.

2.4 Interleukin-5 (IL-5)

IL-5 is an unusually specific growth factor, directed to cells of the eosinophilic and basophilic lineage. In fact, IL-5 appears to be the most important, if not only cytokine involved in eosinophilia (elevated level of eosinophils). Eosinophils are cytotoxic cells capable of killing larger organisms such as helminths, parasitic protozoa, as well as tumour cells and normally constitute a very small population within the immuno-haemopoietic system. In addition to its many eosinophil specific functions (reviewed by Sanderson 1992), IL-5 also has actions on mature basophils (reviewed by Denburg 1992).

IL-5 was first identified as eosinophil differentiation factor and subsequently re-named IL-5. In comparison to GM-CSF and IL-3, IL-5 is a poor stimulator of colony formation in semi-solid assays, whereas IL-5 is much better at stimulating precursor formation in liquid cultures than either GM-CSF or IL-3. In fact whilst GM-CSF and IL-3 can both generate precursors for several different lineages, including eosinophils (see sections 2.2 and 2.11), only IL-5 is specific for eosinophil progenitor formation (Lopez *et al* 1986, Clutterbuck *et al* 1988). It is speculated that IL-5 is only a late acting factor, as suggested by several studies (Sanderson *et al* 1985). GM-CSF and IL-3 can synergise with IL-5 to induce eosinophils (Lu *et al* 1990, Clutterbuck *et al* 1990).

Whilst *in vitro* studies have been unable to demonstrate conclusively, at what stage of progenitor cell IL-5 acts upon, other studies have clearly demonstrated the fundamental importance of IL-5 in vivo. Administration of anti-IL-5 antibody to parasite or helminth infected mice prevented the associated eosinophilia completely (Coffman et al 1989), whilst another study showed that retrovirally mediated overexpression of IL-5 resulted in long-lasting eosinophilia (Vaux et al 1990). Several studies with IL-5 transgenic mice have also determined the in vivo importance of IL-5 to eosinophil production (Dent et al 1990, Tominaga et al 1991). The study of IL-5 transgenic mice has also supported the notion that IL-5 may act upon B lymphocyte development in the mouse system. Mouse IL-5 had previously been reported to function as a promoter of B cell growth and differentiation (Matsumoto et al 1987). Transgenic mice with IL-5 coupled to the metallothionein promoter (mMT-I), showed eosinophilia, but in addition had elevated levels of serum IgM and IgA and an expansion of Ly-1⁺ and Ly-1dull B220⁺ cells in the spleen (Tominaga et al 1991). Ly-1⁺ cells are a minor population of B cell lineage cells, showing a restricted Ig repertoire. In contrast to mouse IL-5, human IL-5 has not been reported to act on Bcells (Clutterbuck et al 1987).

Deletion of both the IL-3 ligand and βc common receptor subunit in mice, removes all ability to respond to GM-CSF, IL-3 and IL-5. Analysis of such mice revealed that these mice exhibited a reduced number of eosinophils and eosinophilic responses to parasitic infections and this was attributable solely to the absence of IL-5 (Nishinakamura *et al* 1996a).

Many questions concerning the role of IL-5 in haemopoiesis remain to be answered and in particular, whether IL-5 is sufficient on its own to stimulate eosinophil progenitor formation or if it must act in synergy with the earlier acting growth factors, GM-CSF and IL-3.

2.5 Interleukin-6 (IL-6)

IL-6, like many other cytokines, exhibits many pleiotropic functions. It was originally identified as a B-cell differentiation factor (BSF-2), that induced maturation of B cells into antibody producing cells (Hirano *et al* 1985, 1986). As well as effects on B cells, IL-6 also acts on T cells, hepatocytes and neuronal cells (reviewed in Kishimoto 1989, Kishimoto *et al* 1992). IL-6's functions as a potent growth factor for human myeloma cells and as a potential autocrine regulator of myeloma, have also been particularly well studied (reviewed in Klein *et al* 1995).

In addition to its roles in human myeloma, IL-6 has also been well characterised as a regulator of haemopoietic progenitor cells. IL-6 synergises with IL-3, to induce the expansion of haemopoietic progenitor cells (Ikebuchi *et al* 1987, Leary *et al* 1988). IL-6 has also been shown to interact with IL-4 or G-CSF to increase GM colony formation (Rennick *et al* 1989), suggesting that IL-6 is able to interact with a variety of factors to regulate the growth of progenitor cells at different stages of lineage commitment and maturation. A role for IL-6 has also been proposed, in the maturation and production of megakaryocytes (Ishibashi *et al* 1989, Suematsu *et al* 1989) and *in vivo* administration of recombinant IL-6 in primates has been demonstrated to result in an increase in the number of platelets (Asano *et al* 1990).

Interestingly, analysis of IL-6 deficient mice has shown that they exhibit defective leukocyte recruitment, and this appears to be associated with a reduced level of chemokine expression (Romano *et al* 1997). This implies that IL-6 may also be involved in amplifying leukocyte accumulation at sites of inflammation *in vivo*.

2.6 Interleukin-7 (IL-7)

IL-7 is a cytokine that is more restricted in its effects, than most other haemopoietic growth factors. IL-7 was originally described as a stroma-derived pre-B cell growth factor (Hunt *et al* 1987). Subsequently to this, IL-7 has been demonstrated to exert it's proliferative effects on B cell progenitors (Namen *et al* 1988, Suda *et al* 1989, Tushinski *et al* 1991), as well as stimulating T cell progenitors (Conlon et al 1989) and mature T cells (Morrissey *et al* 1989). For a long time therefore, IL-7 was

regarded as being purely a lymphoid-lineage-restricted cytokine. Only recently has a role for IL-7 in myelopoiesis emerged, with the demonstration that IL-7 directly enhances colony-stimulating factor and SCF-induced *in vitro* myeloid colony formation from murine bone marrow progenitor cells (Jacobsen *et al* 1993, 1994a and 1994b). IL-7 has also been shown to induce cytokine secretion from monocytes (Alderson *et al* 1991) and bone marrow macrophages have also been reported to express high numbers of IL-7 receptors (Park *et al* 1990b), suggesting that IL-7 may exert effects on more mature myeloid cells and its effects may therefore be more pleiotropic than originally thought.

2.7 Interleukin-8 (IL-8)

IL-8 although designated an interleukin, is actually a member of the chemokine superfamily of cytokines and will therefore be discussed later in section 3.2.1.

2.8 Interleukin-11 (IL-11)

IL-11 was originally cloned from a primate stromal cell line (Paul *et al* 1990) and appears relatively pleiotropic in its effects, on a wide range of cell types. Only a brief synopsis is detailed here and the reader is referred to the various reviews on IL-11 (Du and Williams 1994, 1997).

IL-11 acts synergistically with other early and late acting growth factors, such as IL-3, IL-4, SCF, flt3 ligand (ftl3L) and GM-CSF (Musashi *et al* 1991, Leary *et al* 1992, Lemoli *et al* 1993, Jacobsen *et al* 1995a), to stimulate various stages and lineages of haemopoiesis. It acts with IL-3, thrombopoietin (TPO) or SCF to synergistically stimulate both murine and human megakaryocytopoiesis and thrombopoiesis (Broudy *et al* 1995). IL-11 can also stimulate various stages of erythropoiesis either on its own, or in synergy with IL-3, SCF or Epo (Quesniaux *et al* 1992). IL-11 has also been shown to affect the differentiation and maturation of myeloid progenitor cells. A protective role for IL-11 has also been well documented. For instance, IL-11 has been shown to provide protection of small intestinal clonogenic stem cells against 5-fluorouracil treatment and irradiation (Potten 1995, 1996). In combination with growth factors such as SCF, IL-13, IL-4 and G-CSF, IL-11 effects on B cell growth have also been reported (Hirayama *et al* 1992, 1994).

2.9 Macrophage Colony-Stimulating Factor (M-CSF)

Colony-stimulating factor-1 (CSF-1) or M-CSF, was originally purified on the basis of its ability to stimulate the growth and differentiation of macrophages (Stanley and Heard 1977). M-CSF is somewhat more complex in its gene structure and regulation of expression at RNA and protein levels, than most other cytokines. Soluble M-CSF is a homodimeric glycoprotein and human M-CSF is encoded by a single gene that is transcribed to give a primary mRNA transcript, that is in turn subsequently alternatively spliced and results in mRNAs encoding different forms of M-CSF. These precursor forms of M-CSF are then proteolytically processed to give various forms of the soluble growth factor (Ladner et al 1987). At least two different forms of precursor protein exist, the larger of these is rapidly processed and does not give rise to cell surface M-CSF (Wong et al 1987, Rettenmier and Roussel 1988). The smaller precursor is processed however (Kawasaki et al 1985), but does not always undergo efficient cleavage and may instead be stably expressed on the cell surface (Rettenmier et al 1987). The membrane expressed precursor is then slowly cleaved to yield the soluble homodimeric growth factor. The cell surface form also appears to be biologically active (Stein et al 1990).

M-CSF stimulates the growth, differentiation and proliferation of cells of the monocyte/macrophage lineage (Ralph and Nakoinz 1987). Recombinant M-CSF has been shown to stimulate differentiation of progenitor cells into monoblasts (Becker et al 1987) and also to induce maturation into promonocytes and monocytes (Rosenfeld et al 1990). The effects of M-CSF on human macrophage colony formation in vitro do however appear somewhat limited, as GM-CSF and other growth factors have been shown to be necessary for optimal colony formation (Caracciolo et al 1987, Zhou et al 1988). A role for M-CSF does appear to exist in vivo however, as demonstrated by studies using mice with a homozygous null mutation for the op mutation. These (op/op) or osteopetrotic mice, suffer congenital osteopetrosis as a result of a deficiency in osteoclasts and macrophages. It is now clear that this deficiency in osteoclast formation, lies with a complete absence of M-CSF (Wiktor-Jedrzejczak et al 1990) and furthermore, that the op mutation lies within the coding region for the M-CSF gene (Yoshida et al 1990). Addition of recombinant M-CSF has been shown to correct the haemopoietic abnormalities (Wiktor-Jedrzejczak et al 1991), which are also progressively corrected as the mice age (Begg and Bertoncello 1993). There is also

evidence that M-CSF has roles in reproduction, as female *op/op* mice show impaired fertility (Pollard *et al* 1987, 1991).

In addition to its roles on progenitor cells, M-CSF also exerts a variety of actions on mature cells of the monocyte/macrophage lineage, including phagocytic activity and microbial killing (Karbassi *et al* 1987) and macrophage-mediated tumour cell cytolysis (Wing *et al* 1982). M-CSF also induces expression of many cytokines from monocytes and macrophages, including G-CSF, GM-CSF, IFN- γ , TNF and IL-1 (Moore *et al* 1980, Warren and Ralph 1986, Motoyoshi *et al* 1989).

2.10 Granulocyte Colony-Stimulating Factor (G-CSF)

G-CSF is a growth factor that regulates the production of neutrophilic granulocytes. These are a critical component to the host defence system and it is estimated that as many as 120 billion granulocytes are produced each day, simply to maintain normal numbers and this number may increase ten-fold during infection (Demetri and Griffin 1991). Whilst G-CSF activity was identified by many researchers, it was not until 1983 that G-CSF was purified and characterised (Nicola et al 1983) and the cDNA for a human G-CSF subsequently cloned (Souza et al 1986). G-CSF is produced, in many cases constitutively, by various cell types. Cells of the monocyte/macrophage lineage are a particularly dominant source of G-CSF, along with cells such as vascular endothelial cells and fibroblasts. G-CSF production can be induced by a variety of stimuli, including lipopolysaccharide (LPS) and many other growth factors (Oster et al 1989, Vellenga et al 1988). Evidence for the role of G-CSF in vivo, has come from analysis of mice carrying a homozygous null mutation for G-CSF. These mice have chronic neutropenia, with only approximately 20 % of normal circulating neutrophils and a reduced number of neutrophilic granulocyte and macrophage precursors in their bone marrow (Lieshcke et al 1994). This study, suggests that some role for G-CSF in the production of granulocytes does exist in vivo, although it is one which can be at least partially fulfilled by another factor(s), since G-CSF^{-/-} mice are viable and do still have some neutrophils. This role may be minimal and G-CSF might function instead, to increase neutrophil production during periods of stress, such as during infection. In normal haemopoiesis, the effects of G-CSF are limited to cells of the neutrophilic lineage. These effects will be discussed

only briefly here and the reader is referred to the many reviews on G-CSF (Nicola 1990, Demetri and Griffin 1991).

In vitro, G-CSF stimulates proliferation and differentiation of neutrophil colony-forming cells, as well as affecting several functions of mature neutrophils (Metcalf and Nicola 1983). G-CSF, acts on a relatively mature progenitor cell population, that is mostly committed to neutrophilic differentiation (Ema *et al* 1990a). Whilst the effects of G-CSF alone are relatively limited, in combination with other growth factors, G-CSF can stimulate the survival and proliferation of multi-lineage progenitor cells *in vitro* (Metcalf and Nicola 1983, McNiece *et al* 1989). In fact, G-CSF was originally described as a multi-lineage-stimulating or pluripotent growth factor. However, when accessory cells were depleted from culture this pluripotent activity was shown to disappear (Strife *et al* 1987). G-CSF has been shown to exert synergistic or enhancing actions with IL-3 (Ikebuchi *et al* 1988), IL-4 (Rennick *et al* 1987), Peschel *et al* 1987), IL-1 α (Moore and Warren 1987), and IL-6 (Rennick *et al* 1989).

In addition to stimulating proliferation of progenitor cells, G-CSF also enhances the functions of mature neutrophil effector cells. These range from 'priming' of cells for receptor-mediated responses (Yuo *et al* 1989), production of cytokines such as interferon- α (Shirafuji *et al* 1990), and enhancement of cytotoxic functions (Lopez *et al* 1983).

G-CSF is now widely used clinically to increase neutrophil numbers and to enhance neutrophil responses, during or following chemotherapy or radiotherapy, or following bone marrow transplantation, thus providing a protective role against opportunistic infections.

2.11 Granulocyte-Macrophage Colony-Stimulating Factor (GM-CSF)

GM-CSF, along with G-CSF, M-CSF and IL-3, is one of a number of colony stimulating factors identified to date. It derives its name from the ability to stimulate formation of macroscopic colonies containing neutrophilic granulocytes (neutrophils), eosinophilic granulocytes (eosinophils), and macrophages, or a mixture of these cell types. GM-CSF stimulates the proliferation and maturation of myeloid progenitor cells, as well as having many actions on mature neutrophils, eosinophils and macrophages.

Many different cell types have been reported to produce GM-CSF. These include both immune cells such as T lymphocytes, macrophages and mast cells, as well as adherent cells within the bone marrow microenvironment such as macrophages, fibroblasts and endothelial cells. B lymphocytes and natural killer (NK) cells have also been reported to synthesise GM-CSF (Pistoia *et al* 1987, 1989). The precise nature of this expression is sometimes contradictory, but in most cases GM-CSF does not appear to be constitutively expressed and is instead induced by a variety of stimuli. LPS, IL-1 and TNF- α are amongst the most prevalent and dramatic inducers of GM-CSF expression (Broudy *et al* 1986, Chan *et al* 1986, Sieff *et al* 1988), particularly within the bone marrow microenvironment. In addition to growth factors, other stimuli such as FCS, phagocytosis (Thorens *et al* 1987), and Fc_eRI triggerring by IgE plus antigen in mast cells (Wodnar-Filipowicz *et al* 1989) have also been demonstrated to induce GM-CSF expression.

GM-CSF was first purified from mouse lung-conditioned medium and found to stimulate proliferation and differentiation of myelomonocytic progenitors (Burgess *et al* 1977). Following identification of the complementary cDNA and generation of recombinant protein, the actions of GM-CSF on hemopoietic cells have been studied in great detail (reviewed in Gough and Nicola 1990, Gasson 1991, Baldwin 1992). *In vitro*, GM-CSF is now known to be a mutilineage stimulator for CFU-granulocyte (CFU-G), CFU-GM, CFU-macrophage (CFU-M), and CFU-eosinophil (CFU-Eo) (Tomonaga *et al* 1986). GM-CSF may also support the growth of erythroid and megakaryocytic progenitors *in vitro* (Sieff *et al* 1985, Emerson *et al* 1989).

The effects of GM-CSF on differentiation and maturation are however, less clear. Several studies have suggested that GM-CSF can function to induce differentiation, for instance towards monocytic and eosinophilic lineages in the human myeloid leukemia cell line HL60 (Tomonaga *et al* 1986, Ellias *et al* 1988), and in co-operation with Epo to induce erythroid differentiation in the erythroleukemic cell line K562 (Tawhid *et al* 1989).

In addition to its effects on haemopoietic progenitor cells, GM-CSF also exerts a wide variety of actions on mature, fully differentiated neutrophils, eosinophils and macrophages. These actions can be either 1) direct effects, such as increasing cell survival (Begley *et al* 1986); or 2) indirect or so-called 'priming' effects that occur in conjunction with a secondary stimulus, such as enhancement of neutrophil oxidative, phagocytic and cytolytic functions (Baldwin *et al* 1989), and priming of basophils for enhanced histamine release (Hirai *et al* 1988). As well as granulocyte-directed effects, GM-CSF also has a wide variety of macrophage directed functions, such as increased release of inflammatory mediators and growth factors (Oster *et al* 1992, Takamatsu *et al* 1996). These various effects on mature cells, are too numerous to discuss in full here, and the reader is refered to the excellent review by Baldwin (1992) for a full overview.

Many groups have attempted to examine the role of GM-CSF in vivo. One of the most interesting of these was that of Lang et al (1987), who overexpressed the GM-CSF gene in mice, under a retroviral promoter. These mice had a very unexpected compound phenotype, in that they were born with accumulations of macrophages in their eyes and striated muscle. In fact, the eyes were opaque, with extensive retinal damage that resulted in blindness. Many GM-CSF transgenic mice subsequently died after 2-4 months, with a chronic wasting syndrome. The ocular defects were attributed to autostimulation of macrophages, that are required for normal developmentally programmed tissue remodelling in the mouse eye (Cuthbertson et al 1989, Lang and Bishop 1993). Stanley et al (1994) later reported that GM-CSF was in fact dispensable for haemopoiesis, since GM-CSF deficient mice exhibited apparently normal haemopoiesis. These mice did however, develop an abnormal lung pathology that was characterised by infiltration of lymphocytes and large intraalveolar macrophages and which was reminiscent of the human disease, pulmonary alveolar proteinosis (PAP). Many GM-CSF deficient mice also developed inflammatory lung infections. A similar phenotype has been reported in mice deficient for the common component shared by the GM-CSF, IL-3 and IL-5 receptors, the ßc component (Nishinakamura et al 1995). This phenotype is reversed following bone marrow transplantation, suggesting a haemopoietic origin (Nishinakamura et al 1996b). In fact, human PAP was recently reported to be associated with a defect in the GM-CSF/IL-3/IL-5 common chain (Bc) (Dirksen et al 1997). Taken together, these reports suggest that GM-CSF is probably dispensible (or even redundant) for normal constitutive haemopoiesis, but that GM-CSF may be important for pro-inflammatory functions in vivo.

2.12 Stem Cell Factor (SCF)

SCF is a pleiotropic cytokine (reviewed in Broudy 1997), that functions primarily in the early stages of haemopoiesis. It was previously known as kit ligand, mast cell growth factor, or steel factor. The gene for SCF is encoded within the Steel or *Sl* locus (Zsebo *et al* 1990), whilst the gene for the SCF receptor, *c-kit*, is encoded by the white spotted or *W* locus (Huang *et al* 1990). A host of naturally occuring mutations at the *W* or *Sl* loci have been identified, and these have led to the identification and further characterisation of both SCF and its receptor. The absence of either SCF or Kit receptor results in death in utero, or perinatally as a result of severe anaemia (Besmer *et al* 1993). This, together with evidence from embryonic expression patterns for SCF (Orr-Urtrger *et al* 1990, Matsui *et al* 1990), suggests a developmental role for SCF.

SCF exists in two forms generated by alternative splicing of mRNA (Flanagan et al 1991); as a soluble form and also as a transmembrane form, which are both biologically active (Anderson et al 1990). Mice with deleted transmembrane and cytoplasmic domains (which results in only the soluble form of SCF) are viable but show several phenotypic abnormalities (Flanagan et al 1991, Brannan et al 1991), suggesting the transmembrane form does play a role in vivo. Recently, it was reported that overexpression of the membrane associated form of human SCF in mice (which in murine cells acts as an antagonist, through interference with the interaction between murine SCF and its receptor), resulted in impaired melanocyte development. Moreover, compound heterozygous mice that carried both a mutated W or Sl allele and the transgene for hSCF (hSCF²²⁰) (i.e hSCF²²⁰ / +; W /+ or hSCF²²⁰ / +; Sl / +), showed a significantly reduced number of dermal mast cells, and defective thymocyte development (Kapur et al 1997). This study suggests that transmembrane SCF is required for the migration or survival/proliferation of melanocytes and dermal mast cells, and that c-kit may be involved in thymocyte differentiation. In fact, the ratio of soluble to transmembrane SCF varies considerably between different tissues (Huang et al 1992). In the bone marrow, stromal fibroblasts predominantly express the soluble form of SCF (Heinrich et al 1993a, Linenberger et al 1995). In addition to expression by fibroblasts, SCF is constitutively expressed by endothelial cells, another important cell type in the stromal microenvironment. Keratinocytes and gut epithelial cells have also been reported to express SCF (Longley et al 1993, Klimpel et al 1995) and SCF

mRNA has also been detected in enriched populations of stem cells and progenitor cells (Ratacjczak *et al* 1995). Whilst inflammatory stimuli such as IL-1 and TNF are strong inducers of many growth factors from the bone marrow (Broudy *et al* 1986), they are much less potent at stimulating the expression of SCF (Heinnich *et al* 1993a, Linenberger *et al* 1995).

SCF exhibits a wide variety of haemopoietic functions. It has been reported to accelerate the entry of haemopoietic stem cells into cell cycle (Leary et al 1992) and has been suggested to maintain the survival of long-term repopulating cells in vivo (Li and Johnson 1994). SCF, in conjunction with IL-3, also enhances the production of CFU-S cells (see section 1.3) in vitro (de Vries et al 1991). Addition of a blocking antibody to Kit receptor was shown to reduce the survival of CFU-S cells (Kodama et al 1992) but still allowed the survival of LTC-IC (see section 1.4.1), suggesting that SCF is dispensible for LTC-IC survival. Several studies have reported that SCF can act upon very primitive pre-CFU-C type cells, that give rise to colony forming cells. In addition, SCF in conjunction with IL-3 and other haemopoietic growth factors enhanced the number of BFU-E, CFU-GM and CFU-GEMM in vitro (Bernstein et al 1991, Migliaccio et al 1992). In fact, SCF now forms the basis for an increasing number of ex vivo expansion protocols (Brandt et al 1992), in conjunction with a number of different cytokines. In such protocols, enriched populations of primitive precursor cells are removed from the donor and the number of resulting progenitor cells is expanded in vitro with a cocktail of growth factors, before re-infusion. This can result in as much as a 200-fold expansion of BFU-E, CFU-GM, and CFU-GEMM from enriched haemopoietic stem cells (Brugger et al 1993). It is also suggested that SCF can promote haemopoietic progenitor cell survival, in the absence of cell division (Keller et al 1995).

In vitro, SCF has been reported to have many synergistic actions with other growth factors to support direct colony formation (McNiece *et al* 1991, Metcalf and Nicola 1991). SCF can synergise with factors, such as Tpo and IL-3, to promote the growth (but not differentiation) of megakaryocyte progenitors and cell lines (Broudy *et al* 1995).

SCF appears to be particularly important for mast cell production. The tissues of mice with mutations in the *Sl* locus (*Sl/Sl*^d) or W locus (*W/W*^v) contain less than 1 % of the normal numbers of mast cells (Kitamura *et al* 1978, 1979), although mast

cell progenitors have been identified in the bone marrow of these mice. The most primitive mast cell progenitor, the pro-mastocyte, proliferates and differentiates in response to SCF plus IL-3 (Rodewald *et al* 1996) and mast cells can be cultured in vitro from a variety of sources, using SCF in conjunction with IL-3 (Durand *et al* 1994).

Evidence also exists for a role for SCF in B and T lymphocyte development. SCF can synergise with IL-7 to stimulate proliferation of pre-B cells (Billips *et al* 1992), but it does not enhance the proliferation of the more primitive pro-B cells. (Faust et al 1993). Actions of SCF on NK and dendritic cell progenitors have also been documented (Shibuya *et al* 1995, Szabolcs *et al* 1995).

In addition to its many effects on haemopoietic stem and progenitor cells, SCF also exerts many functions on mature cells of the mast cell lineage and to a lesser extent, basophil lineage. For instance, SCF enhances secretion of mediators from mast cells, as well as their chemotaxis and adhesion (Columbo *et al* 1992, Dastych *et al* 1994) and also induces IgE-dependent histamine release from basophils (Columbo *et al* 1992).

SCF therefore, is a haemopoietic growth factor that acts primarily during the earlier stages of haemopoiesis, interacts with a variety of growth factors to regulate stem / progenitor cell proliferation and which is particularly important to cells of the mast cell lineage.

2.13 Leukemia Inhibitory Factor (LIF)

LIF is an unusual factor, that exerts its actions on haemopoietic and mature cells. LIF was characterised by several groups under various different names; such as differentiation inducing factor (DIF or D-factor), differentiation inhibitory activity (DIA), human interleukin for DA cells (HILDA) and cholinergic neuronal differentiation factor (CNDF), but these are now more commonly known as LIF (reviewed in Metcalf 1991). LIF shows sequence homology with OSM, IL-6, IL-11 and CNTF, and like LIF these all signal through the common gp130 subunit. LIF receptors are found on cells of the monocyte/macrophage lineage (Hilton *et al* 1991) and the number of LIF receptors was found to increase with maturation. LIF receptors are also found on megakaryocytes and on a subclass of T lymphocytes (Hilton *et al*

1991), as well as on osteoblasts (but not osteoclasts), fibroblasts, endothelial cells, adipocytes and embryonic stem cells (reviewed in Metcalf 1991).

LIF expression is normally almost undetectable. No expression can be detected in normal serum by bioassay, nor in tissues by northern blot analysis. One consistent exception to this, is the expression of LIF by metrial gland cells developing in the uterine wall at the site of blastocyst implantation (Croy *et al* 1991). LIF has however, been reported to be expressed by a variety of cell types following induction by a variety of stimulae, such as IL-1, TNF- α , and TGF- β (Wetzler *et al* 1990). Fibroblasts (Tomida *et al* 1984), T lymphocytes (Gearing *et al* 1987), monocytes and macrophages (Anegon *et al* 1990), stromal cells (Wetzler *et al* 1990) and astrocytes (Wesselingh *et al* 1990) have all been reported to express LIF.

The reported effects of LIF on haemopoietic stem cells in vitro, are relatively limited. LIF was originally reported not to have any effect on the survival or proliferation of adult haemopoietic cells (Metcalf et al 1988). It has however been reported to stimulate the proliferation of murine myeloid DA-1 cells (Moreau et al 1988) and also to promote the IL-3 induced proliferation of megakaryocytes and BFU-E cells (Leary et al 1990). The origin of the name, leukemia inhibitory factor, lies with the ability of LIF to suppress the self-renewal of clonogenic murine M1 leukemic cells (Metcalf et al 1988, Metcalf 1989). This suppression occurred in conjunction with the simultaneous differentiation of M1 cells by LIF. These cells differentiated into cells with macrophage morphology, expressed Fc receptors and exhibited other macrophage-like characteristics such as phagocytosis and an increased dependency upon M-CSF for their survival. (Metcalf 1989, Metcalf et al 1988). In addition to M1 cells, LIF can also suppress colony formation from HL-60 or U937 cells, but with negligible changes in differentiation (Maekawa et al 1989). More recent studies have however, suggested that LIF can in fact effect the survival of haemopoietic progenitor cells (Escary et al 1993). LIF deficient mice were generated, which showed dramatically decreased numbers of stem cells in the spleen and bone marrow. In the spleen, pluripotent stem cells (CFU-S), committed erythroid progenitors (BFU-E), and GM progenitors (GM-CFC) were all reduced by 70-90 %. In the bone marrow CFU-S numbers were decreased by 60 %, whilst the BFU-E and GM-CFC were less affected. This decrease in the CFU-S pool could be restored by addition of exogenous LIF. Primitive stem cells from LIF deficient mice were able to reconstitute the

haemopoietic systems of lethally irradiated normal mice, suggesting that LIF was not affecting the differentiation potential of stem cells and was instead required to maintain stem cell numbers. Female LIF deficient mice were also infertile, due to a defect in embryo implantation. It has since been reported, that LIF is able to directly synergise with growth factors to promote the proliferation of purified progenitor cells (Keller *et al* 1996). These reports support previous *in vivo* studies on LIF and suggest that the roles of LIF *in vivo*, may be more extensive than originally envisaged (Metcalf *et al* 1990).

The most striking function of LIF however, is perhaps its ability to prevent differentiation commitment of murine embryonic stem (ES) cells. ES cells are normally totipotent and will give rise to cells of all murine tissues when injected into a blastocyst, but in culture ES cells undergo differentiation and lose their totipotency. LIF was believed to be unique in its ability to block this predisposed differentiation, yet still maintain the totipotent abilities of ES cells (Williams *et al* 1988), but it is now evident that IL-6, IL-11, CNTF and OSM can all prevent ES cell differentiation (Conover *et al* 1993, Nichols *et al* 1994). LIF has proved of immense use in the development of ES cell based technologies, such as those employed in the generation of transgenic and knockout mice.

2.14 The Transforming Growth Factor Beta (TGF-β) Superfamily

The TGF- β superfamily consists of over 20 factors involved in the growth, differentiation and morphogenesis of cells from almost all lineages. This superfamily can be divided into several different sub-families; the TGF- β family, the activins and inhibins, Mullerian inhibitory substance and the DVR group that includes the bone morphogenetic proteins (reviewed in Massague 1990). The various members of this superfamily are structurally related to the prototypic TGF- β 1 molecule, sharing a common precursor structure and conservation of seven or nine cysteine residues in the mature sequence. There is considerable variation at the level of sequence identity with members within a subfamily showing considerable sequence identity e.g. TGF- β 1, β 2 and β 3 share a 64-82 % homology in their amino acid sequence, whilst the homology between members in different subfamilies is more disparate. As a full discussion of all

the members of the TGF- β superfamily is beyond the scope of this work, only the TGF- β subfamily will be discussed hereafter, with a particular emphasis on TGF- β 1.

2.14.1 The TGF- β family

This subfamily consists of at least five members at present. Three distinct mammalian forms of TGF- β (TGF- β 1, β 2 and β 3) have been identified from various species. Two other non-mammalian forms have also been reported, TGF- β 4 in chickens (Jakowlew *et al* 1998) and TGF- β 5 in Xenopus Laevis (Kondaiah *et al* 1990). The TGF- β s are amongst the most pleiotropic of cytokines, and their functions include proliferative and inhibitory effects upon cell growth and differentiation, wound repair and fibrosis, and inflammation. The most abundant form of TGF- β in mammals is TGF- β 1.

2.14.2 Structure of TGF- β : TGF- β exists in a latent form

TGF- β 1 is considered to be the prototypic form of TGF- β . The mature form of TGF- β 1 is a 25 kD disulphide linked polypeptide, composed of two 12.5 kD homodimers. TGF- β 1 is expressed as a precursor, 412 amino acids (412aa) in length (Derynck *et al* 1985), which then undergoes intracellular proteolytic cleavage. The precursor is composed of an N-terminal signal sequence, a pro-region, and a C-terminal region that contains the active domain. TGF- β is synthesised in the form of a latent complex which is inactive and unable to bind to TGF- β receptors (Pircher *et al* 1986). This latent TGF- β complex can exist in two forms, either as a small latent TGF- β complex or as large latent TGF- β complex (Wakefield *et al* 1988, and reviewed in Harpel *et al* 1992).

The active form of TGF- β 1 consists of a disulphide-bonded dimer of the Cterminal 112aa of the TGF- β 1 precursor polypeptide. In the small latent TGF- β 1 complex, the active TGF- β 1 dimer complex is non-covalently associated with a dimer of the amino-terminal remnant from the proteolytically cleaved precursor (the proregion) (Gentry *et al* 1988, Wakefield *et al* 1989). This remnant is known as the latency associated peptide (LAP) and purified LAP can reassociate with mature TGF- β and inactivate it (Gentry *et al* 1987). *In vitro*, small latent TGF- β complex has been reported to be expressed in the culture medium of several cell types, including bone cultures (Jennings and Mohan 1990, Bonewald et al 1991) and an erythroleukaemic cell line (Lioubin et al 1991).

In addition to the small latent complex, latent TGF- β also exists in the form of large latent TGF- β complex. Platelets have been shown to express this (Wakefield *et al* 1988), it has also been found in cell conditioned medium from several cell lines (Olofsson *et al* 1992, Lioubin *et al* 1991) and in cultures of smooth muscle and endothelial cells (Flaumenhaft *et al* 1993). This large latent TGF- β complex has been purified from platelets (Miyazono *et al* 1988) and is a 235 kD complex, composed of the small latent TGF- β complex (as described above) and an additional 125-160 kD protein called the latent TGF- β binding protein (LTBP). This LTBP is linked to the small latent TGF- β complex via disulphide bonds (Kanzaki *et al* 1990). The precise function of LTBP is unclear, although roles in assisting assembly and secretion of the large latent TGF- β complex (Miyazono *et al* 1991), activation of the latent TGF- β complex (Flaumenhaft *et al* 1993), assisting in the association of TGF- β to the ECM (Taipale *et al* 1994), or in stability (Brown *et al* 1990) have all been proposed.

The complex system that appears to have evolved to prevent TGF- β activation, strongly suggests that inappropriate expression of TGF- β is highly undesirable. However, the in vivo mechanisms by which the latent forms of TGF-B are converted to the mature, active form have yet to be fully elucidated. In vitro, latent TGF-B can be activated by a variety of chemical and physical treatments, including extreme changes of pH, heating, urea, or by the use of detergents (Lawrence et al 1985, Brown et al 1990). Many reports have also demonstrated that various enzymatic treatments will also activate latent TGF-B (Lyons et al 1988). Indeed proteolysis of latent TGF-B remains the most likely mechanism for activation and several mechanistic models have been proposed (reviewed in Harpel et al 1992). One of these proposed mechanisms suggests that the activation of TGF- β may involve the activation of plasminogen to plasmin. Apolipoprotein (a) inhibits the activation of plasminogen to plasmin and this was shown to result in the suppression of TGF-B activation (Grainger et al 1993). Further evidence for this came from the observation that the activation of TGF- β was inhibited in mice overexpressing the human apolipoprotein (a) gene (Grainger et al 1994).

One interesting twist to the functioning of TGF- β is the observation that almost all mature active TGF- β found in plasma, is associated with α_2 -macroglobulin (α_2 M) (O'Connor-McCourt and Wakefield 1987). α_2 M is a highly abundant serum glycoprotein that is secreted by the liver and other cells, such as macrophages and which interacts with other cytokines (reviewed in James 1990). α_2 M binds only to mature TGF- β and its function remains uncertain. Several mechanisms have been proposed, for example α_2 M could simply function as a mechanism to remove excess TGF- β from the circulation (Lamarre *et al* 1991), alternatively it may actively scavenge active TGF- β .

2.14.3 Expression of TGF-β

TGF- β has been reported to be expressed by a very wide variety of cells in culture. The expression of the different mammalian isoforms of TGF- β varies according to the cell type. The widespread expression of TGF- β during many stages of embryonic development suggests an important role for TGF- β during development and morphogenesis (Lehnert and Akhurst 1988, Wilcox and Derynck 1988). Expression of TGF- β is also widespread in the adult system, with expression of TGF- β 1, β 2 or β 3 mRNA detectable in most tissues and organs (Thompson *et al* 1989).

The predominant source of TGF- β is platelets (Assoian *et al* 1983), but a number of other mature haemopoietic cell types have also been shown to synthesise TGF- β , including activated macrophages (Assoian *et al* 1987) and B and T lymphocytes (Kehrl *et al* 1986a, 1986b).

2.14.4 The role of TGF- β as an immunoregulator

As indicated above, the expression of TGF- β appears widespread across the course of embryonic development. The production of TGF- β by mature immune cells such as macrophages suggests however, that TGF- β may also act as an immunological mediator. In fact, TGF- β is a very potent chemotactic agent for neutrophils (Brandes *et al* 1991, Reibman *et al* 1991) as well as for monocytes and macrophages (Wahl *et al* 1987), fibroblasts (Postelwaite *et al* 1987) and mast cells (Gruber *et al* 1994). The immunological functions of TGF- β appear to be mostly anti-inflammatory. TGF- β inhibits the activation of T cells and macrophages (Wahl *et al* 1988, Tsunawaki *et al*

1988) and inhibits the production of cytokines (Musso *et al* 1990), as well as inhibiting the production and secretion of IgM and IgG antibodies from B cells (Kehrl *et al* 1986a, 1991).

TGF- β has been reported to provide a protective effect in animal models of certain autoimmune diseases. Administration of TGF-B1 was shown to protect against collagen induced arthritis and experimental autoimmune encephalomyelitis (models for reactive arthritis and multiple sclerosis respectively) (Kuruvilla et al 1991, Racke et al 1991, Johns et al 1991). TGF-B has also been implicated in the impaired immune response associated with autoimmune deficiency syndrome (AIDS), where elevated TGF- β levels were observed in cultures of peripheral blood mononuclear cells derived from HIV-infected donors (Kekow et al 1990). In addition, TGF-B has also been suggested suppress HIV expression and replication to in infected monocyte/macrophages (Poli et al 1991), suggesting that TGF-B may be involved both in modulating viral infection and in facets of the immune dysfunction associated with AIDS.

The most compelling evidence for a direct role for TGF- β in vivo has come from studies in which mice were generated with a homozygous deletion of the TGF-B1 allele (TGF-\beta1^{-/-}) (Shull et al 1992, Kulkarni et al 1993). In these initial studies, approximately 50 % of TGF- β 1^{-/-} embryos experienced prenatal lethality, but the remainder survived and were born normally with no apparent developmental abnormalities. Subsequent reports have examined the reasons for the observed prenatal lethality and have suggested that TGFB1 null embryos do in fact show some defects in haemopoiesis, as well as in embryonic vasculogenesis (Dickson et al 1995). The observed lethality may however, be dependent upon the genetic background (Dickson et al 1995). After approximately 2 weeks, perinatally surviving mice begin to develop a progressive wasting syndrome that results in death with 3-4 weeks of age. Histological analysis revealed that this wasting syndrome was accompanied by a multifocal, mixed inflammatory cell infiltrate and tissue necrosis. The infiltrate was observed in many organs including heart, lungs and liver. Comparative analysis of wild type and homozygous TGF- β 1^{-/-} mice revealed that several pro-inflammatory cytokines were upregulated, including TNF- α , IFN- γ and MIP-1 α (Shull et al 1992). Furthermore, elevated levels of MHC class I and class II expression have also been observed in TGF- $\beta 1^{-/-}$ mice (Geiser *et al* 1993, Nakabayashi *et al* 1997). Recently, it was shown that treatment of TGF- $\beta 1^{-/-}$ mice with the immunosuppressive agent rapamycin led to ablation of the wasting syndrome, gain of weight and an almost complete inhibition of inflammation (Borkowski *et al* 1996), supporting the notion that some of the defects observed in the compound phenotype of TGF- $\beta^{-/-}$ mice are attributed to an inflammatory role for TGF- β *in vivo*.

Although some mice do die in utero, the survival of the remainder initially suggested that TGF-B1 was at least partially dispensible for normal development, or was compensated for, by another isoform. However it is now clear that this postnatal survival is in fact due to maternal transfer of TGF-B1 (Letterio et al 1994). TGF-B1 can be detected in homozygous null mice derived from a heterozygous mother, but none can be detected in null mice derived from a homozygous null mother. Moreover, homozygous null pups born to homozygous null females (that were kept alive by administration of dexamethasone) died perinatally due to severe cardiac abnormalities, suggesting a role for TGF- β 1 in cardiac development. Furthermore, the authors demonstrated that TGF- β 1 was able to cross the placenta and was taken up by homozygous null foetal cells. Additional evidence for an in vivo role for TGF-B1 has arisen from the observation that TGF- $\beta 1^{-/-}$ mice have a complete absence of epidermal Langerhans cells and suggests that TGF-B1 may also be involved in Langerhans cell generation (Borkowski et al 1996, 1997). Additional evidence that TGF-B may be involved in the development of Langerhans cell, is suggested by recent in vitro studies in which TGF-B1 promoted the development of dendritic cells from CD34+ progenitors, by blocking apoptosis (Strobl et al 1996, Riedl et al 1997).

2.14.5 TGF- β as a regulator of cell proliferation

Perhaps more study has been directed towards the role of TGF- β in the control of cell proliferation, than any other of its many functions. TGF- β can act either to inhibit or to stimulate cell proliferation, though it appears to mostly function as an inhibitor. In fact TGF- β is one of the most potent growth inhibitors identified to date and shows activity on a wide variety of both normal and transformed cell types. These include epithelial cells, endothelial cells, fibroblasts, keratinocytes, T and B lymphocytes, neuronal cells and haemopoietic progenitor cells (Tucker *et al* 1984, Moses *et al* 1985, Roberts *et al* 1985, Shipley *et al* 1986, Kehrl *et al* 1986a, Kehrl *et al* 1986b, Ohta *et al* 1987). The degree of inhibition is highly variable between cell types, ranging from induction of apoptosis (Bursch *et al* 1993, Alam *et al* 1994a), to either almost complete growth arrest in lung epithelial cells (Tucker *et al* 1984) or a lengthening of the G1 phase of the cell cycle (reviewed in Massague *et al* 1992, or in Alexandrow and Moses 1995).

2.14.6 The role of TGF- β in haemopoiesis

TGF- β functions as a bidirectional regulator of haemopoiesis, inhibiting proliferation of primitive multipotential progenitors (Keller *et al* 1988, 1990) and yet also synergising with specific growth factors to enhance proliferation of more committed progenitors (Keller *et al* 1991). Autocrine production of TGF- β has been demonstrated to inhibit the growth factor-induced proliferation of haemopoietic progenitors (Hatzfeld *et al* 1991, Ploemacher *et al* 1993). TGF- β can induce the apoptosis of myeloid leukaemic cells, which can be reversed by certain haemopoietic growth factors (Lotem and Sachs 1992, Taetle *et al* 1993). TGF- β has also been reported to induce apoptosis of primitive murine haemopoietic progenitor cells (Jacobsen *et al* 1995b). This effect appears to be at least partially reversible, since a combination of IL-1, IL-6 and IL-11 reduced the negative effects of TGF- β and supported previous studies, demonstrating that TGF- β -induced inhibition can be overcome by a combination of growth factors (Jacobsen *et al* 1994c).

As mentioned above, TGF- β can also function as a stimulator of more committed progenitor cells. Keller *et al* (1991) showed that TGF- β stimulated GM-CSF-induced granulopoiesis by enhancing the proliferation and differentiation of progenitor cells. In general the effects of the three different mammalian isoforms of TGF- β on haemopoietic progenitors tend to be fairly similar. One reported exception is the study by Jacobsen *et al* (1991b) that showed TGF- β 1 and - β 2 were both able to bidirectionally modulate the proliferation of human progenitor cells, whilst TGF- β 3 was unable to stimulate the proliferation of granulopoiesis (but it was a more potent inhibitor of human haemopoiesis than TGF- β 1 or - β 2).

It is now evident that the control of stem cell proliferation is regulated through a balance of both positive and negative regulators. Some regulators such as TGF- β

and MIP-1 α appear to be able to function as both inhibitors and stimulators, depending upon the state of maturity and the growth factors present. Thus, adding yet another twist to understanding interactions occuring within the complex network of regulatory cytokines.

2.15 The Role of Growth Factors in the Regulatory Control of Haemopoiesis

Clearly it is impossible to discuss all the haemopoietic growth factors outlined in table 1.8.1, in detail. In addition to those already discussed, many others have more lineage-restricted effects such as erythropoietin on erythroid cells (reviewed in Krantz 1991), whilst the actions of other recently identified factors such as ftl3 ligand (Lyman et al 1993) and thrombopoietin (reviewed in Kaushansky 1995) are only just beginning to be understood. However, as outlined in the preceding sections of this chapter, growth factors and cytokines exert a wide variety of functions on cells of various stages of differentiation, from uncommitted progenitors through to mature terminally differentiated cell types. Whilst many of these have been strongly implicated in the control of haemopoiesis, their precise role remains highly controversial. At one extreme it has been proposed that these haemopoietic growth factors (HGFs) may only function as survival factors. At the other extreme, these HGFs have been suggested to actively instruct the haemopoietic differentiation program. One can summarise these mechanisms as being either stochastic or instructive, though there are clearly many variations on these two models. It is beyond the scope of this work to provide a detailed explanation of the various models for haemopoietic differentiation. Therefore, the following sections are merely meant to give a very brief summary of the potential role of HGFs in haemopoietic differentiation.

2.15.1 The stochastic model of differentiation

In this model, proposed in its original form by Till *et al* (1964), stem cells undergo self-renewal and differentiate in a random or stochastic manner. Ogawa and colleagues have proposed models involving either random restriction of a stem cell to a monopotential lineage, or through random lineage restriction of oligopotential progenitors (reviewed by Ogawa 1993). The role of HGFs within this stochastic model is proposed to be restricted to the survival and proliferation of stem cells and their

progenitors. Therefore in this model, the process of differentiation must occur through the action of intrinsic factors. Indeed, studies have shown that cytokines can prevent cells from apoptotic death and therefore act as "happiness factors" (Williams et al 1990). As also discussed previously in chapter 2, other studies have clearly implicated HGFs in the control of the cycling status of stem cells and their progenitors. One of the strongest pieces of evidence in support of a stochastic mechanism involving intrinsic control factors is that of Fairbairn et al (1993), using the IL-3 dependent murine FDCPmix cell line. These cells have the capacity to undergo differentiation along multiple myeloid lineages, but in the absence of IL-3 these cells undergo apoptosis. However, introduction of the gene for the anti-apoptotic factor bcl-2 into these cells, resulted in a significant delay in the time taken for the onset of apoptosis to occur. Moreover, this was accompanied by a spontaneous differentiation of cells along a variety of haemopoietic lineages. This suggests that in FDCPmix cells, IL-3 may act to maintain cell viability and to block differentiation. Recently it was reported that monocytes may have an intrinsic program of differentiation toward macrophages, that can occur in the absence of M-CSF (Lagasse and Weissman 1997). In this study op/op mice lacking functional M-CSF and exhibiting depressed macrophage levels, were crossed with mice engineered to overexpress Bcl-2 in the myeloid lineage. Suppression of monocyte apoptosis in serum-free and cytokine-free medium in vitro was sufficient to allow complete macrophage differentiation. In addition, this Bcl-2 overexpression also rescued diverse macrophage populations in vivo and resulted in a less severe form of osteopetrosis in these mice. This suggests that M-CSF acts to augment monocyte survival and allows them to respond to an intrinsic differentiation program.

2.15.2 The instructive or deterministic model of differentiation

In contrast to the above stochastic model, other researchers subscribe to an instructive or deterministic model (reviewed by Morrison *et al* 1997). In this model it is proposed that HGFs directly influence the differentiation program. Evidence in support of this model was eloquently provided by studies on the G-CSF receptor (G-CSF R). Dong *et al* (1994) reported the existence of a nonsense mutation in the G-CSF R, in patients with chronic neutropenia. The authors demonstrated that this mutation caused a truncation in a differentiation-promoting domain within the G-CSF R, resulting in the ablation of G-CSF induced differentiation but not G-CSF induced proliferation. Further

support for a direct effect of HGFs on differentiation (as opposed to indirect effects such as induction or promotion of cell survival, or proliferation of a progenitor cell after commitment) have been reported by Shah *et al* (1994, 1996). In these reports it was shown that glial growth factor or TGF- β superfamily members acted in an instructive manner, to restrict multipotent neural progenitors to a glial lineage. Johe *et al* (1996) have also reported the involvement of instructive mechanisms in the differentiation of CNS stem cells to astrocytes. Therefore strong evidence exists for instructive mechanisms, particularly in the differentiation of cells within the CNS.

2.16 Summary

Clearly the processes involved in the regulatory control of haemopoiesis are highly complex. As indicated in the above sections, various models have been proposed to explain the involvement of growth factors and a large amount of evidence exists to support each model. As to which model is correct, remains uncertain and in controversy? Indeed they may not be mutually exclusive, as it is theoretically feasible for aspects of both stochastic and instructive mechanisms to be involved in this regulation. However, regardless of whether haemopoiesis is the result of intrinsic factors operating in a stochastic manner or is the result of an instructive environment, an interacting network of cytokines and their receptors is implicated in the regulation of haemopoiesis. This network may or may not exhibit redundancy and clearly shows many overlapping actions, whilst others may be unique to individual cytokines. It is evident that the balance of actions between the different members of this network, is important for the survival, proliferation, differentiation and maturation of haemopoietic cells and thus the normal running of the immuno-haemopoietic system.

CHAPTER 3: MIP-1 α and the Chemokine Superfamily

The role of MIP-1 α as a haemopoietic stem cell regulator was first identified in 1990, when Graham *et al* (1990) purified a reversible inhibitor of CFU-A and CFU-S stem cell proliferation from the J774.2 murine macrophage cell line. They called this activity Stem Cell Inhibitor (SCI) and sequence analysis demonstrated that it was identical to a previously described molecule, macrophage inflammatory protein-1 α (MIP-1 α).

MIP-1 α is a member of a rapidly expanding family of <u>chemo</u>tactic cyto<u>kines</u>, known as the chemokines. The biological functions of these chemokines are diverse and range from haemopoietic regulation to proinflammatory functions. The field has recently become a focus of attention and excitement, following the discovery of the critical role certain chemokines and their receptors play in the pathogenesis of HIV. An overview of chemokine biology (including recent advances), with a particular emphasis on the role of MIP-1 α in the immuno-haemopoietic system follows.

3.1 The Chemokine Superfamily: Structure and Family Subdivision

As indicated above, MIP-1 α belongs to the chemokine superfamily, which has expanded in a little under a decade from a handful of family members (Schall 1991, Oppenheim et al 1991) to now number in excess of 30-40 members (reviewed by Rollins 1997, Mackay 1997). In fact some researchers estimate that as many as 50-100 unique chemokines may eventually be identified. Table 3.1 lists the chemokines identified to date. The chemokines are all small inducible proteins, 8-18 kDa in size and are related to each on the basis of sequence homology and the presence of a characteristic cysteine motif in their N-terminus. Originally the family was subdivided into 2 classes, on the basis of the spacing of the first two N-terminal cysteine residues. Thus, in the "CC" or β subfamily these two cysteines are adjacent to each other, whilst in the "CXC" or α subfamily they are separated by a single residue. More recently, this designation has been complicated by the identification of unique members of two novel putative chemokine subfamilies. These are designated the C subfamily, with a sole member Lymphotactin (Kelner et al 1994), due to the presence of only a single N- CX_3C subfamily with its sole member terminal cysteine and the



Table 3.1 : The Chemokine Superfamily....so far

Fractalkine/Neurotactin (Bazan *et al* 1997, Pan *et al* 1997), in which the two amino terminal cysteines are separated by three amino acid residues. Members of each these four subfamilies, will be discussed in more detail in the following sections.

3.2 The CXC Chemokines

The CXC family of chemokines numbers at least 11 members to date (listed in table 3.1) and includes IL-8, PF-4, IP-10, Mig, the GRO proteins, NAP-2, ENA-78 and SDF-1. It is possible to further subdivide the CXC chemokines, based on the presence or absence of a so-called "ELR-motif", located in the N-terminal region just prior to the CXC motif. The ELR-chemokines such as IL-8, NAP-2, the GRO proteins, ENA-78, and GCP-2 are all potent neutrophil chemoattractants, whilst the non-ELR chemokines include PF-4, IP-10, Mig, and SDF-1 and show little or no neutrophil directed actions. Modification of these ELR residues prevents receptor binding and thus also neutrophil chemotaxis (Beall *et al* 1992, Clark-Lewis *et al* 1993) and insertion of an ELR motif into the N-terminus of non-ELR chemokines enables them to become neutrophil chemoattractants (Clark-Lewis *et al* 1993).

3.2.1 Interleukin-8 (IL-8)

IL-8 or neutrophil activating protein (NAP)-1 as it was originally called, is perhaps considered to be the prototypic CXC chemokine and is certainly the most extensively studied. To date, no rat or murine homologue of IL-8 has been identified. However, antibodies to human IL-8 inhibit lung inflammation in rats, suggesting a rodent IL-8 homologue does exist *in vivo*. Human IL-8 is produced by a wide variety of cell types, including monocytes, neutrophils, fibroblasts, endothelial cells, epithelial cells, and T lymphocytes and can be induced by a number of inflammatory stimuli such as IL-1, TNF, and LPS (reviewed by Stoekle and Barber 1990, Oppenheim *et al* 1991). IL-8 was originally identified as an activity that was directed towards neutrophils, but not monocytes (Yoshimura *et al* 1987, Shroeder *et al* 1987) and which may also chemoattract T lymphocytes (Larsen *et al* 1989). In addition to chemoattraction of neutrophils, IL-8 also induces degranulation and the respiratory burst in neutrophils (Walz *et al* 1987). IL-8 has also been suggested to play a role in angiogenesis (Koch *et al* 1992, Smith *et al* 1994a).

There is increasing evidence that IL-8 may also function as an inhibitor of haemopoietic stem cell proliferation. Broxmeyer et al (1993) showed that in addition to MIP-1 α , other chemokines such as PF-4, IL-8 and MCP-1 can all function to suppress colony formation from GM-CSF plus SCF stimulated progenitor cells. Laterveer et al (1995, 1996a) reported that a single injection of IL-8 rapidly mobilised haemopoietic progenitors and that SCF augmented this mobilisation (Laterveer et al 1996b). Although no murine homologue of IL-8 has been identified to date, a putative murine IL-8 receptor homologue (mIL-8Rh) has, and mice have been generated with a homozygous null mutation in this gene (Cacalano et al 1994). These IL-8Rh^{-/-} mice had an increased number of neutrophils, suggesting a negative role for the IL-8Rh in regulating neutrophil numbers. In addition, there was also an increased number of B cells in their lymph nodes, as well as enlarged spleens due to enhanced proliferation of specific myeloid elements and megakaryocytes. There also appeared to be enhanced myelopoiesis in the bone marrow. Thus, providing further evidence supporting a possible role for IL-8 or related chemokines in the regulation of haemopoiesis in vivo. Further evidence for the involvement of the IL-8Rh in the negative regulation of myeloid progenitor cells, was also reported in a more recent study (Broxmeyer et al 1996). This study compared normal mice and mice deficient in mIL-8Rh (mIL-8Rh^{-/-}) that were raised in either a normal or a germ-free environment. Myeloid progenitors from mIL-8Rh^{-/-} mice were insensitive to inhibition by hIL-8 and MIP-2, a ligand for mIL-8Rh. A large expansion of myeloid progenitors was observed in these mice and this expansion was decreased in mice raised under germ-free conditions, suggesting that it was environmentally inducible.

3.2.2 Stromal Cell Derived Factor-1 (SDF-1)

The chemokine SDF-1 was originally cloned from a 'signal sequence trap library' (designed to enrich for proteins containing a signal peptide sequence) (Tashiro *et al* 1993). It was also cloned from a stromal cell line, as pre-B-cell growthstimulating factor (PBSF) and as the name suggests, it stimulated the growth of B-cell progenitors *in vitro* (Nagasawa *et al* 1994). Unusually for a chemokine it is constitutively expressed in a broad range of tissues. The importance and apparent uniqueness of SDF-1 is only just beginning to emerge. Although SDF-1 is grouped with the CXC chemokines, it is equally distant/related to both CC and CXC
chemokines. Furthermore it has a unique degree of conservation with the murine and human homologues sharing 99 % amino acid identity, suggesting that of all the chemokines to date, SDF-1 is the least diverged from any prospective primordial ancestor.

An insight into the probable role of SDF-1 in vivo has come from the generation of mice in which the gene for SDF-1 has been disrupted (Nagasawa et al 1996a). These SDF-1^{-/-} mice died perinatally, showing a severe reduction in the number of B cell progenitors in both the fetal liver and bone marrow and a reduction in myelopoietic progenitors in the bone marrow (but not the fetal liver), thus suggesting SDF-1 is a regulator of B cell lymphopoiesis and bone marrow myelopoiesis. These SDF-1^{-/-} mice also exhibited a defect in the development of the cardiac ventricular septum. It has since been reported that SDF-1 is a low-potency, high efficacy lymphocyte chemoattractant in vitro (Bleul et al 1996a) and that it also chemoattracts monocytes, but not neutrophils (it is a non-ELR chemokine). Most recently, Aiuti et al (1997) demonstrated that SDF-1 was a chemoattractant for human CD34⁺ progenitor cells. These progenitors consisted of both moderately primitive and more lineage-committed CD34⁺ cells. This suggests that SDF-1 may be important in the migration and homing of bone marrow CD34⁺ progenitors to different organs during development and to different niches during the differentiation and maturation of haemopoietic progenitor cells. Such a role certainly provides a feasible explanation for the myelopoietic defects observed in the SDF-1^{-/-} mice. The role of SDF-1 in HIV pathogenesis is discussed in section 3.9.

3.2.3 IP-10 and Mig

Interferon-inducible Protein-10 (IP-10) was originally isolated in 1985, as an IFN- γ induced product from U937 cells (Luster *et al* 1985). IP-10 is expressed by a variety of cell types, including monocytes, keratinocytes, fibroblasts, endothelial cells and T lymphocytes (Luster *et al* 1985). IP-10 is a non-ELR chemokine and this is in keeping with its poor neutrophil activity (Dewald *et al* 1992), but it does appear to be chemotactic for monocytes (Taub *et al* 1996) and smooth muscle cells (Wang *et al* 1996). At the moment there is conflicting evidence as to whether IP-10 does, or does not chemoattract T lymphocytes (Taub *et al* 1996, Luster and Leder 1993, Roth *et al* 1995). In addition to chemotactic functions, IP-10 also appears to be an inhibitor of

angiogenesis (Angiolillo et al 1995), in contrast to IL-8 which appears to stimulate angiogenesis (Smith et al 1994a).

Monokine Induced by Interferon- γ (Mig) is a recently identified chemokine and like IP-10, it is also an IFN- γ inducible protein that was isolated from macrophages (Liao *et al* 1995). Mig is a chemoattractant *in vitro* for tumour infiltrating lymphocytes (TIL) and for peripheral blood lymphocytes (PBLs) (Liao *et al* 1995).

Mig and IP-10 can cross-desensitise each other in receptor activation (Liao *et al* 1995) and in fact, they appear to share the same receptor, CXCR3 (Loetscher *et al* 1996a). Like IL-8 (see section 3.2.1), both Mig and IP-10 have also been reported to act as inhibitors of haemopoietic progenitor cell proliferation (Sarris *et al* 1993, Sarris *et al* 1996, Schwartz *et al* 1997).

3.2.4 Other CXC Chemokines

The early study of PF-4 as an immunoregulator has been well documented (Stoekle *et al* 1990). PF-4 has also been reported to inhibit megakaryocytopoiesis (Gewirtz *et al* 1989, Han *et al* 1990, Gewirtz *et al* 1995) and more recently it was proposed that PF-4 (and IL-8 and NAP-2) supported the viability and survival of normal (but not leukaemic) progenitor cells (Han *et al* 1997). Little is known about the functions of other CXC chemokines like ENA-78 (Walz *et al* 1991) or some of the more recently identified members such as LIX (Smith and Herschman 1995) and granulocyte chemotactic protein-2 (GCP-2) (Proost *et al* 1993), beyond the initial reports of their isolation and characterisation.

3.3 The CC Chemokines

The number of CC chemokines appears to be much larger and their effects more diverse, than those of the CXC chemokine subfamily. Whilst many members were initially characterised as monocyte/macrophage chemoattractants, it is now evident that they act upon an extremely wide range of cell types, including basophils, mast cells, eosinophils, B and T lymphocytes and NK cells (reviewed by Rollins 1997). In general they do not exert neutrophil directed effects, although there are a few examples of specific chemokines that may be exceptions to this.

3.3.1 The Monocyte Chemotactic Proteins

3.3.1.1 Monocyte Chemotactic Protein-1 (MCP-1) / JE

Human monocyte chemotactic protein-1 or MCP-1 was first isolated in 1989 from monocytic cell lines and activated monocytes (Yoshimura *et al* 1989) and its amino acid sequence found to be identical to that of the previously cloned human homologue of the mouse JE gene (Rollins *et al* 1989). JE was originally reported as a gene that was highly inducible by platelet-derived growth factor (PDGF) in the 3T3 fibroblast cell line (Rollins *et al* 1988). It is now generally assumed that JE is the murine homologue of the human MCP-1 gene and JE is therefore often commonly also refered to as MCP-1 (although as discussed in section 3.3.1.3, recent evidence suggests this may not necessarily be the case).

MCP-1 is an extremely potent monocyte chemoattractant (Yoshimura et al 1989), but has also been reported to function in vitro as a chemoattractant for activated T lymphocytes (Carr et al 1994) and NK cells (Taub et al 1995a). MCP-1 is also a potent inducer of mast cell migration (but not degranulation) (Taub et al 1995b). In addition, MCP-1 has also been reported to induce IL-1 and IL-6 expression in monocytes (Jiang et al 1992). Interestingly, a very recent report suggested that MCP-1 may also be involved in vivo with other CXC and CC chemokines in neutrophil recruitment (but does not appear to do so on its own), suggesting the existence of a cooperative chemokine network in vivo (Tessier et al 1997). In an elegant study, Weber et al (1996) demonstrated that MCP-1 could be converted from a potent inducer of mediator release from basophils to an eosinophil chemoattractant, by deleting the Nterminal amino acid residue from full length mature MCP-1. They also demonstrated that deletion of another residue led to inactivity with both basophils and eosinophils, whilst deletion of three or four residues resulted in mutants that were active on both cell types. This strongly suggests that N-terminal processing is a potential mechanism for regulating MCP-1 cell selectivity and function. Interestingly, an antagonist of MCP-1 (9-76) derived by deletions in the N-terminus was recently shown to inhibit arthritis in the MRL-lpr model, a murine model in which mice spontaneously develop a chronic inflammatory arthritis similar to human RA (Gong et al 1997).

Several laboratories are currently trying to address the true *in vivo* role of MCP-1, through the generation and study of MCP-1 transgenic and "knockout" mice.

In one recent report the murine MCP-1 gene was coupled to the strong mouse mammary tumour virus (MMTV) promoter, resulting in mice (MMTV-MCP-1) with a high level of MCP-1 expression that corresponded to an increased susceptibility to intracellular pathogens (Rutledge et al 1995). No monocyte infiltrates were observed in any of these transgenic mice and this was proposed to be the result of monocyte unresponsiveness to locally produced MCP-1, due either to receptor desensitisation or disruption of monocyte chemoattractant gradients. In a second study by this same group, transgenic mice expressing MCP-1 under the control of the rat insulin promoter (RIP) were reported to elicit monocyte recruitment into pancreatic islets without the development of insulitis (diabetes) (Grewal et al 1997). When these RIP-MCP-1 transgenic mice were crossed with the MMTV-MCP-1 mice discussed above (thus generating mice expressing both transgenes), the resulting mice did develop a mild form of insulitis. This suggests that high systemic levels of MCP-1 prevent monocytes from responding to local MCP-1 and therefore the ability of MCP-1 to elicit monocyte infiltration will be dependent on its expression at lows levels and in it being restricted to the appropriate site.

There is increasing evidence that MCP-1 (and other chemokines) may be critically important mediators of inflammation in the central nervous system (CNS). Chemokine expression, particularly that of MCP-1 and MIP-1 α , have been reported in many animal models of CNS inflammation (Berman *et al* 1996, reviewed by Ransohoff *et al* 1996).

3.3.1.2 MCP-2 and MCP-3

MCP-2 and MCP-3 are part of a small subgroup of CC chemokines with extensive sequence homology to MCP-1 (>60 %). In comparison to MCP-1 and MCP-3, MCP-2 action is still poorly understood. MCP-2 and MCP-3 were both identified through their monocyte chemoattractant abilities (Van Damme *et al* 1992) and they are both as effective as MCP-1 in this respect. Like MCP-1, they have both been reported to function as T lymphocyte chemoattractants. It appears that the biological effects of MCP-2 and MCP-3 partially overlap with those of MCP-1, yet others are clearly distinct. They often appear however, to be less effective and are produced at lower levels than MCP-1 (reviewed by Proost *et al* 1996). Whilst all three MCPs chemoattract basophils, MCP-3 is a particularly effective stimulator of basophil chemotaxis and release (Alam *et al* 1994b). The putative murine homologue of human MCP-3 has been identified (Minty *et al* 1993), and was originally cloned as MARC from an IgE plus antigen stimulated mast cell line (Kulmburg *et al* 1992) and subsequently from 3T3 fibroblasts as fibroblast induced cytokine (*fic*) (Heinrich *et al* 1993b).

3.3.1.3 MCP-4 and MCP-5

MCP-4 and MCP-5 are the two most recently described MCPs. MCP-4 shares closest homology to MCP-3 and another CC chemokine, eotaxin. In fact MCP-4 appears to share similar functional properties to MCP-3 and eotaxin. Like MCP-3 and eotaxin it is also a strong chemoattractant for eosinophils, as well as for monocytes and T lymphocytes (Uguccioni *et al* 1996) and it is also reported to stimulate histamine release from basophils (Garcia-Zepeda *et al* 1996).

To date, only murine MCP-5 has been identified (Jia *et al* 1996, Sarafi *et al* 1997). MCP-5 appears to share greatest homology with human MCP-1 (66 % amino acid identity) and is also structurally very similar too. In fact this similarity is even greater than that shown between human MCP-1 and the supposed murine homologue, JE (55 % identity). This suggests that JE may not in fact be the murine homologue of human MCP-1, although Jia *et al* (1996) did find evidence for a potential human MCP-5 gene that was not MCP-1.

3.3.2 RANTES

RANTES (Regulated on Activation Normal T-cell Expressed and Secreted) was identified by Schall *et al* in 1988, and found to be a gene inducible by antigenic or mitogenic stimulation in a variety of T cell lines and circulating T lymphocytes (Schall *et al* 1988). RANTES is produced by a variety of cell types including platelets (Kameyoshi *et al* 1992) and eosinophils (Lim *et al* 1996). RANTES is a potent chemoattractant of monocytes and subsets of T lymphocytes, particularly of the CD4⁺, CD45RO⁺ memory phenotype (Schall *et al* 1993), but is also a very potent chemoattract and activate eosinophils (Rot *et al* 1992) and basophils (Bischoff *et al* 1993), as well as NK cells (Taub *et al* 1995a).

3.3.3 Other CC Chemokines

There are now over twenty known, distinct CC chemokines, many only recently discovered and no doubt others remain to be identified still. Briefly, others include Eotaxin and Eotaxin-2 (also known as MPIF-2), which are potent chemoattractants and activators of only eosinophils and basophils (Jose et al 1994, Forssmann et al 1997). I-309 and its putative murine homologue TCA-3 (Miller et al 1989, Burd et al 1987) are produced by activated mast cells and T cells and are unusual in having an extra pair of cysteines. TCA-3 is activated on a wide variety of cell types, including neutrophils. Other novel chemokines include C10 (Orlofsky et al 1991), MIP-1y (Mohamadzadeh et al 1996), CCF18 or MIP-related protein-2 (MRP-2) (Youn et al 1995, Hara et al 1995). Recently two laboratories isolated a novel chemokine with an extended C-terminus that includes two extra cysteines, known as 6C-Kine (Hedrick and Zlotnick 1997) or Exodus-2 (Hromas et al 1997a). Another interesting recent discovery, was that of a novel chemokine known as thymus expressed chemokine or TECK, that appears to be specifically expressed by thymic dendritic cells and which may be involved in T cell development (Vicari et al 1997). In most cases the functions of these novel chemokines are unknown.

3.4 The C Subfamily: Lymphotactin

To date Lymphotactin (Lptn) is the only member of the so called C subfamily. It only has 2 cysteines, which correspond to cysteines 2 and 4 in the CXC and CC chemokines. Lptn was originally cloned from activated pro-T cells (Kelner *et al* 1994) and the human homologue, SCM-1, was subsequently isolated (Yoshida *et al* 1995). Lptn was shown initially to attract T lymphocytes, particularly of type CD4⁺ and CD8⁺, as well as thymocytes (with the exception of CD4⁺CD8⁺ thymocytes) (Kelner *et al* 1994). More recently, Lptn has also been reported to chemoattract NK cells both *in vitro* and *in vivo* (Hedrick *et al* 1997). In addition to pro-T cells and activated T cells, Lptn is now known to be produced by NK cells (Hedrick *et al* 1997) and IgE plus antigen activated mast cells (Rumsaeng *et al* 1997).

3.5 The CX₃C Subfamily: Fractalkine / Neurotactin

The recent identification of a putative fourth chemokine subfamily has caused considerable interest. This is because fractalkine, the only member identified to date,

appears unique in being the only known membrane-anchored chemokine. Fractalkine was cloned by Bazan *et al* (1997) from non-haemopoietic cells and shown to be a 373amino acid glycoprotein that carried a chemokine domain on a long extended mucinlike stalk. Fractalkine exists either in a membrane bound form or as a secreted protein. The secreted form attracts T cells and monocytes, whilst the anchored form promotes leukocyte adhesion. The expression of fractalkine is also unique amongst chemokines, with little or no expression detectable in peripheral blood leukocytes. Fractalkine can however, be strongly induced by IL-1 and TNF. Fractalkine was also cloned independently as neurotactin and shown to be predominantly expressed in mouse brain, with elevated expression in mice with EAE or in LPS treated mice (Pan *et al* 1997). However, in contradiction to Bazan *et al*, neurotactin was reported to attract neutrophils. The precise function of fractalkine / neurotactin is still uncertain, but it may well be involved in leukocyte adhesion and CNS inflammation.

3.6 MIP-1α

MIP-1 α and MIP-1 β are clearly distinct molecules. However, they do share some striking similarities in their functions, barring a few specific exceptions. Therefore, the following discussion is largely limited to that of MIP-1 α , unless stated otherwise.

3.6.1 Characterisation of MIP-1a

MIP-1 α and MIP-1 β were originally isolated as MIP-1 from the macrophage cell line RAW264.7, stimulated by LPS (Wolpe *et al* 1988). MIP-1 ran as a doublet composed of two peptides with very similar molecular weights of ~8 kDa. This doublet was subsequently found to comprise two distinct proteins, which were renamed MIP-1 α and MIP-1 β (Sherry *et al* 1988). The cDNA previously cloned as MIP-1 corresponded to that of MIP-1 α (Davatelis *et al* 1988). The amino acid sequences of MIP-1 α and MIP-1 β are 59.8 % identical, whilst their cDNAs share 56.7 % identity. The human homologues of MIP-1 α and MIP-1 β had previously been identified as LD78 (Obaru *et al* 1986) and Act-2 (Lipes *et al* 1988) respectively. The MIP-1 proteins are encoded for by distinct genes and clustered on chromosome 17 in humans, and on chromosome 11 in mice, along with most other CC chemokines.

One of the original observations made during the isolation of MIP-1 was its propensity to undergo self-aggregation (Wolpe et al 1988). In fact this self aggregation is a phenomenon noted for both human and murine MIP-1 α and MIP-1 β . The aggregation results from non-covalent and electrostatic interactions between molecules and occurs to such an extent that multimers may result with molecular masses in excess of 100 kDa in physiological buffers in vitro (Graham et al 1992) and even greater aggregation occurs at higher concentrations (>10⁶ kDa). Further understanding of this aggregation has been achieved through the derivation of three MIP-1 α mutants, in which the C-terminal acidic amino acids have been sequentially neutralised, resulting in the generation of non-aggregating forms of MIP-1a (Graham et al 1994). These three mutants formed tetramers, dimers and monomers respectively, which were equipotent to each other and to wild type MIP-1 α , in both stem cell inhibition assays and monocyte shape-change assays. This suggested that aggregation was not necessary for the function of MIP-1 α in these assays and implied that the aggregated forms spontaneously disaggregate and are ultimately functional as monomers, as has been confirmed in more recent studies. However, it should be noted that these results are in contrast to those obtained by Mantel et al (1993) in a previous study. In this study, it was proposed that the inhibitory activity of aggregated MIP-1a resided in small amounts of monomeric MIP-1 α in the aggregated preparations and it was further suggested, that the monomeric MIP-1 α therefore had 1000-fold higher activity per unit weight than the aggregated forms. Graham et al (1994) reported however, that the aggregated form (separated from any monomeric MIP-1 α) was as active as the monomeric form and therefore concluded that the aggregated form spontaneously disaggregated under the assay conditions used.

Recently, the actions of stem cell inhibition and monocyte chemoattraction have been uncoupled. A MIP-1 α mutant has been generated in which the proteoglycan binding domain had been disrupted by mutagenesis of two basic amino acids within this site (Graham *et al* 1996). This mutant (HepMut) was unable to bind proteoglycans or bind to the human or murine MIP-1 α receptor, CCR1, and was also severely compromised in its ability to function in the monocyte shape change assay. In contrast however, HepMut was still fully active in both human and murine stem cell inhibition assays. Therefore, it appears that the proteoglycan binding region mutated in this study is critical for binding of MIP-1 α to CCR1, but not for inhibition of stem cell proliferation and suggests, that inhibition may occur through another receptor distinct from CCR1.

3.6.2 Normal Expression of MIP-1a

MIP-1 α mRNA has been reported to be expressed by a wide variety of haemopoietic cell-types, including monocytes and macrophages (Davatelis *et al* 1988), T lymphocytes (Obaru *et al* 1986), mast cells (Burd *et al* 1989), eosinophils (Costa *et al* 1993), basophils (Li *et al* 1996), neutrophils (Kasama *et al* 1993), NK cells (Bluman *et al* 1996) and epidermal langerhans cells (Heufler *et al* 1992). In addition, several non-haemopoietic cell-types such as fibroblasts (Nakao *et al* 1990) and microglial cells (Hayashi *et al* 1995) have also been reported to express MIP-1 α .

MIP-1 α has been reported to be an inhibitor of haemopoietic stem cell proliferation (Graham et al 1990) and although expression of MIP-1a mRNA is detectable by PCR based techniques in normal bone marrow, the levels of MIP-1 α are low. This is a situation which is in fact common for other haemopoietic cytokines (see section 1.6.1). One explanation for these low levels, is that MIP-1 α protein is sequestered and held in a local microenvironment through interactions with proteoglycans and therefore only a low level of expression is necessary for an effect within a localised microenvironment. Although as indicated above, such proteoglycan interaction is not apparently necessary for stem cell inhibition. Another explanation, is that under normal circumstances the expression of MIP-1 α is held in check and is only released under conditions of stress, such as inflammation. One possible candidate for the *in vivo* suppression of MIP-1 α , is TGF- β . Reports from our laboratory (Maltman et al 1993, 1996) have defined an endogenous reciprocal relationship in bone marrow derived macrophages between MIP-1 α and TGF- β , whereby elevation of MIP-1 α will lead to the increased expression of TGF- β , which in turn will result in a suppression of MIP-1 α expression. Thus, in the bone marrow, where levels of TGF- β are readily detectable, TGF-B would act as the predominant partner and keep the expression of MIP-1 α in check. Outside of the bone marrow, several other cytokines have also been shown to downregulate the expression of MIP-1 α , including IL-10 (Kasama et al 1994) and IL-4 (Standiford et al 1993a).

MIP-1 α was originally identified as an LPS-inducible gene in macrophages, but a number of other stimuli also induce MIP-1 α expression. These include IgE in mast cells (Burd et al 1989) and IL-1, IL-2, and IL-6 in monocytes (Maciejewski et al 1992). Several studies have tried to address the transcriptional elements involved in inducible MIP-1 α expression, but with limited success. Comparative analysis of the promoters for murine MIP-1 α and β revealed a conserved CK-1 element (Widmer et al 1993) previously identified in the murine GM-CSF gene (Stanley et al 1985), whilst Ritter et al (1995) demonstrated that a novel transcription factor named MIP-1 α nuclear protein (MNP) was required for transcription of human MIP-1a. Analysis of LPS induction has revealed that members of the C/EBP, NF-KB, and c-ets transcription factor families may be involved in LPS induced MIP-1a transcription (Grove and Plumb 1993). Another report, by Proffitt et al (1995), suggested that ATF/CREB and AP-1 related proteins may be involved in regulating MIP-1ß expression following LPS induction, implying that whilst MIP-1 α and MIP-1 β are coexpressed in many cell types they may exhibit differential transcription under certain circumstances.

3.6.3 Expression in Disease and Inflammation

Elevated MIP-1 α expression has been observed in a number of disorders and disease states. For instance, a significant increase in MIP-1 α mRNA expression was observed in bone marrow nucleated cells derived from patients with aplastic anaemia (AA) and myelodysplastic syndrome (MDS) (Maciejewski *et al* 1992). It is probable that this aberrant expression results from dysregulated expression of other cytokines within the bone marrow as a result of the disease state, though it is also possible that aberrant MIP-1 α expression might play an etiological role in the development of AA and MDS.

Increased expression of MIP-1 α has been observed in both human interstitial lung disease (Standiford *et al* 1993b) and also in a murine model of interstitial lung disease (Smith *et al* 1994b), as well as in several other animal models of human diseases such as murine endotoxemia (a model for sepsis) (Standiford *et al* 1995).

MIP-1 α expression is also elevated in TGF- $\beta^{-/-}$ mice (Shull *et al* 1992). This finding is certainly in keeping with the known endogenous reciprocal relationship with

TGF- β (see previous section) and suggests that this interaction may indeed occur *in vivo*. It is not clear however, whether MIP-1 α is a direct mediator of the multifocal inflammation observed in these mice or if it is merely a secondary consequence of it.

In addition to classical inflammatory disease, MIP-1 α has also been implicated in several auto-immune diseases, most notably in rheumatoid arthritis (RA), as well as in inflammatory neurological disease. RA is characterised by elicitation and activation of various leukocyte populations within the synovial space and in joint tissue. These recruited leukocytes are known to be critical participants in the resulting pathology. A number of chemokines have been implicated in the recruitment of leukocytes in RA (reviewed by Kunkel et al 1996), one of these being MIP-1a (Koch et al 1994). Aberrant MIP-1 α expression has also been documented by many researchers in a variety of inflammatory neurological disease states. One of the most intensively studied areas has been that of multiple sclerosis (MS), characterised by focal T cell and macrophage infiltration into white matter and resulting demyelination. Significant elevation of MIP-1 α has been reported in MS patients undergoing relapse (Miyagishi et al 1995). Experimental allergic encephalomyelitis (EAE) is an experimental model frequently used to study MS. Investigation of EAE, has revealed that the kinetics of MIP-1 α expression (and certain other chemokines) parallels that of leukocyte recruitment and disease severity (Miyagishi et al 1997). MIP-1 α has been reported to be produced by microglial cells within the central nervous system (CNS) (Hayashi et al 1995) and by glial and neuronal cells within the brain (Ishizuka et al 1997). Perhaps most conclusive, is the report by Karpus et al (1995), in which they demonstrated that administration of anti-MIP-1 α antibodies prevented the development of EAE. Taken together these studies strongly implicate MIP-1 α (and other chemokines) as important mediators of inflammation within the CNS.

3.6.4 Pro-inflammatory Functions of MIP-1 α

During the initial characterisation of MIP-1 α , as MIP-1, Wolpe *et al* (1988) reported that injection of MIP-1 into the footpads of mice resulted in inflammation, characterised primarily by a neutrophil infiltrate. MIP-1 was also shown to attract neutrophils and induce a respiratory burst *in vitro*. Since that initial report the precise relationship between MIP-1 α and neutrophils has remained confused. Firstly, the

report of Wolpe et al (1988) used a supraphysiological dose of MIP-1. A later report by McColl et al (1993) suggested that human MIP-1 α had only very minimal effects on neutrophil attraction and activation. In contrast however, was a later report that murine MIP-1a was a mediator of neutrophil recruitment in murine endotoxemia (Standiford *et al* 1995). One possible explanation is that MIP-1 α does not play a role in neutrophil function in humans, but that it does in rodents. In support of this is the observation that rodent, but not human, neutrophils express abundant CCR1, a MIP- 1α receptor. In addition, IL-8 which is a potent neutrophil chemoattractant has not been identified in rodents, nor has its receptor CXCR1 been detected on rodent neutrophils. It is possible therefore that in rodents, MIP-1 α and CCR1 can fulfil the roles played by IL-8 and CXCR1 in humans. This notion is however, difficult to reconcile with the identification of a putative murine IL-8Rh that has been implicated neutrophil function (Cacalano et al 1994). At present the role of MIP-1 α in in neutrophil function is unclear, but it does underline the fact that caution that should be used in assuming that there is a direct correlation between human and murine chemokines, their receptors, and their functions.

MIP-1 α is unquestionably a potent monocyte/macrophage chemoattractant (Standiford *et al* 1993b), but in addition it is also involved in regulating the effects of many other immune cell types. Macrophages are not only a predominant source of MIP-1 α expression, they are also themselves targets for the action of MIP-1 α . Fahey *et al* (1992) reported that native MIP-1 purified from LPS stimulated macrophages (containing both MIP-1 α and MIP-1 β), was able to enhance the antibody independent macrophage cytotoxicity for tumour cells and stimulate the proliferation of mature tissue macrophages. In addition MIP-1 α , but not MIP-1 β , was able to induce secretion of TNF, IL-1 α and IL-6. An excess of MIP-1 β was in fact able to suppress the MIP-1 α induced production of TNF.

It is clear, that whilst MIP-1 α and MIP-1 β appear to be co-expressed from many cell types and show many similar functions, they do also exert some distinct ones too. This is most apparent in their abilities to regulate stem cell proliferation (see below), but also in their effects on different lymphocyte populations. Taub *et al* (1993) showed that human MIP-1 α preferentially attracted activated CD4⁺ T cells, whilst human MIP-1 β was preferential for activated CD8⁺ T cells. This differential attraction

60

was confirmed by Schall *et al* (1993), who also demonstrated that MIP-1 α was a more potent B cell chemoattractant than MIP-1 β . Although proteoglycan binding appears dispensable for the function of MIP-1 α in stem cell assays, a report by Tanaka *et al* (1993) suggests that proteoglycan binding may be important for inflammatory function. MIP-1 β was shown to be present on lymph node endothelium and this proteoglycan bound MIP-1 β induced the binding of T cell to VCAM-1 *in vitro*. This supports the notion that proteoglycans are involved in the localisation of chemokines to endothelial and other cell surfaces, where they act to recruit immune cells from the periphery to sites of inflammation.

In addition to actions on monocytes, T cells and neutrophils, MIP-1 α has also been reported to activate basophils (Alam *et al* 1992), chemoattract (but not degranulate) IgE plus antigen activated mast cells (Taub *et al* 1995b) and induce the migration and activation of eosinophils (Rot *et al* 1992). MIP-1 α has also been reported to enhance IgE and IgG4 production from human B cells (Kimata *et al* 1996) and to up-regulate synthesis of the latent metalloproteinase, pro-metalloproteinase-9 (proMMP-9) from lymphocytes (Johnatty *et al* 1997). Taken together, these reports demonstrate an important role for MIP-1 α in the recruitment of specific immune celltypes into various tissues during inflammation. In addition MIP-1 α appears to play a modulatory role on a variety of functions exerted by these recruited cells. MIP-1 α also plays a key role in the inflammatory cascade, through its ability to induce production of other cytokines such as TNF and IL-1, that may in turn induce additional MIP-1 α production and thus further amplify the local inflammatory response.

The most conclusive evidence verifying the *in vivo* importance of MIP-1 α has come from the recent generation of MIP-1 α "knockout" or "null" mice (MIP-1 α^{-1}), in which the MIP-1 α gene was disrupted (Cook *et al* 1995). These mice had no overt haemopoietic abnormalities, suggesting that either MIP-1 α is dispensable for normal haemopoiesis or is functionally redundant and its actions are replaced by another factor in these MIP-1 α^{-1} mice. In contrast however, MIP-1 α was required for generation of an inflammatory response to viral infection. Homozygous null mice were resistant to coxsackievirus-induced myocarditis compared with infected wild type mice, which did develop myocarditis. A reduced pneumonitis and delayed viral clearance was also observed in influenza virus infected MIP-1 α^{-1} mice compared with infected wild type mice. Thus, MIP-1 α is involved in the inflammation associated with viral infection *in vivo*.

3.6.5 The Role of MIP-1a as a Haemopoietic Regulator

The first report on the ability of MIP-1 α to function as a regulator of haemopoietic stem cell proliferation, was by Broxmeyer et al (1989). They used purified MIP-1, containing both MIP-1 α and MIP-1 β , and observed that addition of MIP-1 led to enhanced proliferation of M-CSF or GM-CSF stimulated committed progenitors (GM-CFC). MIP-1 had no effect on G-CSF stimulated colony formation. Subsequently, Graham et al (1990) purified the previously identified stem cell inhibitor (SCI) or NBME-IV activity (Lord et al 1976) and discovered that it was identical to MIP-1 α . They were able to demonstrate that SCI/MIP-1 α inhibited colony formation from CFU-A type stem cells and its in vivo correlate, CFU-S, but not that of more committed GM-CFC progenitors. Neutralising polyclonal serum raised against MIP-1 was also able to block the inhibitory activity of both SCI and recombinant MIP-1 α . MIP-1 β however, was unable to inhibit stem cell proliferation, at the concentrations used in these studies. Broxmeyer et al (1990) subsequently confirmed the inhibitory action of MIP-1 α and the inability of MIP-1 β to function as an inhibitor. This group also reported that MIP-1 β was in fact able to block the suppressive effects of MIP-1a on BFU-E, CFU-GEMM and CFU-GM colony formation (Broxmeyer et al 1991). However, this block only occurred when MIP-1 β was added in an excess to MIP-1 α , as a 1:1 ratio did not result in a block of MIP-1 α inhibition (explaining why MIP-1, composed of a 1:1 ratio of MIP-1 α and β , was active as an inhibitor in the earlier study by Broxmeyer et al 1990).

The ability of MIP-1 α to function *in vivo* as an inhibitor of stem cell proliferation has also been addressed in several studies. Dunlop *et al* (1992) demonstrated that human MIP-1 α was active in a dose-dependent manner *in vitro* on d12 CFU-S cells, and to a lesser extent on more mature d8 CFU-S cells. In the same study, the authors reported that administration of multiple doses of MIP-1 α was sufficient to protect the CFU-A / S compartment, following treatment with the cytotoxic drug cytosine arabinoside (Ara-C). Lord *et al* (1992) have reported that MIP-1 α will also protect CFU-S cells against the cytotoxic effects of hydroxyurea

(HU). However, a later study by Quesniaux *et al* (1993), showed that more primitive stem cells were not protected by human MIP-1 α . In this model, a primary injection of 5-FU was used to eliminate late progenitors and cause the more primitive, and normally quiescent, LTR stem cells (see section 1.3) to enter the cell cycle and therefore become sensitive to administration of a second dose of 5-FU. MIP-1 α did not prevent the depletion of LTR stem cells following the second 5-FU injection, suggesting that it did not prevent cycling of the LTR stem cells. MIP-1 α was however, able to inhibit the formation of more mature haemopoietic progenitors (GM-CFC, BFU-E, CFC-_{multi}, and pre-CFC_{multi}). It should be noted that cytotoxic drugs (such as Ara-C, HU, and 5-FU) that target specific stages of the cell cycle, are not the only drugs utilised in chemotherapy. They are often used in combination with non-S phase drugs, such as cyclophosphamide. Studies such as those recently reported by Marshall *et al* (1997), are currently being undertaken to address whether or not MIP-1 α is a potentially useful myeloprotective agent against therapeutic regimes that are more applicable to current clinical settings.

The therapeutic value of MIP-1 α against leukaemic cells has also been investigated. Eaves et al (1993) have demonstrated that MIP-1 α does not inhibit the cycling of primitive chronic myeloid leukaemia (CML) cells supported in long-term cultures (LTC). MIP-1 α did however, inhibit the activation of normal primitive progenitors (but not more mature ones) in similar LTCs. This provides hope that MIP- 1α could prove useful in a therapeutic role, by providing protection of normal primitive stem cells against any cytotoxic agents, such as those used in chemotherapy, whilst simultaneously leaving CML cells susceptible. Chasty et al (1995) also reported that CML cells were relatively refractory to inhibition by MIP-1a. Moreover, the authors demonstrated that these CML cells expressed similar levels of MIP-1a receptors as did normal CD34⁺ cells. This suggests the inability of MIP-1 α to inhibit CML cells may lie in a downstream signalling event. Interestingly, Ferrajoli et al (1994) reported that MIP-1 α could inhibit the proliferation of subsets of acute myelogenous leukaemia (AML) progenitors. However, subsequently Owen-Lynch et al (1996) have shown that sub-sets of AML cells are not inhibited by MIP-1 α . In this study, autonomous AML blast cells were refractory to MIP-1 α inhibition, whilst cytokine responsive AML cells were inhibited by MIP-1 α in the presence of either GM-CSF or SCF. However, when GM-CSF and SCF were used in combination, only a small number were inhibited by MIP-1 α with the remainder remaining refractory. Interestingly, in the earlier study reported by Ferrajoli *et al* (1994) only single cytokine additions were used. In contrast, the later study of Owen-Lynch *et al* (1996) used either single cytokine additions (in which the CML cells were inhibitable), or a combination of cytokines (in which most samples were refractory to MIP-1 α inhibition), or no cytokine treatment to allow autonomous growth. The use of multiple cytokines is more likely to be representative of the environment *in vivo*, and therefore this study suggests that in addition to CML cells, autonomous AML cells and some subsets of cytokine responsive AML cells may be potential targets for therapeutic regimes, in conjunction with MIP-1 α as a protective agent.

In addition to the inhibition of bone marrow haemopoiesis, MIP-1 α also functions *in vitro* as a potent, reversible inhibitor of both human and murine epidermal keratinocyte proliferation (Parkinson *et al* 1993). MIP-1 α mRNA was previously shown to be produced by epidermal langerhans cells (Heufler *et al* 1992) and it has been proposed, that within the skin, epidermal stem cells are held quiescent by MIP-1 α derived from langerhans cells located within the same epidermal proliferative unit (Graham and Pragnell 1992b).

Inhibition by MIP-1 α , seems to occur on a relatively restricted subset of haemopoietic stem cells, represented by the CFU-A / S compartment. MIP-1 α will not inhibit the most primitive stem cells, nor more mature ones. In fact it appears to function as a stimulator of more mature progenitors. In this respect, MIP-1 α shares many similarities with TGF- β . Indeed Keller *et al* (1994) have shown that MIP-1 α and TGF- β show overlapping, as well as distinct functions on haemopoietic stem cells. MIP-1 α and MIP-1 β have extensive homology to each other, but MIP-1 α is more effective as an inhibitor and MIP- β may play a role in blocking this inhibition. It is also becoming evident that many other chemokines may function as inhibitors of haemopoietic inhibitors including Exodus (or MIP-3 α or LARC) (Hromas *et al* 1997b), Exodus-2 (also known as 6C-Kine or SLC) (Hromas *et al* 1997a), myeloid progenitor inhibitory factor-2 (MPIF-2) (Patel *et al* 1997) and MIP related protein-2 (MRP-2) also known as MPIF-1 or CCF18 (Youn *et al* 1995). With the exception of

MPIF-2, all these reports of actions as putative inhibitory factors have come from the Broxmeyer laboratory. Broxmeyer *et al* (1993) have also previously suggested that in addition to MIP-1 α , MCP-1 (and some CXC chemokines) can also function as an inhibitor of haemopoietic progenitor cell proliferation. This observation is in contradiction to a previous report from our laboratory (Graham *et al* 1993), in which only MIP-1 α , and MIP-1 β to a lesser extent, are active as inhibitors. Exactly why these contradictions exist is uncertain, but they may reflect differences resulting from the different types of assay used by each group. Nevertheless it remains a possibility that other future reports will confirm that many of these novel CC chemokines also function as haemopoietic inhibitors.

3.7 Chemokine Receptors

A great deal of research has been directed at trying to understand how various chemokines interact with their cognate receptors. It is now evident that the chemokine receptors belong to the superfamily of seven transmembrane spanning G protein-coupled receptors (7TMsGPCR) and a multitude of chemokine receptors have now been cloned. Recently, the fields of chemokine biology and virology have surprisingly merged. This has arisen through many outstanding contributions and resulted in the identification of an essential role for chemokines and chemokine receptors in the pathogenesis of HIV infection. A brief overview of the various known chemokines, together with their significance to viral pathogenesis is presented in the following sections.

3.7.1 Early characterisation of receptors

The first chemokine receptors to be identified were those for IL-8, and were named IL-8R_A (Holmes *et al* 1991) and IL-8R_B (Murphy and Tiffany). IL-8 binds to both with high affinity, but IL-8R_A is specific for IL-8, whereas IL-8R_B binds other CXC chemokines (see table 3.2). With the cloning of these receptors it became possible to design degenerate oligonucleotide PCR primers, based on conserved sequences within the IL-8 receptors and other 7TMsGPCRs. Neote *et al* (1993a) and Gao *et al* (1993), subsequently cloned a receptor, CC-CKR1, from U937 cells and PMA differentiated HL60 cells, which bound MIP-1 α , MIP-1 β , RANTES and MCP-1

Table 3.2 : The Human Chemokine Receptor Family and their Ligands

<u>Receptor</u>	Ligands that Bind		
CCR1	MIP-1 α , RANTES , MIP-1 β , MCP-3, MCP-2		
CCR2	MCP-1, MCP-2, MCP-3, MCP-5		
CCR3	Eotaxin, RANTES, Eotaxin 2, MCP-2 ?, MCP-4 ?		
CCR4	TARC, MDC (MIP-1 α , RANTES, MIP-1 β , MCP-1)		
CCR5	RANTES, MIP-1 α , MIP-1 β		
CCR6	MIP-3a		
CCR7	ΜΙΡ-3β		
CCR8	I-309		
D6	MIP-1 α , MIP-1 β , MCP-1, 2, 3 and 4, Eotaxin, HCC-1 RANTES		
CXCR1	IL-8		
CXCR2	IL-8, Gro-α, Gro-β, Gro-γ, NAP-2, ENA-78, GCP-2		
CXCR3	IP-10, Mig		
CXCR4	SDF-1		

CX₃CR1 Fractalkine

• •

with various affinities and this became the first CC chemokine receptor. Many human and murine receptors have been cloned since and the nomenclature was changed in 1996, to accommodate these. Receptors are now prefixed with CXC or CC, depending on whether they bind CC or CXC chemokines. Thus, IL-8R_A and IL-8R_B became CXCR1 and CXCR2 respectively, whilst CC-CKR1 became CCR1. It is now evident that many CC and CXC receptors bind multiple ligands, with varying affinities. To date no CC chemokine has been reported to bind to a CXC receptor and vice versa, with the exception of the Duffy antigen receptor (discussed in section 3.8) and certain virally encoded proteins. The assignation of ligands is therefore complex and made further so by differences between species. To simplify this, the known human CC and CXC chemokine receptors and their known ligands have been summarised in table 3.2.

3.7.2 The 7TMsGPCR superfamily

The chemokine receptors identified to date are all members of the 7TMsGPCR superfamily that also includes receptors for other proinflammatory mediators such as C5a, as well as the best characterised member, the β 2-adrenergic receptor. The proposed structure of one such chemokine receptor, CCR1, is shown in figure 3.1. Binding of the cognate ligand results in activation of heterotrimeric G protein complexes, followed by dissociation of the α subunit from the $\beta\gamma$ subunit which leads to activation of phospholipase C (PLC) β 1 and β 2. PLC activation then results in the generation of the second messengers diacylglycerol (DAG) and inositol (1,4,5)triphosphate (IP₃). These secondary messengers then trigger a signalling cascade, in which DAG activates protein kinase C, whilst IP3 mobilises calcium from intracellular stores in the cytoplasm and ultimately gives rise to responses such as chemotaxis, degranulation and the respiratory burst. Many different forms of G protein subunit exist and fall into two classes, the G_q class includes $G\alpha_q$, $G\alpha 11$, $G\alpha 14$, $G\alpha 15$ and $G\alpha 16$ (Simon et al 1991) and the G_i class that includes G_i2 and G_i3. The Gq class are all sensitive to pertussis-toxin (PTX), whilst the G_i proteins are PTX-insensitive and appear to directly activate PLCB2 through the Gi-GBy-PLCB2 pathway via release of Gβγ. In general chemokine receptors are PTX-sensitive, though there are exceptions such as the IL-8 receptors which can function in a PTX-insensitive manner

Figure 3.1: Structure of CCR1: A 7 transmembrane spanning G-protein coupled receptor



(Wu *et al* 1993). Research is currently ongoing to further understand the interaction between chemokines, their receptors and the different G proteins. Recent work by Kuang *et al* (1996), suggests that the CC receptors show different specificities in coupling G α class proteins (unlike the CXC receptors) and implies that they may have distinct functions *in vivo*, a notion supported by the fact that there is obvious differential expression of G α proteins.

3.7.3 Receptors for the CC chemokines

3.7.3.1 CCR1

As can be seen from table 3.2 CCR1 is a receptor for multiple chemokines. In general it appears to be a receptor for MIP-1 α and RANTES (Neote *et al* 1993a), although MCP-2 and MCP-3 are functional ligands too (Ben-Baruch et al 1995). CCR1 appears to be expressed on a wide variety of cell types, including primary monocytes/macrophages and cell lines, neutrophils, B and T lymphocytes, eosinophils and astrocytes (Neote et al 1993a, Post et al 1995, Tanabe et al 1997). A couple of studies have attempted to address the in vivo function of CCR1, through generation of CCR1 homozygous null mice. In the first of these, by Gao et al (1997), the mice were phenotypically normal. There was evidence however, for disordered steady state, in vitro induced trafficking and proliferation of myeloid progenitors in CCR1^{-/-} mice. Furthermore, mature neutrophils from CCR1^{-/-} mice failed to under chemotaxis in vitro in response to MIP-1 α . These results suggest that CCR1 has a role in haemopoiesis and inflammation in vivo. The haemopoietic abnormalities are surprising in the light of the normal haemopoiesis shown by MIP-1 α^{-1} mice (Cook et al 1995) and studies from our laboratory that suggest CCR1 is not the receptor through which MIP-1 α inhibits stem cell proliferation (Graham et al 1996). One reason for these apparent discrepancies is that the study by Gao et al used an in vitro stimulus (LPS) to assess the trafficking and proliferation of myeloid progenitor cells. Therefore this does not prove that CCR1 is critical for in vivo control of myeloid progenitor cell trafficking and proliferation. In another study, performed by Gerard and collegues using CCR1 deficient mice, CCR1 was shown to be important for progression of inflammation (Gerard et al 1997). The authors reported that deletion of CCR1 was associated with

protection from Pancreatitis-associated lung injury and moreover that this was associated with decreased levels of TNF- α .

3.7.3.2 CCR2

The CCR2 receptor was first cloned by Charo *et al* (1994) from the MonoMac6 cell line. The receptor appears to exist in two forms, CCR2A and CCR2B, generated by alternative splicing of a single gene. They appear to differ only in their C-terminal tails, although CCR2A is not as well expressed on the cell surface. CCR2B appears to be an MCP-specific receptor and does not bind either MIP-1 α , MIP-1 β , or RANTES (Myers *et al* 1995). Through the use of truncation mutants and site-directed mutagenesis, the Charo group has been very productive in determining regions involved in interactions between CCR2 and in particular, MCP-1 (Monteclaro and Charo 1996, Monteclaro and Charo 1997, Wong *et al* 1997, Arai *et al* 1997). Most recently, a non-redundant role for CCR2 in macrophage recruitment and host defence against bacterial pathogens was reported (Kurihara *et al* 1997, Kuziel *et al* 1997)

3.7.3.3 CCR3

CCR3 is a receptor with a more restricted expression pattern. It appears highly expressed on eosinophils (Kitaura *et al* 1996) and basophils (Uguccioni *et al* 1997). Limited expression has also been reported on elicited macrophages and neutrophils (Gao *et al* 1996), although the functional relevance of this is questionable. Human CCR3 is very specific in binding only eotaxin, MCP-3, MCP-4, and RANTES, all of which are eosinophil chemoattractants. This further reinforces the notion of CCR3 being an eosinophil/basophil specific receptor. However, a very recent study suggests a wider role in the allergic response, in that CCR3 appears to be expressed on T_h2 cells and moreover, that eotaxin was able to stimulate increases in intracellular calcium and chemotaxis of CCR3⁺ T cells (Sallusto *et al* 1997). This finding is particularly interesting, because it suggests that eotaxin can attract eosinophils, basophils and T_h2 cells, three of the most important cells for development of an inflammatory allergic response.

3.7.3.4 CCR4

Human CCR4 was originally cloned from a human immature basophilic cell line (Power *et al* 1995) and was found to be expressed strongly in the thymus, as well as in T and B lymphocytes, peripheral blood monocytes and in IL-5 stimulated basophils. Originally, human CCR4 was reported to be a functional receptor for MIP- 1α , RANTES and MCP-1, however it is now uncertain whether this is indeed true. The murine homologue was subsequently cloned (Hoogewerf *et al* 1996) and shown to have similar binding characteristics and tissue expression as its human homologue. Imai and collegues have demonstrated that the T cell directed CC chemokine TARC, was a specific ligand for CCR4 and did not bind to any other known receptors (Imai *et al* (1997). More recently this group reported that macrophage-derived chemokine (MDC) was also a functional ligand for CCR4 Imai *et al* (1998).

3.7.3.5 CCR5

CCR5 was first cloned by Samson *et al* (1996a) and subsequently by Raport *et al* (1996) and Combadiere *et al* (1996). CCR5 is abundantly expressed on macrophages and T cells and appears to be a relatively specific receptor, binding only MIP-1 α , MIP-1 β and RANTES (Raport *et al* 1996). Recently, Samson *et al* (1997) demonstrated that a region within the second extracellular loop of CCR5 was critical for high affinity binding of these ligands and subsequent functional response. CCR5 is now the focus of intense study, as a result of its recent identification as a co-factor for HIV entry and this is discussed further in section 3.9.

3.7.3.6 Other CC Receptors

CCRs 1-5 are the most studied CC chemokine receptors to date, but several other receptors have recently been identified. CCR6 appears to show relatively restricted expression in B and T lymphocytes and was strongly expressed in CD34⁺- derived dendritic cells, but not on monocyte-derived ones, nor on monocytes, granulocytes or NK cells (Baba *et al* 1997, Greaves *et al* 1997). CCR7 was recently reported as a receptor for the novel chemokine MIP-3 β or ELC (Yoshida *et al* 1997) and is strongly expressed in B and T lymphocytes (Birkenbach *et al* 1993). The receptor for I-309, a human CC chemokine was recently identified and named CCR8 (Tiffany *et al* 1997, Stuber Roos *et al* 1997). D6 is a recently characterised receptor,

cloned in our laboratory (Nibbs *et al* 1997). It is unusual in having an altered DRYLAIV motif in the second intracellular loop, which is normally highly conserved and which appears critical for signalling. Murine D6 is the highest affinity receptor for MIP-1 α identified to date. The human homologue of this was recently cloned (Nibbs *et al* 1998) and is also a high affinity receptor for MIP-1 α , and also binds MIP-1 β , MCPs 1-4, eotaxin and HCC-1. Unusually however, D6 does not appear to flux calcium in HEK293 cells. The functional significance of this and the other unusual characteristics of D6 are unclear at present.

3.7.4 Receptors for the CXC Chemokines

As indicated in section 3.7.1, CXCR1 is a receptor specific for IL-8, whilst CXCR2 is also a receptor for IL-8, as well as for other CXC chemokines. Recently, expression of CXCR2 was demonstrated on subsets of neurons within the CNS (Horuk *et al* 1997). Loetscher *et al* (1996) cloned the receptor (CXCR3) for IP-10 and Mig, that appears to be mainly expressed in activated T cells and which does not bind to any of the other known CXC chemokines. Murine CXCR4, cloned by Nagasawa *et al* (1996b) was shown to be the receptor for SDF-1 and the murine homologue of the human fusin HIV entry cofactor identified by Feng *et al* (1996). The role of CXCR4 in HIV entry will be discussed in section 3.9. CXCR4, like CCR2, appears to be expressed in two forms (CXCR4A and B) derived from a single gene by alternative splicing (Heesen *et al* 1996).

3.8 The Duffy Antigen for Chemokines (DARC)

One of the most unusual identified chemokine receptors is DARC. The Duffy antigen is an erythrocyte cell surface receptor for the malarial parasite, *plasmodium vivax* (reviewed by Hadley and Peiper 1997). Surprisingly, a promiscuous chemokine receptor that bound both CXC and CC chemokines (with the exception of MIP-1 α) was also identified on erythrocytes (Neote *et al* 1993b) and subsequently shown to be identical to the Duffy antigen (Horuk *et al* 1993). DARC as it was named, is also expressed on the endothelial cell lining some postcapillary venules and on Purkinje cells in the cerebellum of the brain (Horuk *et al* 1996). The precise function of DARC *in vivo* is still unclear, as it does not appear to present chemokines in an active form to receptors, nor does it appear to signal. It may instead function to soak up excess chemokines from the circulation. Its strong conservation between species and the observation that Duffy blood group negative individuals (who are malaria resistant) do still express Duffy on endothelial cells of postcapillary venules, suggests that it does serve a critical role *in vivo*. Very recently, it was reported by Rot and his colleagues, that IL-8 is actually internalised by venular endothelial cells (EC) and transcytosed to the luminal surface where it is presented to adherent leukocytes on the membrane of EC (Middleton *et al* 1997). It is possible that DARC plays a role in this chemokine transcytosis and presentation.

3.9 Chemokines, their Receptors and Significance to HIV Pathogenesis

Beginning in late 1995, the fields of chemokine and AIDS research have rapidly merged, in what has already become one of the most significant breakthroughs in AIDS research over the last 15 years. The review by D'Souza and Harden (1996) provides an excellent summary of this. The following section briefly overviews some of the reports that led to the discovery of the critical role of chemokines and their receptors in HIV pathogenesis.

One of the great elusive mysteries of AIDS research has been the missing cofactor / receptor required for entry of HIV-1 into cells in addition to the previously identified CD4 molecule (Dalgleish et al 1984). Cocchi et al (1995) demonstrated that MIP-1 α , MIP-1 β , RANTES were three soluble factors released by CD8⁺ T cells, that suppressed the entry of primary strains of HIV-1 and provided the first clues to solving this puzzle. In 1996, Feng et al made a seminal discovery and identified the second receptor necessary for HIV-1 entry into T cell lines, so called T-tropic viruses. They named this protein fusin, because it aided HIV fusion (Feng et al 1996). Fusin is identical to CXCR4, the receptor for SDF-1 (Bleul et al 1996b), and was only permissive for entry of T-tropic strains and not for macrophage-tropic (M-tropic) viruses. Subsequently, five groups simultaneously reported that the cofactor for Mtropic viruses was the CCR5 receptor and that CCR2b and CCR3 also functioned as cofactors, though less effectively (Dragic et al 1996, Deng et al 1996, Alkhatib et al 1996, Doranz et al 1996, Choe et al 1996). It is now clear that during infection some viruses evolve to become dual-tropic and can utilise both CC and CXC receptors (Doranz et al 1996), whilst some strains of HIV-2 can use CXCR4 in a CD4 independent fashion (Endres et al 1996). Intense study is now directed at determining critical interactions between different regions of CD4, HIV and various receptors. It now appears that the gp120 protein of HIV induces a complex between CD4, gp120 and CXCR4 (Lapham *et al* 1996). Others have shown similar interactions for CCR5, in which gp120 interacts directly with CCR5, whilst CD4 appears to promote these interactions (Wu *et al* 1996). The V3 domain of HIV gp120 appears critical for this interaction (Cocchi *et al* 1996).

It has been demonstrated that receptor signalling and internalisation are not required for CCR5 to function as an HIV-1 entry co-factor (Gosling *et al* 1997, Farzan *et al* 1997a, Aramori *et al* 1997). Interestingly, Weissman *et al* (1997) have suggested that recombinant M-tropic HIV and SIV envelope proteins can induce a signal through CCR5 on CD4⁺ T cells, which mediates T cell chemotaxis. Davis *et al* (1997) recently reported that chemokines and HIV-1 envelope glycoproteins induced tyrosine phosphorylation of the protein tyrosine kinase, Pyk2 (Davis *et al* 1997). This response required CXCR4 and CCR5 to be accessible at the cell surface. The significance of these intracellular signalling events are unclear at present.

Several reports have established a genetic correlation of protection. Paxton et al (1996) reported that the CD8⁺ T cells of individuals who remained uninfected, despite multiple high-risk exposures, had greater antiviral activity than those of unexposed controls. Their CD4⁺ T Cells also appeared less susceptible to M-tropic viral infection. This protection appeared to relate to the activity of the suppressive factors MIP-1 α , MIP-1 β , and RANTES. Several groups subsequently demonstrated a genetic basis for protection of multiply-exposed high-risk individuals against HIV-1 infection. Some exposed, uninfected individuals were shown to have a homozygous defect in their CCR5 gene. This was the result of a 32 bp deletion in the CCR5 allele, resulting in expression of a truncated CCR5 protein, that is not expressed on the cell surface (Samson et al 1996b, Liu et al 1996). A heterozygous allelic defect may or may not provide a limited protective effect (Dean et al 1996, Huang et al 1996). It is clear however, that the protective role of the CCR5 deletion allele is not absolute. Although rare, individuals who are homozygous for this defect can be infected (Biti et al 1997), perhaps if the original infecting virus evolves to utilise CXCR4 (Michael et al 1997a). Reports have also identified allelic variants of CCR2, though the existence of any protective role is unclear (Michael et al 1997b, Smith et al 1997).

As well as genetic variants in cofactor receptors, the role of endogenous suppressive chemokines in protection against HIV is crucial. Unfortunately determination of their roles is not so simple, whilst it is clear that SDF-1 can suppress T-tropic infection via CXCR4 (Bleul et al 1996b) and the CC chemokines MIP-1a, MIP-1ß and RANTES suppress M-tropic infection (Cocchi et al 1996), it is unlikely that these are the only suppressive chemokines (Moriuchi et al 1997). Indeed Pal et al (1997) recently identified another CC chemokine, macrophage derived chemokine (MDC), as being suppressive. Most recently a rare genetic variant of SDF-1 was identified (Winkler et al 1998). This polymorphism (SDF-1 3'A) was in the 3' untranslated region of SDF-1 and the homozygous variant provided a protective effect against the onset of AIDS, whilst the recessive variant was also protective, particularly during the later stages of disease progression. This protective effect was even more effective than those conferred by the CCR5 (CCR5 del32) and CCR2 (CCR2-641) receptor variants discussed previously in this section. It was speculated that this protective effect may be due to an enhanced level of SDF-1 production, though this remains to be proven.

Although CCR5 and CXCR4 appear to be the primary cofactors along with CCR2b, and CCR3 in the CNS (He *et al* 1997), other HIV entry co-receptors will no doubt be identified. Indeed, several orphan receptors have already been identified (Farzan *et al* 1997b), along with an unusual chemokine-like receptor, STRL33 that appears to function as a fusion-cofactor for both M and T-tropic HIV-1 (Liao *et al* 1997).

Clearly the field of HIV-related chemokine biology is in its infancy, but already chemokine antagonists and other strategies are being developed in attempts to produce therapeutic solutions for intervention and perhaps ultimately a cure for HIV infection.

3.10 Virally Encoded Chemokine and Chemokine Receptor Homologues

As a result of the recent interest in chemokines and viral pathogenesis, it has now emerged that many viruses encode homologues of chemokine and chemokine receptor-like genes. The first evidence for this was reported in 1994, with the identification of a chemokine receptor-like gene in the US28 open reading frame of the human cytomegalovirus (CMV). The US28 gene encodes a receptor that was functional in response to MIP-1 α , RANTES and MCP-1 (Gao and Murphy 1994). A viral homologue of the CXCR2 receptor is encoded by the ECRF3 gene of herpesvirus saimiri (Ahuja and Murphy 1993).

More recently, human herpesvirus 8 (HHV8) or Kaposi's sarcoma-associated herpesvirus (KSHV) has been shown to encode several viral homologues. vMIP-I and vMIP-II have homology to MIP-1 α , whilst the virus also encodes a viral form of IL-6, vIL-6 (Moore *et al* 1996). Arvanitakis *et al* (1997) have recently identified a KSHV encoded G protein-coupled receptor-like gene, with homology to the IL-8 receptors and ECRF3 gene. This receptor-like protein appears to bind both CC and CXC chemokines. Unusually, this homologue appears to be constitutively active and stimulated cell proliferation, suggesting it could be a potential viral oncogene. This oncogenic potential was recently confirmed, by Bais *et al* (1998), who demonstrated that signalling via this G-protein-coupled KSHV encoded receptor resulted in cell transformation, tumourigenicity and a switch to an angiogenic phenotype mediated by vascular endothelial growth factor (VEGF). vMIP-II was recently shown to function as a broad spectrum antagonist and inhibited HIV-1 entry via CCR3, CCR5 and CXCR4 (Kledal *et al* 1997), as well as being a chemoattractant for CCR3⁺ eosinophils (Bosshoff *et al* 1997).

Molluscum contagiosum virus (MSV) types I and II were recently shown to encode a chemokine-like gene. The products were able to block the chemotactic responses of monocytes to MIP-1 α , but were not chemotactic themselves. These proteins were also active as inhibitors of human progenitor cell proliferation, with even greater potency than human MIP-1 α (Krathwohl *et al* 1997).

Lalani *et al* (1997) purified a gamma interferon receptor homologue, M-T7, encoded by myxoma virus and demonstrated that it interacted with the heparin binding domains of both CC and CXC chemokines. M-T7 has no homology with any known chemokine receptors and the significance of it's apparent abilities are unclear at present.

Clearly many viruses have evolved or acquired chemokine-like and chemokine receptor-like genes. In general, viruses do not have the spare coding capacity to encode such genes without good reason, but at present the role that these play in viral pathogenesis is unknown. Further study of the virally encoded homologues may well enable further understanding of the role of chemokines and their receptors in the immuno-haemopoietic system.

3.11 In Summary

It should be evident by now, that the control of the immuno-haemopoietic system is enormously complex. Central to its regulation, are a multitude of growth factors and cytokines, together with their cognate receptors. These form an interacting network, that ultimately functions at every level from early development, through to the control of constitutive and inducible haemopoiesis. Within this network, the chemokine MIP-1 α , is a particularly important mediator. It functions as a regulator of both stem cell proliferation and inflammation. Recent studies have highlighted the importance of MIP-1 α and other chemokines to viral pathogenesis and enabled great advances in our understanding of MIP-1 α , particularly concerning interactions with its various receptors. However, little is known regarding the mechanisms that operate to regulate the expression of MIP-1 α and its receptors. This is particularly true in the context of haemopoiesis and the interaction of MIP-1 α with other growth factors within the bone marrow microenvironment. This therefore presents an exciting area for further research.

AIMS OF THESIS

In order to fully understand the complex process of haemopoiesis, it is necessary to understand the role that haemopoietic growth factors and cytokines play in regulating this process. It is now evident that a large interacting network of such regulatory cytokines exists and these have been implicated in various aspects of haemopoietic control, within the haemopoietic stem cell compartment. It is also apparent that both stimulatory and inhibitory molecules function within this network. Our group has a particular interest in one such inhibitor of haemopoietic stem cell proliferation, namely macrophage inflammatory protein-1 alpha (MIP-1 α). Current evidence suggests that MIP-1 α is member of this network and therefore in order to fully understand the role of MIP-1 α in haemopoietic stem cell inhibition, it is also necessary to further understand how its expression within the bone marrow is regulated. Recently our laboratory identified an endogenous, reciprocal relationship in bone marrow-derived macrophages between MIP-1a and transforming growth factorbeta (TGF- β), in which it appears that TGF- β acts to suppress MIP-1 α expression. Whilst the level of MIP-1 α expression within the bone marrow is very low, it is detectable and this implies that stimulatory mechanisms may exist to overcome this TGF-β mediated suppressive block. The aim of this study therefore, was to investigate the potential effects of haemopoietic growth factors on the regulation of MIP-1 α expression. It was believed probable that within the bone marrow microenvironment, positively acting factors were also functional in the regulatory control of MIP-1 α expression. Investigation of the effects of such positively acting factors and any relationship between them, MIP-1 α and TGF- β , might enable further understanding of the circumstances under which MIP-1 α expression might be altered, and thus provide further insight into its function within the haemopoietic stem cell or inflammatory cell compartments.

PART II: MATERIALS AND METHODS

-

CHAPTER 4: MATERIALS

. -

Table 4.1: Plasmids for Generation of Probe Fragments

<u>Plasmid name</u>	<u>Identity of Insert</u>	<u>Insert size</u>	Excision sites	<u>Source</u>	<u>Address</u>
pSK.β-Actin	Murine β-Actin cDNA	1200 bp	Pst I	D. Jarmin	Beatson Institute
pSK.LD78	Human MIP-1α cDNA	~300 bp	Pst I	S. Wylie	Beatson Institute
pMTX-MIP-1a	Murine MIP-1a cDNA	~300 bp	EcoR I / Xho I	S. Wylie	Beatson Institute
pMTX-MIP-1β	Murine MIP-1β cDNA	~300 bp	EcoR I / Xho I	S. Wylie	Beatson Institute
pSK.mCCR1	Full murine CCR1 cDNA	1346 bp	BamH I / Not I	Dr. R. Nibbs	Beatson Institute
pSK.mCCR3	Full murine CCR3 cDNA	1392 bp	BamH I / Not I	Dr. R. Nibbs	Beatson Institute
pSK.mCCR4	Full murine CCR4 cDNA	1364 bp	BamH I / Not I	Dr. R. Nibbs	Beatson Institute
pSK.mCCR5	Full murine CCR5 cDNA	1095 bp	BamH I / Not I	Dr. R. Nibbs	Beatson Institute
pSK.mD6	Full murine D6 cDNA	1621 bp	BamH I / Not I	Dr. R. Nibbs	Beatson Institute
pMARC	Full murine MCP-3 (MARC) cDNA	776 bp	BamH I / Xho I	Dr. T. Baumruker	Novartis F.I. (Vienna, Austria)
pcJE-1	Full JE cDNA	578 bp	EcoR I	Dr. B. Rollins	(Dana-Farber Cancer Institute MA, USA)
pcDNA-3 hCCR1	Full human CCR1 cDNA	1065 bp	Hind III / Not I	Dr. T. Schall	MMRI (Palo-Alto, CA, USA)

Table 4.2: Plasmids for Use in RT-PCR Analysis

Plasmid name	Identity of Insert	<u>Source</u>	Address
pCDM8.mIL-5Rα	Full murine IL-5R-α chain cDNA	Dr. G. Plaetnick	Roche Research (Gent, Belgium)
pCRII.rGAPDH	Rat glyceraldehyde-3- phosphate dehydrogenase (GAPDH) cDNA (containing sequences between bp 300 - 1020)	M. Walker	Beatson Institute

pSK.mCCR1 pSK.mCCR3 pSK.mCCR4 pSK.mCCR5 pSK.mD6

~.

Details as per Table 4.1

79

Table 4.3: PCR Primers- Identity

. .

<u>Name</u>	Orientation	<u>Amplifies</u>	Comments
DJ03	5'	Murine IL-5Rα	Amplifies both membrane bound and
DJ04	3'		soluble IL-5R α chain fragments
DJ09	5'	Murine MIP-1a	Crosses 1 st Intron/exon boundary
DJ10	3'		(247 bp mRNA, ~1 kb genomic)
DI11	5,	Murine GAPDH	
DII	·		
DJ12	3		
DJ17	5'	Human MIP-1α	
DJ18	3'		
DJ23	5'	Murine D6	
DJ24	3'		
DJ25	5'	Murine CCR1	
DJ26	3'		
DJ27	5'	Murine CCR3	
DJ28	3'		
DJ29	5'	Murine CCR4	
DJ30	3'		
DJ31	5'	Murine CCR5	
DJ32	3'		
DJ33	5'	Murine CCR2	
DJ34	3'		
RNPCR.βA5	5'	Murine and human	Mouse mRNA= 247 bp
RNPCR.BA3	3'	β-Actin	
		-	

. -

DJ03	5'	GCCATTGACCAAGTGAATCC	3'
DJ04	5'	GTGGAATTTCCCATGACTTC	3'
DJ09	5'	ATGAAGGTCTCCACCACTGCC	3'
DJ10	5'	TCAGGCATTCAGTTCCAGGTC`	3'
DJ11	5'	TGAGTATGTCGTGGAGTCTAC	3'
DJ12	5' · ·	GGCCATGTAGGCCATGAGGTC	3'
	-		
DJ17	5'	CACTIGCIGCIGACACG	3'
DJ18	5'	CAACCAGTCCATAGAAGAGG	3'
D123	5'	GTTTTCTTCATGCTGTGGTTCC	3,
D123	5,	GGAGAGAGAGAGTAATGAGTAAGGC	37
DJZ4	5	COARCACACITATOR TRADUC	
DJ25	5'	CTTCTATTCTTCCTCCTCTGGAC	3'
DJ26	5'	GAGCTCATGTTCTCCTGTGGA	3'
DJ27	5'	CTTGCAGGACTGGCAGCATTG3'	
DJ28	5'	CAAAAATAAGACGGATGGCCTTG	3,
DJ29	5'	GAATGAGAAGAAGAACAGAG	3'
DJ30	5'	CTGGACATGTCAGCCGAGTAGAC	3'
DJ31	5'	CTTTACCAGATCTCAGAAAGAAGG	3'
DJ32	5'	GAGTCTCTGTTCCTGCATGG	3'
D133	5'		3,
D134	5,	GAAATGTTGGGTTGGCTC	3'
L/JJ4	5		5
RNPCR.6A5	5'	TCCATCATGAAGTGTGACGT	3'
RNPCR RA3	- 5'	ТАСТССТӨСТТӨСТӨАТССАС	3,
in or hus	5	moreorgorigorionicone	-

4.5 Tissue Culture Supplies

· · · · · · · · · ·

-.

Supplier	<u>Address</u>	
Beatson Institute	(Glasgow, UK)	Sterile phosphate buffered saline (PBS)
Central Services		Sterile glassware and pipettes
R2, Beatson Institute	(Glasgow, UK)	L929 CM
Becton Dickinson, UK Ltd.	(Plymouth, UK)	Falcon tubes (15 ml and 50 ml)
Costar	(Cambridge, MA	6 and 24 well culture plates
	USA)	Cell scrapers
Gelman Sciences	(Northampton, UK)	Sterile acrodisc syringe filters
		(0.2 µm and 0.45 µm)
Gibco Life Technologies	(Paisley, UK)	MEM alpha stock
		L-glutamine (200 mM)
		Sodium pyruvate (100 mM)
		Sodium bicarbonate (7.5 %)
		MEM Non essential amino acids (100x)
		Hepes (1 M)
		Trypsin (2.5 %)
Greiner Labortechnik	(Stonehouse, UK)	Endotoxin-free plugged pipetteman tips
Nalge Nunc International	(Roskilde, Denmark)	Tissue Culture Flasks
		Cryotubes
Sigma Chemical Co.	(Poole, UK)	Endotoxin-free Dulbecco's PBS (10 x)
		2-Mercaptoethanol (2mM stock in PBS)
		Donor Horse Serum
		Bovine Serum Albumin, FractionV, 7.5%
		RPMI-1640 Medium
		Dulbecco's MEM
TCS Biologicals Ltd.	(Buckingham, UK)	Foetal Calf Serum

.
4.6 Cytokines and Antibodies

-

The murine MIP-1 α PM2 deletion mutant was the form of MIP-1 α used in these studies and was as described in Graham *et al* (1994).

<u>Supplier</u>	Address	
R&D Systems Europe	(Abingdon, UK)	Recombinant anti-murine MIP-1α antibody Recombinant human TGF-β1 Recombinant murine GM-CSF Recombinant human GM-CSF Recombinant murine IL-3 Recombinant murine IL-5 Recombinant murine G-CSF
		Recombinant murine IL-6
Sigma Immunochemicals	(Poole, UK)	Anti-goat IgG (HRP-conjugated)
Dr. N. Hole	(University of Durham, UK)	COS-cell derived murine LIF
4.7 Kits		
Supplier	Address	
Amersham International	(Amersham, UK)	ECL Western blotting detection kit
Invitrogen BV	(Leek, The Netherlands)	Micro-FastTrack mRNA isolation kit
Perkin Elmer	(Foster City, CA, USA)	GeneAmp® RNA PCR core kit
Pharmacia Biotech	(St Albans, UK)	Ready-To-Go DNA labelling kit
Qiagen Inc.	(Chatsworth, CA, USA)	QIAGEN plasmid preparation kits QIAquick gel extraction kit

4.8 Membranes, Paper and X-ray Film

. -

Supplier	Address	· · · ·
Amersham International	(Amersham, UK)	Hybond N nylon membranes
Genetics Research, Ltd	(Felstead Dumnow, UK)	Saranwrap
Millipore Corp.	(Bedford, USA)	Immobilon-P transfer membranes
Vernon-Carus, Ltd.	(Preston, UK)	Gauze swabs
Whatman International Ltd.	(Maidstone, UK)	3MM blotting paper
Kodak Scientific Imaging Sytems Ltd.	(Cambridge, UK)	X-OMAT AR X-ray film
Technical Photo Systems	(Cumbernauld, UK)	Fuji RX Medical X-ray film
4.9 Nucleotides		
Supplier	Address	
Amersham International	(Amersham, UK)	[α- ³² P]-dCTP: 3000 Ci/mmol
Dupont NEN	(Stevenage, UK)	¹²⁵ Iodine: 100 mCi/ml
4.10 Protein and DNA Mar		
	kers	
<u>Supplier</u>	kers <u>Address</u>	
<u>Supplier</u> Amersham International	kers <u>Address</u> (Amersham, UK)	Rainbow coloured protein molecular weight markers: (High m.w. : 14,300-220,000 Da) (Low m.w. : 2,300-46,000 Da)

Low DNA mass ladder 123 bp DNA ladder 1 Kb DNA ladder

4.11 Enzymes, Enzyme Buffers and Enzyme Inhibitors

<u>Supplier</u>	Address	
Gibco BRL	(Paisley, UK)	All Restriction enzymes and buffers
		(unless stated otherwise)
		T4 DNA ligase (1 U/μl)
		5x ligation buffer
Boehringer Mannheim	(Mannheim, Germany)	RNase free DNase I (10 U/µl)
Invitrogen BV	(Leek, The Netherlands)	RNaseZap (RNase inhibitor)
Pharmacia Biotech	(St Albans, UK)	One-phor-all plus [™] restriction enzyme buffer

4.12 Columns

-.

<u>Supplier</u>	Address	
Pharmacia Biotech	(St Albans, UK)	NICK columns

4.13 Chemicals

Supplier	<u>Address</u>	
Boehringer Mannheim	(Mannheim, Germany)	MOPS
DIFCO Laboratories	(Michigan, USA)	Bactoagar
Fisons Scientific Equipment	(Loughborough, UK)	Formaldehyde (38 % w/v) Glycerol

GIBCO BRL	(Paisley, UK)	Agarose TRIzol reagent Phenol/Chloroform/Isoamyl alcohol (25:24:1, v/v) (pH 8.0)
James Burroughs Ltd.	(Witham, UK)	Ethanol
Pierce and Warriner	(Chester, UK)	Iodogen reagent
Severn Biotech Ltd.	(Kidderminster, UK)	Design-a-gel 30 % (w/v) Acrylamide, 0.8 % (w/v) bis-Acrylamide solution
Sigma chemical Co.	(Poole, UK)	Ampicillin Bromophenol blue Diethylpyrocarbonate (DEPC) Dithriothreitol (DTT) Ethidium bromide DNA samples (10mg/ml) RNA samples (1mg/ml, in DEPC treated water) Fura-2-AM TEMED Tween 20
Thornton and Ross	(Huddersfield, UK)	Liquid Paraffin
Fisher Scientific UK Ltd.	(Loughborough, UK)	All other chemicals not listed above
4.14 Bacterial Culture		
4.14.1 Solutions		
Supplier	Address	
Beatson Institute	(Glasgow, UK)	Luria-Broth

. -

Central Services

4.14.2 Host strains

. .

<u>Supplier</u>	Address	
GIBCO BRL	(Paisley, UK)	DH5a competent cells
4.15 Solutions		
DEPC-water		0.1 % DEPC in sterile distilled water,
		shaken and left overnight at room temperature. Autoclaved and allowed to cool.
4.15.1 Agarose gel electrophores	is	
1x TE (pH 8.0)		10 mM Tris.HCl (pH 8.0) 1 mM EDTA
50x TAE		2 M Tris-acetate 5 mM EDTA (pH 8.0)
DNA sample loading buffer		50 % Glycerol (v/v)
		1x TAE Bromophenol blue to colour
4.15.2 DNase I digestion		
DNase I restriction buffer (10x)		50 mM Sodium acetate (pH 5.2) 1 M Magnesium chloride made up in DEPC-water
4.15.3 Northern blotting and hyb	oridisation	
10x MOPS buffer (pH 7.0)		0.4 M MOPS (pH 7.0) 0.1 M Sodium acetate 10 mM EDTA
RNA sample loading buffer		50 % Glycerol (v/v) 1x MOPS

Bromophenol blue to colour
made up in DEPC-water
3 M Sodium chloride
0.3 M Sodium citrate
0.25 M Sodium phosphate (pH 7.0)
7 % SDS (w/v)

4.15.3 SDS-Polyacrylamide Gel Electrophoresis (SDS/PAGE) and Western blotting

. .

17.5 % Acrylamide mix (w/v)
375 mM Tris.HCl (pH 8.8)
0.1 % SDS (v/v)
0.1 % Ammonium persulphate (w/v)
0.04 % TEMED (v/v)
5 % Acrylamide mix (w/v)
125 mM Tris.HCl (pH 6.8)
0.1 % SDS (w/v)
0.1 % Ammonium persulphate (w/v)
0.01 % TEMED (v/v)
5 μl 10 % SDS / DTT (sample buffer)
7 μl 50 % glycerol (with Bromophenol
blue to colour)
200 mM Glycine
25 mM Tris-acetate
0.1 % SDS (w/v)
40 mM Glycine
50 mM Tris.HCl
0.04 % SDS (w/v)
20 % Methanol (v/v)

Binding buffer

1x MEM alpha medium:
[20 % MEM alpha stock (v/v),
1 mM L-Glutamine,
0.1 % Sodium bicarbonate (v/v),
25 % DHS (v/v),
20 % L-929 CM (v/v)].
25mM HEPES (pH 7.4)
0.2 % Sodium azide (w/v)

4.15.5 Calcium mobilisation studies

HACM buffer

. .

125 mM Sodium chloride
5 mM Potassium chloride
0.025 % BSA, fraction V (w/v)
1 mM Magnesium chloride
20 mM HEPES (pH 7.4)
0.5 mM D-Glucose

CHAPTER 5: METHODS

5.1 Mice

Female B6D2F1 mice, aged between 4-6 weeks at arrival, were used for all experimental procedures. Mice were purchased from Harlan UK, Ltd. (Oxon, UK) and housed within the animal facility at the Beatson Institute.

5.2 Cell Lines

Cell lines were obtained from the sources indicated in table 5.1. All cell lines were maintained at 37° C in a dry atmosphere of 5% CO₂ in air.

5.2.1 THP-1

The human pro-monocytic cell line THP-1, derived from a patient with acute monocytic leukaemia (Tsuchiya *et al* 1980), was maintained in RPMI 1640 + 2 mM glutamine / 0.02 mM β -mercaptoethanol / 10 % FCS in Nunc 75 cm² tissue culture flasks. THP-1 cells grow in suspension and were subcultured every 2-3 days, by resuspension in fresh medium at a concentration of 2-9 x 10⁵ cells/ml.

5.2.2 BCL1 (clone 13.20-3B3)

The murine chronic B-cell leukaemia cell line BCL1 clone 13.20-3B3, derived from the BCL1 tumour cell line (Gronowicz *et al* 1980) was maintained in RPMI 1640 + 10 mM HEPES / 2 mM glutamine / 0.1 mM non-essential amino acids / 1 mM Sodium pyruvate / 0.05 mM β -mercaptoethanol / 10 % FCS in Nunc 75 cm² tissue culture flasks. BCL1 cells are a loosely adherent cell line and were subcultured upon reaching 75-90 % confluence (approximately every 2-3 days), by sharply tapping the flask against the palm of the hand 3-4 times. The detached cells were then resuspended in fresh medium at a subcultivation ratio of 1:3 to 1:5.

5.2.3 HEK 293-hCCR5 stable transfectants

Human embryonic kidney (HEK) 293 cells stably transfected with human CCR5 were maintained in DMEM + 4 mM glutamine / 10 % FCS in Nunc 175 cm^2 tissue culture flasks. Cells were subcultured every 2-3 days, upon reaching 75-90 %

Table 5.1: Tissue Culture Cell Lines

-

Supplier	Address	<u>Cell line</u>	<u>Type of Cell line</u>
European Collection of Animal	(Salisbury,	THP-1	Human Monocytic
Cell cultures (ECACC)	UK)		leukaemia
American Type Culture	(Rockville,	BCL1 (clone	Murine B-cell
Collection (ATCC)	MD, USA)	13.20-3B3)	leukaemia
Dr. R. J. Nibbs	(Beatson Institute)	HEK hCCR5	Human Embryonic kidney cells, stably transfected with hCCR5

Table 5.2: Analysis of Bone Marrow-Derived Macrophage Population

Day	Percentage of non-specific
	Esterase Positive Cells
4	82 %
5	96 %
6	98.5 %
7	99 %
8	100 %

confluence, by detaching with trypsin and resuspending in fresh medium at a concentration of 5×10^5 to 1×10^6 cells/ml.

5.3 Bone Marrow-Derived Macrophages

5.3.1 Harvesting of bone marrow

4-6 wk old female B6D2F1 mice were used throughout for the harvesting of bone marrow. Mice were killed by cervical dislocation. The femora were excised using surgical scissors and excess muscle removed from the bone shaft and ends using a tissue soaked in ethanol. Both ends of the femora were then cut off with scissors and the bone marrow cells expelled by flushing through a 21 gauge needle with 2.5 ml per femur of α -modified MEM supplemented with 25 % DHS. Approximately 6-9 x 10⁶ bone marrow cells were routinely obtained from each femur using this method.

5.3.2 Derivation of bone marrow-derived macrophages

Bone marrow cells were obtained (as above) and resuspended at 5×10^4 /ml in α -modified MEM supplemented with 25 % DHS and 20 % L929 CM as source of M-CSF. Following incubation for 7 days at 37°C in a dry atmosphere of 5 % CO₂, the remaining adherent cells consisted of a homogeneous (>99 %) population of bone marrow-derived macrophages (BMM), as determined by staining of cytocentrifuge preparations with the macrophage-specific enzyme alpha-napthyl acetate esterase (see table 5.2). These BMM cells were proliferating as assessed by ³H incorporation analysis. Day 7 was selected as the optimal day for usage and unless stated otherwise, all procedures using BMM were performed using this time point.

5.4 Stimulation of Cell lines and Bone Marrow-Derived Macrophages

All reasonable precautions were taken to ensure that contamination of experiments with endotoxin / lipopolysaccharide (LPS) did not occur. These included purchase of endotoxin-free recombinant cytokines and antibodies, as well as the usage of sterile plastic tissue culture pipettes and aerosol resistant plugged pipetteman tips. All recombinant cytokines and antibodies were reconstituted according to the manufacturers guidelines in entoxin free PBS / 0.1 % BSA (v/v). Cells were

stimulated as indicated in the figure legends accompanying each figure, with either recombinant cytokines/antibodies or with PBS / 0.1 % BSA (v/v) for negative controls.

5.5 DNA Methodology

5.5.1 Transformation of bacterial host cells with plasmid DNA

Plasmid constructs were obtained from the sources indicated in tables 4.1 and 4.2, and commercially available DH5 α competent cells were transformed with the appropriate plasmid construct. Briefly, DH5 α cells were removed from -70 °C and thawed on ice. 50 µl aliquots of thawed cells were dispensed into pre-chilled Falcon 2059 tubes, 1 µl of plasmid construct DNA (10-50 ng/µl) added and the mixture very gently swirled, before incubating on ice for 30 minutes. The cells were then heat shocked at 42 °C for exactly 45 seconds, returned to ice for 2 minutes and then 90 µl of Luria broth added. This mixture was then shaken at 225 rpm for 45 minutes at 37 °C, to permit sufficient time for the ampicillin resistance gene to be expressed. 100 µl and 50 µl of transformed cells were then spread onto prewarmed LB-Amp plates (6.5 g of Bactoagar in 400 ml Luria Broth, mixed and autoclaved, allowed to cool to ~37 °C. The agar mix was then supplemented with 100 µg/ml ampicillin and 10-20 ml poured into 10 cm petri dishes. Plates were then allowed to set completely). The spread plates were allowed to stand for 5-10 minutes at room temperature, then inverted and incubated overnight at 37 °C.

5.5.2 Growth of plasmids

A 25 ml sterile universal container containing 5 ml of Luria Broth plus ampicillin (100 μ g/ml) was innoculated with a single colony of cells (from LB-AMP plates prepared in section 5.5.1) using a sterile innoculating loop. This was then incubated for 8 hours at 37 °C with shaking at 225 rpm. This starter culture was then added to 100 ml of Luria Broth containing ampicillin (100 μ g/ml) in a 1 litre conical flask and incubated overnight at 37 °C with shaking at 225 rpm.

5.5.3 Plasmid preparation

Plasmids were prepared using the commercially available QIAGEN midi or maxi plasmid kits, according to the manufacturer's instructions; -briefly as follows. Bacterial suspensions (derived as in section 5.5.2) were transferred to 250 ml centrifuge bottles (Sorvall instruments) and spun at 3000 rpm for 5 minutes in a GS-3 rotor, using a Sorvall RC-5B superspeed centrifuge. The supernatants were discarded and any residual liquid drained off briefly by inversion. The bacterial pellets were then resuspended in either 4 ml (midi) or 10 ml (maxi) of P1 resuspension buffer (50 mM Tris.HCl, 10 mM EDTA pH 8.0, 100 µg/ml RNase A). Once the pellet was suspended in P1, the bottles were gently rocked on an orbital shaker to allow complete resuspension of the bacterial pellet. The suspension was then transferred to 50 ml centrifuge tubes (Sorvall instruments) and either 4 ml (midi) or 10 ml (maxi) of P2 lysis buffer (200 mM NaOH, 1 % SDS) added. Tubes were immediately capped, gently inverted 4-6 times to mix and incubated at room temperature for 5 min. 4 ml (midi) or 10 ml (maxi) of pre-chilled P3 neutralisation buffer (3 M KAc, pH 5.5) was then added, mixed gently by inverting 5-6 times and incubated on ice for 15 minutes (midi) or 20 minutes (maxi). The tubes were then spun for 30 minutes in an SS-34 rotor at 10000 rpm. The supernatant was then carefully decanted, through a double layered surgical gauze swab to remove particulate material and prevent subsequent blocking of QIAGEN columns. During the above spin, QIAGEN columns were equilibrated with buffer QBT (750 mM NaCl, 50 mM MOPS, 15 % ethanol, pH 7.0, 0.15 % Triton X-100)- 4 ml QBT for tip 100 (midi) or 10 ml QBT for tip 500 (maxi). The filtered supernatant was then applied to the column and allowed to drain through by gravity flow. The column was then washed with 2 x 10 ml (midi) or 2 x 30 ml (maxi) of QC wash buffer (1 M NaCl, 50 mM Tris.HCl, 15 % ethanol, pH 7.0). The DNA was eluted from the column with either 5 ml (midi) or 15 ml (maxi) of QF buffer (1.25 M NaCl, 50 mM Tris.HCl, 15 % ethanol, pH 8.5). The DNA was then precipitated with 0.7 volumes of isopropanol and centrifuged at 10000 rpm at 4 °C for 30 minutes as described above. The supernatant was carefully discarded and the pellet resuspended in 300 µl of distilled water. This was transferred to a 1.5 ml microfuge tube and the DNA re-precipitated with 0.1 volume of 3 M Sodium acetate (pH 5.2) and 2.5 volumes of ethanol and spun at 13500 rpm in a MSE bench-top microfuge, at room temperature for 15 minutes. The pellet was washed carefully with 500 µl of 70 % ethanol and respun at 13500 rpm for 5 minutes. The supernatent was discarded and the pellet was then air-dried for 10 minutes and resuspended in a suitable quantity of sterile distilled water. The amount of DNA present was then quantified as follows. 495 μ l of distilled water was mixed with 5 μ l of plasmid DNA and the optical density measured at 260 nm and 280 nm, using a Beckman spectrophotometer. A single OD unit at 260 nm (OD₂₆₀ = 1) is equivalent to 50 ug/ml of double stranded DNA, thus allowing the concentration of DNA present in samples to be determined with accuracy. Plasmid DNA was stored at -20 °C before usage.

5.5.4 Digestion of plasmid DNA with restriction enzymes

20-50 µg of plasmid DNA was digested in a total reaction volume of less than 100 µl, as follows. The appropriate amount of DNA to be digested was added to a 1.5 ml screw cap microfuge tube. 0.1 volume (of the total reaction volume) of the appropriate 10 x ReactTM buffer was then added. Dependent upon which restriction enzyme(s) were selected, the ReactTM buffer was chosen to give the optimal conditions for the selected enzyme. Where two different restriction enzymes were used, a compromise was reached and a buffer chosen that enabled the highest possible efficiency for both enzymes. If this was not possible, then the One-Phor-all plusTM buffer (Pharmacia, UK) was used instead. The restriction enzyme(s) was then added, making sure the volume did not exceed 0.1 volume of the final reaction volume in order to prevent inhibition of the reaction by glycerol in the enzymes. The reaction mix was then made up to the final desired reaction volume with distilled water and incubated at 37 °C for 60 minutes. Reactions were then either used immediately for downstream applications or were stored at -20 °C until use.

5.5.5 Agarose gel electrophoresis of DNA

The following protocol was used for the routine analysis of DNA derived from plasmid preparations, restriction enzyme digestion of plasmid DNA, quantification of DNA probe fragments and visualisation of products generated by polymerase chain reaction (PCR);- as follows. DNA samples were resolved on a non-denaturing agarose gel. 1-2 % agarose w/v was added to a conical flask and made up to a final volume of either 75 ml for small casting trays or 150 ml for large ones, with distilled water. The

agarose was melted for 2-3 minutes in a standard 750 W microwave, 50 x TAE buffer was then added to give a final working concentration of 1 x TAE and 2.5 μ l or 5 μ l (small and large gels respectively) of ethidium bromide (10 mg/ml) added. The agarose was then allowed to cool to the point at which it could be safely hand held, before pouring into casting trays. Gels were allowed to set for 30-60 minutes before use). DNA samples were added to 2.5 μ l of DNA gel loading buffer (see section 4.15.1) and the volume made up to 10 μ l with distilled water (for larger loading volumes the quantities were adjusted accordingly). The DNA was separated by running the gels at a constant 90 volts. The gels were then checked after 1-2 hours by visualisation of the ethidium bromide stained DNA bands on a UV transilluminator (λ =312 nm). The gel image was then recorded using an Appligene imager (Appligene Oncor, UK). For derivation of probe fragments the appropriate band(s) was excised using a sterile scalpel at this point. For Southern blotting of DNA (see section 5.7), the gel was image was recorded alongside a fluorescent ruler and then excess agarose trimmed off prior to blotting.

5.5.6 Isolation of probe fragments for generating radiolabelled probes

Plasmid DNA encoding the selected DNA probe was digested with the appropriate restriction enzymes (see section 5.5.4), resolved by agarose gel electrophoresis and the appropriately sized band of DNA excised from the gel with a sterile scalpel (see section 5.5.5). The gel fragment was collected in a 1.5 ml screw cap tube and the weight of agarose measured. The agarose was removed and the DNA eluted, using the QIAquick gel extraction kit (Qiagen). Briefly, 3 volumes (0.1 g of agarose = approximately 100 μ l = 1 volume) of extraction buffer were added to the excised band. This was heated at 55 °C for 10 minutes and the tube inverted every 2 minutes to ensure complete dissolution. This solution was loaded onto a QIAquick spin column and spun at 13500 rpm for 1 min. The flow-through was discarded and the column washed with 750 μ l of wash buffer by spinning at 13500 rpm for 1 min. The column was then spun again for an additional minute to remove traces of wash buffer. The DNA was eluted from the column by placing in a clean 1.5 ml eppendorf and adding 30 μ l of TE, pH 8.0 to the centre of the spin column. This was left for 1 minute, before the column was spun for 1 minute at 13500 rpm and the DNA

collected. The eluted DNA was visualised and quantified by agarose gel electrophoresis of 2 μ l of DNA on a 1 % non-denaturing agarose gel (see section 5.5.5) and comparison against a known quantity of low DNA mass ladder run on the same gel.

5.6 RNA Methodology

5.6.1 RNase free environment

In order to maintain the integrity of RNA samples during isolation, storage and subsequent downstream handling, stringent precautions were taken to prevent degradation by RNases. Disposable sterile plastic pipettes, plugged sterile nuclease-free aerosol resistant pipetteman tips and disposable sterile plastic tubes were used for handling of RNA samples. All such plasticware, as well as chemical reagents were designated exclusively for use with RNA. Bench top surfaces and non-disposable equipment were wiped with RNase-Zap (Invitrogen, Netherlands), a commercially available solution for inhibiting RNases. DEPC treated sterile distilled water was used for making up RNA sample loading buffer, ethidium bromide and for resuspension of RNA samples.

5.6.2 Isolation of total RNA

Cells were stimulated with appropriate growth factors (or PBS / 0.1 BSA (v/v)) and total RNA was then isolated using TRIzol[™] (Gibco Life Technologies, UK), a commercially available acid guanidinium thiocyanate-phenol-chloroform based extraction mixture (Chomcyznski *et al* 1987), according to the manufacturer's instructions. Briefly as follows. For RNA isolation from adherent cells in 75 cm² flasks, growth medium was removed and 10 ml of Trizol[™] added. For isolation of RNA from cells grown in suspension, the cells were spun down at 1000 rpm, the growth medium removed and 10 mls of TRIzol[™] added. RNA from adherently and suspension grown cells was isolated in an identical way from this point. The flask was swirled (or the tube gently mixed) to ensure complete coverage of the full cell surface area with TRIzol[™] and then left for 5 minutes at room temperature, with occasional swirling of flasks (or mixing of tubes) every 2 minutes. The solution was then pipetted

several times, using an automatic pipette aid, to ensure complete cell lysis. The solution was then transferred to a 15 ml "snap cap" Falcon 2059 tube and 2 ml of chloroform added. Tubes were shaken vigorously for 15 seconds, allowed to stand at room temperature for 2-3 minutes and then spun in a Sorvall RC-5B supercentrifuge at 3000 rpm for 15 minutes using HB4 or HB6 swing-out rotors. The resulting aqueous layer was then promptly removed to a fresh 15 ml tube and 5 ml of isopropanol added. The solution was mixed by inversion and left to stand at room temperature for 10 minutes, after which it was spun at 3000 rpm for 10 minutes. The isopropanol supernatant was discarded and the pellet resuspended in 10 ml of 70 % ethanol, to wash the RNA. Tubes were spun at 3000 rpm for 5 minutes, the ethanol wash discarded and the tubes inverted on the bench for 10 minutes to drain off excess ethanol, before air drying the pellets to remove all remaining ethanol. The RNA was resuspended in 50-200 µl of DEPC treated water, and transferred to sterile 1.5 ml screw cap tubes. The RNA was then quantified by adding 5 µl of RNA to 495 µl of distilled water and measuring the OD_{260} nm (as in section 5.5.3, except that an OD_{260} of 1 is equivalent to 40 µg/ml for RNA). RNA samples were stored at -20 °C until use.

5.6.3 Isolation of PolyA⁺ mRNA

PolyA⁺ mRNA was isolated from snap-frozen eye and leg muscle tissue samples or from frozen cell suspensions of peritoneal cavity cells and bone marrow cells derived from either wild type or GM-CSF transgenic mice, using the commercially available Micro-FastTrackTM kit (Invitrogen). Briefly as follows, a mortar and pestle was cleaned with RNase-Zap, placed on dry ice and a piece of snapfrozen tissue was placed in the mortar. The tissue was then ground in liquid nitrogen, before transferring to a clean RNase-free tube. The tissue was then resuspended in 1 ml of lysis buffer, preheated to 45 °C. For frozen cell suspensions the cell pellet was resuspended in 5 ml of lysis buffer and incubated at 45 °C until thawed, before dispersing into 5 x 1 ml aliquots. The cell suspension was then homogenised by passing through a syringe fitted with an 18-21 gauge needle. This was repeated until the entire suspension could be drawn up into the syringe. The mRNA isolation protocol was then followed according to the manufacturer's instructions. The PolyA⁺ mRNA was eluted in a final volume of 10 µl of elution buffer. 5 µl of mRNA was then Northern blotted and the membrane hybridised, as described in sections 5.6.4, 5.7 and 5.8.

5.6.4 Agarose gel electrophoresis of RNA

Total RNA was separated and visualised in a similar fashion to that used for DNA samples, the main difference being the usage of denaturing conditions. Total RNA was isolated and quantified (see section 5.6.2.) and 20 μ g of RNA was added to sterile 1.5 ml screw cap tubes and freeze-dried in a speedivac. The pellet was then resuspended in 20 μ l of RNA sample loading buffer (see section 4.15.2), denatured in a boiling water bath for 5 minutes and samples placed on ice for 2 minutes to prevent renaturation of RNA strands. Tubes were briefly spun in a microfuge and 1 μ l of ethidium bromide added (10 mg/ml stock). Samples were loaded and resolved on a denaturing agarose gel (1.4 % w/v agarose, 1 x MOPS buffer, 2.2 M formaldehyde and made up to 150 ml with sterile distilled water). Gels were run at a constant 90 volts, until the Bromophenol blue band had migrated almost 2/3 of the distance of the gel (approximately 3 hours). At this time the integrity of the RNA was checked under UV transillumination and the gel image recorded alongside a fluorescent ruler to allow subsequent sizing of RNA species (as described in section 5.5.5). Gels were then trimmed of excess agarose and Northern blotted (see section 5.7).

5.7 Southern and Northern Blotting of DNA and RNA

DNA and RNA resolved by agarose gel electrophoresis can be transferred to nitrocellulose membranes by capillary transfer. For both Southern and Northern blotting the procedure is essentially the same (except for the use of RNase free solutions). Briefly, a glass tank was partially filled with 500 ml of 10 x SSC and a rectangular piece of glass was raised above the tank, to form a platform. A larger rectangular piece of Whatman 3MM blotting paper was then soaked in 10 x SSC and placed on the glass platform, with the ends of the paper in the tank of SSC thus forming a wick. A second piece of SSC soaked 3MM paper, the same size as the glass, was then placed on top. The agarose gel was then placed with the open wells face down on the platform. A sheet of Hybond-N membrane (Amersham, UK) cut to the same size as the gel was soaked briefly in sterile distilled water, followed by 10 x SSC (RNase free for RNA gels) and then layered over the gel. Clean, redundant strips of

autoradiographic film were then used to cover the exposed platform area around the gel, to prevent bypass of the paper wick by subsequent layers. The membrane was then layered with 2 membrane sized pieces of 3MM paper soaked in SSC and any air excluded from the layers by gently rolling with a 1/2 size plastic pipette. 3-4 layers of paper towelling, each layer 6 towels thick and arranged in a perpendicular direction to the previous layer, were then placed on top. Finally a sheet of glass was used as a platform for the addition of a moderately heavy weight and the gels blotted overnight. Following blotting the membranes were briefly soaked in nuclease-free 10 x SSC to remove any pieces of agarose. The RNA was then fixed on to the membranes by UV crosslinking, using a UV Stratalinker 1800 (Stratagene). Membranes were wrapped in Saranwrap and stored at 4 °C prior to use.

5.8 Probing of Blots with Radiolabelled Probes

Essentially the same procedures were used for probing of both Southern and Northern blots.

5.8.1 Preparation of radiolabelled probes

Probe fragments generated by restriction enzyme digestion were isolated (see section 5.5.6) and radiolabelled using a commercially available random priming kit, Ready-to-Go (Pharmacia, UK), according to the manufacturer's instructions. Briefly, 25-50 ng denatured DNA probe fragment was added to a rehydrated Ready-to-Go tube. 5 μ l of [α -³²P]-dCTP (3000 Ci/mmol) was added and the reaction incubated at 37 °C for 15 minutes. Unincorporated ³²P-labelled nucleotides were separated from the radiolabelled probe by passing the reaction through a NICK column (Pharmacia), according to the manufacturer's instructions. The radiolabelled probe was then denatured by heating in a boiling water bath for 2-3 minutes and placed on ice for 2 minutes prior to use.

5.8.2 Pre-hybridisation and hybridisation conditions for Northern and Southern blots.

Pre-hybridisation and hybridisation of membranes (see section 5.7 for generation) were both carried out at 65 °C, using 0.25 M Na₂HPO₄ / 7 % SDS, pH 7.0 as both pre-hybridisation and hybridisation buffers. These steps were performed using

hybridisation bottles and carried out in a rotary hybridisation oven (Hybaid, UK). Briefly, membranes were layered on top of hybridisation mesh (Hybaid) and then rolled up and placed in the bottles, with up to 4 blots per bottle. 10-20 ml of pre-hybridisation fluid was then added and the blots incubated at 65 °C for at least 1 hour. Following pre-hybridisation the denatured radiolabelled probe was added and hybridisation continued overnight at 65 °C.

5.8.3 Washing of blots and exposure to X-ray film

Membranes were washed once in 2 x SSC / 0.1 % SDS for 30 min at room temperature, followed by 20 min at 65 °C with fresh solution and then with 0.5 x SSC / 0.1 % SDS for 30 min at 65 °C. The membranes were then monitored and additional washing performed if required (20 min at 65°C with 1 x SSC / 0.1% SDS and finally if necessary, once for 10 min with 0.01 x SSC / 0.1 % SDS at 65 °C). After washing the membranes were wrapped in Saranwrap and transferred to an exposure cassette incorporating an intensifying screen. Membranes were exposed to Kodak X-ray film (X OMAT-AR) at -70 °C for 4 hours initially (1 hour for Southerns), then for longer periods (overnight if needed or for up to 7 days as appropriate).

5.8.4 Stripping of blots

Blots were stripped to remove bound radiolabelled probes, by boiling the membranes in 300-500 ml of 0.1 % SDS for 10 min. Membranes could then be used in further experimentation.

5.8.5 Scanning densitometric analysis of blots

Northern and Southern blots were scanned using a flat bed laser densitometer and analysed using the Quantity One software program (both purchased from Protein Databases Inc., NY, USA).

5.9 Protein Detection

5.9.1 Preparation of cell conditioned medium

Bone marrow-derived macrophages were grown for 7 days in 15 ml of α -MEM conditioned medium supplemented with 25 % DHS and 20 % L929 CM, as described in section 5.3. Cells were then either treated with recombinant murine GM-CSF at a concentration of 10 ng/ml, or with PBS / 0.1 % BSA (v/v) for controls. Medium was collected after 12, 24, 48 and 72 hours of treatment.

5.9.2 Purification of MIP-1 α from bone marrow-derived macrophage cell conditioned medium

Bone marrow-derived macrophages were grown and treated as described in section 5.9.1. Cell conditioned medium was removed after 12, 24, 48 and 72 hours. 10 ml of each sample of cell conditioned medium was then purified using a combination of heparin and blue sepharose affinity chromatography, followed by reverse phase high performance liquid chromatography (HPLC). This purification of MIP-1 α protein was performed by Dr. J. deBono, whom also performed the Western blot shown in figure 6.3.6. His assistance with this purification is very gratefully acknowledged.

5.9.3 SDS polyacrylamide gel electrophoresis

SDS Polyacrylamide gel electrophoresis was performed using a minigel system. 17.5 % polyacrylamide gels were prepared using a commercially available acrylamide / bis-acrylamide stock solution (30 % w/v acrylamide / 0.8 % w/v bis-acrylamide) (Severn Biotech Ltd.), as detailed in section 4.15.3. This solution was then poured between two mini-gel glass plates. The surface of this was then layered with water saturated isobutanol, to create a smooth interface between stacking and resolving gels (as well as preventing oxidisation of the acrylamide mixture and thus speeding up setting). The gel was left to set at room temperature. A 5 % stacking gel was then prepared in a similar way as previously for the resolving gel (see section 4.15.3), except that a smaller quantity was prepared. The water saturated iso-butanol was poured off and the top of the resolving gel washed 3 x with distilled water. Residual water was removed with a small piece of 3MM paper. The 5 % stacking gel was then poured between the glass plates, on top of the resolving gel and a comb of

the appropriate size inserted to create the loading wells. This was allowed to set at room temperature (approximately 30-60 minutes) and the comb carefully removed. Protein samples were prepared for electrophoresis in 1.5 ml screw cap tubes, by adding 20 μ l of sample to 15 μ l of SDS/DTT sample loading buffer (see section 4.15.3) and samples reduced and denatured in a boiling water bath for 5 minutes. Samples were immediately loaded onto the gel, along with molecular weight rainbow markers (Amersham) to enable subsequent sizing of bands and to verify transfer of proteins during Western blotting. Electrophoresis of gels was carried out at room temperature in SDS/PAGE running buffer (see section 4.15.3) at 150 V, until the blue dye front was at 4 cm from the bottom of the resolving gel.

5.9.4 Western blotting

Western blotting was then used to analyse the purified MIP-1 α content of cell conditioned medium derived from bone marrow-derived macrophages. Proteins separated by SDS/PAGE (see section 5.9.5, above) were transferred to Immobilon-P membrane by the process of semi-dry blotting using an electroblotter (Biogel Electophoresis Systems, UK). Gels were blotted for 60 minutes at a constant current of 8 mA/cm² gel area, and was performed as follows. Pieces of Whatman 3MM blotting paper were cut to the same size as the gel to be blotted. Four pieces of paper were soaked in transfer buffer (see section 4.15.3) and placed on the base plate of the electroblotter. A piece of Immobilon-P, also cut to the dimensions of the polyacrylamide gel, was pre-soaked briefly in methanol followed by brief washing in distilled water and was soaked in transfer buffer for 2 minutes before layering on top of the blotting paper. The polyacrylamide gel was then placed on top of the Immobilon-P, followed by 3 pieces of blotting paper soaked in transfer buffer. Any air bubbles trapped in the layers were removed by gentle rolling of the top layer with a plastic pipette. The top plate of the electroblotter was moistened with transfer buffer and the apparatus assembled. The electroblotter was then switched on and run at the conditions mentioned above. After blotting the apparatus was disassembled and the membrane incubated overnight in BLOTTO solution (5 % dried milk in PBS, 0.1 % NP40) at 4 °C, to block non-specific binding sites. The membrane was then washed for 1 hour at room temperature in several changes of fresh BLOTTO. The membrane was then exposed to the primary polyclonal antibody, for 1 hour, at the appropriate

dilution (for goat anti-MIP-1 α (R&D systems) usually 1:1000 in 10 ml of BLOTTO). The membrane was washed for 45 minutes in at least 3 changes of BLOTTO, before incubation for 40 - 60 minutes with the secondary anti-goat IgG antibody conjugated to horse radish peroxidase (Sigma), in 10-50 ml BLOTTO. The membrane was then washed in several changes of PBS / 0.1 % tween 20 for 1 hour. Binding of antibody to membranes was visualised using the enhanced chemiluminescence (ECL) kit (Amersham). Briefly, the membrane was removed from the final wash step, excess solution drained off and the membrane was then agitated in combined ECL solutions for 60 seconds. Excess solution was drained off from one corner of the membrane, using a paper towel before the membrane was wrapped in Saranwrap and placed in an exposure cassette. An initial exposure was carried out for 2 minutes, using Fuji RX Medical X-ray film, additional exposures were then carried as appropriate to enable protein bands to be visualised.

5.10 Reverse Transcriptase Polymerase Chain Reaction (RT-PCR) Methodology

5.10.1 DNase I digestion of total RNA

Total RNA samples were treated with DNase I in order to eliminate the potential for residual DNA contaminating total RNA samples, to act as a template for PCR amplification. Briefly, 40 μ g of total RNA was digested with 2 μ l of DNase I (10 U/ μ l stock) in the presence of 50 mM sodium acetate (pH 5.2) and 1 M magnesium chloride and made up to 200 μ l with DEPC-water. The reaction was incubated at room temperature for 15 minutes, before heating at 95 °C for 5 minutes and then placing on ice for 10 minutes. The reaction was then extracted twice with phenol / chloroform / isoamyl-alcohol (25:24:1), pH 8.0. The RNA was precipitated with 0.1 volume of 3M NaAC and 2.5 volumes of 100 % ethanol for 60 minutes at -70 °C, before spinning down at 13500 rpm for 15 minutes. The pellet was washed with 500 μ l of 70 % ethanol and spun down at 13500 rpm for 10 minutes. After brief air drying the pellet was resuspended in 30 μ l of DEPC-water (approximately 1 μ g/ μ l).

5.10.2 RT-PCR methodology I (amplification from total RNA)

The commercially available Gene Amp® RNA PCR Core kit (Perkin Elmer, CA, USA) was used to detect the expression of RNA transcripts from total RNA samples. Total RNA was DNase I digested as described in section 5.10.1. The manufacturer's instructions were followed with minor adjustments. Briefly as follows, reverse transcription reactions were set up as 20 μ l reactions from a master mix (5 mM MgCl₂, 1 x PCR buffer II, 1 U/ml RNase inhibitor, 2.5 µM Oligo d(T)16, 1.75 µM random hexamers). 1 μ l of DNase I digested RNA (~1 μ g/ μ l) was added to 10 μ l of master mix, mixed by pipetting and then incubated at 70 °C for 5 minutes. Then 8 µl of dNTP mix (1 mM each of dGTP, dATP, dCTP and dTTP) was added, followed by 1 µl of MuLV reverse transcriptase (2.5 U/µl) or 1 µl of DEPC-water. The reaction was then incubated for 10 minutes at room temperature to allow the Oligo $d(T)_{16}$ to be extended by the reverse transcriptase. The reaction tubes were then transferred to a Perkin Elmer Thermal Cycler 480, and incubated at 42 °C for 20 minutes, followed by 5 minutes at 99 °C and 15 minutes at 5 °C. A PCR master mix was then prepared (0.5 mM MgCl₂, 1 x PCR buffer II, AmpliTaq® DNA polymerase (2.5 U/100 µl)) and 74 μ l added to 20 μ l of reverse transcription reaction from above. 3 μ l each of 5' and 3' primers (200 ng/ul) were then added (see tables 4.3 and 4.4 for identity and sequences). The whole reaction mix was mixed by pipetting and then overlaid with 100 µl of liquid paraffin. Tubes were incubated in a Thermal Cycler for 1 minute at 94 °C and then for 30-35 cycles at the following conditions. A) For amplification of MIP-1α sequences: 1 minute at 94 °C, 2 minutes at 60 °C, 2 minutes at 72 °C, followed by soaking at 4 °C. B) For amplification of CC chemokine receptor sequences: 1 minute at 94 °C, 1 minute at 55 °C, 1 minute 30 seconds at 72 °C, followed by soaking at 4 °C. C) For amplification of IL-5Ra chain fragments: 1 minute at 94 °C, 2 minutes at 56 °C, 3 minutes at 72 °C, followed by soaking at 4 °C. Samples were stored at 4 °C prior to visualisation of 20 µl of amplified DNA product, by agarose gel electrophoresis on a 1 % agarose gel.

5.10.3 RT-PCR methodology II (amplification from polyA⁺ mRNA)

The Gene Amp® RNA PCR Core kit was again used, this time for the detection of RNA transcripts from polyA⁺ mRNA. PolyA⁺ mRNA was derived (as

described in section 5.6.3) and 0.5 μ l (eye and leg muscle) or 2 μ l (bone marrow or peritoneal cells) of this used as a template for the generation of cDNA, prior to PCR amplification. The basic RT-PCR methodology was exactly as described for total RNA, with the following modifications: 1) the reactions were performed in 10 μ l reaction volumes for the reverse transcriptase step and 50 µl for the PCR step, 2) the use of random hexamers was omitted, 3) A single master mix was prepared for the reverse transcriptase step, and then split into 2 tubes prior to the addition of either Wild type or GM-CSF transgenic polyA⁺ mRNA. The reverse transcriptase step was then carried out as before. 4) The 2 reverse transcriptase reactions were then split into 2 again and a PCR master mix added to each of the 4 tubes. 5) 5' and 3' primers for the detection of either MIP-1 α or β -Actin were then added to each reaction as appropriate. 6) Each of the 4 reactions (wildtype or GM-CSF with either MIP-1 α or β -Actin primers) was then split into 5 tubes. Tubes were then incubated in a Thermal Cycler for 1 minute at 94 °C and then at the following conditions 1 minute at 94 °C, 2 minutes at 60 °C, 2 minutes at 72 °C. Tubes were removed after 15, 20, 25, 30 and 35 cycles respectively and stored at 4 °C prior to analysis on a 2 % agarose gel.

5.11 Receptor Binding Studies

5.11.1 ¹²⁵I labelling of MIP-1α protein

Labelling of MIP-1 α protein with Na¹²⁵I (Dupont NEN) using Iodo-gen, a commercially available iodination reagent, was performed by Dr. R.J.B. Nibbs as described previously (Graham *et al* 1993). The assistance of Dr. Nibbs in this labelling and during other aspects of these studies is very gratefully acknowledged.

5.11.2 MIP-1α receptor binding studies on bone marrow-derived macrophages

BMM were grown as described in section 5.3. At day 5 of growth, cells were scraped free and replated at 2 x 10^5 cells in 24 well plates and then incubated at 37 °C for a further 24 h to allow cell re-adherence. After this time the cells were stimulated with 10 ng/ml recombinant murine GM-CSF or with PBS / 0.1 % BSA (v/v) for 16 h. Treatment of plates was staggered at 30 minute intervals, to allow sufficient time to process each plate. After 16 h treatment the plates were washed with PBS and the cells

incubated with 250 μ l of binding buffer (α -modified MEM supplemented with 25 % DHS, 20 % L929 CM, 0.2 % sodium azide and 25 mM Hepes (pH 7.4)) containing 4 nM ¹²⁵I labelled MIP-1 α and variable concentrations of unlabelled MIP-1 α competitor or PBS, up to a final volume of 260 μ l. Cells were then washed 3 times with 2 ml of ice-cold PBS. The amount of bound radioactivity was determined by lysing the washed cells in 0.5 ml 1 % SDS and counting the lysate in a gamma counter (Beckman Gamma 5500B). Each point was assayed in triplicate and experiments performed at least twice. Analysis of binding results was then performed using the LIGAND software program (Munson and Robard 1980).

5.12 Calcium Flux Studies

5.12.1 Calcium flux studies in bone marrow-derived macrophages

BMM were grown as described in section 5.3. Cells were either stimulated on day 7 with 10 ng/ml recombinant murine GM-CSF or with PBS for 16 h. Cells were then washed once with PBS and scraped free into HACM buffer (see section 4.15.5) and spun down at 1000 rpm for 5 minutes. Cells were resuspended at 1×10^6 cells/ml and washed twice in HACM buffer. The tubes containing the cell suspensions were then covered in aluminium foil and the light sensitive fluorescent dye Fura-2 AM added to the cell suspension, at a concentration of 2.5 µM. The cell suspension was then incubated at 37 °C for 1 hour. The tubes were gently inverted to mix the suspension after 20 and 40 minutes of incubation. The cells were spun down at 1000 rpm for 5 minutes and washed twice with warmed HACM buffer. Cells were then resuspended at 2 x 10⁶ /ml in warmed HACM buffer. Cells were kept at 37 °C and used immediately for fluxing studies, briefly as follows. A spectrophotometer cuvette, containing a magnetic stirrer bar was placed in a Perkin Elmer luminescence spectrophotometer LS50 and 2 ml of cell suspension was aliquoted directly into the cuvette. After 15 seconds 20 µl of CaCl₂ (100 mM stock) was added and the cuvette incubated at 37 °C for 2 minutes, with a high level of stirring. The time drive was then started on the spectrophotometer and after 15 seconds, MIP-1a (final concentration 100 ng/ml) was added directly to the cell suspension in the cuvette. Overall changes in the intensity of FURA-2 AM were then assessed, thus enabling any changes in intracellular calcium mobilisation to be determined.

5.12.2 Calcium flux studies in THP-1 cells

THP-1 cells were grown in RPMI 1640 / 10 % FCS to a concentration of 5 x 10^5 /ml, as described in section 5.2.1. Cells were then stimulated with either 10 ng/ml recombinant human GM-CSF or PBS / 0.1 % BSA (v/v), for 16 h. Cells were spun down and washed once in PBS and then resuspended at 1 x 10^6 cells/ml and washed twice in HACM buffer. Calcium flux studies were then performed exactly as described in section 5.12.1 above.

PART III: RESULTS AND DISCUSSION

-

CHAPTER 6: RESULTS

6.1 Overview

In order to investigate the potential of haemopoietic growth factors to regulate the expression of MIP-1 α , it was first necessary to find a cell type that provided a system suitable for such investigation. As described in section 1.4.1, the macrophage is one of the primary cell types within the stromal microenvironment. It is therefore likely to be an important source of MIP-1 α within the bone marrow and in peripheral tissues. Murine 'bone marrow-derived macrophages' (BMM) provide a homogeneous population, previously used by others as a model system for marrow macrophages (Temeles *et al* 1993) and which are suitable for investigations on bone marrowderived MIP-1 α . These cells have also previously been successfully used by our laboratory, in other studies on MIP-1 α regulation (Maltman *et al* 1993, 1996). BMM were therefore selected as a model cell system, with which to investigate the regulation of MIP-1 α expression by haemopoietic growth factors.

It is widely recognised that a large interacting network functions within the bone marrow to regulate the processes of constitutive and inducible haemopoiesis (see introduction) and current evidence suggests that MIP-1 α is a member of this network. At present, little is known about the regulatory factors controlling MIP-1 α expression within the bone marrow. Initially therefore, known haemopoietic growth factors with previously identified effects on macrophage functions, were investigated for their ability to modulate MIP-1 α expression in BMM.

6.2 Effect of G-CSF, LIF and IL-6 on the Expression of MIP-1 α in BMM

G-CSF, LIF and IL-6 are all important regulators of various facets of haemopoiesis (see introduction). LIF receptors are known to be expressed on both normal and leukaemic macrophages (Hilton *et al* 1988), whilst G-CSF receptors have been reported to be expressed on macrophages (Shieh *et al* 1991), as have receptors for IL-6 (Lotem and Sachs 1987, Shabo *et al* 1988). Their effect on the expression of MIP-1 α mRNA in day 7 BMM was therefore investigated. This was performed by Northern blot analysis of total RNA isolated from BMM after 4 h treatment with various concentrations of G-CSF, LIF or IL-6. As shown in Figure 6.1,

Figure 6.1: Northern blot analysis of the effect of G-CSF, IL-6 and LIF on the expression of MIP-1 α mRNA in BMM.

A) Effect of G-CSF. 1) control; 2) 4 h 1 ng/ml; 3) 4 h 10 ng/ml; 4) 4 h 50 ng/ml.

B) Effect of IL-6. 1) control; 2) 4 h 1 ng/ml; 3) 4 h 10 ng/ml; 4) 4 h 50 ng/ml.

C) Effect of LIF. 1) control; 2) 4 h 10 U/ml; 3) 4 h 100 U/ml. Top blot in each panel shows a blot hybridised with a probe for MIP-1 α , whilst the bottom blot in each panel shows the same blot, stripped and reprobed for β -Actin. This figure represents a single experiment. The numbers in the table below show the relative levels of MIP-1 α mRNA for this experiment, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA.

	Control	1 ng/ml	10 ng/ml	50 ng/ml
G-CSF	1	1.32	1.42	1.28
IL-6	1	0.36	0.53	0.68
	Control	10 U/ml	100 U/ml	
LIF	1	0.48	0.56	



1

1

2

3

4

B

A

MIP-1α

β–Actin

С

MIP-1a

1

2

3

β–Actin

a low basal level of MIP-1 α expression was detectable in unstimulated BMM. Addition of G-CSF, LIF or IL-6 had no apparent effect on this level of expression, compared with control cells, as determined by densitometric analysis of Northern blots and normalisation for RNA loading using the house-keeping gene β -Actin. This demonstrated that G-CSF, IL-6 and LIF do not appear to modulate MIP-1 α expression in BMM.

6.3. Effect of GM-CSF on the Expression of MIP-1α mRNA in BMM

6.3.1 GM-CSF induces MIP- α expression

The cytokine GM-CSF has been widely studied for its effects on various haemopoietic cell types, particularly those of the granulocyte / macrophage lineages. It was therefore considered a potential candidate for regulation of macrophage-derived MIP-1 α expression within the bone marrow. The effects of GM-CSF on MIP-1 α mRNA expression in BMM were investigated by Northern blot analysis and as previously, a low level of basal MIP-1 α expression was observed. However, following treatment of BMM with recombinant murine GM-CSF (10 ng/ml) for 4 h, this expression appeared to be elevated by approximately 10-fold, as indicated in figure 6.2. Ethidium bromide staining confirmed that loading of RNA in each lane was equal (data not shown). In all future Northern blot analysis, blots were stripped and reprobed with β -Actin to allow normalisation for any loading differences between lanes. The striking level of induction observed in GM-CSF functions as a modulator of MIP-1 α expression and therefore warranted further investigation.

6.3.2 Induction of MIP- α expression by GM-CSF is rapid, but transient

To further assess the effect of GM-CSF on MIP-1 α expression, BMM were treated with GM-CSF for various lengths of time and then the levels of MIP-1 α mRNA were analysed as previously, by Northern blotting and densitometric analysis. Figure 6.3 shows one such experiment. MIP-1 α expression was rapidly induced, by approximately 3-fold compared to control cells, within 1 h following GM-CSF addition (compare lanes 1 and 2). In fact, in some other experiments increased



MIP-1α

Figure 6.2: Northern blot analysis of the effect of GM-CSF on the expression of MIP-1 α mRNA in BMM.

1) control; 2) 4 h 10 ng/ml GM-CSF. Loading levels were confirmed by ethidium bromide staining. This figure represents a single experiment. Scanning densitometric analysis revealed that the level of MIP-1 α mRNA in GM-CSF treated cells (lane 2) was increased by approximately 9.63-fold relative to that of control treated cells (lane 1).

Figure 6.3:

This figure represents data for a single experiment. The numbers in the table below show the relative levels of MIP-1 α mRNA for this experiment, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in 1 h control treated cells. * = data shown in the figure opposite.

• • • •	1 h Control	1 h GM-CSF	2 h GM-CSF	4 h GM-CSF	6 h GM-CSF	
	*1.00	*3.34	*3.50	*13.7	*17.4	
	1.00	3.64	5.00	22.9	9.03	
Average	1.00	3.49	4.25	18.3	13.2	



Figure 6.3: Northern blot analysis of the effect of GM-CSF on a time course of MIP-1 α expression in BMM.

1) 1 h control; 2) 1 h GM-CSF (10 ng/ml) ; 3) 2 h control; 4) 2 h GM-CSF; 5) 4 h control 6) 4 h GM-CSF; 7) 6 h control; 8) 6 h GM-CSF. Top panel shows a blot hybridised with a probe for MIP-1 α , the bottom panel shows the same blot, stripped and reprobed for β -Actin.

expression was visible within 30 minutes of GM-CSF stimulation (data not shown). The level of MIP-1 α mRNA increased between 1 and 4 h following GM-CSF stimulation, with up to a 20-fold increase relative to control cells (compare lanes 5 and 6). Comparison of lanes 7 and 8 shows that MIP-1 α expression was still elevated after 6 h, though not more so than at 4 h. In most experiments however, this peak was reached 4 h after stimulation with GM-CSF and subsequently declined within 6 h (data not shown). Treatment of BMM with GM-CSF for 4 h gave the most consistent increase in MIP-1 α expression.

6.3.3 Induction of MIP- α mRNA expression in response to a titration of GM-CSF concentration

In order to assess the potency of the inductive effect of GM-CSF on MIP-1 α mRNA expression, BMM were stimulated with a titration of GM-CSF (1 pg/ml to 10 ng/ml) and the level of MIP-1 α mRNA assessed by Northern blot analysis 4 h after addition of GM-CSF. Figure 6.4 shows MIP-1 α mRNA expression to be induced in a dose dependent manner, and densitometric analysis demonstrated that as little as 0.1 ng/ml of GM-CSF resulted in approximately a 3-fold increase in the level of MIP-1 α mRNA compared to control values (compare lanes 1 and 4). The increase in MIP-1 α mRNA appeared to plateau with between 5 and 10 ng/ml GM-CSF, resulting in approximately a 10-fold increase in this experiment (compare lane 1, with lanes 7 and 8). However, it should be noted that the MIP-1 α blot is overexposed and the increase in MIP-1 α is perhaps therefore likely to be greater than that suggested by the scanning densitometric analysis. As 10 ng/ml appeared to be just beyond the "shoulder" of this plateau and therefore likely to give a full induction, all future experiments were performed using 10 ng/ml GM-CSF for 4h (unless stated otherwise).

6.3.4 Induction of MIP-α mRNA expression is not dependent on factors present in serum or L929 conditioned medium

BMM are grown in the presence of high levels of both serum and M-CSF (in the form of L929 conditioned medium). These cells will enter a quiescent state if deprived of serum for 16 h and absolutely require the presence of a source of M-CSF for their differentiation, continued proliferation and survival (Becker *et al* 1987). In

Figure 6.4:

This figure represents data for a single experiment. The numbers in the table below show the relative levels of MIP-1 α mRNA for this experiment, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells.

	Concentration of GM-CSF added								
Control	1 pg/ml	10 pg/ml	0.1 ng/ml	0.5 ng/ml	1 ng/ml	5 ng/ml	10 ng/ml		
1.00	1.10	1.74	3.1	4.4	8.1	11	9.20		


β-Actin

Figure 6.4: Northern blot analysis of the effect of a titration of GM-CSF on MIP-1 α mRNA expression in BMM.

BMM were treated for 4 h with the following concentrations of GM-CSF: 1) control; 2) 1 pg/ml; 3) 10 pg/ml; 4) 0.1 ng/ml; 5) 0.5 ng/ml; 6) 1 ng/ml; 7) 5 ng/ml; 8) 10 ng/ml. Top panel shows a blot hybridised with a probe for MIP-1 α , the bottom panel shows the same blot, stripped and reprobed for β -Actin. order to assess whether either serum or L929 conditioned medium (L-929 CM) played a co-operative part in the induction of MIP-1 α expression by GM-CSF, cells were stimulated for 4 h with GM-CSF in the presence or absence, of either serum or L-929 CM.

Panel A of figure 6.5 shows that similar levels of MIP-1 α mRNA were induced in either the presence, or absence of serum. Densitometric analysis revealed that this induction was approximately 4-fold, in both the presence or absence of serum.

Panel B of figure 6.5 shows that the absence of L-929 CM did not result in the abolition of GM-CSF induced MIP-1 α expression. In fact, densitometric analysis revealed that in the absence of L-929 CM the induction of MIP-1 α mRNA appeared to increase by approximately 170 %, rising from 11-fold in the presence of L-929 CM to 29-fold in the absence of L-929 CM. This suggests that MIP-1 α suppressive factors may be present in the conditioned medium of L-929 cells.

These results demonstrate that GM-CSF does not require the presence of factors present in either serum or L-929 CM, to induce MIP-1 α mRNA expression.

6.3.5 GM-CSF directly induces the expression of MIP-1a mRNA

It has previously been demonstrated using neutrophils that LPS is able to synergise with GM-CSF to induce MIP-1 α expression, although GM-CSF was unable to induce MIP-1 α expression when used as a sole agent (Kasama *et al* 1993). In order to exclude the presence of LPS as far as possible and therefore rule out a role for LPS in the observed induction of MIP-1 α expression, BMM were stimulated under essentially LPS-free conditions. To demonstrate that GM-CSF was indispensable for this observed induction, the effects of GM-CSF on MIP-1 α expression were assessed in the presence of a GM-CSF specific neutralising antibody. Figure 6.6 shows that as previously, GM-CSF increased the level of MIP-1 α mRNA by approximately 10-fold relative to control values. However, in the presence of a GM-CSF neutralising antibody this induction was almost completely abrogated. Densitometric analysis of blots revealed that the basal level of MIP-1 α expression did not appear to be significantly different between the control and those cells treated with neutralising antibody. This suggests that there is no autocrine regulation of MIP-1 α expression in

Figure 6.5; Northern blot analysis of the effect of GM-CSF on MIP-1 α mRNA expression in BMM, in the absence or presence of either serum or L-929 CM.

A) Effect of serum. BMM were treated as follows: 1) control plus serum; 2) 4 h GM-CSF (10 ng/ml) plus serum; 3) control (serum-free); 4) 4 h GM-CSF (serum-free). B) Effect of L-929 CM. 1) control plus L-929 CM; 2) 4 h GM-CSF (10 ng/ml) plus L-929 CM; 3) control (L-929 CM-free); 4) 4 h GM-CSF (L-929 CM-free). Top blot in each panel shows a blot hybridised with a probe for MIP-1 α , whilst the lower blot in each panel shows the same blot stripped and reprobed for β -Actin. This figure represents single experiments. Tables A and B below, show the relative levels of MIP-1 α mRNA for the experiments shown in panels A and B (opposite), as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA.

A

Plus	Serum	Seru	m Free
Control	GM-CSF	Control	GM-CSF
1	4.0	1	3.2

B

Plus L-	929 CM	Minus	L-929 CM
Control	GM-CSF	Control	GM-CSF
1	10.7	1	28.7



B



Figure 6.6;

The numbers in the table below show the relative levels of MIP-1 α mRNA for these 2 experiments, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells. * = data shown in the figure opposite

1	Control	GM-CSF	GM-CSF/Nab	- Nab
	*1.00	*23.0	*4.23	*3:41
	1.00	34.0	2.60	0.86
	1.00	29.5	2.40	0.14



Figure 6.6: Northern blot analysis of the effect of a neutralising GM-CSF antibody on GM-CSF induced MIP-1 α mRNA expression in BMM, under LPS-free conditions.

BMM were treated as follows: 1) control; 2) 4 h 10 ng/ml GM-CSF; 3) 10 ng/ml GM-CSF plus neutralising antibody (5 μ g/ml); 4) neutralising antibody. Top panel shows a blot hybridised with a probe for MIP-1 α , the bottom panel shows the same blot, stripped and reprobed for β -Actin. This figure is representative of **2** separate experiments.

These results suggest that GM-CSF directly induces the expression of MIP-1 α mRNA and because these experiments were performed using conditions and reagents known to be essentially LPS-free, it is highly unlikely that GM-CSF is acting in synergy with LPS.

6.3.6 GM-CSF induces MIP-1α protein expression in BMM

To assess whether increases in MIP-1 α expression at mRNA level were also reflected by increases in MIP-1 α protein expression, BMM were either treated with GM-CSF (10 ng/ml) or PBS / 0.1 % BSA (v/v), and conditioned medium collected after various lengths of time. The samples were collected later than for total RNA, to allow sufficient time for MIP-1 α mRNA to be translated to protein and to allow time for the levels of protein to accumulate in the macrophage growth medium to a level at which they could be detected. The content of MIP-1 α protein in these BMM supernatants was then measured by Western blot analysis. In initial experiments, no MIP-1 α protein could be detected in supernatants from either control or GM-CSF treated BMM, although a recombinant MIP-1 α control was detectable. These observations suggested either that the level of MIP-1 α protein in these cells was too low to detect by Western blot analysis, or that GM-CSF stimulation does not result in an increase in MIP-1 α protein expression.

It has previously been demonstrated that the level of MIP-1 α protein in unstimulated BMM is extremely low and often undetectable by Western blot analysis (Maltman *et al* 1993). Therefore, in order to assess whether MIP-1 α protein expression was induced in BMM by GM-CSF, MIP-1 α present in supernatants from control and GM-CSF stimulated BMM was purified by a combination of heparin and blue sepharose affinity chromatography, followed by reverse phase HPLC. This purification simultaneously also resulted in approximately a 100-fold concentration of protein. The level of purified MIP-1 α protein in these supernatants was then assessed by Western blotting analysis. Figure 6.7 shows that MIP-1 α protein was undetectable in unstimulated BMM over the time course of the experiment, but that treatment with GM-CSF (10 ng/ml) resulted in an increased level of MIP-1 α protein expression. This increase was just visible 24 h after addition of GM-CSF and increased steadily with



Figure 6.7: Western blot analysis of the effect of GM-CSF on MIP-1α protein expression in BMM.

BMM cell conditioned medium was collected after the appropriate time as denoted at the top of each lane. BMM were treated as follows: PBS /0.1 % BSA control (left hand 4 lanes) or GM-CSF (10 ng/ml) (right hand 4 lanes). Proteins were resolved by SDS/PAGE on a 17.5 % gel. MIP-1 α protein was detected using a murine anti-MIP-1 α antibody and ECL reagents, as detailed in the Materials and Methods.

This figure represents data for a single experiment and both the protein purification and the Western blot analysis were performed by Dr. Johann de Bono, as detailed in the materials and methods. The reproduction of his work here, is very gratefully acknowledged.

MIP-1 α

time up to 72 h. This data suggests that whilst the level of MIP-1 α protein in BMM is very low, a finding in accordance with previous reports from our laboratory (Maltman *et al* 1993), the level of MIP-1 α protein does increase following GM-CSF treatment. This demonstrates that induction of MIP-1 α expression observed at an mRNA level, is also reflected at the level of MIP-1 α protein expression and suggests that the induction of MIP-1 α by GM-CSF may be physiologically important.

6.4 Effect of IL-3 on MIP-1α mRNA Expression in BMM

GM-CSF is a member of a cytokine family and exhibits some functional overlap with IL-3. It is speculated that this is the result of the shared usage of a receptor component, the common β subunit (or β c), found as a component of both GM-CSF and IL-3 heterodimeric receptors (see introduction). Moreover, IL-3 has previously been reported, along with GM-CSF, to affect bone marrow-derived macrophage functions (Takamatsu *et al* 1996). It was therefore speculated that IL-3 might also exert an inductive effect on MIP-1 α expression in BMM, in a similar fashion to that of GM-CSF.

6.4.1 Titration of the inductive effect of IL-3 on MIP-1 α mRNA expression

To assess the ability of IL-3 to regulate MIP-1 α mRNA expression, BMM were treated with a titration of IL-3 (1 pg/ml to 10 ng/ml) for 4 h. Panel A of figure 6.8 shows MIP-1 α mRNA expression to be increased in a dose dependent manner, and densitometric analysis revealed that concentrations as low as 0.1 to 1 ng/ml of IL-3 were sufficient to induce substantial expression of MIP-1 α mRNA. As with GM-CSF, the expression of MIP-1 α plateaued with between 5 ng/ml and 10 ng/ml of IL-3, resulting in approximately a 20 to 25-fold increase in the levels of MIP-1 α mRNA relative to control values. A time course of MIP-1 α expression in response to IL-3 (10 ng/ml), also revealed a similar time course to that seen GM-CSF (data not shown). As a dose of 10 ng/ml appeared to be just beyond the shoulder of this plateau, all future stimulations using IL-3 were performed using a dose of 10 ng/ml for 4 h. These results indicate that in addition to GM-CSF, IL-3 is also an inducer of MIP-1 α expression.

Panel B of figure 6.8 shows that treatment of BMM with IL-3 increased the level of MIP-1 α mRNA by approximately 23-fold relative to control values, but that

Figure 6.8: Northern blot analysis of the effect of IL-3 on MIP-1 α mRNA expression in BMM.

A) Effect of a titration of IL-3 on MIP-1 α mRNA expression. BMM were treated for 4 h with the following concentrations of IL-3: 1) control; 2) 1 pg/ml; 3) 10 pg/ml; 4) 0.1 ng/ml; 5) 0.5 ng/ml; 6) 1 ng/ml; 7) 5 ng/ml; 8) 10 ng/ml. B) Effect of a neutralising IL-3 antibody on IL-3 induced MIP-1 α mRNA expression. BMM were treated for 4 h as follows: 1) control; 2) 10 ng/ml IL-3; 3) 10 ng/ml IL-3 plus neutralising antibody (5 µg/ml); 4) neutralising antibody. Top blot in each panel shows a blot hybridised with a probe for MIP-1 α , the bottom blot in each panel shows the same blot, stripped and reprobed for β -Actin. This figure is representative of a single, and 2 separate experiments respectively. Table A below shows the relative levels of mRNA for MIP-1 α (as shown in panel A, opposite) and calculated from the data obtained by scanning densitometric analysis and normalisation against the calculated level of β -Actin mRNA. Table B shows the relative levels of mRNA for MIP-1 α , calculated from 2 separate experiments (* = data for experiment shown in panel B, opposite).

A

	Concentration of IL-3						
Control	1 pg/ml	10 pg/ml	0.1 ng/ml	0.5 ng/ml	1 ng/ml	5ng/ml	10ng/ml
1.00	1.20	1.00	2.80	10.0	25.0	18.0	24.0

B

	Control	IL-3	IL-3/NAb	NAb
	*1.00	*23.2	*0.55	*0.53
	1.00	24.8	6.40	3.16
Awaraga	1.00	24.0	2.49	2 (0

A

in the presence of an IL-3 neutralising antibody this induction was reduced to the same level as that of the control cells. As with GM-CSF, densitometric analysis of blots showed that the basal level of MIP-1 α expression appeared to be similar in both control and IL-3 neutralising antibody treated cells, suggesting that there is no involvement of autocrine IL-3 in regulating the overall expression of MIP-1 α in macrophages. In summary, these data suggest that, like GM-CSF, IL-3 directly induces the expression of MIP-1 α mRNA. This experiment was performed using conditions and reagents known to be essentially LPS-free and IL-3 is therefore unlikely to be acting in synergy with LPS.

6.5 Effect of IL-5 on MIP-1α mRNA Expression

IL-5 is also a member of the same small subfamily of cytokines, comprised of GM-CSF and IL-3, and in addition shares the βc component in its receptor. The ability of IL-5 to regulate MIP-1 α expression was therefore investigated.

6.5.1 IL-5 does not induce expression of MIP-1 α mRNA in BMM

BMM were treated with a titration of IL-5 (1 pg/ml to 10 ng/ml) and the level of MIP-1 α mRNA analysed after 4 h. Figure 6.9 shows a Northern blot of one such experiment. Densitometric analysis revealed that the level of MIP-1 α expression was not altered following treatment of BMM with IL-5, at any of the concentrations tested. This data suggests either that IL-5 has no effects on the expression of MIP-1 α , at the concentrations tested, or that BMM do not have the capacity to respond to IL-5. The observation that no increase in MIP α expression occurred in the presence of IL-5, also suggests that LPS was not present in these experiments and confirms that this system is very clean with respect to the potential presence of LPS.

6.5.2 BMM do not express mRNA for the IL-5R α chain

To attempt to explain the inability of BMM to respond to IL-5, expression of the IL-5 specific α chain receptor subunit (IL-5R α) was examined in BMM by Northern blot analysis and RT-PCR analysis. No expression of IL-5R α chain was detectable in BMM by Northern blot analysis (data not shown). Figure 6.10 shows that IL-5R α chain expression was also undetectable in BMM by RT-PCR analysis (lane 3).

Figure 6.9:

i.

The numbers in the table below show the relative levels of MIP-1 α mRNA for these 3 experiments, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells. * = data shown in the figure opposite

· · · ·			Concentration of IL-5						
	Control	1 pg/ml	10 pg/ml	0.1ng/ml	0.5ng/ml	1 ng/ml	5 ng/ml	10 ng/ml	
	*1.0	*1.41	*1.10	*1.44	*0.60	*0.98	*0.90	*2.10	
	1.0	1.54	0.77	1.09	1.37	2.20	0.90	1.50	
	1.0	0.60	1.10	1.20	1.90	2.20	4.40	1.60	
Average	1.0	1.18	0.99	1.24	1.29	1.79	1.93	1.73	



Figure 6.9: Northern blot analysis of the effect of a titration of IL-5 on MIP-1 α mRNA expression in BMM.

BMM were treated for 4 h with the following concentrations of IL-5: 1) control; 2) 1 pg/ml; 3) 10 pg/ml; 4) 0.1 ng/ml; 5) 0.5 ng/ml;6) 1 ng/ml; 7) 5 ng/ml; 8) 10 ng/ml. Top panel shows a blot hybridised with a probe for MIP-1 α , the bottom panel shows the same blot stripped and reprobed for β -Actin. This figure is representative of at least 3 separate experiments. 124



Figure 6.10: RT-PCR analysis of the expression of the IL-5Ra chain in BMM.

Total RNA was isolated and analysed for the expression of either GAPDH or IL-5R α chain as follows: Lanes 1-3 show reactions performed using BMM RNA, whilst lanes 4-6 show reactions performed using BCL-1 RNA (a positive control cell line for mIL-5R α chain expression). Reactions were performed using either GAPDH primers (lanes 1-3) or with primers specific for mIL-5R α chain (lanes 4-7). Lanes 2 and 5 are negative control reactions performed without the addition of reverse transcriptase. The reaction shown in lane 7 is a positive control, performed using mIL-5R α chain cDNA. The IL-5R α chain primers used for the PCR reaction define fragments of the following size: arrow a) 548 bp; arrow b) 452 bp; arrow c) 369; arrow d) a possible alternatively spliced variant. This figure represents data from a single experiment.

Also included was a positive control cell line, BCL-1 (lane 6), previously shown to express the IL-5R α chain. Four bands were observed in this and they corresponded exactly to those previous identified by Takaki *et al* (1990) as follows. Arrow a) membrane bound form; arrows b) and c) soluble IL-5R; arrow d) probably corresponds to an alternatively spliced mRNA variant. Also shown is an internal control for the housekeeping gene GAPDH (lanes 1 and 4). This data strongly suggests that BMM are unable to respond to IL-5, due to the absence of IL-5R α chain and therefore a functional IL-5 receptor. It does not however preclude the possibility of IL-5 receptors.

6.5.3 IL-5 does not induce expression of MIP-1a mRNA in bone marrow cells

The bone marrow consists of a heterogeneous population of cells which includes cells other than macrophages, some of which are known to be IL-5 responsive. As MIP-1 α expression was undetectable in unfractionated murine bone marrow cells by Northern blot analysis, the expression of MIP-1 α mRNA was examined by RT-PCR analysis in order to determine whether cells within the bone marrow could alter their expression of MIP-1 α in response to IL-5. Figure 6.11 shows that MIP-1 α is readily detectable in untreated bone marrow cells, but that this level of expression does not appear to be significantly altered following treatment with IL-5. This suggests that cells within the bone marrow do not readily increase their expression of MIP-1 α mRNA, in response to IL-5. It is possible however, that cells may respond but that the number of responsive cells is too low to result in a significant change in the overall level of MIP-1 α mRNA expression.

6.6 Effect of GM-CSF on the Expression of mRNA for other CC chemokines

MIP-1 α is a member of a large superfamily of cytokines known as the chemokines which are divided into several subfamilies, with MIP-1 α being a member of the CC subfamily (see introduction). It had been previously demonstrated that IL-8, a member of the CXC chemokine subfamily, was inducible in monocytes by GM-CSF and IL-3 stimulation (Takahashi *et al* 1993). Therefore, the effect of GM-CSF and



MIP-1α



β-Actin

Figure 6.11: RT-PCR analysis of the effect of IL-5 on the expression of MIP-1 α mRNA in murine bone marrow cells.

Bone marrow cells harvested from murine femora were treated for 4 h as follows: Lane 1) Control treated cells; lane 2) IL-5 treated (10 ng/ml). RT-PCR analysis was then used to examine the expression of MIP-1 α . The reactions shown in lanes 3 and 4 are positive PCR controls performed using genomic DNA and MIP-1a cDNA, respectively. Top panel shows products of reactions performed with MIP-1 α primers, whilst bottom panel shows those performed with β -Actin primers. This figure represents data from a single experiment. IL-3 on the expression of other murine members of the CC subfamily was investigated in BMM, to examine if GM-CSF induction was specific to MIP-1 α .

6.6.1 GM-CSF induces expression of MIP-1 β mRNA

MIP-1 β is very similar to MIP-1 α , both in its expression and in many of its functions. BMM were therefore treated with either GM-CSF or IL-3 (both at 10 ng/ml) for 4 h and then the level of murine MIP-1 β mRNA analysed by Northern blot analysis, using a MIP-1 β -specific probe. Figure 6.12 shows that MIP-1 β mRNA is induced by GM-CSF and IL-3 and densitometric analysis revealed this induction to be approximately 15 to 20-fold, a similar magnitude to that observed for MIP-1 α .

6.6.2 GM-CSF and IL-3 induce expression of JE and MARC mRNA

MARC and JE are two other murine CC chemokines known to be expressed by monocytes and macrophages (see introduction). The expression of JE and MARC mRNA was examined using an identical protocol to that used for MIP-1 β expression. Figures 6.13 A and 6.13 B show that both JE and MARC mRNA are substantially induced by both GM-CSF and IL-3.

These results suggest that expression of at least 3 other chemokines, in addition to MIP-1 α , can be regulated by GM-CSF and implies that additional chemokines not investigated here may also be regulated in a similar fashion.

6.7. Effect of TGF- β 1 on the GM-CSF and IL-3 Induced Expression of CC Chemokines

Previous work in our laboratory has documented the existence of an endogenous reciprocal relationship in BMM, between MIP-1 α and TGF- β 1 (Maltman *et al* 1993, 1996). It is proposed that TGF- β 1 may function to suppress the expression of MIP-1 α in the bone marrow, and yet MIP-1 α expression is detectable (Cluitmans *et al* 1995), suggesting that other endogenous factors may act to oppose this block. In the light of the observations documented here, that GM-CSF and IL-3 are both strong inducers of MIP-1 α expression in BMM, the potential relationship between TGF- β 1, GM-CSF, IL-3 and MIP-1 α expression was investigated.

Figure 6.12:

.

.

The numbers in the table below show the relative levels of MIP-1 β mRNA for these 2 experiments, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells. * = data shown in the figure opposite

	Control	GM-CSF	Control	IL-3
	*1.00	*14.7	*1.00	*18.0
	1.00	14.5	1.00	22.0
Average	1.00	14.6	1.00	20.0

MIP-1β β-Actin 1 2 MIP-1β β-Actin

1

2

A

Figure 6.12: Northern blot analysis of the effect of GM-CSF and IL-3 on the expression of MIP-1 β mRNA in BMM.

A) Effect of GM-CSF on MIP-1 β expression. BMM were treated for 4 h as follows: 1) control; 2) GM-CSF (10 ng/ml). B) Effect of IL-3 on MIP-1 β expression. BMM were treated as follows: 1) control; 2) IL-3 (10 ng/ml). The top blot of each panel shows a blot hybridised with a probe for MIP-1 β , the bottom blot of each panel shows the same blot, stripped and reprobed for β -Actin. This figure is representative of 2 separate experiments.

Figure 6.13: Northern blot analysis of the effect of GM-CSF and IL-3 on the expression of JE and MARC mRNA in BMM.

A) Effect of GM-CSF and IL-3 on JE expression. BMM were treated as follows: 1) control; 2) 4 h GM-CSF (10 ng/ml); 3) 4 h IL-3 (10 ng/ml). B) Effect of GM-CSF and IL-3 on MARC expression. BMM were treated as in A above. Top blot of each panel shows a blot hybridised with a probe for either JE or MARC, the bottom blot in each panel shows the same blot stripped and reprobed for b-Actin. This figure is representative of 3 separate experiments. The table below shows the relative levels of mRNA for JE and MARC for 3 separate experiments (*= data for those experiments shown opposite in panels A and B, respectively), as assessed by scanning densitometric analysis and normalisation against the calculated level of β -Actin mRNA.

		Control	GM-CSF	IL-3
		*1.00	*33.0	*52.0
JE		1.00	25.0	62.0
		1.00 · ·	. 59.0	33.0
·····	Average	1.00	39.0	49.0
		1.00	9.78	10.4
MARC		1.00	10.4	7.00
		*1.00	*16.0	*25.0
	Average	1.00	12.1	14.1

A

JE

2

2

1

1

3

3

β-actin

B

MARC



6.7.1 TGF-β1 can suppress the induction of MIP-1α mRNA expression induced by GM-CSF and IL-3

The ability of TGF- β 1 to suppress MIP-1 α expression in the presence of GM-CSF or IL-3 was assessed. BMM were stimulated with GM-CSF or IL-3 as described above, either in the presence or absence of TGF-β1 (20 ng/ml). A previous study in our laboratory had shown suppression of MIP-1a expression, with around 1 pM TGFβ1 (Maltman et al 1993). A dose of 20 ng/ml was selected for the experiments reported herein, as this was in excess of the doses used in the previous studies and therefore likely to result in suppression of MIP-1 α expression, despite the presence the strong inducing stimulus of GM-CSF. Figure 6.14 shows one such experiment. As previously, densitometric analysis revealed that GM-CSF and IL-3 potently induced MIP-1a mRNA expression, approximately 20-fold, relative to control values (compare lane 1 with lanes 2 and 4). However, in the presence of TGF- β 1, this induction was reduced by approximately 55-85 % (compare lane 2 with 3, and lane 4 with 5). As previously reported, TGF- β 1 alone was able to suppress the level of MIP- 1α expression below the basal level normally seen in unstimulated cells (compare lane 1 with 6). These data indicate that TGF- β 1 is indeed able to suppress the induction of MIP-1a mRNA expression induced by GM-CSF and IL-3. However, this suppression does not appear complete despite the very high level of TGF- β 1 used, suggesting that GM-CSF and IL-3 are indeed strong inducers of MIP-1 α expression and may therefore be able to overcome TGF- β 1 suppression of MIP-1 α expression in vivo. These results indicate that the opposing regulatory stimuli exerted by GM-CSF and TGF- β on MIP- 1α expression are probably not exclusive of each other and are also unlikely to be an "all or nothing effect". The interactions occuring in vivo are therefore likely to be dynamic and dependent upon the relative levels of GM-CSF, IL-3 and TGF- β present within the bone marrow microenvironment at any given time.

6.7.2 The suppression of GM-CSF induced MIP-1 α expression by TGF- β 1 is dose-dependent

In order to further examine the relationship between TGF- β 1, GM-CSF and MIP-1 α , BMM were treated with GM-CSF in the presence of a titration of TGF- β 1. Figure 6.15 shows that as above, TGF- β 1 suppressed the induction of MIP-1 α mRNA

Figure 6.14:

The numbers in the table below show the relative levels of MIP-1 α mRNA for this experiment, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells. * = data shown in the figure opposite.

	Control	GM-CSF	GM-CSF + TGF-β1	IL-3	IL-3 + TGF-β1	TGF-β1
	*1.0	*21.0	*3.30	*22.8	*10.4	N/M
	1.0	6.0	1.90	7.40	3.22	N/M
Average	1.0	13.5	2.60	15.1	6.81	N/M

N/M = not measurable



Figure 6.14: Northern blot analysis of the effect of TGF- β 1 on the induction of MIP-1 α mRNA expression by GM-CSF and IL-3.

BMM were stimulated for 4 h with the following additions: 1) control; 2) GM-CSF (10 ng/ml); 3) GM-CSF plus TGF- β 1 (20 ng/ml); 4) IL-3 (10 ng/ml); 5) IL-3 plus TGF- β 1; 6) TGF- β 1. Top panel shows a blot hybridised with a probe for MIP-1 α , whilst the bottom panel shows the same blot stripped and reprobed for β -Actin. This figure is representative of **2** separate experiments.

Figure 6.15:

This figure represents data from a single experiment. The numbers in the table below show the relative levels of MIP-1 α mRNA for this experiment, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells.

Control	GM-CSF	GM + 20ng/ml	GM +2ng/ml	GM+200pg	GM+20pg/
		TGF-β1	TGF-β1	/ml TGF-β1	ml TGF-β1
1.00	11.9	2.49	3.70	10.8	5.70



Figure 6.15: Northern blot analysis of the effect of a titration of TGF- β 1 on the induction of MIP-1 α mRNA expression by GM-CSF

BMM were stimulated for 4 h with the following additions: 1) control; 2) GM-CSF (10 ng/ml); 3) GM-CSF plus TGF- β 1 (20 ng/ml); 4); GM-CSF plus TGF- β 1 (2 ng/ml); 5) GM-CSF plus TGF- β 1(200 pg/ml); 6) GM-CSF plus TGF- β (20 pg/ml). Top panel shows a blot probed with MIP-1 α , whilst the lower panel shows the same blot stripped and reprobed with a probe for β -Actin.

by GM-CSF, by approximately 80 % (compare lane 1 with lanes 2 and 3). As the amount of TGF- β 1 was decreased, so the level of MIP-1 α expression rose (compare lanes 4, 5, and 6 with lane 3). The level of expression observed with the lowest dose of TGF- β 1 (0.02 ng/ml) was unexpectedly lower than that seen with a 10-fold higher dose of TGF- β 1 (lanes 6 and 5 respectively). The reason for this is unclear, but may be the result of a slightly unequal loading of RNA between lanes as indicated by the β -Actin blot. These results confirm the observations made in the previous experiment, that TGF- β 1 is a suppressor of MIP-1 α expression, but that GM-CSF may be sufficiently potent to at least partially overcome TGF- β 1 mediated suppression.

6.7.3 TGF-β1 can also suppress the induction of JE and MARC mRNA expression induced by GM-CSF and IL-3

As described in section 6.6.2, GM-CSF and IL-3 can induce the expression of several chemokines in addition to that of MIP-1 α . This observation suggested that TGF- β might have the potential to suppress the induction of these additional chemokines. The expression of MARC and JE mRNA following GM-CSF or IL-3 treatment was therefore examined in the presence of TGF- β 1 (20 ng/ml). Figure 6.16 shows that like MIP-1 α , the induction of both MARC and JE by GM-CSF or IL-3, can be suppressed in the presence of TGF- β 1 (by approximately 50-75 %). As before, this TGF- β 1 suppression was not absolute and some induction of both MARC and JE expression was still evident in the presence of GM-CSF or IL-3, relative to control values. As with MIP-1 α , in the absence of either GM-CSF or IL-3, the basal levels of both MARC and JE appeared to be suppressed by TGF- β . These results indicate that, as with MIP-1 α , the basal or the induced expression of other CC chemokines can be negatively regulated by TGF- β 1.

6.8 Effect of GM-CSF, IL-3 and TGF-β1 on the Expression of MIP-1α Receptors

As outlined in the introduction, the last few years have seen a rapid growth in the understanding of the interactions between chemokines and their cognate receptors. Many chemokines, including MIP-1 α , are somewhat unusual in their ability to bind and functionally activate more than one receptor. In fact, as indicated in table 3.1, MIP-1 α has been reported to be a functional ligand for at least 4 receptors. To date,

Figure 6.16: Northern blot analysis of the effect of TGF- β 1 on the induction of MARC and JE mRNA expression by GM-CSF and IL-3.

A) Effect of TGF- β 1 on the induction of MARC mRNA by GM-CSF and IL-3. BMM were treated for 4 h under the following conditions 1) control; 2) GM-CSF (10 ng/ml); 3) GM-CSF plus TGF- β 1 (20 ng/ml); 4) IL-3 (10 ng/ml); 5) IL-3 plus TGF- β 1; 6) TGF- β 1. B) Effect of TGF- β 1 on the induction of JE mRNA by GM-CSF and IL-3. BMM were treated as A) above. Top blot in each panel shows a blot hybridised with a probe for either MARC or JE, whilst the bottom blot in each panel shows the same blot stripped and reprobed for β -Actin. This figure is representative of 2 separate experiments. The table below shows the relative levels of mRNA for MARC and JE for these 2 separate experiments (*= data for those experiments shown opposite in panels A and B, respectively), as assessed by scanning densitometric analysis and normalisation against the calculated level of β -Actin mRNA.

	Control	GM-CSF	GM-CSF/TGF- β	IL-3	ΙL-3/TGF-β	TGF-β
	1.00	122	70.3	121	51.4	N/M
JE	*1.00	*62.3	*31.4	*92.0	*25.0	N/M
· · · ·	• • • •		· ,			· · · ·
Average	1.00	92.1	50.9	49.0	38.2	N/M
	*1.00	*119	*48.6	*120	*65	N/M
MARC	1.00	210	69.3	325	143	N/M
Average	1.00	165	58	223	104	N/M



B

JE

β-Actin

whilst some progress has been made in understanding the regulation of MIP-1 α expression, very little is known regarding the regulatory controls governing the expression of the various MIP-1 α receptors. Given the potent inductive effect of GM-CSF and IL-3 on MIP-1 α expression reported in the previous sections, it was decided to examine whether GM-CSF and IL-3 could also affect the expression of MIP-1 α receptors.

6.8.1 GM-CSF and IL-3 increase the expression of CCR1 in BMM, but not that of CCR5

As discussed in the introduction, the CC chemokine receptors CCR1 and CCR5 are known to be widely expressed on cells of the macrophage lineage. Therefore the expression of CCR1 and CCR5, together with that of CCR3, CCR4 and D6, was examined in BMM using northern blot analysis. No expression of CCR3, CCR4, and D6 was visible in BMM, as assessed by northern blot analysis. Densitometric analysis revealed that the expression of CCR1 mRNA was increased approximately 3-5 fold following GM-CSF or IL-3 stimulation of BMM, relative to control values (see figure 6.17 panel A, compare lanes 2 and 3 with lane 1). This was the typical level of expression observed following GM-CSF stimulation, although as much as a 10-fold increase in expression was observed in some experiments. In contrast however, no change in the level of CCR5 expression was observed following treatment with GM-CSF or IL-3 (see figure 6.17 panel B, compare lanes 2 and 3 with lane 1).

6.8.2 GM-CSF and IL-3 also regulate the expression of CCR1, but not that of CCR5, in splenic cells

Splenic cells were also used, in an crude attempt to examine the expression of CCR1 and CCR5 in other cell types. Figure 6.18 shows that a very similar pattern of expression for CCR1 and CCR5 is evident in splenic cells, as is observed in BMM. As in BMM, CCR1 expression is elevated following stimulation of splenic cells with GM-CSF or IL-3, but CCR5 expression remains unaltered. The normal murine spleen is comprised mainly of a mixture of B and T cells and erythrocytes. Given the few numbers of macrophages present in normal mouse spleen, it is highly unlikely that the observed increase in CCR1 expression observed by Northern blotting, was

Figure 6.17: Northern blot analysis of the effect of GM-CSF and IL-3 on the expression of CCR1 and CCR5 mRNA in BMM.

A) Effect of GM-CSF and IL-3 on CCR1 expression. 1) control; 2) 4 h GM-CSF (10 ng/ml); 3) 4 h IL-3 (10 ng/ml). B) Effect of GM-CSF and IL-3 on CCR5 expression. BMM were treated as A) above. Top blot of each panel shows a blot hybridised with a probe for either CCR1 or CCR5, whilst the bottom blot in each panel shows the same blot, stripped and reprobed for β -Actin. This figure is representative of 3 separate experiments. The table below shows the relative levels of mRNA for CCR1 and CCR5 for these 3 separate experiments (*= data for those experiments shown opposite in panels A and B, respectively), as assessed by scanning densitometric analysis and normalisation against the calculated level of β -Actin mRNA.

		Control	GM-CSF	IL-3
		*1.00	*3.10	*4.60
CCR1		1.00	4.76	6.80
• • • •		1.00	9.06	8.96
	Average	1.00	5.64	6.79
		*1.00	*1.00	*0.85
CCR5		1.00	1.16	0.91
		1.00	1.43	1.24
	Average	1.00	1.20	1.00



Figure 6.18: Northern blot analysis of the effect of GM-CSF and IL-3 on the expression of CCR1 and CCR5 mRNA in splenic cells.

A) Effect of GM-CSF and IL-3 on CCR1 expression. Splenic cells were treated as follows 1) control; 2) 4 h GM-CSF (10 ng/ml); 3) 4 h IL-3 (10 ng/ml). B) Effect of GM-CSF and IL-3 on CCR5 expression. Splenic cells were treated as A) above. Top blot of each panel shows a blot hybridised with a probe for either CCR1 or CCR5, whilst the bottom blot in each panel shows the same blot, stripped and reprobed for β -Actin. This figure is representative of 2 separate experiments. The table below shows the relative levels of mRNA for CCR1 and CCR5 for these 2 separate experiments (*= data for those experiments shown opposite in panels A and B, respectively), as assessed by scanning densitometric analysis and normalisation against the calculated level of β -Actin mRNA.

		Control	GM-CSF	IL-3
		*1.00	*4.98	*3.20
CCR1		· · 1.00· ·	2.47	· ·1.49· ·
	Average	1.00	3.73	2.36
CCR5		· · ·		
		*1.00	*1.44	*1.43
		1.00	1.82	0.93
	Average	1.00	1.63	1.18



macrophage derived. This data therefore suggests that increased CCR1 expression may also occur outwith the bone marrow in non-macrophage cell types, in response to GM-CSF and IL-3.

6.8.3 GM-CSF and IL-3 do not affect the expression of other MIP-1 α receptors in BMM

As indicated above, northern blot analysis of BMM failed to reveal detectable expression of mRNA for other MIP-1 α receptors, namely CCR4 and D6. Nor was expression of murine CCR3 detected, a receptor which may, or may not, be a functional MIP-1 α receptor and which was previously reported to be expressed in monocytes (Graham *et al* 1996) (data not shown). The highly sensitive technique of RT-PCR was therefore used to examine the expression of these receptors in BMM and splenic cells. Panel A of figure 6.19 shows that no expression of either CCR3 or CCR4 was detected in BMM, either in the presence, or absence of GM-CSF or IL-3. Included were positive controls for CCR3 (lane 8) and CCR4 (lane 12), as well as those for the housekeeping gene GAPDH (lanes 1-4), to show that the RT-PCR had indeed worked successfully and allow some inferences to be made concerning the loading between lanes. A similar observation was made in splenic cells (data not shown).

Panel B of figure 6.19 shows that expression of CCR2, which does not bind MIP-1 α , was detectable in BMM but its expression was not altered by the addition of GM-CSF (compare lane 4 with lane 5). As before, a positive control for CCR2 (lane 6) and internal controls for β -Actin (lanes 1-3) were included.

Panel C of figure 6.19 shows that expression of D6 was also undetectable in BMM, either in the presence or absence of GM-CSF (compare lane 4 with lane 5). Also included was a positive control for D6 (lane 6) and internal controls for β -Actin (lanes 1-3).

6.8.4 TGF-β1 does not markedly alter the expression of CCR1 and CCR5 in BMM

As outlined in previous sections, TGF- β 1 appears to be a potent suppressor of GM-CSF and IL-3 induced MIP-1 α expression. With the knowledge that GM-CSF
Figure 6.19: RT-PCR analysis of the expression of CCR3, CCR4, D6 and CCR2 following stimulation with GM-CSF or IL-3.

BMM were treated for 4 h with GM-CSF and IL-3, total RNA isolated and RT-PCR analysis performed to determine the expression of various CC chemokine receptors as follows:

A) CCR3 and CCR4 expression. Lanes 1, 5 and 9 show reactions performed with control RNA; lanes 2, 6 and 10 show reactions performed with RNA derived from cells treated for 4 h with GM-CSF (10 ng/ml); lanes 3, 7 and 11 show reactions performed with RNA derived from cells treated for 4 h with IL-3 (10 ng/ml); lanes 4, 8 and 12 show positive control reactions performed with cDNAs for GAPDH, CCR3, and CCR4 respectively. Lanes 1-4 show reactions performed with GAPDH primers, whilst lanes 5-8 and 9-12 show reactions performed with primers for CCR3 and CCR4 respectively.

B) CCR2 expression. Lanes 1 and 4 show reactions performed with control RNA; lanes 2 and 5 show reactions performed with RNA derived from cells treated for 4 h with GM-CSF (10 ng/ml); lanes 3 and 6 show positive control reactions performed with cDNAs for β -Actin and CCR2 respectively. Lanes 1-3 show reactions performed with primers for β -Actin, whilst lanes 4-6 show reactions performed with primers for CCR2.

C) D6 expression. Lane assignments as for B) above, except that reactions in lanes 4-6 were performed with primers for D6.

This figure is representative of 3 separate experiments.



B

A







and IL-3 also induce the expression of CCR1, but not CCR5, the effect of TGF- β 1 on CCR1 and CCR5 expression was also examined. Figure 6.20 shows that whilst TGF- β 1 does decrease the basal expression of both CCR1 and CCR5 following addition of GM-CSF or IL-3, the expression of CCR1 was induced as previously in the presence of GM-CSF (compare lanes 2 and 3, and lanes 4 and 5, with that of lane 1). This basal suppression of approximately 10-30 % was minimal in comparison to the 50-80 % suppression observed previously. TGF- β 1 appears to suppress only the basal level of expression of both CCR1 and CCR5. This basal suppression is most obvious when cells treated with TGF- β 1 are compared to control cells (compare lane 6 with the negative control, lane 1), an observation also previously noted for MIP-1 α . These data suggests that whilst TGF- β 1 may act to suppress basal expression of CCR1 and CCR5, it does not appear to markedly alter their expression following stimulation with GM-CSF or IL-3.

6.9 Effect of GM-CSF on MIP-1a Receptor Levels on BMM

To assess whether the observed increase in CCR1 mRNA expression that occurred following GM-CSF stimulation was of functional relevance, the effect of GM-CSF on the BMM cell surface expression of MIP-1 α receptors was examined. BMM were either stimulated with GM-CSF (10 ng/ml) or with PBS / 0.1 % BSA (v/v) for 16 h and then radioligand displacement studies were performed in the presence of 4 nM ¹²⁵I labelled MIP-1 α and variable amounts of cold MIP-1 α competitor. The extended GM-CSF treatment time of 16 h, compared with the 4 h used for Northern blot analysis, was to allow sufficient time for CCR1 protein expression on the cell surface to reflect the mRNA levels. One such experiment is shown in figure 6.21 and shows a Scatchard analysis of displacement of radiolabelled MIP-1a from control and GM-CSF treated BMM. This Scatchard analysis was performed using the Ligand software program, widely used for the analysis of receptor binding data. The plot indicates several things. Firstly, the binding of MIP-1 α to cell surface receptors appeared to occur with relatively similar affinities, with K_d's (dissociation constant)of 0.67 nM and 0.2 nM for control and GM-CSF treated BMM respectively (as indicated by the gradients of the slopes of the lines). Secondly, the number of MIP-1 α receptors increased following GM-CSF treatment, from over 20,000 to 46,000 receptors per

Figure 6.20:

The numbers in the table below show the relative levels of CCR1 and CCR5 mRNA for these 3 experiments, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells. * = data shown in the figure opposite.

	Control	GM-CSF	GM-CSF + TGF-β1	IL-3	Π3+ 	TGF-β1
CCR1	*1.00	*3.10	*2.70	*4.60	*3.60	*0.09
	1.00	2.33	1.60	1.9 0	1.89	0.14
	1.00	4,09	4.05	3.77	1.77	0.73
). 				
Average	1,00	3.17	2,78	3.42	2.42	0.32
CCR5	*1.00	*1.43	*1.06	*1.11	*0.86	*0.47
	1.00	1.10	0.58	0.91	0.65	0.30
	1.00	1.80	0.77	1.34	0.56	0.32
Average	1.00	1,44	0.80	1.12	0.69	0.36



Figure 6.20: Northern blot analysis of the effect of TGF-β1 on the expression of CCR1 and CCR5 mRNA, in the presence of GM-CSF and IL-3.

BMM were treated for 4 h with GM-CSF or IL-3, in the presence or absence of TGF- β 1, as follows: A) CCR1 expression. 1) control; 2) GM-CSF (10 ng/ml); 3) GM-CSF plus TGF- β 1 (20 ng/ml); 4) IL-3 (10 ng/ml); 5) IL-3 plus TGF- β 1 (20 ng/ml); 6) TGF- β 1. B) CCR5 expression. As A) above. Top blot of each panel shows a blot hybridised with a probe for either CCR1 or CCR5, whilst the bottom blot in each panel shows the same blot stripped and reprobed for β -Actin. This figure is representative of 3 separate experiments.



Figure 6.21 Scatchard analysis of the effect of GM-CSF on the binding of ¹²⁵I MIP-1 α to MIP-1 α receptors on BMM.

BMM were treated for 16 h with either a PBS/0.1 % BSA control or with GM-CSF (10 ng/ml). Radioligand displacement studies were then performed either with or without the addition of variable amounts of cold competitor MIP-1 α . To eliminate self-aggregation, the MIP-1 α used in this experiment was the non-aggregating PM2 mutant (Graham *et al* 1994). This analysis represents data derived from a single experiment.



Figure 6.22 Plot of the effect of GM-CSF on BMM MIP-1α receptor numbers.

Data obtained from radiolabelled binding studies (as shown in figure 6.9.1), was analysed using the 'ligand' software package, from which the number of cell surface MIP-1 α receptors was determined (as indicated on the Y axis).

cell, equating to approximately a 130 % increase (summarised in figure 6.22). It was not possible to determine exactly which MIP- α receptor was being elevated on BMM. following GM-CSF treatment, using the binding studies as performed herein. However, given the fact that BMM appear to express only CCR1 and CCR5, it is highly likely that the observed increase in MIP-1a receptor numbers could only be attributable to expression of one of these two receptors. Further support for this is provided by the similarity of K_d values observed in this study, to those previously reported for murine CCR1 (Nibbs et al 1996). Given the fact that the Scatchard analysis suggested only a single class of MIP-1 α receptor, coupled with the previous failure to detect any increase in the expression of other known MIP-1 α receptors, it is most likely that an increase in the level of CCR1 was responsible for the observed increase in number of macrophage MIP-1a cell surface receptors following GM-CSF treatment. It is also theoretically possible however, that this increase could have been due in part to an increase in an as yet unidentified MIP-1 α receptor present on BMM, though the apparent existence of only a single class of MIP-1 α receptor would suggest otherwise.

6.10 Effect of GM-CSF on the Ability of Cells Expressing MIP-1α Receptors to Mobilise Calcium in Response to MIP-1α

6.10.1 Effect of GM-CSF on the ability of BMM to mobilise calcium in response to MIP-1α

The data obtained from the binding studies on BMM, as indicated in the above section, suggested that treatment of BMM with GM-CSF resulted in an increase in the total number of MIP-1 α receptors. This increase was in the region of 130 % and it is certainly feasible to speculate that an increase of this magnitude might be sufficient to result in an alteration in the functional response of GM-CSF stimulated BMM to MIP-1 α , relative to control BMM. One widely used standard for assessing the functional interaction between a chemokine and its receptor, is the ability of the activated receptor to mobilise calcium from intracellular stores within the cytoplasm. This can be measured *in vitro*, using the technique of 'calcium fluxing'. Therefore, the effect of

a 16 h GM-CSF stimulation on the ability of BMM to mobilise calcium was measured, according to previously detailed methods (see methods).

HEK-293 cells transfected with human CCR5, were included as a positive control and to verify the functional ability of the exogenously added recombinant MIP-1 α . Figure 6.23 shows one such experiment. The addition of MIP-1 α (100 ng/ml), as indicated by an arrow, resulted in only a tiny flux peak in BMM and this was observed in both unstimulated and GM-CSF stimulated BMM (Panels B and C). This is more evident when the fluxes obtained with BMM were compared to those obtained with the control HEK-293 cells transfected with hCCR5 (panel A). Although the fluxes did appear to be extremely small, the results of the experiment shown in figure 6.23, suggest that GM-CSF may have a potential effect on the ability of BMM to mobilise calcium in response to MIP-1 α . BMM did flux, with a peak of 6 intensity units (see figure legend 6.23) in the presence of GM-CSF, compared to a peak of 2 units observed for the control BMM. It should be noted however, that the experiment shown in figure 6.23 was the only one to show such an increase and this result was not reproducible. In fact, in most cases BMM failed to flux at all.

This data implies that whilst BMM may have the potential ability to flux calcium and that this may be enhanced by GM-CSF, this increase was small and inconclusive. BMM appear therefore, to be an inappropriate system in which to measure calcium mobilisation in response to MIP-1 α .

6.10.2 GM-CSF induces the expression of CCR1 mRNA in THP-1 cells

As discussed in the above section, BMM appear to be unsuitable cell choice for performing calcium flux studies in response to MIP-1a. Therefore, in order to observed effect of GM-CSF further analyse the on the ability of macrophages/monocytes to mobilise calcium in response to MIP-1 α , another cell type was selected. The human pro-monocytic cell line THP-1 had been previously used with success for MIP-1a calcium flux studies in our laboratory (R.J.B. Nibbs, personal communication) and was therefore considered as a suitable alternative. The effect of a 4 h addition of GM-CSF (10 ng/ml) on the expression of CCR1 mRNA in THP-1 cells was therefore measured by Northern blot analysis and one such experiment is shown in figure 6.24. Densitometric analysis showed that treatment of

Figure 6.23 Effect of GM-CSF stimulation on the ability of BMM to mobilise calcium, in response to MIP-1 α .

BMM were treated with GM-CSF (10 ng/ml) as indicated below and after this time Ca²⁺ studies were performed. A) shows HEK-293 cells transfected with human CCR5, as a positive control; B) control BMM; C) BMM plus 16 h GM-CSF. The point at which MIP-1 α was added is indicated by an arrow. The Y axis shows the intensity of Fura-2-AM emission at 500 nm. This figure is one^A_A3 separate experiments.



B







1

2

β-Actin

Figure 6.24: Northern blot analysis of the effect of human GM-CSF on the expression of human CCR1 mRNA in THP-1 cells.

THP-1 cells were treated as follows: 1) control 2) 4 h with human GM-CSF (10 ng/ml). The top blot shows a blot hybridised with a probe for human CCR1, the bottom panel shows the same blot, stripped and reprobed for β -Actin. This figure is representative of 3 separate experiments. The numbers in the table below show the relative levels of CCR1 mRNA for these 2 experiments, as assessed by scanning densitometric analysis and normalisation against the level of β -Actin mRNA in control treated cells. * = data shown in the figure opposite

	Control	GM-CSF
hCCR1	*1.00	*1.75
	1.00	1.90
	1.00	1.59
Average	1.00	1.75

THP-1 cells with human GM-CSF resulted in approximately a 2-fold increase in the level of CCR1 mRNA expression, which is of a slightly lower magnitude than that observed in BMM. This suggests that THP-1 cells are similar to BMM, in that they respond to GM-CSF by upregulating the expression of CCR1 and therefore may be a more suitable choice of cell, in which to examine the effect of GM-CSF on calcium mobilisation.

6.10.3 GM-CSF Does Not Alter the Ability of THP-1 cells to Mobilise Calcium in Response to MIP-1α

As previously described for BMM, the effect of a 16 h GM-CSF stimulation on the ability of THP-1 cells to mobilise calcium was measured. As before, HEK-293 cells transfected with human CCR5 were included as a positive control (see figure 6.25 panel A). Figure 6.25, shows that unlike BMM, unstimulated THP-1 cells respond well to addition of recombinant MIP-1 α . However, no increase in the ability of THP-1 cells to mobilise calcium was observed following GM-CSF stimulation. GM-CSF treated THP-1 cells responded to MIP-1 α with a peak flux of approximately 20 units, compared to the peak of 23 units observed with control THP-1 cells (compare panel B with panel C). This data suggests that whilst GM-CSF does increase the expression of CCR1, this increased expression does not alter the ability of those cells to mobilise calcium and it is possible therefore that the 2-fold increase in CCR1 expression is not functionally relevant. Although as discussed previously in section 6.10.1, a functional increase in BMM in response to MIP-1 α still remains a theoretical possibility, as the experiments shown in figure 6.25 were performed in THP-1 cells.

6.11 Analysis of MIP-1α Expression in GM-CSF Transgenic Mouse Tissues

6.11.1 Overview: A potential link between MIP-1 α and the pathologies observed in GM-CSF transgenic mice

So far, the results presented in this study have indicated a role for GM-CSF in the regulation of MIP-1 α expression *in vitro*. Therefore, it was decided to examine the possibility that GM-CSF might be an important regulator of MIP-1 α function *in vivo*. As discussed in the introduction, derivation and analysis of GM-CSF transgenic mice has previously been reported (Lang *et al* 1987). These mice are characterised by

Figure 6.25 Effect of GM-CSF stimulation on the ability of THP-1 cells to mobilise calcium, in response to MIP-1 α .

THP-1 cells were treated with human GM-CSF (10 ng/ml) as indicated below and after this time Ca ²⁺ studies were performed. *A*) shows HEK-293 cells transfected with human CCR5, as a positive control; *B*) control THP-1; *C*) THP-1 plus 16 h GM-CSF. The point at which MIP-1 α was added is indicated by an arrow. This figure is representative of 3 separate experiments. The Y axis shows the intensity of Fura-2-AM emission at 500 nm.



С



Time/sec

several unusual phenotypic abnormalities, many of which involve the actions of macrophages. Perhaps the most striking of these, is the macrophage infiltration and defects associated with the abnormal development of the mouse eye in GM-CSF transgenic mice (Cuthbertson and Lang 1989). In GM-CSF transgenic mice, macrophages are autostimulated and produce elevated levels of IL-1a, bFGF and TNF (Lang et al 1992). In addition to the ocular defects, GM-CSF transgenic mice also exhibit infiltration of inflammatory cells into the pleural and peritoneal cavities and within striated muscle. As discussed in the introduction, MIP-1 α is clearly an important mediator of both haemopoietic and inflammatory functions and is known to be expressed in many disease states, particularly within the lungs. Fahey et al (1992) also previously demonstrated that MIP-1 α was capable of increasing the expressing of proinflammatory cytokines, such as IL-1 and TNF. The finding that GM-CSF is a potent stimulator of MIP-1 α expression in macrophages, together with the importance of macrophages as a source of MIP-1 α and the predominant involvement of macrophages in the various pathologies observed in GM-CSF transgenic mice, suggests that MIP-1 α might be expressed at an elevated level in these GM-CSF transgenic mice and maybe involved in the migration of monocytes and macrophages in this animal model. It is therefore possible that MIP-1 α could be involved in the pathologies observed in these transgenic mice. The expression of MIP-1 α in selected tissues of wild type and GM-CSF transgenic mice was therefore examined, to determine if the expression of MIP-1a was elevated in transgenic mice relative to wild type mice.

6.11.2 RT-PCR analysis of MIP-1 α expression in GM-CSF transgenic tissues

Using mRNA isolated from tissue samples frozen in liquid nitrogen immediately following excision, the highly sensitive technique of RT-PCR analysis was used to investigate the expression of MIP-1 α mRNA in eye, bone marrow, leg muscle and peritoneal cells derived from wild type and GM-CSF transgenic mice. The expression of the house keeping gene β -Actin was also examined in parallel, in an attempt to determine the relative amount of RNA input in each reaction. To allow any observed changes in expression to be more accurately assessed and observed during the log phase of amplification, samples were analysed after 15, 20, 25, 30 and 35 cycles by agarose gel electrophoresis and subsequent Southern blotting of gels. Densitometric analysis was performed on the resulting autoradiographs and the level of MIP-1 α expression determined in GM-CSF transgenic samples, relative to wild type samples and to the level of β -Actin expression observed.

6.11.3 MIP-1 α expression appears elevated in the eyes of GM-CSF transgenic mice

Panel A of figure 6.26 shows the expression of MIP-1 α in wild type and transgenic mouse eye, as detected by RT-PCR analysis, whilst panel B below shows the expression of β -Actin. RT-PCR analysis revealed that expression of MIP-1 α was only barely visible in wild type eye after 35 cycles of PCR, although an additional band approximately 120 bp larger than expected appeared to be amplified in the wild type eye after 30-35 cycles (indicated by the arrows in figure 6.26, panel A; lanes 4 and 5). Expression of bands of the expected size, but not of the larger size observed in the wild type eye, were however clearly visible in the GM-CSF transgenic eye after 30 to 35 cycles (see figure 6.26, panel A; lanes 9 and 10; position indicated by lower arrow). Also included was a BMM RNA positive control for MIP-1 α (figure 6.26, panel A; lane 11). Figure 6.26, panel B; shows the expression of β -Actin to be very similar between the wild type and transgenic samples. The weak, but nevertheless detectable, expression of MIP-1 α observed in the wild type eye, was highly unexpected and the source of this expression remains unknown.

Figure 6.27 shows a subsequent Southern blot analysis of the gels shown in figure 6.26, with probes specific for MIP-1 α (panel A) and β -Actin (panel B). Analysis of these blots, in conjunction with the results shown in figure 6.26, revealed several things. Firstly, the MIP-1 α specific probe appeared only to hybridise to the band corresponding to the expected size of MIP-1 α and the larger band observed in the wild type eye by RT-PCR analysis was not detectable, even after prolonged exposure of the autoradiographs (see figure 6.27, panel A; lanes 4 and 5). Secondly, in support of the similar observation made in the RT-PCR analysis, the level of MIP-1 α expression appeared to be elevated in the GM-CSF transgenic eye when compared to that of the wild type (see figure 6.27, panel A; compare lanes 9 and 10 with those of lanes 4 and 5). Thirdly, some hybridisation was visible to bands corresponding to the

Figure 6.26: RT-PCR analysis of MIP-1 α expression in wild type and GM-CSF transgenic mouse eye.

Panel A, shows the results of an RT-PCR analysis for the detection of MIP-1 α mRNA, resolved by agarose gel electrophoresis. The arrow to the left of the gel indicates the position of the MIP-1 α mRNA product (expected size of ~267 bp), whilst the arrow above this, marked with a ?, indicates the relative position of a second unidentified band. Lanes 1-5 show analysis of wild type mouse eye and 6-10 that of GM-CSF transgenic eye. They represent PCR reactions performed for 15, 20, 25, 30 and 35 cycles respectively. Lane 11 is a positive control for MIP-1 α , that shows a reaction performed with BMM RNA, whilst lane 12 shows a positive control for the PCR reaction performed using genomic DNA. Panel B, shows an identical analysis, performed in parallel, using primers specific for the detection of β -Actin. The lane assignments for lanes 1-10 are as indicated for A above. This figure represents a single experiment.





B



Figure 6.27: Southern blot analysis of MIP-1 α and β -Actin expression detected by RT-PCR analysis, in wild type and GM-CSF transgenic mouse eye.

Southern blot analyses were performed on the agarose gels shown in figure 6.26. *Panel A*, shows a blot hybridised with a probe for MIP-1 α . The lane assignments for lanes 1-11 are exactly as in the legend for figure 6.26. The lower arrow to the left of the blot indicates the position of the MIP-1 α band, the upper arrow (marked with GC) indicates the expected size of genomic DNA (~1 Kb). *Panel B*, shows a blot hybridised with a probe for β -Actin. The lane assignments for lanes 1-10 are as indicated in the legend for figure 6.26. The numbers in the table below represent an approximation of the relative levels of MIP-1 α product, as assessed by scanning densitometric analysis and normalisation to calculated β -Actin levels. They represent data from a single experiment.

	Wild Type	GM-CSF transgenic
30 Cycles	1.0	2.14
35 Cycles	1.0	5.50

A

1 2 3 4 5 6 7 8 9 10 11

GC →

MIP-1 $\alpha \rightarrow$

B

1 2 3 4 5 6 7 8

β–Actin –



10

expected size of genomic DNA, suggestive of contamination (see figure 6.27 panel A lanes 4 and 5). Indeed, very weak expression of bands corresponding to the expected size of genomic MIP-1 α were visible on the original gel, though these are not in fact visible in the image shown in figure 6.26. Figure 6.27; panel B, shows that expression of β -Actin was similar between wild type and GM-CSF samples, suggesting that the initial amount of RNA input to the RT-PCR reactions was very similar. However, it should be noted that the observed expression of β -Actin, as shown in figure 6.27, does not exactly reflect that observed in the gel analysis shown in figure 6.26. It is possible that this reduction may be the result of incomplete transfer of DNA from the gel, during the blotting process. Densitometric analysis showed that this increase in MIP-1 α expression was between approximately 2 and 6-fold. Attempts to sub-clone and sequence the larger sized band observed only in the wild type eye, proved unsuccessful and therefore the exact nature of this unidentified band remains unknown.

In summary, these results suggested that the expression of MIP-1 α mRNA may be elevated in the eyes of GM-CSF transgenic mice.

6.11.4 MIP-1 α expression may be elevated in the bone marrow of GM-CSF transgenic mice

Insufficient bone marrow material was available to allow, either the derivation of BMM, or the sorting of cell populations from total bone marrow. Therefore, the expression of MIP-1 α was examined by RT-PCR analysis using total bone marrow cells derived from wild type and GM-CSF transgenic mice, in a manner identical to the analysis performed in section 6.11.3. Products were resolved as before, by agarose gel electrophoresis and figure 6.28 shows the results of this analysis. Panel A shows the expression of MIP-1 α , whilst panel B shows that of β -Actin. Expression of MIP-1 α was just visible in wild type bone marrow after 30 cycles of PCR and was clearly visible after a further 5 cycles (figure 6.28, panel A; lanes 4 and 5). In GM-CSF transgenic bone marrow this expression appeared to be very slightly elevated relative to the expression observed in wild type samples, with a small increase after 30 cycles and a more visibly evident increase after 35 cycles (see figure 6.28, panel A; compare lane 4 with lane 9, and lane 5 with lane 10). Also included was a genomic control

(lane 11) and a positive PCR control (lane 12). It should also be noted that some genomic MIP-1 α expression appears evident in wild type bone marrow samples. though this is not evident from the gel as reproduced here. The presence of this may have resulted in sequestration of reaction components and thus a reduced amplification relative to that of GM-CSF transgenic samples. Panel B of figure 6.28 shows that the expression of β -Actin was not the same in wild type and GM-CSF transgenic samples and the reason for this discrepancy is unclear. Figure 6.29 shows a Southern blot analysis of the same gels as shown in figure 6.28. Unexpectedly, this shows that the greatest level of MIP-1 α expression appeared to occur after 30 cycles, and was actually decreased after 35 cycles (see figure 6.29, panel B; lanes 4-5 and 9-10). The expression appeared to be increased in GM-CSF transgenic bone marrow, relative to that observed in the wild type (compare lanes 4 and 9). Lane 11 shows the positive control (previously lane 12 in figure 6.28). Panel B of figure 6.29 shows the expression of β -Actin and clear differences in the pattern of β -Actin expression are evident, compared to that previously observed in panel B of figure 6.28. The reasons for the inconsistencies observed in this experiment are inexplicable and it is therefore difficult to draw any firm conclusions from this data.. It should be born in mind however, that this experiment could only be performed once due to the limited amount of sample material available and these results are therefore clearly not conclusive. It is possible however, that the results obtained from the RT-PCR analysis are indeed correct. If this were the case, then this would indicate that an elevated level of MIP-1 α expression may occur in the bone marrow of GM-CSF transgenic mice, compared to that observed for wild type. If true this result would be surprising, in the light of the reported observation that the GM-CSF transgene was not detectable in the bone marrow of GM-CSF transgenic mice (Lang et al 1987). However, it should be noted that elevated levels of GM-CSF were previously reported to be detectable in the serum of GM-CSF transgenic mice compared to control mice (Lang et al 1987) and this could provide an explanation for any increase in MIP-1 α expression that might theoretically occur. Due to the very limited availability of sample material and the inconclusive results obtained, this experiment should clearly be repeated with fresh material.

Figure 6.28: RT-PCR analysis of MIP-1 α expression in wild type and GM-CSF transgenic mouse bone marrow.

Panel A, shows the results of an RT-PCR analysis for the detection of MIP-1 α mRNA, resolved by agarose gel electrophoresis. The arrow to the left of the gel indicates the position of the MIP-1 α mRNA product. Lanes 1-5 show analysis of wild type mouse bone marrow and 6-10 that of GM-CSF transgenic bone marrow. They represent PCR reactions performed for 15, 20, 25, 30 and 35 cycles respectively. Lane 11 shows a positive control for the PCR reaction performed using genomic DNA, whilst the reaction shown in lane 12 is a positive control for MIP-1 α performed using MIP-1 α cDNA. Panel B, shows an identical analysis, performed in parallel, using primers specific for the detection of β -Actin. The lane assignments for lanes 1-10 are as indicated for A above. This figure represents a single experiment.



Figure 6.29: Southern blot analysis of MIP-1 α and β -Actin expression detected by RT-PCR analysis, in wild type and GM-CSF transgenic bone marrow.

Southern blot analyses were performed on the agarose gels shown in figure 6.28 *Panel A*, shows a blot hybridised with a probe for MIP-1 α . The lane assignments for lanes 1-10 are exactly as in the legend for figure 6.28, whilst lane 11 represents the positive control performed using MIP-1 α cDNA. The lower arrow to the left of the blot indicates the position of the MIP-1 α band, the upper arrow (marked with GC) indicates the expected size of genomic DNA. *Panel B*, shows a blot hybridised with a probe for β -Actin. The lane assignments for lanes 1-10 are as indicated in the legend for figure 6.28. The numbers in the table below represent an approximation of the relative levels of MIP-1 α product, as assessed by scanning densitometric analysis and normalisation to calculated β -Actin levels. They represent data from a single experiment.

	Wild Type	GM-CSF transgenic
30 Cycles	1.0	5.60
35 Cycles	1.0	0.60



1 .

6.11.5 Expression of MIP-1 α does not appear elevated in striated muscle or peritoneal cells of GM-CSF transgenic mice

Results obtained from eye samples derived from GM-CSF transgenic mice indicated that MIP-1 α expression might be elevated in these mice, compared to wild type mice. Whilst these results were only preliminary and not conclusive, they did however suggest that examination of other tissues was warranted. The expression of MIP-1 α was therefore examined in samples of leg muscle and peritoneal cells, since macrophage infiltrates and dramatically increased numbers of cells have been observed in the striated muscle and peritoneal cavities of transgenic mice respectively.

Figure 6.30 shows an RT-PCR analysis of the expression of MIP-1 α (panel A) and β -Actin (panel B) in normal and GM-CSF transgenic leg muscle. A similar level of MIP-1 α expression was observed in GM-CSF transgenic leg muscle, compared to that of wild type (see figure 6.30, panel A; compare lanes 1-4 with lanes 5-8). Parallel analysis of β -Actin expression revealed that this expression was relatively similar between normal and transgenic leg muscle (see figure 6.30, panel B; compare lanes 1-4 with lanes 5-8). Positive controls were included in lanes 9-11 and 9-10 of panels A and B respectively. A Southern blot of the above experiment confirmed those observations, but also suggested that there was slight contamination of samples with genomic DNA (see figure 6.31) and this was particularly evident in the GM-CSF transgenic leg muscle. The impact of this contamination on a particular RT-PCR reaction can not be assessed, but it remains a possibility that heavy genomic contamination may have sequestered primers and therefore reduced the amplification potential of that particular RT-PCR reaction for available MIP-1 α mRNA templates.

Figure 6.32 shows an RT-PCR analysis of the expression of MIP-1 α (panel A) and β -Actin (panel B) in normal and GM-CSF transgenic peritoneal cells (PC). Visible MIP-1 α expression was only observed in wild type PC after 35 cycles of amplification (see figure 6.32, panel A; lane 5). Parallel analysis of β -Actin expression revealed that the expression was lower in transgenic PC compared to wild type PC (see figure 6.32, panel B; compare lanes 6-10 with lanes 1-5). These differences in β -Actin expression may explain why no MIP-1 α expression was observed in the RT-PCR analysis of GM-CSF transgenic PC. Positive controls were included in lanes 11 and 12. Figure 6.33 shows a Southern blot analysis of the same experiment, with panel

Figure 6.30: RT-PCR analysis of MIP-1 α expression in wild type and GM-CSF transgenic mouse leg muscle.

Panel A, shows the results of an RT-PCR analysis for the detection of MIP-1 α mRNA, resolved by agarose gel electrophoresis. The arrow to the left of the gel indicates the position of the MIP-1 α mRNA product. Lanes 1-4 show analysis of wild type mouse leg muscle and 5-8 that of GM-CSF transgenic leg muscle. They represent PCR reactions performed for 20, 25, 30 and 35 cycles respectively. Lane 9 is a positive control for the RT reaction performed using BMM RNA, lane 10 is a positive control for the PCR reaction performed using genomic DNA, whilst lane 11 is another positive control performed using genomic DNA, *Panel B*, shows an identical analysis, performed in parallel, using primers specific for the detection of β -Actin. The lane assignments are as indicated for A above, with the exception of positive controls for the PCR performed with genomic DNA (lane 9) and β -Actin cDNA (lane 10). This figure represents a single experiment.

A



B



β-Actin →

Figure 6.31: Southern blot analysis of MIP-1 α and β -Actin expression detected by RT-PCR analysis, in wild type and GM-CSF transgenic leg muscle.

Southern blot analyses were performed on the agarose gels shown in figure 6.12.4.1. *Panel A*, shows a blot hybridised with a probe for MIP-1 α . The lane assignments for lanes 1-9 are exactly as in the legend for figure 6.30, with the exception of lanes 10 and 11 which have not been included. The lower arrow to the left of the blot indicates the position of the MIP-1 α band, the upper arrow (marked with GC) indicates the expected size of genomic DNA. *Panel B*, shows a blot hybridised with a probe for β -Actin. The lane assignments for lanes 1-8, are as indicated in the legend for figure 6.30. The numbers in the table below represent an approximation of the relative levels of MIP-1 α product, as assessed by scanning densitometric analysis and normalisation to calculated β -Actin levels. They represent data from a single experiment.

	Wild Type	GM-CSF transgenic	
35 Cycles	1.0	0.60	



Figure 6.32: RT-PCR analysis of MIP-1α expression in wild type and GM-CSF transgenic mouse peritoneal cells.

Panel A, shows the results of an RT-PCR analysis using primers specific for the detection of MIP-1 α mRNA, resolved by agarose gel electrophoresis. The arrow to the left of the gel indicates the position of the MIP-1 α mRNA product. Lanes 1-5 show analysis of wild type mouse peritoneal cells and 6-10 that of GM-CSF transgenic peritoneal cells. They represent PCR reactions performed for 15, 20, 25, 30 and 35 cycles respectively. Lanes 11 is a positive control for the PCR reaction performed with genomic DNA, whilst lane 12 shows a positive control reaction for MIP-1 α performed using MIP-1 α cDNA. Panel B, shows an identical analysis, performed in parallel, using primers specific for the detection of β -Actin. The lane assignments for lanes 1-10 are as indicated for A above. This figure represents a single experiment. 1 2 3 4 5 6 7 8 9 10 11 12

1 2 3 4 5 6 7 8 9 10



A

B

n 🔸

Figure 6.33: Southern blot analysis of MIP-1 α and β -Actin expression detected by RT-PCR analysis, in wild type and GM-CSF peritoneal cells.

Southern blot analyses were performed on the agarose gels shown in figure 6.32 *Panel A*, shows a blot hybridised with a probe for MIP-1 α . The lane assignments for lanes 1-10 are exactly as in the legend for figure 6.32. The arrow to the left of the blot indicates the position of the MIP-1 α band. *Panel B*, shows a blot hybridised with a probe for β -Actin. The lane assignments for lanes 1-10 are as indicated in the legend for figure 6.32. The numbers in the table below represent an approximation of the relative levels of MIP-1 α product, as assessed by scanning densitometric analysis and normalisation to calculated β -Actin levels. They represent data from a single experiment.

	Wild Type	GM-CSF transgenic	
35 Cycles	1.0	0.60	



1 2 3 4 5 6 7 8 9 10

β-Actin -

A

B
A showing MIP- α expression and panel B that of β -Actin. This indicates that some expression of MIP-1 α was visible in both normal and GM-CSF transgenic PC after 35 cycles, although this GM-CSF transgenic expression was not visible in the RT-PCR analysis as previously shown in figure 6.32. The southern blot analysis of β -Actin expression (as shown in figure 6.33) support the RT-PCR analysis of β -Actin expression, confirming lower β -Actin expression in the GM-CSF transgenic PC compared to wild type PC. This suggests that differences in the original amount of input mRNA may have occurred during the setting-up of the RT-PCR reactions. No genomic DNA contamination was detected in any of the lanes. Densitometic analysis of the levels of MIP-1 α expression (see figure 6.33, panel A; compare lane 5 with lane 10) with those of β -Actin expression (see panel B) revealed that essentially the same levels of MIP-1 α expression occurred in PC from GM-CSF transgenic mice, as did in PC from normal mice.

In summary, these results indicate that whilst the level of MIP- α expression may be elevated in the eye and bone marrow of GM-CSF transgenic mice, compared to wild type mice, the level does not appear to be elevated in the leg muscle and peritoneal cells. It should be noted, that due to the very limited amount of sample material, these experiments were only performed once and therefore the analysis of the results can not be totally conclusive. They do however suggest that elevated levels of MIP-1 α may exist in specific tissues within GM-CSF transgenic mice, compared to normal mice and a more extensive analysis of these observations would perhaps be informative. In the light of the preliminary observations indicated previously in section 6.11, it was decided to further analyse the expression of MIP-1 α in GM-CSF transgenic tissues, using the technique of *in situ* hybridisation. Analysis of frozen tissue sections from eye, leg muscle and peritoneal cells was performed using this technique, but unfortunately proved uninformative.

CHAPTER 7: DISCUSSION

7.1 Introduction

One of the most obvious conclusions resulting from the research that has accumulated over the past 20 years or so, into the functions of the multitude of haemopoietic growth factors and cytokines that act upon the immunohaemopoietic system, is the enormous complexity of the regulatory mechanisms in which they are involved. As discussed previously in section 1.8, there exists a large interacting network of cytokines that exhibit a large degree of functional overlap and potential for redundancy. Our laboratory is particularly interested in the role of MIP-1 α in the regulation of stem cell proliferation and previous research has attempted to dissect the interactive relationship between MIP-1 α and another inhibitor of haemopoietic stem cell proliferation, TGF- β 1. As a result of this investigation it is now believed that MIP- 1α and TGF- β 1 have an endogenous, reciprocal relationship in the bone marrow, in which TGF- β 1 probably functions to suppress the expression of MIP-1 α (Maltman et al 1993, 1996). A study by Cluitmans et al (1995) has however, demonstrated the simultaneous presence of both TGF- β 1 and MIP-1 α in the bone marrow and peripheral blood of normal individuals. Clearly therefore, in addition to suppressive factors such as TGF- β 1, the expression of MIP-1 α also must be regulated by positively acting factors. To date, our understanding of the regulatory mechanisms controlling the expression of MIP-1 α is minimal, particularly with regard to its regulation within the bone marrow. The focus of this thesis has therefore been to attempt to further understand the regulation of MIP-1 α expression, with a particular emphasis on its role within this complex network of regulatory cytokines. One very important point that must be considered in such an investigation, is that MIP-1 α (as well as many other cytokines involved in this network) functions as both a haemopoietic and inflammatory regulator. Any conclusions resulting must therefore consider both these aspects, as conclusions regarding inflammatory functions may be informative about haemopoietic functions and vice versa. For the purposes of this discussion, although inflammation is a natural part of haemopoiesis, they will be considered as being distinct. Thus in this discussion, the term haemopoiesis will refer to constitutive and inducible haemopoiesis, such as that occurring during the generation of mature end cells. The term

inflammation will be used in the context of 'inflammatory functions', such as histamine release, phagocytosis, etc. and particularly with regard to mature immune cells.

The primary source of MIP-1 α within the bone marrow is considered to be the macrophage. Previous studies in our laboratory have successfully used the well characterised bone marrow-derived macrophage (BMM) and this was therefore selected as a cell system in which to examine the regulation of MIP-1 α expression. The initial investigations sought to identify haemopoietic cytokines as potential regulators of MIP-1 α expression. The results shown in figure 6.1, demonstrate that neither G-CSF, IL-6, nor LIF had any visible effects on the expression of MIP-1 α in BMM. Receptors for LIF, G-CSF and IL-6 have all been reported to be expressed on macrophages (Hilton et al 1988, Shieh et al 1991, Lotem and Sachs 1987. This suggests that the lack of response of BMM to LIF, G-CSF and IL-6 is not due to the absence of expression of receptors for these cytokines. In contrast however, figure 6.2 shows that GM-CSF did affect MIP-1a expression, resulting in a marked increase in the levels of MIP-1 α mRNA. Given the many known haemopoietic and inflammatory functions of GM-CSF, particularly with regard to cells of the monocyte / macrophage lineage, this observation was of considerable interest. It was therefore decided to examine the role of GM-CSF and related cytokines in the regulation of MIP-1a expression, in greater detail.

7.2 Induction of MIP-1a Expression by GM-CSF and IL-3

Initial experiments suggested that GM-CSF is a strong inducer of MIP-1 α expression and to confirm this observation, further investigation was therefore performed. The results shown in figures 6.3 and 6.4, demonstrate that GM-CSF rapidly induces MIP-1 α expression and that this is dose dependent. It is likely that this induction occurs directly, rather than via induction of a secondary cytokine, since the induction is visible within 30 minutes. This is however, merely suggestive and remains to be determined. The observed induction even at a low concentration of GM-CSF, suggests that GM-CSF is relatively potent in its effects. Figure 6.5 demonstrates that the continued presence of either serum or L929 CM is not a prerequisite for the inductive effects of GM-CSF stimulation. This implies that GM-CSF is not synergising with additional factors present in the serum or L929 CM. Kasama *et al*

(1993) previously demonstrated that GM-CSF was able to synergise with LPS to increase the level of MIP-1 α expression in neutrophils, but that GM-CSF alone was unable to induce MIP-1 α expression over a wide range of concentrations. This is in contrast to the observations shown in figures 6.2 to 6.5, in which GM-CSF rapidly induces MIP-1 α expression over a wide range of concentrations. The fact that addition of a GM-CSF-specific neutralising antibody resulted in the substantial abolishment of the GM-CSF induced expression of MIP-1a expression, shows that GM-CSF was indispensable for this induction. Furthermore, because these experiments were carried out under conditions which were designed (as far as possible) to be essentially LPS free, it is therefore unlikely that LPS plays a role in this observed induction of MIP-1 α mRNA by GM-CSF. The contrast in these results to those of Kasama et al (1993), suggest that clear differences exist between neutrophils and macrophages in the GM-CSF regulated expression of MIP-1 α expression. It should also be noted that their study was performed using human neutrophils, whilst the macrophages used in this study were murine in origin. Although the promoters of murine MIP-1 α and the various forms of human MIP-1a (LD78) are similar (Widmer et al 1991), with approximately 75 % homology within the first 350 bp of proximal promoter sequences, it is possible that differences may exist in the regulatory control of MIP- α and LD78 acting in the two species. If so, this could be one explanation for the observed differences between this study and that of Kasama et al (1993).

It is recognised that increases in mRNA expression are not always followed by a corresponding increase in protein expression, which casts doubt on the physiological and functional relevance of any changes in mRNA levels that are not reflected at protein level. Efforts were therefore made to examine the expression of MIP-1 α protein, following GM-CSF stimulation. The observation that MIP-1 α protein was only detectable following concentration of BMM supernatants (see figure 6.7) was in keeping with previous data from our laboratory by Maltman *et al* (1993) who previously reported that the expression of MIP-1 α protein in unconcentrated BMM medium was undetectable using Western blot analysis. Whilst it is unclear whether or not full translation of MIP-1 α mRNA into protein occurs in BMM following GM-CSF stimulation, the level of protein seems low compared to the level of mRNA. It is theoretically possible therefore that for full translation of mRNAs to occur, an additional signal is required. Indeed there is evidence that such mechanisms may operate for the CC chemokine RANTES and other cytokines, during the induction phase of T cell anergy (Schall et al 1992). Certainly one can speculate that it may be desirable to limit expression of a protein with potent inflammatory effects, such that this expression only occurs when strictly needed. Indeed one can stretch this hypothesis to suggest that expression could be limited to specific niches and locations. One possible hypothetical mechanism for such control, could be through the action of adhesion molecules and other matrix associated molecules. Expression of these could be limited to particular niches or circumstances, such that only when a cell is interacting with them, do they provide the necessary secondary stimulus. Therefore, in the context of GM-CSF stimulation (for example in the immediate locale of a inflammatory infection), GM-CSF would act to prime cells for increased expression of MIP-1 α , but this would only occur following the appropriate secondary signal. It is clear however, that some increased protein expression does result from GM-CSF stimulation and suggests that regulation of MIP-1 α expression by GM-CSF is physiologically relevant.

As indicated in the introduction, GM-CSF is in fact a member of a small subfamily of cytokines and shares extensive functional homology with IL-3, as well as the shared usage of the common β -subunit in their respective receptors. In fact GM-CSF and IL-3 receptors have previously been suggested to be co-ordinately regulated in macrophages (Fan et al 1992), providing further evidence for linkage in their actions. The reported overlap in specific functions of GM-CSF and IL-3 suggested that IL-3 might also therefore be a candidate regulator of MIP-1 α expression. Figures 6.8 A and 6.8 B, demonstrate that IL-3 did indeed appear to exert a very similar effect upon MIP-1a expression, as did GM-CSF. The kinetics of this expression were identical and as with GM-CSF, the dose of IL-3 required to exert a visible increase in MIP-1a expression was low. Thus, both GM-CSF and IL-3 appear to be strong inducers of MIP-1 α expression. One interesting possibility, reported during the writing of this thesis, was evidence for the expression of a pre-formed human GM-CSF receptor complex that could be activated by GM-CSF, IL-3 or IL-5 (Woodcock et al 1997). This suggests that it is theoretically possible that the observed induction of MIP-1 α expression by IL-3 could be occurring through such a pre-formed receptor

complex, rather than through the expected IL-3 receptor complex., GM-CSF As indicated below, IL-5 was unable to effect MIP-1 α expression, which suggests that this pre-formed receptor complex is not present on BMM. However it is unknown if a similar preformed murine receptor complex actually exists.

7.3 Effect of IL-5 on MIP-1a Expression

Given the shared ability of GM-CSF and IL-3 to induce MIP-1a expression, their shared usage of the common β subunit and because IL-5 is a member of this same subfamily, with some similar functions and that it also signals through the common β subunit, it was considered a possibility that IL-5 may also be able to regulate MIP-1 α expression, in a similar fashion to that observed for GM-CSF and IL-3. However, the results presented in figures 6.9 to 6.11, suggest that IL-5 does not regulate MIP-1 α expression in macrophages, or other cells within the murine bone marrow. That IL-5 does not effect MIP-1 α expression in macrophages is perhaps obvious and is clearly due to the absence of functional IL-5 receptors. Indeed, Takagi et al (1995) previously reported that the lineage specificity of IL-5 was mainly due to the limited expression of IL-5R α chain. Results not presented in this thesis have suggested that IL-5 has no apparent effect on MIP-1a expression in murine B cells lines. It is possible however, that in the B cell lines examined MIP-1 α may be transcriptionally silent. In hindsight, a more appropriate model would perhaps have been to analyse the expression of MIP-1 α in primary human eosinophils. Attempts were in fact made to perform this analysis, though they were unsuccessful, due to the logistical difficulties involved in obtaining samples, particularly from patients with eosinophilia. Therefore, the question of whether IL-5 functions as a regulator of MIP- 1α expression remains unanswered. Recent evidence from a report by Li *et al* (1996) does suggest however, that IL-5 may have some limited effects on MIP-1 α expression in basophils, though this was reported to be not significant.

7.4 Induction of other CC chemokines by GM-CSF and IL-3

Previously, Takahashi *et al* (1993) showed that in monocytes, both GM-CSF and IL-3 were potent inducers of IL-8 expression, another member of the chemokine superfamily. This suggested the possibility that additional members of the chemokine superfamily might also be regulated by GM-CSF and IL-3. As can be seen in figures 6.12 and 6.13, expression of the CC chemokines MIP-1B, JE and MARC were all elevated following GM-CSF and IL-3 treatment. Current evidence suggests that that both JE/MCP-1 and MARC/MCP-1 are functionally important as inflammatory cytokines (Huffnagle et al 1995, Stafford et al 1997). This would suggest that their induction by GM-CSF and IL-3 is related to a role in inflammatory functions, though it should be noted that this does not preclude the involvement of GM-CSF and IL-3 in regulating MIP-1 α expression in the context of haemopoiesis. Several studies have attempted to define regulatory elements involved in controlling the expression of various chemokine genes, such as MIP-1a, MCP-1, RANTES and MARC (Grove and Plumb 1993, Ray et al 1997, Moriuchi et al 1997, Jarmin et al 1994). Current evidence suggests that many of these genes may share the presence of common transcription factor binding sequences within their promoters. It is certainly feasible therefore, that factors such as NF-kB, previously identified as being potentially involved in regulating the expression of MIP-1a and other CC chemokines (Grove and Plumb 1993, Moriuchi et al 1997, Ueda et al 1997), may be involved in the regulation of other genes within the same family. If this were indeed the case, it is certainly possible to hypothesise that one important function of GM-CSF and IL-3, may be to nonspecifically upregulate chemokine expression. As discussed earlier, this may relate to a potential priming role of GM-CSF and IL-3. The regulatory control of the GM-CSF gene appears to be fairly complex. For instance in T cells, multiple signals appear to be required for GM-CSF expression to occur and it is evident that elements containing binding motifs for transcription factors such as ETS1, AP1, NFAT, NFATp and NF- κB and NF- κB -like are just a few of those that have been implicated in the regulation of GM-CSF expression in a variety of cell types (Jenkins et al 1995, Thomas et al 1997). Indeed, mice deficient in c-rel (a subunit of the NF-kB-like family of transcription factors) exhibit defects in the production of GM-CSF and IL-3 (Gerondakis et al 1996), underlining the importance of this family of transcription factors in the regulation of GM-CSF and IL-3 expression. Also of interest is the observation that some of these factors have been shown to be involved in the regulation of chemokine expression. Examples include NFAT, in the control of MARC expression in allergically triggered murine mast cells (Jarmin et al 1994,

Prieschel *et al* 1995) and as already mentioned above, NF- κ B in the regulation of other chemokines. If these transcription factors are involved in the regulation of both GM-CSF or IL-3 and the genes they induce, then this could provide additional amplification.

Interestingly, direct support for the observation that GM-CSF and IL-3 were potent inducers of selective chemokine gene expression in BMM had been previously reported, in a very similar study by Orlofsky et al (1994). In this report, the authors investigated the effects of GM-CSF on expression of the CC chemokine C10 in BMM and also noted that expression of MIP-1a, JE and RANTES was induced in BMM. There are several discrepancies observed, between results reported in this study and that of Orlofsky et al (1994). They reported that expression of MIP-1 α was only weakly induced by GM-CSF, whilst in contrast the results reported in this work show that expression of MIP-1 α mRNA was markedly induced by GM-CSF. One possible explanation for this is that Orlofsky et al stimulated their BMM for 24 h with GM-CSF, whereas the results reported herein were achieved with only 4 h stimulation. The results discussed in section 6.3.2 suggest that MIP-1 α expression had declined within 6 h, following GM-CSF stimulation. It is therefore quite possible that expression of MIP-1 α in response to GM-CSF had declined by 24 h, as reported by Orlofsky et al. Another contrast, is that in this study no expression of RANTES was detectable in BMM, in concordance with the previous study of Maltman et al (1996), whilst Orlofsky and colleagues reported that RANTES was also induced by GM-CSF. However, they did acknowledge that this induction could have been due to LPS contamination. During the course of all experiments performed and reported in this thesis, precautions were taken to ensure that as far as was possible, no LPS was present. The authors also reported similar effects by IL-3, but presented no data. Whilst some discrepancies exist between this study and that of Orlofsky et al, probably due to experimental differences, they do both suggest that GM-CSF and IL-3 are stimulators of chemokine expression. Moreover, the results presented herein extend previous observations, to include the novel observation that MIP-1 β and MARC are both induced by GM-CSF and IL-3. Whilst it is probable that the widespread expression of chemokines in response to GM-CSF and IL-3 is likely to be the result of a role in an inflammatory context, confirmation of this may only occur when the effects of GM-CSF and IL-3 on the expression of a much wider range of chemokines are examined. In fact one such novel chemokine, known as MPIF-2, was recently cloned and reported to be weakly induced in GM-CSF-treated macrophages (Patel *et al* 1997), providing further evidence that this GM-CSF/IL-3 mediated induction may be relatively widespread amongst members of the chemokine superfamily.

7.5 TGF- β 1 Suppression of Members of the MIP-1 α Superfamily

As discussed in section 3.6.2, previous work in our laboratory has discovered that an endogenous, reciprocal relationship exists in BMM, between TGF-B1 and MIP-1 α . In the light of the findings raised during the course of the work reported in this thesis, the effect of TGF- β 1 on GM-CSF and IL-3 induced MIP-1 α expression was therefore of significant interest. The results in figures 6.14 and 6.15 confirmed that TGF- β 1 was a suppressor of GM-CSF and IL-3 induced MIP-1 α expression. This suppression was however not complete, suggesting that GM-CSF and IL-3 might be sufficiently potent in their effects to overcome the endogenous block exerted by TGF- β 1 on MIP-1 α expression. However, it should be noted that it is also equally possible to speculate the reverse, namely that TGF- β is sufficient enough to at least partially overcome the inductive stimulus of GM-CSF or IL-3. Either of these may hold true, and they are not necessarily exclusive. This finding was however of significant interest, given the previous report of Cluitmans et al (1995), suggesting that some MIP-1 α expression was detectable in the bone marrow, despite the presence of TGF- β 1. It is therefore possible that in vivo GM-CSF and IL-3 may act to overcome this TGF- β 1 mediated block. However, the relationship between TGF- β , GM-CSF, IL-3 and MIP-1 α in vivo, is likely to be dynamic and the precise level of MIP-1 α expression may therefore depend upon the relative levels of GM-CSF, IL-3 and TGF- β at any given time. Indeed, the constitutive expression of GM-CSF within the haemopoietic microenvironment has been reported to be somewhat variable. Some reports have suggested that constitutive GM-CSF expression does not occur (Cluitmans et al 1995), whilst others have documented expression of GM-CSF by adherent cells within bone marrow cultures (Charbord et al 1991). Clearly however, GM-CSF is expressed by a wide variety of cell types and in response to a variety of inducing stimuli. More certain is the observation that constitutive IL-3 expression is undetectable. However, as discussed previously in the introduction, it is possible that such expression may occur in localised niches within the microenvironment or following conditions of extreme stress.

The observations made in figure 6.16, that TGF- β 1 also suppressed JE and MARC expression following GM-CSF stimulation suggests, that the suppressive effect of TGF- β 1 is more widespread and therefore probably represents the well recognised ability of TGF- β 1 to function as an immuno-suppressive factor. However, this observation does not preclude the possibility that one role of TGF- β 1 may be to suppress the induction of chemokine expression in the bone marrow microenvironment, under non-inflamed conditions.

7.6 Expression of CCR1 in BMM is Regulated Differently to that of CCR5, in Response to GM-CSF and IL-3

Recently, a greater understanding of the interaction of chemokines with their cognate receptors has been achieved. One of the most important reasons for this recent increase in understanding, has been due to the explosion of interest in the roles of chemokines and chemokine receptors in the progression of HIV pathogenesis. However, one area which has remained poorly understood is that of the regulation of chemokine receptor expression. Understanding this is clearly important, not just for novel aspects of chemokine biology such as HIV infection, but also for understanding inflammatory and haemopoietic aspects of chemokine function.

The data shown in figures 6.17 to 6.19 show that the expression of CCR1 is induced by GM-CSF and IL-3 in BMM and splenic cells, whilst CCR5 expression is unaltered. This suggests that the expression of CCR1 and CCR5 is regulated differently. One inference from these results, is that CCR1 and CCR5 may have distinct functions. It is possible that expression of CCR5 may be under a more restricted and constitutive control. The reason for this different regulation of expression is unclear, but it may reflect some importance of CCR1 over that of CCR5, under conditions in which GM-CSF or IL-3 are expressed (i.e. under inflammatory conditions). Indeed evidence from recent studies in which CCR1 was deleted in mice, have provided clear evidence that CCR1 has a crucial inflammatory role *in vivo* (Gao *et al* 1997, Gerard *et al* 1997). Attempts to demonstrate a functional significance of

this increase in CCR1 expression, by analysis of potential changes in calcium mobilisation, were unsuccessful. However as discussed in the results, this remains to be determined and may reflect experimental difficulties, rather than a lack of effect of GM-CSF on CCR1 expression, in a functional context. The effect of TGF-B1 on CCR1 and CCR5 expression appears to be minimal, apparently only affecting the basal level of expression. The significance of this observation is unclear. Different mechanisms for the regulatory control of chemokine receptor expression is not a new observation and several reports were published during the course of these studies, demonstrating that IL-2 regulated the expression of CCR1, CCR2 and CCR5, but did not alter the expression of CXCR1, CXCR2 or CXCR4 (Loetscher et al 1996b, Bleul et al 1997). Several reports have also documented suppressive effects of various agents, such as LPS, IFN- γ , TNF- α and IL-1 α on chemokine receptor expression (Sica et al 1997, Tangirala et al 1997). In fact, Tangirala et al reported that treatment of THP-1 cells with GM-CSF resulted in a marked reduction in the expression of hCCR2 mRNA. Interestingly, in this same study the authors also reported that differentiation of THP-1 cells with phorbol myristate acetate (PMA) resulted in an almost complete loss of hCCR2 mRNA expression. As indicated in figure 6.19 B, GM-CSF did not appear to affect mCCR2 expression. The reason for the difference between this finding and that reported by Tangirala et al is unclear, but may be due to the different cell systems used and perhaps the difference in differentiation state of BMM compared to THP-1. A recent report, published during the course of writing this thesis, has provided further and intriguing evidence of the potential complexity of chemokine receptor expression. This report by Mummidi et al (1997), provided evidence for multiple transcripts for human CCR5 (derived by alternative splicing and usage of multiple transcriptional start sites) and the usage of multiple promoters. Previous reports have documented alternative splicing for CCR2 (Charo et al 1994), and CXCR4 (Heesen et al 1997), suggesting that a complex regulation of gene expression and organisation may occur in other chemokine receptors. Future investigation of the promoters of the various chemokine receptor genes will no doubt yield further understanding of the regulation of their expression. In the context of the results presented in this thesis, a comparative analysis of the promoters for murine CCR1 and CCR5 following GM-CSF treatment would be a very useful approach to further understanding why they appear to be differently regulated.

No stem-cell inhibitory specific receptor has been reported for MIP-1 α as yet, and current evidence from our laboratory suggests that the inhibitory actions of MIP-1 α are not mediated through any of the reported MIP-1 α receptors (Graham *et al* 1996). It is probable that the observed effects of GM-CSF and IL-3 on CCR1 expression as reported in this work, relate to involvement in an inflammatory mechanism. Further understanding and confirmation of this must however, await identification of this stem cell inhibitory receptor.

7.7 Analysis of MIP-1α Expression in GM-CSF Transgenic Mouse Tissues

The studies discussed in the previous sections, strongly suggest a potent effect of GM-CSF and IL-3 on MIP-1 α expression *in vitro*. However they have provided little insight into any role of GM-CSF or IL-3 *in vivo*. As discussed in various sections in the introduction, analysis of knock out and transgenic mice have in many cases proved a useful approach to further understanding the potential *in vivo* role(s) of growth factors and cytokines. As discussed previously in section 2.12, analysis of mice overexpressing GM-CSF, as performed by Lang *et al* (1987), revealed some unusual phenotypic abnormalities particularly with regard to macrophage function. It was speculated by us, that the levels of MIP-1 α might be elevated in these mice as a result of over expression of GM-CSF and indeed increased expression was previously reported for other GM-CSF inducible growth factors in these transgenic mice (Lang *et al* 1992). MIP-1 α may therefore be ultimately responsible for aspects of the GM-CSF transgenic phenotype. It was therefore decided to perform a brief examination of the expression of MIP-1 α in these mice.

It should be pointed out here, that only a very limited amount of material was available for examination and therefore any investigation is, by necessity, of a purely preliminary nature. However, as can be seen in figures 6.26 to 6.29, RT-PCR analysis and subsequent Southern blot analysis did suggest that the level of MIP-1 α expression was elevated in the eye, and perhaps also in the bone marrow, of a GM-CSF transgenic mouse, relative to that observed in a wild type littermate control. The precise source of this MIP-1 α in wild type mice eyes is unknown, but may result from MIP-1 α expressing cells within the vasculature of the eye or perhaps even infiltrating leukocytes. The elevated expression in the eye of the transgenic mouse was

unsurprising, given the potent effect of GM-CSF on MIP-1 α expression. It must be pointed out that this observed increase in MIP-1 α may simply be the result of an overall increase in the number of macrophages within the eye, rather than a change in the actual level of expression as a result of GM-CSF over expression. An answer to this and a determination of the precise cellular source, will have to await *in-situ* hybridisation studies.

Although, as discussed in section 6.11.4, the analysis of the GM-CSF transgenic bone marrow sample was inconclusive, the RT-PCR analysis did suggest that MIP-1 α expression might be elevated in the GM-CSF transgenic bone marrow compared to the wild type. If this was indeed the case, then this would be somewhat unexpected, for the following reason. In the original characterisation of the GM-CSF transgenic mice reported by Lang et al (1987), no expression of the GM-CSF transgene was reported in the bone marrow of transgenic mice and therefore an increase in MIP-1 α expression may be unexpected. It should be noted however, that this original analysis of transgene expression was characterised by Northern blot analysis rather than by the more sensitive technique of RT-PCR analysis and it is therefore possible that the GM-CSF transgene is indeed expressed at low levels in the transgenic mice. Furthermore, the serum levels of GM-CSF were reported to be elevated, by a much as 40-fold in GM-CSF transgenics (Lang et al 1987). This could provide a possible explanation, if increased MIP-1a expression does occur in transgenic bone marrow, relative to that of wild type bone marrow. Clearly the above discussion is merely speculative and can only be confirmed by repeating this analysis.

The observation that the level of MIP-1 α expression was unaltered in the striated muscle of a transgenic mouse, as indicated in figures 6.30 and 6.31, is perhaps surprising in the light of observations made during the initial characterisation of GM-CSF transgenic mice by Lang *et al* 1987). In this, the authors reported that the expression of the transgene did not occur in non-infiltrated muscle of GM-CSF transgenic mice. Furthermore, this transgene expression was age dependent and was actually reported to follow, rather than precede the development of lesions in transgenic mice exhibiting infiltrates in their striated muscle. As a result of these observations, the authors proposed that macrophage accumulation was unlikely to be the direct cause of local GM-CSF expression and suggested that GM-CSF might

merely prolong the survival of macrophages already present within the striated muscle, thus leading to an indirect accumulation of macrophages. The results discussed in section 6.11.5 suggest that MIP-1 α expression is not elevated in the striated muscle of GM-CSF transgenic mice. The reason for the discrepancy between this result and that expected is unclear. However, as noted previously in section 6.11.5, some genomic contamination was present in the GM-CSF transgenic samples analysed and this may have resulted in the sequestration of primers. If this is indeed the case then this might explain the absence of any elevated level of MIP-1 α expression in striated muscle, particularly if the magnitude of any increase was relatively small.

GM-CSF transgenic mice show a very striking increase in the cellularity of their peritoneal cavities, with 90 % of these peritoneal cells being of an enlarged macrophage phenotype and showing characteristics of a functionally activated state. Given the potent effect of GM-CSF on macrophages as described during the course of this work and the reported expression of the GM-CSF transgene in the peritoneal cavity, one would perhaps expect to see an increased expression of MIP-1 α in peritoneal cells derived from a transgenic mouse, compared to the level of expression observed in a littermate control. As seen in figures 6.32 and 6.33, this was apparently not the case. It should be noted that BMM were used throughout the course of the studies reported herein and these may be quite distinct in comparison to other macrophages, such as peritoneal macrophages (reviewed by Gordon 1995). Obviously, it would be necessary to first compare the effect of GM-CSF on BMM and peritoneal macrophage expression of MIP-1 α , in order to answer this question. It is therefore theoretically possible that peritoneal macrophages may respond differently to GM-CSF, compared to resident BMM.

Whilst this examination of MIP-1 α in GM-CSF transgenic mice is only preliminary, it does offer some tentative suggestion that MIP-1 α expression may be elevated in specific tissues of transgenic mice, compared to wild type mice. It should however be noted that the results of such RT-PCR analysis should always be treated with caution, due to the extreme sensitivity of RT-PCR. These results do however, suggest that further examination of larger numbers of GM-CSF transgenic mice would be warranted, perhaps in conjunction with *in situ* hybridisation analysis.

7.8 The Role of GM-CSF, IL-3 and TGF- β in Regulating the Expression of MIP-1 α in BMM

Previous work in our laboratory identified an endogenous reciprocal relationship between MIP-1 α and TGF- β in BMM. The studies performed during the course of this thesis have added yet another layer of regulation to those previously identified. The results suggest, that in addition to the suppressive effects of TGF- β , stimulation of MIP-1 α expression is mediated by GM-CSF and IL-3. Moreover, this inductive effect may be potent enough to overcome the suppressive block exerted on these cells by TGF- β . Taken together this implies that under the appropriate conditions, the expression of GM-CSF and IL-3 may be induced and this may be sufficient to overcome the TGF- β block on MIP-1 α expression.

So are these observations pertinent to the regulation of MIP-1 α during haemopoiesis? BMM are an excellent in vitro model cell system, roughly equivalent to the resident bone marrow macrophage population that is believed to be the primary source of MIP-1 α in the bone marrow and therefore are ideally suited to the study of the regulation of MIP-1 α expression. Interestingly, the levels of GM-CSF mRNA have been reported to increase in the bone marrow and spleens of mice, during the recovery period following sublethal irradiation (Chang et al 1995), whilst the levels of TGF-B were reported to remain constant. It would be of interest to determine if the level of MIP-1 α expression was also increased, perhaps with a delayed kinetic response relative to that of GM-CSF expression? (see section 7.10). If the expression of MIP- 1α was delayed with respect to that of GM-CSF, this may suggest that MIP- 1α expression is induced via GM-CSF. If this were indeed the case, then one could speculate a role for MIP-1 α in 'dampening down' the proliferation of haemopoietic stem cells following irradiation or cytotoxic drug treatment. One interesting approach would be to investigate this relationship in both normal and MIP-1 α deficient mice and compare their recovery. Such investigation, coupled with in-situ hybridisation studies would also enable the precise cellular sources of this increased GM-CSF expression to be identified.

The previous study of Maltman *et al* (1996) suggested that the suppressive effect of TGF- β was relatively restricted to MIP-1 α and MIP-1 β . Studies reported in this work have suggested however, that this suppression is probably more widespread,

179

with TGF- β 1 acting to suppress the GM-CSF/IL-3 induced expression of at least two additional chemokines, JE and MARC. Current evidence suggests that JE and MARC are not functional as inhibitors of stem cell proliferation and therefore the suppressive effect of TGF-B may be related to its widely documented role as an immunosuppressive molecule. It is equally possible therefore, that the observed ability of GM-CSF and IL-3 to induce the expression of JE and MARC in addition to that of MIP-1a and MIP-1 β , may also be more related to pro-inflammatory processes than to stem cell proliferative control. However this observation does not rule out the involvement of GM-CSF or IL-3, in stem cell proliferative control. As discussed in the introduction, GM-CSF, IL-3 and TGF- β are just three such factors and there are a multitude of additional growth factors, cytokines, receptors, adhesion molecules and other regulatory mechanisms, that could impinge upon their interactions. One must also consider the fact that these factors may function within highly localised niches in the bone marrow microenvironment, possibly through interaction with molecules such as proteoglycans that may act to sequester or even actively present growth factors and chemokines to target cells. Clearly, an in vitro culture system such as BMM can not provide a full re-creation of the complex environment of the bone marrow. Additional levels of control may also exist, such as through endogenous interactions between MIP-1a, GM-CSF and IL-3, and between TGF-B, GM-CSF and IL-3 and their receptors. Additionally, other cells besides macrophages are likely to be important in this co-ordinated regulation. The use of long-term bone marrow cultures, could provide a better model system in which to investigate the importance of these additional factors and cellular interactions.

The precise implications of the differences observed in the regulation of MIP-1 α receptor expression remains unclear. No firm functional support could be observed for the observations made regarding the effect of GM-CSF at the level of mRNA expression, as determined by comparison of the ability to mobilise calcium in GM-CSF treated BMM. This may however, merely reflect technical problems. Indeed, the increase in CCR1 mRNA expression was reflected by an increase in the binding of MIP-1 α to BMM, revealing that an increase in the cell surface expression of MIP-1 α receptor numbers did occur in GM-CSF stimulated BMM. The potential effects of a 2-3 fold increase in receptor numbers on BMM function is unclear and warrants further investigation. It is not possible to say with certainty whether the observed effects of GM-CSF and IL-3 on CCR1 expression relate to a proinflammatory effect or one involved in stem cell proliferative control, but several points suggest that it is more likely to be the result of a role in an inflammatory context. Firstly, the MIP-1 α stem-cell inhibitory receptor has not yet been identified. Current evidence from our laboratory suggests that it is not CCR1 (Graham *et al* 1996), although others have suggested a role for human CCR1 in BFU-E inhibition (Su *et al* 1997). Secondly, the observation that CCR1 expression was also elevated in splenic cells, suggests that this pattern of expression may also occur in B and T cells (since B cells and to a lesser extent T cells, are a major cellular component of the murine spleen). A full understanding of the involvement of GM-CSF and IL-3 in the regulation of MIP-1 α receptor expression, in the context of stem cell proliferative control, must therefore await identification of the receptor through which MIP-1 α transduces the signal for stem cell inhibition.

So what is the function of GM-CSF, IL-3 and TGF- β in the regulation of MIP- 1α expression? As discussed above, these studies have implicated GM-CSF and IL-3 in the regulatory control of MIP-1 α expression in BMM. In the context of stem cell proliferative control, GM-CSF and IL-3 may function as stimulators of MIP-1a expression, whilst TGF- β may act to suppress this induction of MIP-1 α . As discussed in sections 2.2 and 2.11, GM-CSF and IL-3 clearly function as haemopoietic regulators. So if GM-CSF and IL-3 upregulate MIP-1a expression, why does this not result in inhibition? One explanation maybe that full translation of this increased MIP-1α mRNA may not occur and indeed it is uncertain exactly what level of functional MIP-1 α protein would be required to result in inhibition, in either autocrine or paracrine situations. In addition, it is always important to consider the many other factors that may be present in the bone marrow microenvironment. It is clear that MIP- 1α functions within a relatively narrow window within the stem cell compartment and it may be speculated that MIP-1 α acts as a stem cell inhibitor only during occasions of extreme stress, such as a systemic infection. During such an infection extensive haemopoietic stem cell proliferation may occur and one possible role for MIP-1 α may therefore be to 'dampen down' proliferation of these cells.

7.9 The Importance of MIP-1 α in Disease Pathology

The work presented in this thesis suggests that a complex relationship exists between MIP-1 α and at least three other haemopoietic growth factors. This has important implications not just for the regulation of haemopoietic stem cell proliferation, but in the much wider context of a multitude of diseases and pathological states. Indeed it is now evident, particularly from the study of mice deficient in the functions of GM-CSF, IL-3 and IL-5, that GM-CSF and IL-3 may be important mediators of inflammation, as discussed in sections 2.2 and 2.11. For instance GM-CSF, IL-3 and MIP-1 α have been implicated in the functions of mature eosinophil, basophil and mast cells. These cells are critical to the development of inflammatory disorders such as asthma and in the protection against parasitic infections. This complex picture is further complicated by the existence of over 35 other chemokines (see table 3.1), and others are likely to be identified in the future. Results presented in this thesis suggest that the relationship with GM-CSF and IL-3 appears relatively non-specific and can be extended to other chemokines. Indeed other researchers have shown that C10, MPIF-2 and IL-8 are all inducible by GM-CSF and/or IL-3 (Orlofsky et al 1994, Patel et al 1997, Takahashi et al 1993). Clearly this results in a very large and complex interacting network of inflammatory cytokines with potentially important pathological implications.

As discussed in section 3.6.3, MIP-1 α is implicated in a wide variety of diseases and pathological states. Perhaps the most compelling evidence for the *in vivo* importance of MIP-1 α is the report by Cook *et al* (1995). In this report, the authors generated mice in which the gene for MIP-1 α was disrupted and demonstrated that these MIP-1 α^{-1} mice were resistant to coxsackievirus B3 (CVB3)-induced myocarditis and exhibited a reduced pneumonitis following influenza virus infection. The cardiac damage resulting from CVB3 infection therefore appears to be the result of the actions of recruited inflammatory cells, rather than from a direct effect of the virus. This report clearly demonstrated an *in vivo* role for MIP-1 α in viral infection and has subsequently been followed by a number of other studies. For instance Epstein Barr virus (EBV) has recently been reported to induce the expression of MIP-1 α (and IL-8) in human neutrophils, which may result in the chemoattraction of T and B cells and therefore enhance the ability of EBV to infect T and B cells (McColl *et al* 1997).

Infection of human monocytes with influenza A virus has been shown to result in the expression of MIP-1 α and other CC chemokines (Sprenger *et al* 1996). A critical role for MIP-1 α in NK cell inflammatory responses against murine CMV (MCMV) was also recently reported (Salazar-Mather *et al* 1998). These studies, combined with the mounting evidence from studies on HIV pathogenesis (see section 3.9) suggest that this role in viral pathogenesis is likely to be widespread. Indeed this importance is further underlined by the emergence of various virally encoded homologues of MIP-1 α and chemokine-like receptors (see section 3.10)

MIP-1 α is implicated in several autoimmune diseases, perhaps most notably that of MS. Indeed, administration of anti-MIP-1 α antibodies results in prevention of experimentally induced MS (Karpus *et al* 1995). Intriguingly, TGF- β may also play a protective role in MS (Kuruvilla *et al* 1991) and given the clear relationship between TGF- β and MIP-1 α , as detailed in this thesis, it is tempting to speculate that these may be linked. In a similar way, both GM-CSF and MIP-1 α have been implicated in the development of rheumatoid arthritis (Koch *et al* 1994, Haworth *et al* 1991).

The suppressive actions of TGF- β may also reflect an involvement in antiinflammatory processes and this suggestion is supported by the suppressive effect of TGF- β 1 on other chemokines such as JE and MARC. The observation that MIP-1 α expression is elevated in TGF- β 1 null mice suggests that this suppressive role is likely to be important *in vivo*.

7.10 Conclusions and aspects for further investigation

MIP-1 α , in addition to its roles in the normal function of the immunohaemopoietic system, is clearly implicated in a wide variety of diseases and is important in many pathological circumstances. This underlines the need to further understand how its expression is controlled under normal circumstances. The aim of this thesis was to attempt to further understand the network of regulatory factors controlling the expression of MIP-1 α in the bone marrow. As discussed above, this included a previously identified suppressive role for TGF- β . With the observation that GM-CSF and IL-3 appear to function as stimulators of MIP-1 α mRNA expression in BMM, yet further complexity is now evident in the existing network of factors involved in MIP-1 α expression. In conclusion, both stimulatory and inhibitory

mechanisms appear to regulate the expression of MIP-1 α and other chemokines in BMM, forming a complex interacting network between GM-CSF, IL-3, TGF- β , MIP-1 α , other chemokines and perhaps even some of their receptors. Figure 7.1 provides a summary of the proposed interactions between MIP-1 α , TGF- β , GM-CSF, IL-3 and other factors in BMM, as suggested by work reported in this thesis.

Many others aspects of these interactions remain in question. For instance, is there also a reciprocal relationship between MIP-1 α and GM-CSF / IL-3 and what effect does TGF- β have on the expression of GM-CSF and IL-3? A previous report has suggested that TGF-B downregulates the expression of receptors for GM-CSF and IL-3 on murine haemopoietic progenitor cell lines (Jacobsen et al 1991a). If TGF-ß is also able to suppress GM-CSF and IL-3 receptor expression on BMM, then this could result in a reduced responsiveness of BMM to GM-CSF and IL-3 (with a subsequent reduced expression of MIP-1 α). TGF- β has also been reported to downregulates TNF- α or IL-1-induced GM-CSF expression in human articular chondrocytes (Alsalameh et al 1994). Further investigation is required to ascertain the precise relationship between these different factors and how this affects their individual expression, in both constitutive and induced circumstances. For instance, it is uncertain whether or not BMM constitutively express GM-CSF or IL-3. A study by Temeles et al (1993) suggests that BMM do not express mRNA for either GM-CSF or IL-3. This would imply that GM-CSF and IL-3 do not function in an autocrine manner to regulate MIP- 1α expression in BMM. This observation is in agreement with experiments using neutralising antibodies against GM-CSF and IL-3, the results of which are shown in figures 6.6 and 6.8B respectively. This suggests that under uninduced conditions, BMM are not a source of GM-CSF or IL-3 and that they must be produced by other types of accessory cells within the bone marrow microenvironment. Although in the study reported by Temeles et al (1993), stimulation of BMM with cytokines did not result in the induction of GM-CSF or IL-3 expression, it is possible that induction of GM-CSF and IL-3 may occur in BMM following an appropriate inductive stimulus (such as LPS?). This induced GM-CSF and IL-3 could then operate in an autocrine manner to induce MIP-1a expression. This potential for 'induced autocrine' expression is also illustrated in figure 7.1. It is unlikely that this is the whole story, as other growth factors within the bone marrow microenvironment are almost certain to



affect the expression of GM-CSF, IL-3, TGF- β and MIP-1 α . For example, IL-1, TNF- α and IFN- α have all been reported to **a**ffect GM-CSF expression in a variety of cell types (Fibbe *et al* 1986, Gollner *et al* 1995). This creates the potential for yet further complexity in this network.

A molecular analysis of the precise mechanisms, such as transcription factors and other downstream signal transduction pathways, through which GM-CSF and IL-3 achieve induction of MIP-1 α expression, is also worthy of further investigation. As discussed earlier in section 7.4, such an analysis could prove particularly informative, if coupled with analysis of other chemokine genes, such as JE/MCP-1. Preliminary results, not presented in this thesis, obtained from studies using the protein synthesis inhibitor Cycloheximide have indicated that GM-CSF-induced MIP-1a expression is at least in part transcriptionally regulated. Analysis of the molecular elements involved in the regulatory control of MIP-1 α expression by GM-CSF and IL-3 is clearly an area with extensive potential for further investigation. This could also be coupled with an analysis of those pathways involved in TGF-B1 mediated suppression of MIP-1a expression, with the aim of determining exactly how TGF- β suppresses MIP-1 α expression. In fact, a recent report has suggested that TGF-B suppreses MCP-1 expression in macrophages, possibly by downregulating the expression of c-jun, one of the components of the AP-1 transcription factor complex (Kitamura et al 1997). It would be of interest to examine this observation further in the light of the results presented here in this thesis and this might provide further clues as to why GM-CSF, IL-3 and TGF- β appear to be so non-specific in their effects on chemokines.

One final area worthy of further attention, is an investigation into the *in vivo* interactions between MIP-1 α , GM-CSF, IL-3 and TGF- β 1. It is assumed that the regeneration of the haemopoietic system after sublethal myeloablation involves the action of endogenously produced haemopoietic growth factors. Indeed, following *in vitro* irradiation the levels of GM-CSF protein were previously reported to increase in murine long-term bone marrow cultures and spleen cell cultures (Naparstek *et al* 1985, Onoda *et al* 1980). A study reported by Chang *et al* (1995), demonstrated that a marked increase in GM-CSF mRNA occurred during the first two weeks of the recovery period following sublethal irradiation. This increase in GM-CSF mRNA expression appeared to precede the cellular recovery, which did not appear until

approximately 2 weeks post-irradiation. Therefore, an analysis of MIP-1 α expression, together with that of GM-CSF, IL-3 and TGF- β , in sublethally irradiated mice could prove highly informative. Admittedly such an analysis might however be relatively difficult to undertake, given the relatively low levels of cytokines present in bone marrow and particularly with the very small numbers of cells that would be present during the earlier stages of reconstitution of the bone marrow microenvironment.

REFERENCES

Ahuja, S.K., and P.M. Murphy. 1993. Molecular piracy of mammalian interleukin-8 receptor type B by Herpesvirus Saimiri. J. Biol. Chem. 268:20691.

Aiuti, A., I.J. Webb, C. Bleul, T. Springer, and J.C. Gutierrez-Ramos. 1997. The chemokine SDF-1 is a chemoattractant for human CD34⁺ hematopoietic progenitor cells and provides a new mechanism to explain the mobilization of CD34⁺ progenitors to peripheral blood. *J. Exp. Med.* 185:111.

Alam, R., P.A. Forsythe, M.A. Lett-Brown, and J.A. Grant. 1992. Macrophage inflammatory protein- 1α is an activator of basophils and mast cells. *J. Exp. Med.* 176:781.

Alam, R., P. Forsythe, S. Stafford, and Y. Fukuda. 1994a. Transforming growth factor β abrogates the effects of hematopoietins on eosinophils and induces their apoptosis. *J. Exp. Med.* 179:1041.

Alam, R., P. Forsythe, S. Stafford, J. Heinrich, R. Bravo, P. Proost, and J. Van Damme. 1994b. Monocyte chemotactic protein-2, monocyte chemotactic protein-3, and fibrobalst-induced cytokine. *J. Immunol.* 153:3155.

Alderson, M.R., T.W. Tough, S.F. Zeigler, and K.H. Grabstein. 1991. Interleukin 7 induces cytokine secretion and tumoricidal activity by human peripheral blood monocytes. J. Exp. Med. 173:923.

Alexandrow, M.G., and H.L. Moses. 1995. Transforming growth factor β and cell cycle regulation. *Cancer Res.* 55:1452.

Alkhatib, G., C. Combadiere, C.C. Broder, Y. Feng, P.E. Kennedy, P.M. Murphy, E.A. Berger. 1996. CC CKR5: A RANTES, MIP-1 α , MIP-1 β receptor as a fusion cofactor for macrophage-tropic HIV-1. *Science* 272:1955.

Alsalameh, S., G.S. Firestein, S. Oez, R. Kurrle, J.R. Kalden, and G.R. Burmester. 1994. Regulation of granulocyte macrophage colony-stimulating factor production by human articular chondrocytes. Induction by both tumor necrosis factor- α and interleukin 1, downregulation by transforming growth factor- β and upregulation by fibroblast growth factor. *J. Rheumatol.* 21:993.

Anderson, D.M., S.D. Lyman, A. Baird, J.M. Wignall, J. Eisenman, C. Rauch, C.J. March, H.S. Boswell, S.D. Gimpel, D. Cosman, and D.E. Williams. 1990. Molecular cloning of mast cell growth factor, a hematopoietin that is active in both membrane bound and soluble forms. *Cell* 63:235.

Anegon, I., J.F. Moreau, A. Godard, Y. Jacques, M.A. Peyrat, M.M. Hallet, G. Wong, and J.P. Soulillou. 1990. Production of human interleukin for DA cells (HILDA)/ leukemia inhibitory factor (LIF) by activated monocytes. *Cell Immunol.* 130:50.

Angiolillo, A.L., C. Sgadari, D.D. Taub, F. Liao, J.M. Farber, S. Maheshwari, H.K. Kleinman, G.H. Reaman, and G. Tosato. 1995. Human interferon-inducible protein 10 is a potent inhibitor of angiogenesis in vivo. *J. Exp. Med.* 182:155.

Arai, H., F.S. Monteclaro, C-L. Tsou, C. Franci, and I.F. Charo. 1997. Dissociation of chemotaxis from agonist-induced receptor internalization in a lymphocyte cell line transfected with CCR2B. *J. Biol. Chem.* 272:25037.

Aramori, I., J. Zhang, S.S.G. Ferguson, P.D. Bieniasz, B.R. Cullen, M.G. Caron. Molecular mechanism of desensitization of the chemokine receptor CCR-5: receptor signaling and internalization are dissociable from its role as an HIV-1 co-receptor. *EMBO J.* 16:4606.

Arvanitakis, L., E. Geras-Raaka, A. Varma, M.C. Gershengorn, and E. Cesarman. 1997. Human herpesvirus KSHV encodes a constitutively active G-protein-coupled receptor linked to cell proliferation. *Nature* 385:347.

Asano, S., A. Okano, K. Ozawa, T. Nakahata, T. Ishibashi, K. Koike, H. Kimura, Y. Tanioka, A. Shibuya, T. Hirano, T. Kishimoto, F. Takaku, and Y. Akiyama. 1990. In vivo effects of recombinant human interleukin-6 in primates: Stimulated production of platelets. *Blood* 75:1602.

Assoian, R.K., A. Komoriya, C.A. Meyers, D.M. Miller, and M.B. Sporn. 1983. Transforming growth factor- β in human platelets. *J. Biol. Chem.* 258:7155.

Assoian, R.K., B.E. Fleurdelys, H.C. Stevenson, P.J. Miller, D.K. Madtes, E.W. Raines, R. Ross, and M.B. Sporn. 1987. Expression and secretion of type β transforming growth factor by activated human macrophages. *Proc. Natl. Acad. Sci* USA. 84:6020.

Avaraham, H., E. Vannier, S.Y. Chi, C.A. Dinarello, and J.E. Groopman. 1992. Cytokine gene expression and synthesis by human megakaryocytic cells. *Int. J. Cell Cloning.* 10:70.

Axelrad, A.A. 1990. Some hemopoietic negative regulators. Exp. Hematol. 18:143.

Baba, M., T. Imai, M. Nishimura, M. Kakizaki, S. Takagi, K. Hieshima, H. Nomiyama, and O. Yoshie. 1997. Identification of CCR6, the specific receptor for a novel lymphocyte-directed CC chemokine LARC. J. Biol. Chem. 272:14893.

Bagley, C.J., J.M. Woodcock, F.C. Stomski, A.F. Lopez. 1997. The structural and functional basis of cytokine receptor activation: Lessons from the common β subunit of the granulocyte-macrophage colony-stimulating factor, interleukin-3 (IL-3), and IL-5 receptors. *Blood* 89:1471.

Bais, C., B. Santomasso, O. Coso, L. Avaranitakis, E.G. Raaka, J.S. Gutkind, A.S. Asch, E. Cesarman, M.C. Gerhengorn, and E.A. Mesri. 1998. G-protein-coupled receptor of Kaposi's sarcoma-associated herpesvirus is a viral oncogene and angiogenesis activator. *Nature 391:86*.

Baldwin, G.C., N.D. Fuller, R.L. Roberts, D.D. Ho, and D.W. Golde. 1989. Granulocyte- and granulocyte-macrophage colony-stimulating factors enhance neutrophil cytotoxicity toward HIV-infected cells. *Blood* 74:1673.

Baldwin, G.C. 1992. The biology of granulocyte-macrophage colony-stimulating factor: Effects on hematopoietic and nonhematopoietic cells. *Dev. Biol.* 151:352.

Bartelmez, S.H., R. Sacca, and E.R. Stanley. 1985. Lineage specific receptors used to identify a growth factor for developmentally early hemopoietic cells: assay of hemopoietin-2. *J. Cell Physiol.* 122:362.

Bazan, J.F. 1990. Structural design and molecular evolution of a cytokine receptor superfamily. *Proc. Natl. Acad. Sci. USA* 87:6934.

Bazan, J.F., K.B. Bacon, G. Hardiman, W. Wang, K. Soo, D. Rossi, D.R. Greaves, A. Zlotnik, and T.J. Schall. 1997. A new class of membrane-bound chemokine with a CX_3C motif. *Nature* 385:640.

Beall, C.J., S. Mahajan, and P.E. Kolattukudy. 1992. Conversion of monocyte chemoattractant protein-1 into a neutrophil attractant by substitution of two amino acids. *J. Biol. Chem.* 267:3455.

Becker, S., M.K. Warren, S. Haskill. 1987. Colony-stimulating factor-induced monocyte survival and differentiation into macrophages in serum-free cultures. J. Immunol. 139:3703.

Begg, S.K., and I. Bertoncello. 1993. The hematopoietic deficiencies in osteopetrotic (*op/op*) mice are not permanent, but progressively correct with age. *Exp. Hematol.* 21:493.

Begley, C.G., A.F. Lopez, N.A. Nicola, D.J. Warren, M.A. Vadas, C.J. Sanderson, and D. Metcalf. 1986. Purified colony-stimulating factors enhance the survival of human neutrophils and eosinophils in vitro: A rapid and sensitive microassay for colony-stimulating factors. *Blood* 68:162.

Ben-Baruch, A., L. Xu, P.R. Young, K. Bengali, J.J. Oppenheim, and J.M. Wang. 1995. Monocyte chemotactic protein-3 (MCP3) interacts with multiple leukocyte receptors. *J. Biol. Chem.* 270:22123.

Berman, J.W., M.P. Guida, J. Warren, J. Amat, and C.F. Brosnan. 1996. Localization of monocyte chemoattractant Peptide-1 expression in the central nervous system in experimental autoimmune encephalomyelitis and trauma in the rat. *J. Immunol.* 156:3017.

Bernstein, I.D., R.G. Andrews, and K.M. Zsebo. 1991. Recombinant human stem cell factor enhances the formation of colonies by CD34⁺ and CD34⁺lin⁻ cells, and the generation of colony-forming cell progeny from CD34⁺lin⁻ cells cultured with interleukin-3, granulocyte colony-stimulating factor, or granulocyte-macrophage colony-stimulating factor. *Blood* 77:2316.

Bertoncello, I. 1992. Status of high proliferative potential colony-forming cells in the hematopoietic stem cell hierachy. *Curr. Top. Microbiol. Immunol.* 177:83.

Besmer, P., K. Manova, R. Duttlinger, E.J. Huang, A. Packer, C. Gyssler, and R.F. Bachvarova. 1993. The *kit*-ligand (steel factor) and its receptor *c-kit/W*: Pleiotropic roles in gametogenesis and melanogenesis. *Development (Suppl):125*.

Billips, L.G., D. Petitte, K. Dorshkind, R. Narayanan, C-P. Chiu, and K.S. Landreth. 1992. Differential roles of stromal cells, interleukin-7, and *kit*-ligand in the regulation of B lymphopoiesis. *Blood* 79:1185.

Birchenbach, M., Josefsen, K. R. Yalamanchili, G. Lenoir, and E. Kieff. 1993. Epstein-Barr virus-induced genes: First lymphocyte-specific G protein-coupled peptide receptors. J. Virol. 67:2209.

Bischoff, S.C., M. Krieger, T. Brunner, A. Rot, V. Tscharner, M. Baggiolini, and C.A. Dahinden. 1993. RANTES and related chemokines activate human basophil granulocytes through different G protein-coupled receptors. *Eur. J. Immunol.* 23:761.

Biti, R., R. Ffrench, J. Young, B. Bennetts, G. Stewart, and T. Liang. 1997. HIV-1 infection in an individual homozygous for the *CCR5* deletion allele. *Nature Med.* 3:252.

Bleul, C.C., R.C. Fuhlbrigge, J.M. Casasnovas, A. Aiuti, and T.A. Springer. 1996a. A highly efficacious lymphocyte chemoattractant, stromal cell-derived factor 1 (SDF-1). *J. Exp. Med.* 184:1101.

Bleul, C.C., M. Farzan, H. Choe, C. Parolin, I. Clark-Lewis, J. Sodroski, and T.A. Springer. 1996b. The lymphocyte chemoattractant SDF-1 is a ligand for LESTR/fusin and blocks HIV-1 entry. *Nature 382:829*.

Bleul, C.C., L. Wu, J.A. Hoxie, T.A. Springer, and C.R. Mackay. 1997. The HIV coreceptors CXCR4 and CCR5 are differentially expressed and regulated on human T lymphocytes. *Proc. Natl. Acad. Sci. USA* 94:1925.

Bluman, E.M., K.J. Bartynski, B.R. Avalos, and M.A. Caligiuri. 1996. Human natural killer cells produce abundant macrophage inflammatory protein- 1α in response to monocyte-derived cytokines. J. Clin. Invest. 97:2722.

Bonewald, L.F., L. Wakefield, R.O. Oreffo, A. Escobedo, D.R. Twardzik, and G.R. Mundy. 1991. Latent forms of transforming growth factor-beta (TGF beta) derived from bone cultures: identification of a naturally occurring 100-kDa complex with similarity to recombinant latent TGF beta. *Mol. Endocrinol.* 5:741.

Borkowski, T.A., J.J. Letterio, A.G. Farr, and M.C. Udey. 1996. A role for endogenous transforming growth factor $\beta 1$ in langerhans cell biology: The skin of transforming growth factor $\beta 1$ null mice is devoid of epidermal langerhans cells. *J. Exp. Med.* 184:2417.

Borkowski, T.A., J.J. Letterio, C.L. Mackall, A. Saitoh, X-J. Wang, D.R. Roop, R.E. Gress, and M.C. Udey. 1997. A role for TGFβ1 in langerhans cell biology. J. Clin. Invest. 100:575.

Boshoff, C., Y. Endo, P.D. Collins, Y. Takeuchi, J.D. Reeves, V.L. Schweickart, M.A. Siani, T. Sasaki, T.J. Williams, P.W. Gray, P.S. Moore, Y. Chang, R.A. Weiss. 1997. Angiogenic and HIV-inhibitory functions of KSHV-encoded chemokines. *Science* 278:290.

Bradley, T.R., and G.S. Hodgson. 1979. Detection of primitive macrophage progenitor cells in mouse bone marrow. *Blood 54:1446*.

Bradley, T.R., N. Williams, A.B. Kreigler, J. Fawcett, and G.S. Hodgson. 1989. In vivo effects of interleukin- 1_{α} on regenerating mouse bone marrow myeloid colony-forming cells after treatment with 5-fluorouracil. *Leukemia* 3:893.

Brandes, M.E., U.E.H. Mai, K. Ohura, and S.M. Wahl. 1991. Type I transforming growth factor- β receptors on neutrophils mediate chemotaxis to transforming growth factor- β . J. Immunol. 147:1600.

Brandt, J., R.A. Briddell, E.F. Srour, T.B. Leemhuis, and R. Hoffman. 1992. Role of *c-kit* ligand in the expansion of human hematopoietic progenitor cells. *Blood* 79:634.

Brannan, C.I., S.D. Lyman, D.E. Williams, J. Eisenman, D.M. Anderson, D. Cosman, M.A. Bedell, N.A. Jenkins, and N.G. Copeland. 1991. Steel-Dickie mutation encodes a c-Kit ligand lacking transmembrane and cytoplasmic domains. *Proc. Natl. Acad. Sci. USA 88:4671*.

Breen, F.N., D.A. Hume, and M.J. Weidermann. 1990. The effects of interleukin 3 (IL-3) on cells responsive to macrophage colony-stimulating factor (CSF-1) in liquid murine bone marrow culture. *Br. J. Haematol.* 74:138.

Broudy, V.C. 1997. Stem cell factor and hematopoiesis. Blood 90:1345.

Broudy, V.C., K. Kaushansky, G.M. Segal, J.M. Harlan, and J.W. Adamson. 1986. Tumor necrosis factor type α stimulates human endothelial cells to produce granulocyte/macrophage colony-stimulating factor. *Proc. Natl. Acad. Sci. USA* 83:7467.

Broudy, V.C., N.L. Lin, and K. Kaushansky. 1995. Thrombopoietin (c-mpl ligand) acts synergistically with erythropoietin, stem cell factor and interleukin-11 to enhance murine megakaryocyte colony growth and increases megakaryocte ploidy in vitro. *Blood* 85:1719.

Brown, P.D., L.M. Wakefield, A.D. Levinson, and M.B. Sporn. 1990. Physiochemical activation of recombinant latent transforming growth factor-beta's 1, 2, and 3. *Growth Factors 3:35*.

Broxmeyer, H.E., D.E. Williams, S. Cooper, R.K. Shadduck, S. Gillis, A. Waheed, D.L. Urdal, and D.C. Bicknell. 1987a. Comparative effects in vivo of recombinant murine interleukin 3, natural murine colony-stimulating factor-1, and recombinant murine granulocyte-macrophage colony-stimulating factor on myelopoiesis in mice. J. Clin. Invest. 79:721.

Broxmeyer, H.E., D.E. Williams, G. Hangoc, S. Cooper, S. Gillis, R.K. Shadduck, and D.C. Bicknell. 1987b. Synergistic myelopoietic actions *in vivo* after administration to mice of combinations of purified natural murine colony-stimulating factor 1, recombinant murine interleukin 3, and recombinant murine granulocyte/macrophage colony-stimulating factor. *Proc. Natl. Acad. Sci USA* 84:3871.

Broxmeyer, H.E., B. Sherry, L. Lu, S. Cooper, C. Carow, S.D. Wolpe, and A. Cerami. 1989. Myelopoietic enhancing effects of murine macrophage inflammatory proteins 1 and 2 on colony formation in vitro by murine and human bone marrow granulocyte/macrophage progenitor cells. *J. Exp. Med.* 170:1583.

Broxmeyer, H.E., B. Sherry, L. Lu, S. Cooper, K-O. Oh, P. Tekamp-Olson, B.S. Kwon, and A. Cerami. 1990. Enhancing and suppressing effects of recombinant murine macrophage inflammatory proteins on colony formation in vitro by bone marrow myeloid progenitor cells. *Blood* 76:1110.

Broxmeyer, H.E., B. Sherry, S. Cooper, F.W. Ruscetti, D.E. Williams, P. Arosio, B.S. Kwon, and A. Cerami. 1991. Macrophage inflammatory protein (MIP)-1 β abrogates the capacity of MIP-1 α to suppress myeloid progenitor cell growth. *J. Immunol.* 147:2586.

Broxmeyer, H.E., B. Sherry, S. Cooper, L. Lu, R. Maze, M.P. Beckmann, A. Cerami, and P. Ralph. 1993. Comparitive analysis of the human macrophage inflammatory protein family of cytokines (chemokines) on proliferation of human myeloid progenitor cells. *J. Immunol.* 150:3448.

Broxmeyer, H.E., S. Cooper, G. Cacalano, N.L. Hague, E. Bailish, and M.W. Moore. 1996. Involvement of interleukin (IL) 8 receptor in negative regulation of myeloid progenitor cells in vivo: Evidence from mice lacking the murine IL-8 receptor homologue. J. Exp. Med. 184:1825.

Brugger, W., W. Mocklin, S. Heimfeld, R.J. Berenson, R. Mertelsmann, and L. Kranz. 1993. Ex vivo expansion of enriched peripheral blood CD34⁺ progenitor cells by stem cell factor, interleukin-1 β (IL-1 β), IL-6, IL-3, interferon- γ , and erythropoietin. *Blood* 81:2579.

Burd, P.R., G.J. Freeman, S.D. Wilson, M. Berman, R. DeKruyff, P.R. Billings, and M.E. Dorf. 1987. Cloning and characterization of a novel T cell activation gene. J. Immunol. 139:3126.

Burd, P.R., H.W. Rodgers, J.R. Gordon, C.A. Martin, S. Jayaraman, S.D. Wilson, A.M. Dvorak, S.J. Galli, and M.E. Dorf. 1989. Interleukin 3-dependent and - independent mast cells stimulated with IgE and antigen express multiple cytokines. *J. Exp. Med.* 170:245.

Burgess, A.W., J. Camakaris, and D. Metcalf. 1977. Purification and properties of colony-stimulating factor from mouse lung-conditioned medium. J. Biol. Chem. 252:1998.

Bursch, W., F. Oberhammer, R.L. Jirtle, M. Askari, R. Sedivy, B. Grasl-Kraupp, A.F. Purchio, and R. Schulte-Hermann. 1993. Transforming growth factor- β 1 as a signal for induction of cell death by apoptosis. *Br. J. Cancer* 67:531.

Cacalano, G., J. Lee, K. Kikly, A.M. Ryan, S. Pitts-Meek, B. Hultgren, W.I. Wood, M.W. Moore. 1994. Neutrophil and B cell expansion in mice that lack the murine IL-8 receptor homolog. *Science 265:682*.

Cannistra, S.A., E. Vellenga, P. Groshek, A. Rambaldi, and J.D. Griffin. 1988. Human granulocyte-macrophage colony-stimulating factor and interleukin 3 stimulate monocyte cytotoxicity through a tumor necrosis factor-dependent mechanism. *Blood* 71:672.

Carr, M.W., S.J. Roth, E. Luther, S.S. Rose, and T.A. Springer. 1994. Monocyte chemoattractant protein 1 acts as a T-lymphocyte chemoattractant. *Proc. Natl. Acad. Sci. USA 91:3652.*

Caracciolo, D., N. Shirsat, G.G. Wong, B. Lange, S. Clark, and G. Rovera. 1987. Recombinant human macrophage colony-stimulating factor (M-CSF) requires subliminal concentrations of granulocyte/macrophage (GM)-CSF for optimal stimulation of human macrophage colony formation in vitro. J. Exp. Med. 166:1851.

Chan, J.Y., D.J. Slamon, S.D. Nimer, D.W. Golde, and J.C. Gasson. 1986. Regulation of expression of human granulocyte/macrophage colony-stimulating factor. *Proc. Natl. Acad. Sci. USA* 83:8669.

Chang, C.M., A. Limanni, W.H. Baker, M.E. Dobson, J.F. Kalinich, W. Jackson, and M.L. Patchen. 1995. Bone marrow and splenic granulocyte-macrophage colonystimulating factor and transforming growth factor- β mRNA levels in irradiated mice. *Blood 86:2130*.

Charbord, P., E. Tamayo, S. Saeland, V. Duvert, J. Poulet, A.M. Gown, and P. Herve. 1991. Granulocyte-macrophage colony-stimulating factor (GM-CSF) in human long-term bone marrow cultures: Endogenous production in the adherent layer and effect of exogenous GM-CSF on granulomonopoiesis. *Blood* 78:1230.

Charo, I.F., S.J. Myers, A. Herman, C. Franci, A.J. Connolly, and S.R. Coughlin. 1994. Molecular cloning and functional expression of two monocyte chemoattractant protein 1 receptors reveals alternative splicing of the carboxyl-terminal tails. *Proc. Natl. Acad. Sci. USA 91:2752.*

Chasty, R.C., G.S. Lucas, P.J. Owen-Lynch, A. Pierce, and A.D. Whetton. 1995. Macrophage inflammatory protein- 1α receptors are present on cells enriched for CD34 expression from patients with chronic myeloid leukemia. *Blood* 86:4270.

Chen, B.D-M., M. Mueller, and T. Olencki. 1988. Interleukin-3 (IL-3) stimulates the clonal growth of pulmonary alveolar macrophage of the mouse: Role of IL-3 in the regulation of macrophage production outside the bone marrow. *Blood* 72: 685.

Choe, H., M. Farzan, Y. Sun, N. Sullivan, B. Rollins, P.D. Ponath, L. Wu, C.R. Mackay, G. LaRosa, W. Newman, N. Gerard, C. Gerard, and J. Sodroski. 1996. The β -chemokine receptors CCR3 and CCR5 facilitate infection by primary HIV-1 isolates. *Cell* 85:1135.

Chomczynski, P., and N. Sacchi. 1987. Single-step method of RNA isolation by acid guanidinium thiocyanate-phenol-chloroform extraction. *Anal. Biochem.* 162:156.

Clark-Lewis I., B. Dewald, T. Geiser, B. Moser, M. Baggiolini. 1993. Platelet factor 4 binds to interleukin 8 receptors and activates neutrophils when its N terminus is modified with Glu-Leu-Arg. *Proc. Natl. Acad. Sci. USA 90:3574*.

Cluitmans, F.H.M., B.H.J. Esendam, J.E. Landegent, R. Willemze, and J.H.F. Falkenburg. 1995. Constitutive in vivo cytokine and hematopoietic growth factor gene expression in the bone marrow and peripheral blood of healthy individuals. *Blood* 85:2038.

Clutterbuck, E., J.G. Shields, J. Gordon, S.H. Smith, A. Boyd, R.E. Callard, H.D. Campbell, I.G. Young, and C.J. Sanderson. 1987. Recombinant human interleukin 5 is an eosinophil differentiation factor but has no activity in standard human B cell growth factor assays. *Eur. J. Immunol.* 17:1743.

Clutterbuck, E.J., and C.J. Sanderson. 1988. Human eosinophil hematopoiesis studied in vitro by means of murine eosinophil differentiation factor (IL-5): Production of functionally active eosinophils from normal human bone marrow. *Blood* 71:646.

Clutterbuck, E.J., and C.J. Sanderson. 1990. The regulation of eosinophil precursor production by cytokines: A comparison of recombinant human interleukin-1 (rhIL-1), rhIL-3, rhIL-5, rhIL-6, and rh granulocyte-macrophage colony-stimulating factor. *Blood* 75:1774.

Cocchi, F., A.L. DeVico, A. Garzino-Demo, S.K. Arya, R.C. Gallo, and P. Lusso. 1995. Identification of RANTES, MIP-1 α , and MIP-1 β as the major HIV-suppressive factors produced by CD8⁺ T cells. *Science* 270:1811.

Cocchi, F., A.L. DeVico, A. Garzino-Demo, A. Cara, R.C. Gallo, and P. Lusso. 1996. The V3 domain of the HIV-1 gp120 envelope glycoprotein is critical for chemokinemediated blockade of infection. *Nature Med. 2:1244* Coffman, R.L., B.W.P. Seymour, S. Hudak, J. Jackson, and D. Rennick. 1989. Antibody to interleukin-5 inhibits helminth-induced eosinophilia in mice. *Science* 245:308.

Columbo, M., E.M. Horowitz, L.M. Botana, D.W. MacGlashan Jr., B.S. Bochner, S. Gillis, K.M. Zsebo, S.J. Galli, and L.M. Lichtenstein. 1992. The human recombinant *c-kit* receptor ligand, rhSCF, induces mediator release from human cutaneous mast cells and enhances IgE-dependent mediator release from both skin mast cells and peripheral blood basophils. *J. Immunol.* 149:599.

Combadiere, C., S.K. Ahuja, H.L. Tiffany, and P.M. Murphy. 1996. Cloning and functional expression of CC CKR5, a human monocyte CC chemokine receptor selective for MIP-1 α , MIP-1 β , and RANTES. J. Leukoc. Biol. 60:147.

Conlon, P.J., P.J. Morrissey, R.P. Nordan, K.H. Grabstein, K.S. Prickett, S.G. Reed, R. Goodwin, D. Cosman, and A.E. Namen. 1989. Murine thymocytes proliferate in direct response to interleukin-7. *Blood* 74:1368.

Conover, J.C., N.Y. Ip, W.T. Poueymirou, B. Bates, M.P. Goldfarb, T.M. DeChiara, and G.D. Yancopoulos. 1993. Ciliary neurotrophic factor maintains the pluripotentiality of embryonic stem cells. *Development 119:559*.

Cook, D.N., M.A. Beck, T.M. Coffman, S.L. Kirby, J.F. Sheridan, I.B. Pragnell, and O. Smithies. 1995. Requirement of MIP-1 α for an inflammatory response to viral infection. *Science 269:1583*.

Cosman, Y., S.D. Lyman, R.L. Idzerda, M.P. Beckmann, L.S. Park, R.G. Goodwin, and C.J. March. 1990. A new cytokine receptor superfamily. *Trends. Biochem. Sci.* 15:265.

Costa, J.J., K. Matossian, M.B. Resnick, W.J. Beil, D.T.W. Wong, J.R. Gordon, A.M Dvorak, P.E. Weller, and S.J. Galli. 1993. Human eosinophils can express the cytokines tumour necrosis factor- α and macrophage inflammatory protein-1 α . J. Clin. Invest. 91:2673.

Croy, B.A., L.J. Guilbert, M.A. Browne, N.M. Gough, D.T. Stinchcomb, N. Reed, and T.G. Wegmann. 1991. Characterization of cytokine production by the metrial gland and granulated metrial gland cells. *J. Reproduc. Immunol.* 19:149.

Cuthbertson, R.A., and R.A. Lang. 1989. Developmental ocular disease in GM-CSF transgenic mice is mediated by autostimulated macrophages. *Dev. Biol.* 134:119.

Cuturi, M.C., I. Anegon, F. Sherman, R. Loudon, S.C. Clark, B. Perussia and G. Trinchieri. 1989. Production of hematopoietic colony-stimulating factors by human natural killer cells. *J. Exp. Med.* 169:569.

Dalgleish, A.G., P.C.L. Beverly, P.R. Clapham, D.H. Crawford, M.F. Greaves, and R.A. Weiss. 1984. The CD4 (T4) antigen is an essential component of the receptor for the AIDS retrovirus. *Nature 312:763*.

Dalloul, A.H., M. Arock, C. Fourcade, A. Hatzfeld, J.M, P. Debre, and M.D. Mossalayi. 1991. Human thymic epithelial cells produce interleukin-3. *Blood* 77:69.

Dastych, J., and D.D. Metcalfe. 1994. Stem cell factor induces mast cell adhesion to fibronectin. J. Immunol. 152:213.

Davatelis, G., P. Tekamp-Olson, S.D. Wolpe, K. Hermsen, C. Luedke, C. Gallegos, D. Coit, J. Merryweather, and A. Cerami. 1988. Cloning and characterization of a cDNA for murine macrophage inflammatory protein (MIP), a novel monokine with inflammatory and chemokinetic properties. *J. Exp. Med.* 167:1939.

Davis, C.B., I. Dikic, D. Unutmaz, C.M. Hill, J. Arthos, M.A. Siani, D.A. Thompson, J. Schlessinger, and D.R. Littman. 1997. Signal transduction due to HIV envelope interactions with chemokine receptors CXCR4 or CCR5. *J. Exp. Med.* 186:1793.

Dean, M., M. Carrington, C. Winkler, G.A. Huttley, M.W. Smith, R. Allikmets, J.J. Goedert, S.P. Buchbinder, E. Vittinghoff, E. Gomperts, S. Donfield, D. Vlahov, R. Kaslow, A. Saah, C. Rinaldo, R. Detels, Hemophilia Growth and Development Study, Multicenter AIDS Cohort Study, Multicenter Hemophilia Cohort Study, San Francisco City Cohort, ALIVE Study, and S.J. O'Brien. 1996. Genetic resolution of HIV-1 infection and progression to AIDS by a deletion allele of the *CKR5* structural gene. *Science* 273:1856.

Demetri, G.D., and J.D. Griffin. 1991. Granulocyte colony-stimulating factor and its receptor. *Blood* 78:2791.

Denburg, J.A. 1992. Basophil and mast cell lineages in vitro and in vivo. *Blood* 79:846.

Deng, H., R. Lui, W. Ellmeier, S. Choe, D. Unutmaz, M. Burkhart, P. Di Marzio, S. Marmon, R.E. Sutton, C.M. Hill, C.B. Davis, S.C. Peiper, T. J. Schall, D.R. Littman, and N.R. Landau. 1996. Identification of a major co-receptor for primary isolates of HIV-1. *Nature 381:661*.

Dent, L.A., M. Strath, A.L. Mellor, and C.J. Sanderson. 1990. Eosinophilia in transgenic mice expressing interleukin 5. J. Exp. Med. 172:1425.

Derynck, R., J.A. Jarrett, E.Y. Chen, D.H. Eaton, J.R. Bell, R.K. Assoian, A.B. Roberts, M.B. Sporn, and D.V. Goeddel. 1985. Human transforming growth factor- β complementary DNA sequence and expression in normal and transformed cells. *Nature 316:701.*

Deryugina, E.I., and C.E. Muller-Sieburg. 1993. Stromal cells in long-term cultures: Keys to the elucidation of hematopoietic development? *Crit. Rev. Immunol.* 13:115.

deVries, P., K.A. Brasel, J.R. Eisenman, A.R. Alpert, and D.E. Williams. 1991. The effect of recombinant mast cell growth factor on purified murine hematopoietic stem cells. *J. Exp. Med.* 173:1205.

Dewald, B., B. Moser, L. Barella, C. Schumacher, M. Baggiolini, and I. Clark-Lewis. 1992. IP-10, a γ -interferon-inducible protein related to interleukin-8, lacks neutrophil activating properties. *Immunol. Lett.* 32:81.

Dexter, T.M. 1982. Stromal cell associated haemopoiesis. J. Cell Physiol. (Suppl 1):87.

Dexter, T.M., T.D. Allen, and L.G. Lajtha. 1977a. Conditions controlling the proliferation of haemopoietic stem cells in vitro. J. Cell Physiol. 91:335.

Dexter, T.M., M.A.S. Moore, and A.P.C Sheridan. 1977b. Maintenance of haemopoietic stem cells and production of differentiated progeny in allogeneic and semiallogeneic bone marrow chimeras in vitro. J. Exp. Med. 145:1612.

Dickson, M.C., J.S. Martin, F.M. Cousins, A.B. Kulkarni, S. Karlsson, and R.J. Akhurst. 1995. Defective haematopoiesis and vasculogenesis in transforming growth factor- β 1 knock out mice. *Development* 121:1845.

Dinarello, C.A. 1994. The biological properties of interleukin-1. Eur. Cytokine Netw. 5:517.

Dirksen, U., R. Nishinakamura, P. Groneck, U. Hattenhorst, L. Nogee, R. Murray, and S. Burdach. 1997. Human pulmonary alveolar proteinosis associated with a defect in GM-CSF/IL-3/IL-5 receptor common β chain expression. J. Clin. Invest. 100:2211.

Dong, F., L.H. Hoefsloot, A.M. Schelen, L.C.A.M. Broeders, Y. Meijer, A.J.P. Veerman, I.P. Touw, and B. Lowenberg. 1994. Identification of a nonsense mutation in the granulocyte-colony-stimulating factor receptor in severe congenital neutropenia. *Proc. Natl. Acad. Sci. USA 91:4480.*

Doranz, B.J., J. Rucker, Y. Yi, R.J. Smyth, M. Samson, S.C. Peiper, M. Parmentier, R.G. Collman, and R.W. Doms. 1996. A dual-tropic primary HIV-1 isolate that uses fusin and the β -chemokine receptors CKR-5, CKR-3, and CKR-2b as fusion cofactors. *Cell* 85:1149.

Dragic, T., V. Litwin, G.P. Allaway, S.R. Martin, Y. Huang, K.A. Nagashima, C. Cayanan, P.J. Maddon, R.A. Koup, J.P. Moore, and W.A. Paxton. 1996. HIV-1 entry into CD4⁺ cells is mediated by the chemokine receptor CC-CKR-5. *Nature 381:667*.

D'Souza, M.P., and V.A. Harden. 1996. Chemokines and HIV-1 second receptors. *Nature Med. 2:1293.*

Du, X.X., and D.A. Williams. 1994. Interleukin-11: A multifunctional growth factor derived from the hematopoietic microenvironment. *Blood 83: 2023*.

Du, X.X., and D.A. Williams. 1997. Interleukin-11: Review of molecular biology, cell biology, and clinical use. *Blood* 89:3897.

Dunlop, D., E.G. Wright, S. Lorimore, G.J. Graham, T. Holyoake, D.J. Kerr, S.D. Wolpe, and I.B. Pragnell. 1992. Demonstration of stem cell inhibition and myeloprotective effects of SCI/rhMIP-1 α in vivo. *Blood* 79:2221.

Durand, B., G. Migliaccio, N.S. Yee, K. Eddleman, T. Huima-Byron, A.R. Migliaccio, and J.W. Adamson. 1994. Long-term generation of human mast cells in serum-free cultures of $CD34^+$ cord blood cells stimulated with stem cell factor and interleukin-3. *Blood 84:3667*.

Eaves, C.J., J.D. Cashman, S.D. Wolpe, and A.C. Eaves. 1993. Unresponsiveness of primitive chronic myeloid leukemia cells to macrophage inflammatory protein 1α , an inhibitor of primitive normal hematopoietic cells. *Proc. Natl. Acad. Sci. USA* 90:12015.

Ellias, L., and D.E. Van Epps. 1988. Early effects of G- and GM-CSF upon HL-60 proliferation and differentiation. *Leukemia 2:763*.

Ema, H., T. Suda, Y. Miura, and H. Nakauchi. 1990a. Colony formation of clonesorted human hematopoietic progenitors. *Blood* 75:1941.

Ema, H., T. Suda, K. Nagayoshi, Y. Miura, C.I. Civin, and H. Nakauchi. 1990b. Target cells for granulocyte colony-stimulating factor, interleukin-3, and interleukin-5 in differentiation pathways of neutrophils and eosinophils. *Blood* 76:1956.

Emerson, S.G., S. Thomas, J.L. Ferrara, and J.L. Greenstein. 1989. Developmental regulation of erythropoiesis by hematopoietic growth factors: Analysis on populations of BFU-E from bone marrow, peripheral blood, and fetal liver. *Blood* 74:49.

Endres, M.J., P.R. Clapham, M. Marsh, M. Ahuja, J. Davis Turner, A. McKnight, J.F. Thomas, B. Stoebenau-Haggarty, S. Choe, P.J. Vance, T.N.C. Wells, C.A. Power, S.S. Sutterwala, R.W. Doms, N.R. Landau, and J.A. Hoxie. 1996. CD4-independent infection by HIV-2 is mediated by fusin/CXCR4. *Cell* 87:745.

Escary, J-L., J. Perreau, D. Dumenil, S. Ezine, and P. Brulet. 1993. Leukemia inhibitory factor is necessary for maintenance of haematopoietic stem cells and thymocyte stimulation. *Nature* 363:361.

Fahey, T.J. III, K.J. Tracey, P. Tekamp-Olson, L.S. Cousens, W.G. Jones, G.T. Shires, A. Cerami, and B. Sherry. 1992. Macrophage inflammatory protein 1 modulates macrophage function. *J. Immunol.* 148:2764.

Fairbairn, L.J., G.J. Cowling, B.M. Reipert, and T.M. Dexter. 1993. Suppression of apoptosis allows differentiation and development of a multipotent hemopoietic cell line in the absence of added growth factors. *Cell* 74:823.

Fan, K., Q. Ruan, L. Sesenbrenner, and B.D-M. Chen. 1992. Up-regulation of granulocyte-macrophage colony-stimulating factor (GM-CSF) receptors in murine peritoneal exudate macrophages by both GM-CSF and IL-3. *J. Immunol.* 149:96.

Farzan, M., H. Choe, K.A. Martin, Y. Sun, M. Sidelko, C.R. Mackay, N.P. Gerard, J. Sodroski, and C. Gerard. 1997a. HIV-1 entry and macrophage inflammatory protein- 1β -mediated signaling are independent functions of the chemokine receptor CCR5. *J. Biol. Chem.* 272:6854.

Farzan, M., H. Choe, K. Martin, L. Marcon, W. Hofmann, G. Karlsson, Y. Sun, P. Barrett, N. Marchand, N. Sullivan, N. Gerard, C. Gerard, and J. Sodroski. 1997. Two orphan seven-transmembrane segment receptors which are expressed in CD4-positive cells support simian immunodeficiency virus infection. *J. Exp. Med.* 186:405.

Faust, E.A., D.C. Saffran, D. Toksoz, D.A. Williams, and O.N. Witte. 1993. Distinctive growth requirements and gene expression patterns distinguish progenitor B cells from pre-B cells. J. Exp. Med. 177:915.

Feng, Y., C.C. Broder, P.E. Kennedy, and E.A. Berger. 1996. HIV-1 entry cofactor: Functional cDNA cloning of a seven-transmembrane, G protein-coupled receptor. *Science* 272:872.

Ferrajoli, A., M. Talpaz, T.F. Zipf, C. Hirsch-Ginsberg, E. Estey, S.D. Wolpe, and Z. Estrov. 1994. Inhibition of acute myelogenous leukemia progenitor proliferation by macrophage inflammatory protein $1-\alpha$. *Leukemia* 8:798.

Fibbe, W.E., J. Van Damme, A. Billiau, P.J. Voogt, N. Duinkerken, and J.H.F. Falkenburg. 1986. Interleukin-1 (22-K factor) induces release of granulocytemacrophage colony-stimulating activity from human mononuclear phagocytes. *Blood* 68:1316.

Fibbe, W.E., J.W.M. van der Meer, J.H.F. Falkenburg, M.S. Hamilton, P.M. Kluin, and C.A. Dinarello. 1989. A single low dose of human recombinant interleukin 1 accelerates the recovery of neutrophils in mice with cyclophosphamide-induced neutropenia. *Exp. Hematol.* 17:805.

Flanagan, J.G., D.C. Chan, and P. Leder. 1991. Transmembrane form of the *kit* ligand growth factor is determined by alternative splicing and is missing in the Sl^d mutant. *Cell* 64:1025.

Flaumenhaft, R., M. Abe, Y. Sato, K. Miyazono, J. Harpel, C-H. Heldin, and D.B. Rifkin. 1993. Role of the latent TGF- β binding protein in the activation of latent TGF- β by co-cultures of endothelial and smooth muscle cells. *J. Cell. Biol.* 120:995.

Forssmann, U., M. Uguccioni, P. Loetscher, C.A. Dahinden, H. Langen, M. Thelen, and M. Baggiolini. 1997. Eotaxin-2, a novel CC chemokine that is selective for the chemokine receptor CCR3, and acts like eotaxin on human eosinophil and basophil leukocytes. J. Exp. Med. 185:2171.

Fraser, C., C.J. Eaves, and R.K. Humphries. 1991. Proliferation of marked clones of long-term repopulating hematopoietic stem cells. *Exp. Hematol.* 19:466.
Fujisawa, T., R. Abu-Ghazaleh, H. Kita, C.J. Sanderson, and G.J. Gleich. 1990. Regulatory effect of cytokines on eosinophil degranulation. J. Immunol. 144:642.

Gao, J-L., and P.M. Murphy. 1994. Human cytomegalovirus open reading frame US28 encodes a functional β chemokine receptor. *J. Biol. Chem.* 269:28539.

Gao, J-L., D.B. Kuhns, H.L. Tiffany, D. McDermott, X. Li, U. Francke, and P.M. Murphy. 1993. Structure and functional expression of the human macrophage inflammatory protein 1α /RANTES receptor. *J. Exp. Med.* 177:1421.

Gao, J-L., A.I. Sen, M. Kitaura, O. Yoshie, M.E. Rothenberg, P.M. Murphy, and A.D. Luster. 1996. Identification of a mouse eosinophil receptor for the CC chemokine eotaxin. *Biochem. Biophys. Res. Commun. 223:679*.

Gao, J-L., T.A. Wynn, Y. Chang, E.J. Lee, H.E. Broxmeyer, S. Cooper, H.L. Tiffany, H. Westphal, J. Kwon-Chung, and P.M. Murphy. 1997. Impaired host defence, hematopoiesis, granulomatous inflammation and type 1-type 2 cytokine balance in mice lacking CC chemokine receptor 1. *J. Exp. Med.* 185:1959.

Garcia-Zepeda, E.A., C. Combadiere, M.E. Rothenberg, M.N. Sarafi, F. Lavigne, Q. Hamid, P.M. Murphy, and A.D. Luster. 1996. Human monocyte chemoattractant protein (MCP)-4 is a novel CC chemokine with activities on monocytes, eosinophils, and basophils induced in allergic and nonallergic inflammation that signals through the CC chemokine receptors (CCR)-2 and -3. J. Immunol. 157:5613.

Gartner, S., and H.S. Kaplan. 1980. Long-term culture of human bone marrow cells. *Proc. Natl. Acad. Sci. USA* 71:4756.

Gasparetto, C., J. Laver, M. Abboud, A. Gillio, C. Smith, R.J. O'Reilly, and M.A.S. Moore. 1989. Effects of interleukin-1 on hematopoietic progenitors: Evidence of stimulatory and inhibitory activities in a primate model. *Blood.* 74:547.

Gasson, J.C. 1991. Molecular physiology of granulocyte-macrophage colonystimulating factor. *Blood* 77:1131.

Gearing, D.P., N.M. Gough, J.A. King, D.J. Hilton, N.A. Nicola, R.J. Simpson, E.C. Nice, A. Kelso, and D. Metcalf. 1987. Molecular cloning and expression of cDNA encoding a murine myeloid leukaemia inhibitory factor (LIF). *EMBO J. 6:3995*.

Geiser, A.G., J.J. Letterio, A.B. Kulkarni, S. Karlsson, A.B. Roberts, and M.B. Sporn. 1993. Transforming growth factor $\beta 1$ (TGF- $\beta 1$) controls expression of major histocompatability genes in the postnatal mouse: Aberrant histocompatability antigen expression in the pathogenesis of the TGF- $\beta 1$ null mouse phenotype. *Proc. Natl. Acad. Sci. USA* 90:9944.

Gentry, L.E., N.R. Webb, G.J. Lim, A.M. Brunner, J.E. Ranchalis, D.R. Twardzik, M.N. Lioubin, H. Marquardt, and A.F. Purchio. 1987. Type 1 transforming growth factor beta: Amplified expression and secretion of mature and precursor polypeptides in Chinese hamster ovary cells. *Mol. Cell. Biol.* 7:3418.

Gentry, L.E., M.N. Lioubin, A.F. Purchio, and H. Marquardt. 1988. Molecular events in the processing of recombinant type 1 pre-pro-transforming growth factor beta to the mature polypeptide. *Mol. Cell. Biol.* 8:4162.

Gerard, C., J-L. Frossard, M. Bahtia, A. Saluja, N.P. Gerard, B. Lu, and M. Steer. 1997. Targeted disruption of the β -chemokine receptor CCR1 protects against Pancreatitis-associated lung injury. *J. Clin. Invest.* 100:2022.

Gerondakis, S., A. Strasser, D. Metcalfe, G. Grigoriadis, J-P.Y. Scheerlinck, and R.J. Grumont. 1996. Rel-deficient T cells exhibit defects in production of interleukin 3 and grnaulocyte-macrophage colony-stimulating factor. *Proc. Natl. Acad. Sci. USA* 93:3405.

Gewirtz, A.M., B. Calabretta, B. Rucinski, S. Niewiarowski, and W.Y. Xu. 1989. Inhibition of human megakaryocytopoiesis in vitro by platelet factor 4 (PF4) and a synthetic COOH-terminal PF4 peptide. *J. Clin. Invest.* 83:1477.

Gewirtz, A.M., J. Zhang, J. Ratajczak, M. Ratajczak, K.S. Park, C.Q. Li, Z. Yan, and M. Poncz. 1995. Chemokine regulation of human megakaryocytopoiesis. *Blood* 86:2559.

Gollner, G., M.J. Aman, H.P. Steffens, C. Huber, C. Peschel, and H.G. Derigs. 1995. Interferon- α (IFN- α) inhibits granulocyte-macrophage colony-stimulating factor (GM-CSF) expression at the post-transcriptional level in murine bone marrow stromal cells. *Br. J. Haematol.* 91:8.

Gong, J-H., L.R. Ratkay, J.D. Waterfield, and I. Clark-Lewis. 1997. An antagonist of monocyte chemoattractant protein 1 (MCP-1) inhibits arthritis in the MRL-*lpr* mouse model. J. Exp. Med. 186:131.

Gordon, S. 1995. The macrophage. BioEssays 17:977.

Gordon, M.Y., J.A. King, and E.C. Gordon-Smith. 1980. Bone marrow fibroblasts, fat cells and colony-stimulating activity. *Br. J. Haematol.* 46:151.

Gordon, M.Y., G.P. Riley, S.M. Watt, and M.F. Greaves. 1987. Compartmentalization of a haematopoietic growth factor (GM-CSF) by glycosaminoglycans in the bone marrow microenvironment. *Nature 326:403*.

Gosling, J., F.S. Monteclaro, R.E. Atchison, H. Arai, C-L. Tsou, M.A. Goldsmith, and I.F. Charo. 1997. Molecular uncoupling of C-C chemokine receptor 5-induced chemotaxis and signal transduction from HIV-1 coreceptor activity. *Proc. Natl. Acad. Sci. USA* 94:5061.

Gough, N.M., and N.A. Nicola. 1990. Granulocyte-macrophage colony-stimulating factor. In *Colony-stimulating factors*. M. Dexter, J. Garland and N. Testa, eds. Dekker, New York, NY, p.39.

Graham, G.J. 1997. Growth inhibitors in haemopoiesis and leukaemogenesis. *Bailliere's Clinical Haematology 10:539*.

Graham, G.J., and I.B. Pragnell. 1990. Negative regulators of haemopoiesis- Current advances. *Prog. Growth Factor Res. 2:181*.

Graham, G.J., and I.B. Pragnell. 1992a. The haemopoietic stem cell: properties and control mechanisms. *Seminars in Cell Biol. 3:423*.

Graham, G.J., and I.B. Pragnell. 1992b. SCI/MIP-1a: A potent stem cell inhibitor with potential roles in development. *Dev. Biol.* 151:377.

Graham, G.J., and E.G. Wright. 1997. Haemopoietic stem cells: their heterogeneity and regulation. Int. J. Exp. Path. 78:197.

Graham, G.J., E.G. Wright, R. Hewick, S.D. Wolpe, N.M. Wilkie, D. Donaldson, S. Lorimore, and I.B. Pragnell. 1990. Identification and characterization of an inhibitor of haemopoietic stem cell proliferation. *Nature* 344:442.

Graham, G.J., M.G. Freshney, D. Donaldson, and I.B. Pragnell. 1992. Purification and biochemical characterisation of human and murine stem cell inhibitors (SCI). *Growth Factors* 7:151.

Graham, G.J., L. Zhou, J.A. Weatherbee, M.L-S. Tsang, M. Napolitano, W.J. Leonard, and I.B. Pragnell. 1993. Characterization of a receptor for macrophage inflammatory protein 1α and related proteins on human and murine cells. *Cell Growth and Diff. 4:* 137.

Graham, G.J., J. MacKenzie, S. Lowe, M.L-S. Tsang, J.A. Weatherbee, A. Issacson, J. Medicherla, F. Fang, P.C. Wilkinson, and I.B. Pragnell. 1994. Aggregation of the chemokine MIP-1 α is a dynamic and reversible phenomenon. *J. Biol. Chem.* 269:4974.

Graham, G.J., P.C. Wilkinson, R.J.B. Nibbs, S. Lowe, S.O. Kolset, A. Parker, M.G. Freshney, M.L-S. Tsang, and I.B. Pragnell. 1996. Uncoupling of stem cell inhibition from monocyte chemoattraction in MIP-1 α by mutagenesis of the proteoglycan binding site. *EMBO J.* 15:101.

Grainger, D.J., H.L. Kirschenlohr, J.C. Metcalfe, P.L. Weissberg, D.P. Wade, and R.M. Lawn. 1993. Proliferation of human smooth muscle cells promoted by lipoprotein(a). *Science 260:1655*.

Grainger, D.J., P.R. Kemp, A.C. Liu, R.M. Lawn, and J.C. Metcalfe. 1994. Activation of transforming growth factor- β is inhibited in transgenic apolipoprotein(a) mice. *Nature 370: 460.*

Greaves, D.R., W. Wang, D.J. Dairaghi, M.C. Dieu, B. de Saint-Vis. K. Franz-Bacon, D. Rossi, C. Caux, T. McClanahan, S. Gordon, A. Zlotnik, and T.J. Schall. 1997. CCR6, a CC chemokine receptor that interacts with macrophage inflammatory protein 3α and is highly expressed in human dendritic cells. *J. Exp. Med.* 186:837.

Greenberger, J.S. 1991. The hematopoietic microenvironment. Crit. Rev. Onc / Haematol. 11: 65.

Grewal, I.S., B.J. Rutledge, J.A. Fiorillo, L. Gu, R.P. Gladue, R.A. Flavell, and B.J. Rollins. 1997. Transgenic monocyte chemoattractant protein-1 (MCP-1) in pancreatic islets produces monocyte-rich insulitis without diabetes. *J. Immunol.* 159:401.

Gronowicz, E.S., C.A. Doss, F.D. Howard, D.C. Morrison, and S. Strober. 1980. An in vitro line of the B cell tumour BCL1 can be activated by LPS to secrete IgM1. J. *Immunol.* 125:976.

Grove, M., and M. Plumb. 1993. C/EBP, NF- κ B, and c-Ets family members and transcriptional regulation of the cell-specific and inducible macrophage inflammatory protein 1 α immediate-early gene. *Mol. Cell. Biol.* 13:5276.

Gruber, B.L., M.J. Marchese, and R.B. Kew. 1994. Transforming growth factor- β 1 mediates mast cell chemotaxis. *J. Immunol.* 152:5860.

Guba, S.C., C.I. Sartor, L.R. Gottschalk, J. Ye-Hu, T. Mulligan, and S.G. Emerson. 1992. Bone marrow stromal fibroblasts secrete interleukin-6 and granulocytemacrophage colony-stimulating factor in the absence of inflammatory stimulation: demonstration by serum-free bioassay, enzyme-linked immunosorbent assay, and reverse transcriptase polymerase chain reaction. *Blood* 80:1190.

Guigon, M., and D. Bonnet. 1995. Inhibitory peptides in hematopoiesis. *Exp. Hematol.* 23:477.

Hadley, T.J., and S.C. Peiper. 1997. From malaria to chemokine receptor: The emerging physiologic role of the Duffy blood group antigen. *Blood 89:3077*.

Han, Z.C., L. Sensebe, J.F. Abgrall, and J. Briere. 1990. Platelet factor 4 inhibits human megakaryocytopoiesis in vitro. *Blood* 75:1234.

Han, Z.C., M. Lu, J. Li, M. Defard, B. Boval, N. Schlegel, and J.P. Caen. 1997. Platelet factor 4 and other CXC chemokines support the survival of normal hematopoietic cells and reduce the chemosensitivity of cells to cytotoxic agents. *Blood* 89:2328.

Hapel, A.J, M-C. Fung, I.G. Young, G. Johnson, and D, Metcalf. 1985. Biologic properties of molecularly cloned and expressed murine interleukin-3. *Blood* 65:1453.

Hara, T., K.B. Bacon, L.C. Cho, A. Yoshimura, Y. Morikawa, N.G. Copeland, D.J. Gilbert, N.A. Jenkins, T.J. Schall, and A. Miyajima. 1995. Molecular cloning and functional characterization of a novel member of the C-C chemokine family. *J. Immunol.* 155:5352.

Hara, T., and A. Miyajima. 1996. Function and signal transduction mediated by the interleukin 3 receptor system in hematopoiesis. *Stem Cells* 14:605.

Harpel, J.G., C.N. Metz, S. Kojima, and D.B. Rifkin. 1992. Control of transforming growth factor- β activity: Latency VS. activation. *Prog. Growth Factor Res.* 4:321.

Harrison, D.E., and C.P. Lerner. 1991. Most primitive hematopoietic stem cells are stimulated to cycle rapidly after treatment with 5-Fluorouracil. *Blood* 78:1237.

Hatzfeld, J., M-L. Li, E.L. Brown, H. Sookdeo, J-P. Levesque, T. O'Toole, C. Gurney, S.C. Clark, and A. Hatzfeld. 1991. Release of early human hematopoietic progenitors from quiescence by antisense transforming growth factor $\beta 1$ or Rb oligonucleotides. *J. Exp. Med.* 174:925.

Haworth, C., F.M. Brennan, D. Chantry, M. Turner, R.N. Maini, and M. Feldmann. 1991. Expression of granulocyte-macrophage colony-stimulating factor in rheumatoid arthritis: regulation by tumour necrosis factor- α . *Eur. J. Immunol.* 21:2575.

Hayashi, M., Y. Luo, J. Laning, R.M. Strieter, and M.E. Dorf. 1995. Production and function of monocyte chemoattractant protein-1 and other β -chemokines in murine glial cells. *J. Neuroimmunol.* 60:143.

He, J., Y. Chen, M. Farzan, H. Choe, A. Ohagen, S. Gartner, J. Busciglio, X. Yang, W. Hofmann, W. Newman, C.R. Mackay, J. Sodroski, and D. Gabuzda. 1997. CCR3 and CCR5 are co-receptors for HIV-1 infection of microglia. *Nature* 385:645.

Hedrick, J.A., and A. Zlotnik. 1997. Identification and characterization of a novel β chemokine containing six conserved cysteines. *J. Immunol.* 159:1589.

Hedrick, J.A., V. Saylor, D. Figueroa, L. Mizoue, Y. Xu, S. Menon, J. Abrams, T. Handel, and A. Zlotnik. 1997. Lymphotactin is produced by NK cells and attracts both NK cells and T cells in vivo. *J. Immunol.* 158:1533.

Heesen, M., M.A. Berman, J.D. Benson, C. Gerard, and M.E. Dorf. 1996. Cloning of the mouse fusin gene, homologue to a human HIV-1 co-factor. J. Immunol. 157:5455.

Heinrich, M.C., D.C. Dooley, A.C. Freed, L. Band, M.E. Hoatlin, W.W. Keeble, S.T. Peters, K.V. Silvey, F.S. Ey, D. Kabat, R.T. Maziarz, and G.C. Bagby Jr. 1993a. Constitutive expression of steel factor gene by human stromal cells. *Blood* 82:771.

Heinrich, J.N., R-P. Ryseck, H. MacDonald-Bravo, and R. Bravo. 1993b. The product of a novel growth factor-activated gene, *fic*, is a biologically active "C-C" -type cytokine. *Mol. Cell. Biol.* 13:2020.

Heufler, C., G. Topar, F. Koch, B. Trockenbacher, E. Kampgen, N. Romani, and G. Schuler. 1992. Cytokine gene expression in murine epidermal cell suspensions: Interleukin 1 β and macrophage inflammatory protein 1 α are selectively expressed in langerhans cells but are differentially regulated in culture. *J. Exp. Med.* 176:1221.

Hilton, D.J., N.A. Nicola, and D. Metcalf. 1988. Specific binding of murine leukemia inhibitory factor to normal and leukemic monocytic cells. *Proc. Natl. Acad. Sci. USA* 85:5971.

Hilton, D.J., N.A. Nicola, and D. Metcalf. 1991. Distribution and comparison of receptors for leukemia inhibitory factor on murine hemopoietic and hepatic cells. J. *Cell Physiol.* 146:207.

Hirai, K., Y. Morita, Y. Misaki, K. Ohta, T. Takaishi, S. Suzuki, K. Motoyoshi, and T. Miyamoto. 1988. Modulation of human basophil histamine release by hemopoietic growth factors. *J. Immunol.* 141:3958.

Hirayama, F., J-P. Shih, A. Awgulewitsch, G.W. Warr, S.C. Clark, and M. Ogawa. 1992. Clonal proliferation of murine lymphohemopoietic progenitors in culture. *Proc. Natl. Acad. Sci. USA* 89:5907.

Hirayama, F., S.C. Clark, and M. Ogawa. 1994. Negative regulation of early B lymphopoiesis by interleukin 3 and interleukin 1a. *Proc. Natl. Acad. Sci. USA 91:469.*

Hirano, T., N. Nakano, K. Yasukawa, S. Kashiwamura, K. Shimizu, K. Nakajima, K.H. Pyun, and T. Kishimoto. 1985. Purification to homogeneity and characterization of human B-cell differentiation factor (BCDF or BSFp-2). *Proc. Natl. Acad. Sci. USA* 82:5490.

Hirano, T., K. Yasukawa, H. Harada, T. Taga, Y. Wanatabe, T. Matsuda, S. Kashiwamura, K. Nakajima, K. Koyoma, A. Iwamatsu, S. Tsuanasama, F. Sakiyama, H. Matsui, Y. Takahara, T. Taniguchi, and T. Kishimoto. 1986. Complementary DNA for a novel human interleukin (BSF-2) that induces B lymphocytes to produce immunoglobulin. *Nature 324:73*.

Hodgson, G.S., and T.R. Bradley. 1979. Properties of haematopoietic stem cells surviving 5-fluorouracil treatment: evidence for a pre-CFU-S cell? *Nature 281:381*.

Hodgson, G.S., T.R. Bradley, and J.M. Radley. 1982. The organization of hemopoietic tissue as inferred from the effects of 5-fluorouracil. *Exp. Hematol.* 10:26.

Holmes, W.E., J. Lee, W-J. Kuang, G.C. Rice, and W.I. Wood. 1991. Structure and functional expression of a human interleukin-8 receptor. *Science 253:1278*.

Hoogewerf, A.J., D. Black, A.E.I. Proudfoot, T.N.C. Wells, and C.A. Power. 1996. Molecular cloning of murine CC CKR-4 and high affinity binding of chemokines to murine and human CC CKR-4. *Biochem. Biophys. Res. Commun.* 218:337. Horuk, R., C.E. Chitnis, W.C. Darbonne, T.J. Colby, A. Rybicki, T.J. Hadley, and L.H. Miller. 1993. A receptor for the malarial parasite *Plasmodium vivax*: The erythrocyte chemokine receptor. *Science* 261:1182.

Horuk, R., A. Martin, J. Hesselgesser, T. Hadley, Z. Lu, Z. Wang, and S.C. Peiper. 1996. The Duffy antigen receptor for chemokines: structural analysis and expression in the brain. *J. Leukoc. Biol.* 59:29.

Horuk, R., A.W. Martin, Z. Wang, L. Schweitzer, A. Gerassimides, H. Guo, Z. Lu, J. Hesselgesser, H.D. Perez, J. Kim, J. Parker, T.J. Hadley, and S.C. Peiper. 1997. Expression of chemokine receptors by subsets of neurons in the central nervous system. *J. Immunol.* 158:2882.

Hromas, R., C.H. Kim, M. Klemsz, M. Krathwohl, K. Fife, S. Cooper, C. Schnizlein-Bick, and H.E. Broxmeyer. 1997a. Isolation and characterization of Exodus-2, a novel C-C chemokine with a unique 37-amino acid carboxyl-terminal extension. *J. Immunol.* 159:2554.

Hromas, R., P.W. Gray, D. Chantry, R. Godiska, M. Krathwohl, K. Fife, G.I. Bell, J. Takeda, S. Aronica, M. Gordon, S. Cooper, H.E. Broxmeyer, and M.J. Klemsz. 1997b. Cloning and characterization of Exodus, a novel β-chemokine. *Blood* 89:3315.

Huang, E., K. Nocka, D.R. Beier, T-Y. Chu, J. Buck, H-W. Lahm, D. Wellner, P. Leder, and P. Besmer. 1990. The hematopoietic growth factor KL is encoded by the Sl locus and is the ligand of the *c*-kit receptor, the gene product of the *W* locus. *Cell* 63:225.

Huang, E., K.H. Nocka, J. Buck, and P. Besmer. 1992. Differential expression and processing of two cell associated forms of the kit-ligand: KL-1 and KL-2. *Mol. Biol. Cell.* 3:349.

Huang, Y., W.A. Paxton, S.M. Wolinsky, A.U. Neumann, L. Zhang, T. He, S. Kang, D. Ceradini, Z. Jin, K. Yazdanbakhsh, K. Kunstman, D. Erickson, E. Dragon, N.R. Landau, J. Phair, D.D. Ho, and R.A. Koup. 1996. The role of a mutant CCR5 allele in HIV-1 transmission and disease progression. *Nature Med. 2:1240*.

Huffnagle, G.B., R.M. Strieter, T.J. Standiford, R.A. McDonald, M.D. Burdick, S.L. Kunkel, and G.B. Toews. 1995. The role of monocyte chemotactic protein-1 (MCP-1) in the recruitment of monocytes and CD4⁺ T cells during a pulmonary *Cryptococcus neoformans* infection. J. Immunol. 155:4790.

Hunt, P., D. Robertson, D. Weiss, D. Rennick, F. Lee, and O.N. Witte. 1987. A single bone marrow-derived stromal cell type supports the in vitro growth of early lymphoid and myeloid cells. *Cell* 48:997.

Ikebuchi, K., G.G. Wong, S.C. Clark, J.N. Ihle, Y. Hirai, and M. Ogawa. 1987. Interleukin 6 enhancement of interleukin 3-dependent proliferation of multipotential hemopoietic progenitors. *Proc. Natl. Acad. Sci. USA* 84:9035. Ikebuchi, K., S.C. Clark, J.N. Ihle, L.M. Souza, and M. Ogawa. 1988. Granulocyte colony-stimulating factor enhances interleukin 3-dependent proliferation of multipotential hemopoietic progenitors. *Proc. Natl. Acad. Sci. USA* 85:3445.

Imai, T., M. Baba, M. Nishimura, M. Kakizaki, S. Takagi, and O. Yoshie. 1997. The T cell-directed CC chemokine TARC is a highly specific biological ligand for CC chemokine receptor 4. J. Biol. Chem. 272:15036.

Imai, T., D. Chantry, C.J. Raport, C.L. Wood, M. Nishimura, R. Godiska, O. Yoshie, and P.W. Gray. 1998. Macrophage-derived chemokine is a functional ligand for the CC chemokine receptor 4. *J. Biol. Chem.* 273:1764.

Ishibashi, T., H. Kimura, T. Uchida, S. Kariyone, P. Friese, and S.A. Burstein. 1989. Human interleukin 6 is a direct promoter of maturation of megakaryocytes *in vitro*. *Proc. Natl. Acad. Sci. USA 86:5953*.

Ishizuka, K., R. Igata-Yi, T. Kimura, K. Hieshima, T. Kukita, Y. Kin, Y. Misumi, M. Yamamoto, H. Nomiyama, R. Miura, J. Takamatsu, S. Katsuragi, and T. Miyakawa. 1997. Expression and distribution of CC chemokine macrophage inflammatory protein- 1α /LD78 in the human brain. *NeuroReport 8:1215*.

Jacobsen, S.E.W., F.W. Ruscetti, C.M. Dubois, J. Lee, T.C. Boone, and J.R. Keller. 1991a. Transforming growth factor- β trans-modulates the expression of colony stimulating factor receptors on murine hematopoietic progenitor cell lines. *Blood* 77:1706.

Jacobsen, S.E.W., J.R. Keller, F.W. Ruscetti, P. Kondaiah, A.B. Roberts, and L.A. Falk. 1991b. Bidirectional effects of transforming growth factor β (TGF- β) on colony-stimulating factor-induced human myelopoiesis in vitro: Differential effects of distinct TGF- β isoforms. *Blood* 78:2239.

Jacobsen, F.W., O.P. Veiby, C. Skjønsberg, and S.E.W. Jacobsen. 1993. Novel role of interleukin 7 in myelopoiesis: Stimulation of primitive murine hematopoietic progenitor cells. *J. Exp. Med.* 178:1777.

Jacobsen, F.W., L.S. Rusten, and S.E.W. Jacobsen. 1994a. Direct synergistic effects of interleukin-7 on in vitro myelopoiesis of human $CD34^+$ bone marrow progenitors. *Blood 84:775*.

Jacobsen, F.W., O.P. Veiby, and S.E.W. Jacobsen. 1994b. IL-7 stimulates CSFinduced proliferation of murine bone marrow macrophages and Mac-1⁺ myeloid progenitors in vitro. *J. Immunol.* 153:270.

Jacobsen, S.E.W., F.W. Ruscetti, M. Oritz, J.M. Gooya, and J.R. Keller. 1994c. The growth response of Lin^Thy-1⁺ hematopoietic progenitors to cytokines is determined by the balance between the synergy of multiple stimulators and negative cooperation of multiple inhibitors. *Exp. Hemtol.* 22:985.

Jacobsen, F.W., J.R. Keller, F.W. Ruscetti, O.P. Veiby, and S.E.W. Jacobsen. 1995a. Direct synergistic effects of IL-4 and IL-11 on the proliferation of primitive hematopoietic progenitor cells. *Exp. Hematol.* 23:990.

Jacobsen, F.W., T. Stokke, and S.E.W. Jacobsen. 1995b. Transforming growth factor- β potently inhibits the viability-promoting activity of stem cell factor and other cytokines and induces apoptosis of primitive murine hematopoietic progenitor cells. *Blood 86:2957*.

Jakowlew, S.B., P.J. Dillard, P. Kondaiah, M.B. Sporn, and A.B. Roberts. 1988. Complementary deoxyribonucleic acid cloning of a novel transforming growth factorbeta messenger ribonucleic acid from chicken embryo chondrocytes. *Mol. Endocrinol.* 2:1186.

James, K. 1990. Interactions between cytokines and alpha 2-macroglobulin. *Immun. Today 11:163*.

Jansen, J.H., G-J.H.M. Wientjens, W.E. Fibbe, R. Willemze, and H.C. Kluin-Nelemans. 1989. Inhibition of human macrophage colony formation by interleukin 4. J. Exp. Med. 170:577.

Jarmin, D.I., P.A. Kulmburg, N.E. Huber, G. Baumann, E.E. Prieschl-Strassmayr, and T. Baumruker. 1994. A transcription factor with AP3-like binding specificity mediates gene regulation after an allergic triggering with IgE and Ag in mouse mast cells. *J. Immunol.* 153:5720.

Jenkins, F., P.N. Cockerill, D. Bohmann, and M.F. Shannon. 1995. Multiple signals are required for function of the human granulocyte-macrophage colony-stimulating factor gene promoter in T cells. *J. Immunol.* 155:1240.

Jennings, J.C., and S. Mohan. 1990. Heterogeneity of latent transforming growth factor-beta isolated from bone matrix proteins. *Endocrinology 126:1014*.

Jia, G-Q., J.A. Gonzalo, C. Lloyd, L. Kremer, L. Lu, C. Martinez-A, B.K. Wershil, and J.C. Gutierrez-Ramos. 1996. Distinct expression and function of the novel mouse chemokine monocyte chemotactic protein-5 in lung allergic inflammation. *J. Exp. Med.* 184:1939.

Jiang, Y., D.I. Beller, G. Frendl, and D.T. Graves. 1992. Monocyte chemoattractant protein-1 regulates adhesion molecule expression and cytokine production in human monocytes. *J. Immunol.* 148:2423.

Johe, K.K., T.G. Hazel, T. Muller, M.M. Dugich-Djordjevic, and R.D.G. McKay. 1996. Single factors direct the differentiation of stem cells from the fetal and adult central nervous system. *Genes Dev* 10:3129.

Johnatty, R.N., D.D. Taub, S.P. Reeder, S.M. Turcovski-Corrales, D.W. Cottam, T.J. Stephenson, and R.C. Rees. 1997. Cytokine and chemokine regulation of proMMP-9 and TIMP-1 production by human peripheral blood lymphocytes. *J. Immunol.* 158:2327.

Johns, L.D., K.C. Flanders, G.E. Ranges, and S. Sriram. 1991. Successful treatment of experimental allergic encephalomyelitis with transforming growth factor- β 1. *J. Immunol.* 147:1792.

Johnson, C.S., D.J. Keckler, M.I. Topper, P.G. Braunschweiger, and P. Furmanski. 1989. In vivo hematopoietic effects of recombinant interleukin-1 α in mice: Stimulation of granulocytic, monocytic, megakaryocytic and early erythroid progenitors, suppression of late-stage erythropoiesis, and reversal of erythroid suppression with erythropoietin. *Blood.* 73:678.

Jones, R.J., P. Celano, S.J. Sharkis, and L.L. Sensenbrenner. 1989. Two phases of engraftment established by serial bone marrow transplantation in mice. *Blood* 73: 397.

Jones, R.J., J.E. Wagner, P. Celano, M.S. Zicha, and S.J. Sharkis. 1990. Separation of pluripotent haematopoietic stem cells from spleen colony forming cells. *Nature 347: 188.*

Jose, P.J., D.A. Griffiths-Johnson, P.D. Collins, D.T. Walsh, R. Moqbel, N.F. Totty, O. Truong, J.J. Hsuan, and T.J. Williams. 1994. Eotaxin: A potent eosinophil chemoattractant cytokine detected in a guinea pig model of allergic airways inflammation. J. Exp. Med. 179:881.

Kameyoshi, Y., A. Dorschner, A.I. Mallet, E. Christophers, and J-M. Schroder. 1992. Cytokine RANTES released by thrombin-stimulated platelets is a potent attractant for human eosinophils. *J. Exp. Med.* 176:587.

Kanzaki, T., A. Olofsson, A. Moren, C. Wernstedt, U. Hellman, K. Miyazono, L. Claesson-Welsh, and C-H. Heldin. 1990. TGF- β 1 binding protein: A component of the large latent complex of TGF- β 1 with multiple repeat sequences. *Cell* 61:1051.

Kapur, R., E.T. Everett, J. Uffman, M. McAndrews-Hill, R. Cooper, J. Ryder, T. Vik, and D.A. Williams. 1997. Overexpression of human stem cell factor impairs melanocyte, mast cell, and thymocyte development: A role for receptor tyrosine kinase-mediated mitogen activated protein kinase activation in cell differentiation. *Blood 90:3018*.

Karbassi, A., J.M. Becker, J.S. Forster, and R.N. Moore. 1987. Enhanced killing of Candida albicans by murine macrophages treated with macrophage colony-stimulating factor: evidence for augmented expression of mannose receptors. *J. Immunol.* 139:417.

Karpus, W.J., N.W. Lukacs, B.L. McRae, R.M. Strieter, S.J. Kunkel, and S.D. Miller. 1995. An important role for the chemokine macrophage inflammatory protein- 1α in the pathogenesis of the T cell-mediated autoimmune disease, experimental autoimmune encephalomyelitis. *J. Immunol.* 155:5003.

Kasama, T., R.M. Strieter, T.J. Standiford, M.D. Burdick, and S.J. Kunkel. 1993. Expression and regulation of human neutrophil-derived macrophage inflammatory protein 1α . J. Exp. Med. 178:63.

Kasama, T., R.M. Strieter, N.W. Lukacs, M.D. Burdick, and S.J. Kunkel. 1994. Regulation of neutrophil-derived chemokine expression by IL-10. *J. Immunol.* 152:3559.

Kaushansky, K. 1995. Thrombopoietin: The primary regulator of platelet production. *Blood 86:419*.

Kawasaki, E.S., M.B. Ladner, A.M. Wang, J. Van Arsdell, M.K. Warren, M.Y. Coyne, V.L. Schweickart, M.T. Lee, K.J. Wilson, A. Boosman, E.R. Stanley, P. Ralph, and D.F. Mark. 1985. Molecular cloning of a complementary DNA encoding human macrophage-specific colony-stimulating factor (CSF-1). *Science 230:291*.

Kehrl, J.H., A.B. Roberts, L.M. Wakefield, S. Jakowlew, M.B. Sporn and A.S. Fauci. 1986a. Transforming growth factor β is an important immunomodulatory protein for human B lymphocytes. *J. Immunol.* 137:3855.

Kehrl, J.H., L.M. Wakefield, A.B. Roberts, S. Jakowlew, M. Alvarez-Mon, R. Derynck, M.B. Sporn and A.S. Fauci. 1986b. Production of transforming growth factor β by human T lymphocytes and its potential role in the regulation of T cell growth *J. Exp. Med.* 163:1037.

Kehrl, C. Thevenin, P. Rieckmann, and A.S. Fauci. 1991. Transforming growth factor- β suppresses human B lymphocyte Ig production by inhibiting synthesis and the switch from the membrane form to the secreted form of Ig mRNA. *J. Immunol.* 146:4016.

Kekow, J., W. Wachsman, J.A. McCutchan, M. Cronin, D.A. Carson, and M. Lotz. 1990. Transforming growth factor β and noncytopathic mechanisms of immunodeficiency in human immunodeficiency virus infection. *Proc. Natl. Acad. Sci. USA 87:8321*.

Keller, J.R., C. Mantel, G.K. Sing, L.R. Ellingsworth, S.K. Ruscetti, and F.W. Ruscetti. 1988. Transforming growth factor β 1 selectively regulates early murine hematopoietic progenitors and inhibits the growth of IL-3-dependent myeloid leukemia cell lines. *J. Exp. Med.* 168:737.

Keller, J.R., I.K. McNiece, K.T. Sill, L.R. Ellingsworth, P.J. Quesenberry, G.K. Sing, and F.W. Ruscetti. 1990. Transforming growth factor β directly regulates primitive murine hematopoietic cell proliferation. *Blood* 75:596.

Keller, J.R., S.E.W. Jacobsen, K.T. Sill, L.R. Ellingsworth, and F.W. Ruscetti. 1991. Stimulation of granulopoiesis by transforming growth factor β : Synergy with granulocyte/macrophage-colony-stimulating factor. *Proc. Natl. Acad. Sci.* 88:7190.

Keller, J.R., S.H. Bartelmez, E. Sitnicka, F.W. Ruscetti, M. Ortiz, J. M. Gooya, and S.E.W. Jacobsen. 1994. Distinct and overlapping direct effects of macrophage inflammatory protein-1 α and transforming growth factor β on hematopoietic progenitor/stem cell growth. *Blood* 84:2175.

Keller, J.R., M. Ortiz, and F.W. Ruscetti. 1995. Steel factor (c-kit ligand) promotes the survival of hematopoietic stem/progenitor cells in the absence of cell division. *Blood* 86:1757.

Keller, J.R., J.M. Gooya, and F.W. Ruscetti. 1996. Direct synergistic effects of leukemia inhibitory factor on hematopoietic progenitor cell growth: Comparison with other hematopoietins that use the gp130 receptor subunit. *Blood 88:863*.

Kelner, G.S., J. Kennedy, K.B. Bacon, S. Kleyensteuber, D.A. Largaespada, N.A. Jenkins, N.G. Copeland, J.F. Bazan, K.W. Moore, T.J. Schall, and A. Zlotnik. 1994. Lymphotactin: A cytokine that represents a new class of chemokine. *Science* 266:1395.

Kimata, H., A. Yoshida, C. Ishioka, M. Fujimoto, I. Lindley and K. Furusho. 1996. RANTES and Macrophage Inflammatory Protein 1α selectively enhance immunoglobulin (IgE) and IgG4 production by human B cells. J. Exp. Med. 183:2397.

Kincade, P.W., G. Lee, G. Fernandes, M.A.S. Moore, N. Williams, and R.A. Good. 1979. Abnormalities in clonable B lymphocytes and myeloid progenitors in autoimmune NZB mice. *Proc. Natl. Acad. Sci. USA* 76:3464.

Kindler, V., B. Thorens, S. De Kossodo, B. Allet, J.F. Eliason, D. Thatcher, N. Farber, and P. Vassalli. 1986. Stimulation of haematopoiesis *in vivo* by recombinant bacterial murine interleukin 3. *Proc. Natl. Acad. Sci. USA* 83:1001.

Kishimoto, T. 1989. The biology of interleukin-6. Blood 74:1.

Kishimoto, T., S. Akira, and T. Taga. 1992. Interleukin-6 and its receptor: A paradigm for cytokines. *Science 258:593*.

Kishimoto, T., S. Akira, M. Narazaki, and T. Taga. 1995. Interleukin-6 family of cytokines and gp130. *Blood 86:1243*.

Kitamura, M. 1997. Identification of an inhibitor targeting macrophage production of monocyte chemoattractant protein-1 as TGF- β_1 . J. Immunol. 159:1404.

Kitamura, Y., S. Go, and K. Hatanaka. 1978. Decrease of mast cells in W/W' mice and their increase by bone marrow transplantation. *Blood* 52:447.

Kitamura, Y., and S. Go. 1979. Decreased production of mast cells in Sl/Sl^d anemic mice. *Blood* 53:492.

Kitaura, M., T. Nakajima, T. Imai, S. Harada, C. Combadiere, H.L. Tiffany, P.M. Murphy, and O. Yoshie. 1996. Molecular cloning of human Eotaxin, an eosinophil-selective CC chemokine, and identification of a specific eosinophil Eotaxin receptor, CC chemokine receptor 3. 1996. *J. Biol. Chem.* 271:7725.

Kittler, E.L.W., H. McGrath, D. Temeles, R.B. Crittenden, V.K. Kister, and P.J. Quesenberry. 1992. Biologic significance of constitutive and subliminal growth factor production by bone marrow stroma. *Blood* 79:3168.

Kledal, T.N., M.M. Rosenkilde, F. Coulin, G. Simmons, A.H. Johnsen, S. Alouani, C.A. Power, H.R. Luttichau, J. Gerstoft, P.R. Clapham, I. Clark-Lewis, T.N.C. Wells, T.W. Schwartz. 1997. A broad-spectrum chemokine antagonist encoded by Kaposi's sarcoma-associated herpesvirus. *Science* 277:1656.

Klein, B., X-G. Zhang, Z-Y. Lu, and R. Bataille. 1995. Interleukin-6 in human multiple myeloma. *Blood* 85:863.

Klimpel, G.R., A.K. Chopra, K.E. Langley, J. Wypych, C.A. Annable, D. Kaiserlian, P.B. Ernst, and J.W. Peterson. 1995. A role for stem cell factor and c-kit in the murine intestinal tract secretory response to cholera toxin. *J. Exp. Med.* 182:1931.

Koch, A.E., P.J. Polverini, S.L. Kunkel, L.A. Harlow, L.A. DiPietro, V.M. Elner, and R.M. Strieter. 1992. Interleukin-8 as a macrophage-derived mediator of angiogenesis. *Science 258:1798*.

Koch, A.E., S.L. Kunkel, L.A. Harlow, D.D. Mazarakis, G.K. Haines, M.D. Burdick, R.M. Pope, and R.M. Strieter. 1994. Macrophage inflammatory protein-1a. A novel chemotactic cytokine for macrophages in rheumatoid arthritis. *J. Clin. Invest.* 93:921.

Kodama, H., M. Nose, Y. Yamaguchi, J. Tsunoda, T. Suda, and S. Nishikawa. 1992. In vitro proliferation of primitive hemopoietic stem cells supported by stromal cells: Evidence for the presence of a mechanism(s) other than that involving *c-kit* receptor and its ligand. *J. Exp. Med.* 176:351.

Koike, K., J.N. Ihle, and M. Ogawa. 1986. Declining sensitivity to interleukin 3 of murine multipotential hemopoietic progenitors during their development. Application to a culture system that favors blast cell colony formation. J. Clin. Invest. 77:894.

Koike, K., M. Ogawa, J.N. Ihle, T. Miyake, T. Shimizu, A. Miyajima, T. Yokota, and K. Arai. 1987. Recombinant murine granulocyte-macrophage (GM) colonystimulating factor supports formation of GM and multipotent blast cell colonies in culture: comparison with the effects of interleukin-3. *J. Cell Physiol.* 131:458.

Kondaiah, P., M.J. Sands, J.M. Smith, A. Fields, A.B. Roberts, M.B. Sporn, and D.A. Melton. 1990. Identification of a novel transforming growth factor- β (TGF- β 5) mRNA in *Xenopus laevis*. J. Biol. Chem. 265:1089.

Krathwohl, M.D., R. Hromas, D.R. Brown, and H.E. Broxmeyer. 1997. Functional characterization of the C-C chemokine-like molecules encoded by molluscum contagiosum virus types 1 and 2. *Proc. Natl. Acad. Sci. USA* 94:9875.

Krantz, S.B. 1991. Erythropoietin. Blood 77:419.

Kuang, Y., Y. Wu, H. Jiang, and D. Wu. 1996. Selective G protein coupling by C-C chemokine receptors. J. Biol. Chem. 271: 3975.

Kulkarni, A.B., C-G. Huh, D. Becker, A. Geiser, M. Lyght, K.C. Flanders, A.B. Roberts, M.B. Sporn, J.M. Ward, and S. Karlsson. 1993. Transforming growth factor β 1 null mutation in mice causes excessive inflammatory response and early death. *Proc. Natl. Acad. Sci. USA* 90:770.

Kulmburg, P.A., N.E. Huber, B.J. Scheer, M. Wrann, and T. Baumruker. 1992. Immunoglobulin E plus antigen challenge induces a novel intercrine/chemokine in mouse mast cells. J. Exp. Med. 176:1773

Kunkel, S.L., N. Lukacs, T. Kasama, and R.M. Strieter. 1996. The role of chemokines in inflammatory joint disease. *J. Leukoc. Biol.* 59:6.

Kurihara, T., G. Warr, J. Loy, and R. Bravo. 1997. Defects in macrophage recruitment and host defence in mice lacking the CCR2 chemokine receptor. J. Exp. Med. 186:1757.

Kurimoto, Y., A.L. De Weck, and C.A. Dahinden. 1989. Interleukin 3-dependent mediator release in basophils triggered by C5a. J. Exp. Med. 170:467.

Kurt-Jones, E.A., D.I. Peller, S.B. Mizel, and E.R. Unanue. 1985. Identification of a membrane-associated IL-1 in macrophages. *Proc. Natl. Acad. Sci. USA* 82:1204.

Kuruvilla, A.P., R. Shah, G.M. Hochwald, H.D. Liggitt, M.A. Palladino, and G.J. Thorbecke. 1991. Protective effect of transforming growth factor β 1 on experimental autoimmune diseases in mice. *Proc. Natl. Acad. Sci. USA* 88:2918.

Kuziel, W.A., S.J. Morgan, T.C. Dawson, S. Griffin, O. Smithies, K. Ley, and N. Maeda. 1997. Severe reduction in leukocyte adhesion and monocyte extravasation in mice deficient in CC chemokine receptor 2. *Proc. Natl. Acad. Sci. USA* 94:12053.

Ladner, M.B., G.A. Martin, J.A. Noble, D.M. Nikoloff, R. Tal, E.S. Kawasaki, and T.J. White. 1987. Human CSF-1: Gene structure and alternative splicing of mRNA precursors. *EMBO J. 6:2693*.

Lagasse, E., and I.L. Weissman. 1997. Enforced expression of Bcl-2 in monocytes rescues macrophages and partially reverses osteopetrosis in *op/op* mice. *Cell* 89:1021.

Lalani, A.S., K. Graham, K. Mossman, K. Rajarathnam, I. Clark-Lewis, D. Kelvin, and G. McFadden. 1997. The purified Myxoma virus gamma interferon receptor homolog M-T7 interacts with the heparin-binding domains of chemokines. *J. Virol.* 71:4356.

LaMarre, J., M.A. Hayes, G.K. Wollenberg, I. Hussaini, S.W. Hall, S.L. Gonias. 1991. An alpha 2-macroglobulin receptor-dependent mechanism for the plasma clearance of transforming growth factor-beta 1 in mice. *J. Clin. Invest.* 87:39.

Lang, R.A., and J.M. Bishop. 1993. Macrophages are required for cell death and tissue remodelling in the developing mouse eye. *Cell* 74:453.

Lang, R.A., D. Metcalf, R.A. Cuthbertson, I. Lyons, E. Stanley, A. Kelso, G. Kannourakis, D.J. Williamson, G.K. Klintworth, T.J. Gonda, and A.R. Dunn. 1987. Transgenic mice expressing a hemopoietic growth factor gene (GM-CSF) develop accumulations of macrophages, blindness, and a fatal syndrome of tissue damage. *Cell* 51:675.

Lang, R.A., R.A. Cuthbertson, and A.R. Dunn. 1992. TNF α , IL-1 α and bFGF are implicated in the complex disease of GM-CSF transgenic mice. *Growth Factors* 6:1'31.

Lapham, C.K., J. Ouyang, B. Chandrasekhar, N.Y. Nguyen, D.S. Dimitrov, and H. Gollding. 1996. Evidence for cell-surface association between fusin and the CD4-gp120 complex in human cell lines. *Science* 274:602.

Larsen, C.G., A.O. Anderson, E. Appella, J.J. Oppenheim, and K. Matsushima. 1989. Neutrophil activating protein (NAP-1) is also chemotactic for T lymphocytes. *Science* 243:1464.

Laterveer, L., I.J.D. Lindley, M.S. Hamilton, R. Willemze, and W.E. Fibbe. 1995. Interleukin-8 induces rapid mobilization of hematopoietic stem cells with radioprotective capacity and long-term myelolymphoid repopulating ability. *Blood* 85:2269.

Laterveer, L., I.J.D. Lindley, D.P.M. Heemskerk, J.A.J. Camps, E.K.J. Pauwels, R. Willemze, and W.E. Fibbe. 1996a. Rapid mobilization of hematopoietic progenitor cells in rhesus monkeys by a single intravenous injection of interleukin-8. *Blood* 87:781.

Laterveer, L., J.M. Zijlmans, I.J.D. Lindley, M.S. Hamilton, R. Willemze, and W.E. Fibbe. 1996b. Improved survival of lethally irradiated recipient mice transplanted with circulating progenitor cells mobilized by IL-8 after pretreatment with stem cell factor. *Exp. Hematol.* 24:1387.

Lawrence, D.A., R. Pircher, and P. Jullien. 1985. Conversion of a high molecular weight latent beta-TGF from chicken embryo fibroblasts into a low molecular weight active beta-TGF under acidic conditions. *Biochem. Biophys. Res. Commun.* 133:1026.

Leary, A.G., K. Ikebuchi, Y. Hirai, G.G. Wong, Y-C. Yang, S.C. Clark, and M. Ogawa. 1988. Synergism between interleukin-6 and interleukin-3 in supporting proliferation of human hematopoietic stem cells: Comparison with interleukin-1 α . Blood 71: 1759.

Leary, A.G., G.G. Wong, S.C. Clark, A.G. Smith and M. Ogawa. 1990. Leukemia inhibitory factor / differentiation-inhibitory activity / human interleukin for DA cells augments proliferation of human hematopoietic stem cells. *Blood* 75:1960.

Leary, A.G., H.Q. Zeng, S.C. Clark, and M. Ogawa. 1992. Growth factor requirements for survival in G_0 and entry into the cell cycle of primitive human hemopoietic progenitors. *Proc. Natl. Acad. Sci. USA* 89:4013.

Lehnert, S.A., and R.J. Akhurst. 1988. Embryonic expression pattern of TGF- β type 1 RNA suggests both paracrine and autocrine mechanisms of action. *Development* 104:263.

Lemoli, R.M., M. Fogli, A. Fortuna, M.R. Motta, S. Rizzi, C. Benini, and S. Tura. 1993. Interleukin-11 stimulates the proliferation of human hematopoietic CD34⁺ and CD34⁺CD33⁻DR⁻ cells and synergizes with stem cell factor, interleukin-3, and granulocyte-macrophage colony-stimulating factor. *Exp. Hematol.* 21:1668.

Leslie, K.B., S. Jalbert, P. Orban, M. Welham, V. Duronio, and J.W. Schrader. 1996. Genetic basis of hypo-responsiveness of A/J mice to interleukin-3. *Blood* 87:3186.

Letterio, J.J., A.G. Geiser, A.B. Kulkarni, N.S. Roche, M.B. Sporn, and A.B. Roberts. 1994. Maternal rescue of transforming growth factor-β1 null mice. *Science 264:1936*.

Li, C.L., and G.R. Johnson. 1994. Stem cell factor enhances the survival but not the self-renewal of murine hematopoietic long-term repopulating cells. *Blood.* 84:408.

Li, H., T.C. Sim, J.A. Grant, and R. Alam. 1996. The production of macrophage inflammatory protein- 1α by human basophils. *J. Immunol.* 157:1207.

Liao, F., R.L. Rabin, J.R. Yannelli, L.G. Koniaris, P. Vanguri, and J.M. Farber. 1995. Human Mig chemokine: Biochemical and functional characterization. *J. Exp. Med.* 182:1301.

Liao, F., G. Alkhatib, K.W.C. Peden, G. Sharma, E.A. Berger, J.M. Farber. 1997. STRL33, A novel chemokine receptor-like protein, functions as a fusion cofactor for both macrophage-tropic and T cell line-tropic HIV-1. J. Exp. Med. 185:2015.

Lieschke, G.J., D. Grail, G. Hodgson, D. Metcalf, E. Stanley, C. Cheers, K.J. Fowler, S. Basu, Y.F. Zhan, and A.R. Dunn. 1994. Mice lacking granulocyte colonystimulating factor have chronic neutropenia, granulocyte and macrophage progenitor cell deficiency, and impaired neutrophil mobilization. *Blood* 84:1737. Lim, K.G., H.C. Wan, P.T. Bozza, M.B. Resnick, D.T.W. Wong, W.W. Cruickshank, H. Kornfeld, D.M. Center, and P.F. Weller. 1996. Human eosinophils elaborate the lymphocyte chemoattractants. IL-16 (lymphocyte chemoattractant factor) and RANTES. *J. Immunol.* 156:2566.

Linenberger, M.L., F.W. Jacobsen, L.G. Bennett, V.C. Broudy, F.H. Martin, and J.L. Abkowitz. 1995. Stem cell factor production by human marrow stromal fibroblasts. *Exp. Hematol.* 23:1104.

Lioubin, M.N., L. Madisen, R.A. Roth, and A.F. Purchio. 1991. Characterization of latent transforming growth factor-beta 2 from monkey kidney cells. *Endocrinology* 128:2291.

Lipes, M.A., M. Napolitano, K-T. Jeang, N.T. Chang, and W.J. Leonard. 1988. Identification, cloning, and characterization of an immune activation gene. *Proc. Natl. Acad. Sci. USA* 85:9704.

Liu, R., W.A. Paxton, S. Choe, D. Ceradini, S.R. Martin, R. Horuk, M.E. MacDonald, H. Stuhlmann, R.A. Koup, and N.R. Landau. 1996. Homozygous defect in HIV-1 coreceptor accounts for resistance of some multiply-exposed individuals to HIV-1 infection. *Cell* 86:367.

Loetscher, M., B. Gerber, P. Loetscher, S.A. Jones, L. Piali, I. Clark-Lewis, M. Baggiolini and B. Moser. 1996a. Chemokine receptor specific for IP-10 and Mig: Structure, function, and expression in activated T-lymphocytes. *J. Exp. Med.* 184:963.

Loetscher, P., M. Seitz, M. Baggiolini, and B. Moser. 1996b. Interleukin-2 regulates CC chemokine receptor expression and chemotactic responsiveness in T lymphocytes. *J. Exp. Med.* 184:569.

Longley, B.J. Jr., G.S. Morganroth, L. Tyrrell, T.G. Ding, D.M. Anderson, D.E. Williams, and R. Halaban. 1993. Altered metabolism of mast cell growth factor (*c-kit* ligand) in cutaneous mastocytosis. *N. Engl. J. Med.* 328:1302.

Lopez, A., N.A. Nicola, A.W. Burgess, D. Metcalf, F.L. Battye, W.A. Sewell, and M. Vadas. 1983. Activation of granulocyte cytotoxic function by purified mouse colony-stimulating factors. *J. Immunol.* 131:2983.

Lopez, A.F., C.G. Begley, D.J. Williamson, D.J. Warren, M.A. Vadas, and C.J. Sanderson. 1986. Murine eosinophil differentiation factor. An eosinophil-specific colony-stimulating factor with activity for human cells. *J. Exp. Med.* 163:1085.

Lord, B.I., K.J. Mori, E.G. Wright. 1976. An inhibitor of stem cell proliferation in normal bone marrow. Br. J. Haematol. 34:441.

Lord, B.I., T.M. Dexter, J.M. Clements, M.G. Hunter, and A.J.H. Gearing. 1992. Macrophage-inflammatory protein protects multipotent hematopoietic cells from the cytotoxic effects of hydroxyurea in vivo. *Blood* 79:2605.

Lorimore, S.A., I.B. Pragnell, L. Eckmann, and E.G. Wright. 1990. Synergistic interactions allow colony formation *in vitro* by murine haemopoietic stem cells. *Leukemia Res.* 14:481.

Lotem, J., and L. Sachs. 1991. Regulation of cell-surface receptors for hematopoietic differentiation-inducing protein MGI-2 on normal and leukemic myeloid cells. *Int. J. Cancer.* 40:532.

Lotem, J., and L. Sachs. 1991. The network of hemopoietic regulatory proteins in myeloid cell differentiation. *Cell Growth Differ. 2:421*.

Lotem, J., and L. Sachs. 1992. Hematopoietic cytokines inhibit apoptosis induced by transforming growth factor $\beta 1$ and cancer chemotherapy compounds in myeloid leukemic cells. *Blood* 80:1750.

Lotem, J., Y. Shabo, and L. Sachs. 1989. Induction of dependence on hematopoietic proteins for viability and receptor upregulation in differentiating myeloid leukemia cells. *Blood* 74:579.

Lu, L., Z-H. Lin, R-N. Shen, D.J. Warren, T. Leemhuis, and H.E. Broxmeyer. 1990. Influence of interleukins 3, 5, and 6 on the growth of eosinophil progenitors in highly enriched human bone marrow in the absence of serum. *Exp. Hematol.* 18:1180.

Luster, A.D., J.C. Unkeless, and J.V. Ravech. 1985. γ -Interferon transcriptionally regulates an early-response gene containing homology to platelet proteins. *Nature* 315: 672.

Luster, A.D., and P. Leder. 1993. IP-10, a -C-X-C- chemokine, elicits a potent thymus-dependent antitumor response in vivo. J. Exp. Med. 178:1057.

Lyman, S.D., L. James, T. Vanden Bos, P. de Vries, K. Brasel, B. Gliniak, L.T. Hollingsworth, K. S. Picha, H.J. McKenna, R.R. Splett, F.A. Fletcher, E. Maraskovsky, T. Farrah, D. Foxworthe, D.E. Williams, and M.P. Beckmann. 1993. Molecular cloning of a ligand for the flt3/flk-2 tyrosine kinase receptor: A proliferative factor for primitive hematopoietic cells. *Cell* 75:1157.

Lyons, R.M., J. Keski-Oja, and H.L. Moses. 1988. Proteolytic activation of latent transforming growth factor-beta from fibroblast-conditioned medium. J. Cell. Biol. 106:1659.

Maciejewsky, J.P., J.M. Liu, S.W. Green, C.E. Walsh, M. Plumb, I.B. Pragnell, and N.S. Young. 1992. Expression of stem cell inhibitor (SCI) gene in patients with bone marrow failure. *Exp. Hematol.* 20:1112.

Mackay, C.R. 1997. Chemokines: What chemokine is that? Current Biol. 7:R384.

Maekawa, T., and D. Metcalf. 1989. Clonal suppression of HL-60 and U937 cells by recombinant leukemia inhibitory factor in combination with GM-CSF or G-CSF. *Leukemia 3:270*.

Magli, M.C., N.N. Iscove, and Odartchenko N. 1982. Transient nature of early haematopoietic spleen colonies. *Nature 295:527*.

Maltman, J., I.B. Pragnell, and G.J. Graham. 1993. Transforming growth factor β : Is it a down-regulator of stem cell inhibition by MIP-1 α ? J. Exp. Med. 176:925.

Maltman, J., I.B. Pragnell, and G.J. Graham. 1996. Specificity and reciprocity in the interactions between TGF- β and macrophage inflammatory protein-1 α . J. Immunol. 156:1566.

Mantel, C., Y.J. Kim, S. Cooper, B. Kwon, and H.E. Broxmeyer. 1993. Polymerization of murine macrophage inflammatory protein 1α inactivates its myelosuppressive effects *in vitro*: The active form is a monomer. *Proc. Natl. Acad. Sci. USA 90:2232.*

Marshall, E., L.B. Woolford, B.I. Lord. 1997. Continuous infusion of macrophage inflammatory protein MIP-1 α enhances leucocyte recovery and haemopoietic progenitor cell mobilization after cyclophosphamide. *Br. J. Cancer* 75:1715.

Massague, J. 1990. The transforming growth factor- β family. Annu. Rev. Cell. Biol. 6:597.

Massague, J., S. Cheifetz, M. Laiho, D.A. Ralph, F.M.B. Weis, and A. Zentella. 1992. Transforming growth factor- β . *Cancer Surveys* 12:81.

Matsui, Y., K.M. Zsebo, and B.L.M. Hogan. 1990. Embryonic expression of a haematopoietic growth factor encoded by the *Sl* locus and the ligand for c-kit. *Nature* 347:667.

Matsumoto, M., N. Tominaga, N. Harada, and K. Takatsu. 1987. Role of T cellreplacing factor (TRF) in the murine B cell differentiation: induction of increased levels of expression of secreted type IgM mRNA. J. Immunol. 138:1826.

Matsushima, K., M. Taguchi, E.J. Kovacs, H.A. Young, and J.J. Oppenheim. 1986. Intracellular localization of human monocyte associated interleukin 1 (IL 1) activity and release of biologically active IL 1 from monocytes by trypsin and plasmin. *J. Immunol.* 136: 2883.

McColl, S.R., M. Hachicha, S. Levasseur, K. Neote, and T.J. Schall. 1993. Uncoupling of early signal transduction events from effector function in human peripheral blood neutrophils in response to recombinant macrophage inflammatory proteins- 1α and -1β . *J. Immunol.* 150:4550.

McColl, S.R., C.J. Roberge, B. Larochelle, and J. Gosselin. 1997. EBV induces the production and release of IL-8 and macrophage inflammatory protein- 1α in human neutrophils. *J. Immunol.* 159:6168.

McNiece, I.K., A.B. Kriegler, A.J. Hapel, M-C. Fung, I.G. Young, T.R. Bradley, and G.S. Hodgson. 1984. Recombinant interleukin-3 exhibits synergistic factor activity. *Cell. Biol. Int. Rep.* 8:812.

McNiece, I., R. Andrews, M. Stewart, S. Clark, T. Boone, and P. Quesenberry. 1989. Action of interleukin-3, G-CSF, and GM-CSF on highly enriched human hematopoietic progenitor cells: Synergistic interaction of GM-CSF plus G-CSF. *Blood* 74:110.

McNiece, I.K., K.E. Langley, and K.M. Zsebo. 1991. Recombinant human stem cell factor synergises with GM-CSF, G-CSF, IL-3 and Epo to stimulate human progenitor cells of the myeloid and erythroid lineages. *Exp. Hematol.* 19:226.

Metcalf, D. 1989. Actions and interactions of G-CSF, LIF and IL-6 on normal and leukemic murine cells. *Leukemia 3:349*.

Metcalf, D. 1991. The Leukemia inhibitory factor (LIF). Intl. J. Cell Cloning 9:95.

Metcalf, D. 1993. Hematopoietic regulators: Redundancy or subtlety ? Blood 82: 3515.

Metcalf, D., and N.A. Nicola. 1983. Proliferative effects of purified granulocyte colony-stimulating factor (G-CSF) on normal mouse hemopoietic cells. J. Cell Physiol. 116:198.

Metcalf, D., and N.A. Nicola. 1991. Direct proliferative actions of stem cell factor on murine bone marrow cells *in vitro*: Effects of combination with colony-stimulating factors. *Proc. Natl. Acad. Sci. USA* 88:6239.

Metcalf, D., G.J.V. Nossal, N.L. Warner, J.F.A.P. Miller, T.E. Mandel, J.E. Layton, and G.A. Gutman. 1975. Growth of B lymphocyte colonies in vitro. J. Exp. Med. 142:1534.

Metcalf, D., C.G. Begley, G.R. Johnson, N.A. Nicola, A.F. Lopez, and D.J. Williamson. 1986. Effects of purified bacterially synthesized murine Multi-CSF (IL-3) on hematopoiesis in normal adult mice. *Blood* 68:46.

Metcalf, D., D.J. Hilton, and N.A. Nicola. 1988. Clonal analysis of the actions of the murine leukemia inhibitory factor on leukemic and normal murine hemopoietic cells. *Leukemia 2:216*.

Metcalf, D., N.A. Nicola, and D.P. Gearing. 1990. Effects of injected leukemia inhibitory factor on hematopoietic and other tissues in mice. *Blood.* 76:50.

Michael, N.L., G. Chang, L.G. Louie, J.R. Mascola, D. Dondero, D.H. Birx, and H.W. Sheppard. 1997a. The role of viral phenotype and *CCR5* gene defects in HIV-1 transmission and disease progression. *Nature Med.* 3:338.

Michael, N.L., L.G. Louie, A.L. Rohrbaugh, K.A. Schultz, D.E. Dayhoff, C.E. Wang, and H.W. Sheppard. 1997b. The role of *CCR5* and *CCR2* polymorphisms in HIV-1 transmission and disease progression. *Nature Med.* 3:1160.

Middleton, J., S. Neil, J. Wintle, I. Clark-Lewis, H. Moore, C. Lam, M. Auer, E. Hub, and A. Rot. 1997. Transcytosis and surface presentation of IL-8 by venular endothelial cells. *Cell* 91:385.

Miller, M.D., S. Hata, R. DeWaal Malefyt, and M.S. Krangel. 1989. A novel polypeptide secreted by activated human T lymphocytes. *J. Immunol.* 143:2907.

Migliaccio, G., A.R. Migliaccio, M.L. Druzin, P-J.V. Giardina, K.M. Zsebo, and J.W. Adamson. 1992. Long-term generation of colony-forming cells in liquid culture of $CD34^+$ cord blood cells in the presence of recombinant human stem cell factor. *Blood* 79:2620.

Minty, A., P. Chalon, J.C. Guillemot, M. Kaghad, P. Liauzun, M. Magazin, B. Miloux, C. Minty, P. Ramond, N. Vita, J. Lupker, D. Shire, P. Ferrara, and D. Caput. 1993. Molecular cloning of the MCP-3 chemokine gene and regulation of its expression. *Eur. Cytokine Netw.* 4:99.

Miyagishi, R., S. Kikuchi, T. Fukazawa, and K. Tashiro. 1995. Macrophage inflammatory protein- 1α in the cerebrospinal fluid of patients with multiple sclerosis and other inflammatory neurological diseases. *J. Neurol. Sci.* 129:223.

Miyagishi, R., S. Kikuchi, C. Takayama, Y. Inoue, and K. Tashiro. 1997. Identification of cell types producing RANTES, MIP-1 α and MIP-1 β in rat experimental autoimmune encephalomyelitis by in situ hybridization. J. Neuroimmunol. 77:17.

Miyajima, A., A. L-F. Mui, T. Ogorochi, and K. Sakamaki. 1993. Receptors for granulocyte-macrophage colony-stimulating factor, interleukin-3, and interleukin-5. *Blood 82:1960*.

Miyazono, K., U. Hellman, C. Wernstedt and C-H. Heldin. 1988. Latent high molecular weight complex of transforming growth factor β 1. Purification from human platelets and structural characterization. *J. Biol. Chem.* 263:6407.

Miyazono, K., A. Olofsson, P. Colosetti, and C-H. Heldin. 1991. A role of the latent TGF- β 1-binding protein in the assembly and secretion of TGF- β 1. *EMBO J.* 10.1091.

Mochizuki, D.Y., J.R. Eisenman, P.J. Conlon, A.D. Larsen, and R.J. Tushinski. 1987. Interleukin 1 regulates hematopoietic activity, a role previously ascribed to hemopoietin 1. *Proc. Natl. Acad. Sci. USA* 84:5267.

Mohamadzadeh, M., A.N. Poltorak, P.R. Bergstresser, B. Beutler, and A. Takashima. 1996. Dendritic cells produce macrophage inflammatory protein- 1γ , a new member of the CC chemokine family. *J. Immunol.* 156:3102.

Monteclaro, F.S., and I.F. Charo. 1996. The amino-terminal extracellular domain of the MCP-1 receptor but not the RANTES/MIP-1 α receptor, confers chemokine selectivity. J. Biol. Chem. 271:19084.

Monteclaro, F.S., and I.F. Charo. 1997. The amino-terminal domain of CCR2 is both necessary and sufficient for high affinity binding of monocyte chemoattractant protein 1. J. Biol. Chem. 272:23186.

Moore, M.A.S., and D.J. Warren. 1987. Synergy of interleukin 1 and granulocyte colony-stimulating factor: *In vivo* stimulation of stem-cell recovery and hematopoietic regeneration following 5-fluorouracil treatment of mice. *Proc. Natl. Acad. Sci. USA* 84:7134.

Moore, R.N., J.J. Oppenheim, J.J. Farrar, C.S. Carter Jr., A. Waheed, and R.K. Shadduck. 1980. Production of lymphocyte-activating factor (interleukin 1) by macrophages activated with colony-stimulating factors. *J. Immunol.* 125:1302.

Moore, P.S., C. Boshoff, R.A. Weiss, and Y. Chang. 1996. Molecular mimicry of human cytokine and cytokine response pathway genes by KSHV. *Science 274:1739*.

Moreau, J-F., D.D. Donaldson, F. Bennett, J. Witek-Giannotti, S.C. Clark, and G.G. Wong. 1988. Leukemia inhibitory factor is identical to the myeloid growth factor human interleukin for DA cells. *Nature 336:690*.

Mori, M., Y. Sadahira, S. Kawasaki, T. Hayashi, and M. Awai. 1990. Macrophage heterogeneity in bone marrow culture *in vitro*. J. Cell. Sci. 95:481.

Moriuchi, H., M. Moriuchi, C. Combadiere, P.M. Murphy, and A.S. Fauci. 1996. $CD8^+$ T-cell-derived soluble factors(s), but not β -chemokines RANTES, MIP-1 α , and MIP-1 β , suppress HIV-1 replication in monocyte/macrophages. *Proc. Natl. Acad. Sci. USA* 93:15341.

Moriuchi, H., M. Moriuchi, and A.S. Fauci. 1997. Nuclear factor- κ B potently upregulates the promoter activity of RANTES, a chemokine that blocks HIV infection. *158:3483*.

Morris, C.F., J. Salisbury, M. Kobayashi, P.V. Townsend, and A.J. Hapel. 1990. Interleukin 3 alone does not support the proliferation of bone marrow cells from A/J mice: a novel system for studying the synergistic activities of IL-3. *Br. J. Haematol.* 74:131.

Morrison, S.J., N.M. Shah, and D.J. Anderson. 1997. Regulatory mechanisms in stem cell biology. *Cell* 88:287.

Morrissey, P.J., R.G. Goodwin, R.P. Nordan, D. Anderson, K.H. Grabstein, D. Cosman, J. Sims, S. Lupton, B. Acres, S.G. Reed, D. Mochizuki, J. Eisenman, P.J. Conlon, and A.E. Namen. 1989. Recombinant interleukin 7, pre-B cell growth factor, has costimulatory activity on purified mature T cells. *J. Exp. Med.* 169:707.

Moses, H.L., R.F. Tucker, E.B. Leof, R.J. Coffey, J. Halper, and G.D. Shipley. 1985. Type β transforming growth factor is a growth stimulator and growth inhibitor. In *Cancer Cells*. J. Feramisco, B. Ozanne, and C. Stiles. eds. Cold Spring Harbor Press, New York, p.65.

Mosmann, T.R., M.W. Bond, R.L. Coffman, J. Ohara, and W.E. Paul. 1986. T-cell and mast cell lines respond to B-cell stimulatory factor 1. *Proc. Natl. Acad. Sci. USA* 83:5654.

Motoyoshi, K., K. Yoshida, K. Hatake, M. Saito, Y. Miura, N. Yanai, M. Yamada, T. Kawashima, G.G. Wong, P.A. Temple, A.C. Leary, J.S. Witek-Giannotti, M. Fujisawa, A. Yuo, T. Okabe, and F. Takaku. 1989. Recombinant and native human urinary colony-stimulating factor directly augments granulocytic and granulocyte-macrophage colony-stimulating factor production of human peripheral blood monocytes. *Exp. Hematol.* 17:68.

Mummidi, S., S.S. Ahuja, B.L. McDaniel, and S.K. Ahuja. 1997. The human CC chemokine receptor 5 (CCR5) gene. J. Biol. Chem. 272:30662.

Munson, P.J., and D. Robard. 1980. LIGAND: A versatile computerized approached for the characterization of ligand-binding systems. *Annal. Biochem.* 107:220.

Murphy, P.M., and H.L. Tiffany. 1991. Cloning of complementary DNA encoding a functional human interleukin-8 receptor. *Science* 253:1280.

Musashi, M., Y-C. Yang, S.R. Paul, S.C. Clark, T. Sudo, and M.Ogawa. 1991. Direct and synergistic effects of interleukin 11 on murine hematopoiesis in culture. *Proc. Natl. Acad. Sci. USA* 88:765.

Musso, T., I. Espinoza-Delgado, K. Pulkki, G.L. Gusella, D.L. Longo, and L. Varesio. 1990. Transforming growth factor- β down-regulates interleukin 1 (IL-1)-induced IL-6 production by monocytes. *Blood* 76:2466.

Myers, S.J., L.M. Wong, and I.F. Charo. 1995. Signal transduction and ligand specificity of the human monocyte chemoattractant protein-1 receptor in transfected embryonic kidney cells. *J. Biol. Chem.* 270:5786.

Nagasawa, T., H. Kikutani, and T. Kishimoto. 1994. Molecular cloning and structure of a pre-B-cell growth-stimulating factor. *Proc. Natl. Acad. Sci. USA 91:2305*.

Nagasawa, T., S. Hirota, K. Tachibana, N. Takakura, S. Nishikawa, Y. Kitamura, N. Yoshida, H. Kikutani, and T. Kishimoto. 1996a. Defects of B-cell lymphopoiesis and bone marrow myelopoiesis in mice lacking the CXC chemokine PBSF/SDF-1. *Nature* 382:635.

Nagasawa, T., T. Nakajima, K. Tachibana, H. Iizasa, C.C. Bleul, O. Yoshie, K. Matsushima, N. Yoshida, T.A. Springer, and T. Kishimoto. 1996b. Molecular cloning and characterization of a murine pre-B-cell growth-stimulating factor/stromal cell-derived factor 1 receptor, a murine homolog of the human immunodeficiency virus 1 entry coreceptor fusin. *Proc. Natl. Acad. Sci. USA* 93:14726.

Nakabayashi, T., J.J. Letterio, A.G. Geiser, L. Kong, N. Ogawa, W. Zhao, T. Koike, G. Fernandes, H. Dang, and N. Talal. 1997. Up-regulation of cytokine mRNA, adhesion molecule proteins, and MHC class II proteins in salivary glands of TGF- β 1 knockout mice. *J. Immunol.* 158:5527.

Nakao, M., H. Nomiyama, and K. Shimada. 1990. Structures of human genes coding for cytokine LD78 and their expression. *Mol. Cell. Biol.* 10:3646.

Namen, A.E., S. Lupton, K. Hjerrild, J. Wignall, D.Y. Mochizuki, A. Schmierer, B. Mosley, C.J. March, D. Urdal, S. Gillis, D. Cosman, and R.G. Goodwin. 1988. Stimulation of B-cell progenitors by cloned murine interleukin-7. *Nature* 333:571.

Naparstek, E., T. Donnelly, K. Kase, and J.S. Greenberger. 1985. Biologic effects of in vitro X-irradiation of murine long-term bone marrow cultures on the production of granulocyte-macrophage colony-stimulating factors. *Exp. Hematol.* 13:701.

Neote, K., D. DiGregorio, J.Y. Mak, R. Horuk, and T.J. Schall. 1993a. Molecular cloning, functional expression and signalling characteristics of a C-C chemokine receptor. *Cell* 72:415.

Neote, K., W. Darbonne, J. Ogez, R. Horuk, and T.J. Schall. 1993b. Identification of a promiscuous inflammatory peptide receptor on the surface of red blood cells. *J. Biol. Chem.* 268:12247.

Nibbs, R.J.B., S.M. Wylie, I.B. Pragnell, and G.J. Graham. 1997. Cloning and characterization of a novel murine β chemokine receptor, D6. J. Biol. Chem. 272: 12495.

Nibbs, R.J.B., S.M. Wylie, J. Yang, N.R. Landau, and G.J. Graham. 1998. Cloning and characterization of a novel promiscuous human β -chemokine receptor D6. *J. Biol. Chem.* 272:32078.

Nichols, J., I. Chambers, and A. Smith. 1994. Derivation of germline competent embryonic stem cells with a combination of interleukin-6 and soluble interleukin-6 receptor. *Exp. Cell. Res.* 215:237.

Nicola, N.A. 1990. Granulocyte colony-stimulating factor. In *Colony-stimulating factors*. M. Dexter, J. Garland and N. Testa, eds. Dekker, New York, NY, p.77.

Nicola, N.A., D. Metcalf, M. Matsumoto, and G.R. Johnson. 1983. Purification of a factor inducing differentiation in murine myelomonocytic leukemia cells. Identification as granulocyte colony-stimulating factor. J. Biol. Chem. 258:9017.

Nicola, N.A., L. Robb, D. Metcalf, D. Cary, C.C. Drinkwater, and C.G. Begley. 1996. Functional inactivation in mice of the gene for the interleukin-3 (IL-3)-specific receptor β -chain: Implications for IL-3 function and the mechanism of receptor transmodulation in hematopoietic cells. *Blood* 87:2665.

Nishinakamura, R., N. Nakayama, Y. Hirabayashi, T. Inoue, D. Aud, T. McNeil, S. Azuma, S. Yoshida, Y. Toyoda, K. Arai, A. Miyajima, and R. Murray. 1995. Mice deficient for the IL-3/GM-CSF/IL-5 β c receptor exhibit lung pathology and impaired immune response, while β_{IL-3} receptor-deficient mice are normal. *Immunity 2:211*.

Nishinakamura, R., A. Miyajima, P.J. Mee, V.L.J. Tybulewicz, and R. Murray. 1996a. Hematopoiesis in mice lacking the entire granulocyte-macrophage colony-stimulating factor/interleukin-3/interleukin-5 functions. *Blood* 88:2458.

Nishinakamura, R., R. Wiler, U. Dirksen, Y. Morikawa, K. Arai, A. Miyajima, S. Burdach, and R. Murray. 1996b. The pulmonary alveolar proteinosis in granulocyte macrophage colony-stimulating factor / interleukins $3/5 \beta c$ receptor-deficient mice is reversed by bone marrow transplantation. *J. Exp. Med.* 183:2657.

Obaru, K., M. Fukuda, S. Maeda, and K. Shimada. 1986. A cDNA clone used to study mRNA inducible in human tonsillar lymphocytes by a tumour promoter. *J. Biochem.* 99:885.

O'Connor-McCourt, M.D., L.M. Wakefield. 1987. Latent transforming growth factor- β in serum. A specific complex with α_2 -macroglobulin. J. Biol. Chem. 262:14090.

Ogawa, M. 1993. Differentiation and proliferation of hematopoietic stem cells. *Blood* 81:2844.

Ohta, M., J.S. Greenberger, P. Anklesaria, A. Bassols, and J. Massague. 1987. Two forms of transforming growth factor- β distinguished by multipotential haematopoietic progenitor cells. *Nature 329:539*.

Olofsson, A., K. Miyazono, T. Kanzaki, P. Colosetti, U. Engstrom, and C-H. Heldin. 1992. Transforming growth factor- β 1, - β 2, and - β 3 secreted by a human glioblastoma cell line. Identification of small and different forms of large latent complexes. *J. Biol. Chem.* 267:19482.

Onoda, M., M. Shinoda, K. Tsuneoka, and M. Shikita. 1980. X-ray-induced production of granulocyte-macrophage colony-stimulating factor (GM-CSF) by mouse spleen cells in culture. *J. Cell. Physiol.* 104:11.

Orlofsky, A., M.S. Berger, and M.B. Prystowsky. 1991. Novel expression pattern of a new member of the MIP-1 family of cytokine-like genes. *Cell Regulation 2:403*.

Orlofsky, A., E.Y. Lin, and M.B. Prystowsky. 1994. Selective induction of the β chemokine C10 by IL-4 in mouse macrophages. *J. Immunol.* 152:5084.

Orr-Urtreger, A., A. Avivi, Y. Zimmer, D. Givol, Y. Yarden, and P. Lonai. 1990. Developmental expression of *c-kit*, a proto-oncogene encoded by the *W* locus. *Development 109:911*.

Oppenheim, J.J., R. Neta, P. Tiberghien, R. Gress, J.J. Kenny, and D.L. Longo. 1989. Interleukin-1 enhances survival of lethally irradiated mice treated with allogeneic bone marrow cells. *Blood* 74:2257.

Oppenheim, J.J., C.O.C. Zachariae, N. Mukaida, and K. Matsushima. 1991. Properties of the novel proinflammatory supergene "intercrine" cytokine family. *Ann. Rev. Immunol.* 9:617.

Oster, W., A. Lindemann, R. Mertelsmann, and F. Herrmann. 1989. Granulocytemacrophage colony-stimulating factor (CSF) and multilineage CSF recruit human monocytes to express granulocyte CSF. *Blood* 73:64.

Oster, W., M.A. Brach, H-J. Gruss, R. Mertelsmann, and F. Herrmann. 1992. Interleukin-1 β (IL-1 β) expression in human blood mononuclear phagocytes is differentially regulated by granulocyte-macrophage colony-stimulating factor (GM-CSF), M-CSF, and IL-3. *Blood* 79:1260.

Owen-Lynch, P.J., J.A. Adams, M.L. Brereton, L.G. Czaplewski, A.D. Whetton, and J.A. Liu Yin. 1996. The effect of the chemokine rhMIP-1 α , and a non-aggregating variant BB-10010, on blast cells from patients with acute myeloid leukaemia. *Br. J. Haematol.* 95:77.

Pal, R., A. Garzino-Demo, P.D. Markham, J. Burns, M. Brown, R.C. Gallo, A.L. DeVico. 1997. Inhibition of HIV-1 infection by the β -chemokine MDC. *Science* 278:695.

Pan, Y., C. Lloyd, H. Zhou, S. Dolich, J. Deeds, J-A. Gonzalo, J. Vath, M. Gosselin, J. Ma, B. Dussault, E. Woolf, G. Alperin, J. Culpepper, J.C. Gutierrez-Ramos and D. Gearing. 1997 Neurotactin, a membrane-anchored chemokine upregulated in brain inflammation. *Nature* 387:611.

Paran, M., L. Sachs, Y. Barak, and P. Resnitzky. 1970. *In vitro* induction of granulocyte differentiation in hematopoietic cells from leukemic and non-leukemic patients. *Proc. Natl. Acad. Sci. USA* 67:1542.

Park, L., S. Gillis, and D.L. Urdal. 1990a. Hematopoietic growth-factor receptors. In *Colony-stimulating factors*. M. Dexter, J. Garland and N. Testa, eds. Dekker, New York, NY, p.39.

Park, L.S., D.J. Friend, A.E. Schmierer, S.K. Dowler, and A.E. Namen. 1990b. Murine interleukin 7 (IL-7) receptor. Characterization on an IL-7-dependent cell line. *J. Exp. Med.* 171:1073. Parkinson, E.K., G.J. Graham, P. Daubersies, J.E. Burns, C. Heufler, M. Plumb, G. Schuler, and I.B. Pragnell. 1993. Hemopoietic stem cell inhibitor (SCI/MIP-1 α) also inhibits clonogenic epidermal keratinocyte proliferation. J. Invest. Dermatol. 101:113.

Patel, V.P., B.L. Kreider, Y. Li, H. Li, K. Leung, T. Salcedo, B. Nardelli, V. Pippalla, S. Gentz, R. Thotakura, D. Parmelee, R. Gentz, and G. Garotta. 1997. Molecular and functional characterization of two novel human C-C chemokines as inhibitors of two distinct classes of myeloid progenitors. *J. Exp. Med.* 185:1163.

Paul, W.E. 1991. Interleukin-4: A prototypic immunoregulatory lymphokine. *Blood* 77:1859.

Paul, P.R., F. Bennett, J.A. Calvetti, K. Kelleher, C.R. Wood, R.M. O'Hara Jr., A.C. Leary, B. Sibley, S.C. Clark, D.A. Williams, and Y-C. Yang. 1990. Molecular cloning of a cDNA encoding interleukin 11, a stromal cell derived lymphopoietic and hematopoietic cytokine. *Proc. Natl. Acad. Sci. USA* 87:7512.

Paxton, W.A., S.R. Martin, D. Tse, T.R. O'Brien, J. Skurnick, N.L. VanDevanter, N. Padian, J.F. Braun, D.P. Kotler, S.M. Wolinsky, and R.A. Koup. 1996. Relative resistance to HIV-1 infection of CD4 lymphocytes from persons who remain uninfected despite multiple high-risk sexual exposures. *Nature Med. 2:412*.

Peschel, C., W.E. Paul, J. Ohara, and I. Green. 1987. Effects of B cell stimulatory factor-1 / interleukin 4 on hematopoietic progenitor cells. *Blood* 70:254.

Peschel, C., I. Green and W.E. Paul. 1989. Interleukin-4 induces a substance in bone marrow stromal cells that reversibly inhibits factor-dependent and factor-independent cell proliferation. *Blood* 73:1130.

Petzer, A.L., D.E. Hogge, P.M. Landsdorp, D.S. Reid, and C.J. Eaves. 1996, Selfrenewal of primitive human hematopoietic cells (long-term culture-initiating cells) in vitro and their expansion in defined medium. *Proc. Natl. Acad. Sci. USA* 93:1470.

Piersma, A.H., K.G.M. Brockbank, and R.E. Ploemacher. 1984. Regulation of in vitro myelopoiesis by a hemopoietic stromal fibroblastic cell line. *Exp. Hematol.* 12:617.

Pike, B.L., and W. A. Robinson. 1970. Human bone marrow growth in agar gel. J. Cell Physiol. 76:77.

Pircher, R., J. Jullien, D.A. Lawrence. 1986. Beta-transforming growth factor is stored in human blood platelets as a latent high molecular weight complex. *Biochem. Biophys. Res. Commun.* 136:30.

Pistoia, V., R. Ghio, A. Nocera, A. Leprini, A. Perata, and M. Ferrarini. 1987. Production of colony-stimulating activity by normal and neoplastic B lymphocytes. *Blood 69:1340*.

Pistoia, V., S. Zupo, A. Corcione, S. Roncella, T. Matera, R. Ghio, and M. Ferrarini. 1989. Production of colony-stimulating activity by human natural killer cells: Analysis of the conditions that influence the release and detection of colony-stimulating activity. *Blood* 74:156.

Ploemacher, R.E., and R.H.C. Brons. 1988. Cells with marrow and spleen repopulating ability and forming spleen colonies on day 16, 12, and 8 are sequentially ordered on the basis of increasing Rhodamine 123 retention. J. Cell. Physiol. 136:531.

Ploemacher, R.E., and R.H.C. Brons. 1989. Separation of CFU-S from primitive cells responsible for reconstitution of the bone marrow hemopoietic stem cell compartment following irradiation: evidence for a pre-CFU-S cell. *Exp. Hematol.* 17: 263.

Ploemacher, R.E., P.L. van Soest, and A. Boudewijn. 1993. Autocrine transforming growth factor β 1 blocks colony formation and progenitor cell generation by hemopoietic stem cells stimulated with steel factor. *Stem Cells* 11:336.

Poli, G., A.L. Kinter, J.S. Justement, P. Bressler, J.H. Kehrl, A.S. Fauci. 1991. Transforming growth factor β suppresses human immunodeficiency virus expression and replication in infected cells of the monocyte/macrophage lineage. *J. Exp. Med.* 173:589.

Pollard, J.W., A. Bartocci, R. Arceci, A. Orlofsky, M.B. Ladner, and E.R. Stanley. 1987. Apparent role of the macrophage growth factor, CSF-1, in placental development. *Nature 330:484*.

Pollard, J.W., J.S. Hunt, W. Wiktor-Jedrzejczak, and E.R. Stanley. 1991. A pregnancy defect in the osteopetrotic (*op/op*) mouse demonstrates the requirement for CSF-1 in female fertility. *Dev. Biol.* 148:273.

Post, T.W., C.R. Bozic, M.E. Rothenberg, A.D. Luster, N. Gerard, and C. Gerard. 1995. Molecular characterization of two murine eosinophil β chemokine receptors. *J. Immunol.* 155:5299.

Postlethwaite, A.E., J. Keki-Oja, H.L. Moses, and A.H. Kang. 1987. Stimulation of the chemotactic migration of human fibroblasts by transforming growth factor β . *J. Exp. Med.* 165:251.

Potten, C.S. 1995. Interleukin-11 protects the clonogenic stem cells in murine smallintestinal crypts from impairment of their reproductive capacity by radiation. *Int. J. Cancer.* 62:356.

Potten, C.S. 1996. Protection of the small intestinal clonogenic stem cells from radiation-induced damage by pretreatment with interleukin 11 also increases murine survival time. *Stem Cells* 14:452.

Power, C.A., A. Meyer, K. Nemeth, K.B. Bacon, A.J. Hoogewerf, A.E.I. Proudfoot, and T.N.C. Wells. 1995. Molecular cloning and functional expression of a novel CC chemokine receptor cDNA from a human basophilic cell line. *J. Biol. Chem.* 270:19495.

Pragnell, I.B., E.G. Wright, S.A. Lorimore, J. Adam, M. Rosendaal, J.F. DeLamarter, M. Freshney, L. Eckmann, A. Sproul, and N. Wilkie. 1988. The effect of stem cell proliferation regulators demonstrated with an in vitro assay. *Blood* 72: 196.

Prieschel, E.E., G.G. Prendl, N.E. Harrer, and T. Baumruker. 1995. p21ras links $Fc_{\epsilon}RI$ to NF-AT family members in mast cells. The AP3-like factor in this cell type is an NF-AT family member. *J. Immunol.* 155:4963.

Proffitt, J., G. Crabtree, M. Grove, P. Daubersies, B. Bailleul, E. Wright, and M. Plumb. 1995. An ATF/CREB binding site is essential for cell-specific and inducible transcription of the murine $MIP-1\beta$ cytokine gene. Gene 152:173.

Proost, P., C. de Wolf-Peeters, R. Connings, G. Opdenakker, A. Billiau, and J. Van Damme. 1993. Identification of a novel granulocyte chemotactic protein (GCP-2) from human tumor cells: In vitro and in vivo comparison with natural forms of GRO, IP-10, and IL-8. *J. Immunol.* 150:1000.

Proost, P., A. Wuyts, and J. Van Damme. 1996. Human monocyte chemotactic proteins-2 and -3: structural and functional comparison with MCP-1. *J. Leukoc. Biol.* 59:67.

Quesniaux, V.F.J., S.C. Clark, K. Turner, and B. Fagg. 1992. Interleukin-11 stimulates multiple phases of erythropoiesis in vitro. *Blood* 80:1218.

Quesniaux, V.F.J., G.J. Graham, I. Pragnell, D. Donaldson, S.D. Wolpe, N.N. Iscove, and B. Fagg. 1993. Use of 5-fluorouracil to analyze the effect of macrophage inflammatory protein- 1α on long-term reconstituting stem cells in vivo. *Blood* 81:1497.

Racke, M.K., S. Dhib-Jalbut, B. Cannella, P.S. Albert, C.S. Raine, and D.E. McFarlin. 1991. Prevention and treatment of chronic relapsing experimental allergic encephalomyelitis by transforming growth factor- β 1. *J. Immunol.* 146:3012.

Ralph, P., and I. Nakoinz. 1987. Stimulation of macrophage tumoricidal activity by the growth and differentiation factor CSF-1. *Cell Immunol.* 105:270.

Ransohoff, R.M., A. Glabinski, and M. Tani. 1996. Chemokines in immune-mediated inflammation of the central nervous system. *Cytokine and Growth Factor Reviews* 7:35.

Raport, C.J., J. Gosling, V.L. Schweickart, P.W. Gray, and I.F. Charo. 1996. Molecular cloning and functional characterization of a novel human CC chemokine receptor (CCR5) for RANTES, MIP-1 β , and MIP-1 α . J. Biol. Chem. 271:17161. Ratacjczak, M.Z., W.I. Kuczynski, D.L. Sokol, J.S. Moore, C.H. Pletcher Jr., and A.M. Gewirtz. 1995. Expression and physiologic significance of *Kit* ligand and stem cell tyrosine kinase-1 receptor ligand in normal human CD34⁺, *c-kit*⁺ marrow cells. *Blood 86:2161*.

Ray, P., L. Yang, D-H. Zhang, S.K. Ghosh, and A. Ray. 1997. Selective upregulation of cytokine-induced RANTES gene expression in lung epithelial cells by overexpression of IkBR. *J. Biol. Chem.* 272:20191.

Reibman, J., S. Meixler, T.C. Lee, L.I. Gold, B.N. Cronstein, K.A. Haines, S.L. Kolasinski, and G. Weissmann. 1991. Transforming growth factor- β 1, a potent chemoattractant for human neutrophils, bypasses classic signal-transduction pathways. *Proc. Natl. Acad. Sci. USA* 88:6805.

Rennick, D.M, F.D. Lee, T. Yokota, K-I. Irai, H. Cantor, and G.J. Nabel. 1985. A cloned MCGF cDNA encodes a multilineage hematopoietic growth factor: multiple activities of interleukin 3. *J. Immunol.* 134:910.

Rennick, D., G. Young, C. Muller-Sieburg, C. Smith, N. Arai, Y. Takabe, and L. Gemmell. 1987. Interleukin 4 (B-cell stimulatory factor 1) can enhance or antagonize the factor-dependent growth of hemopoietic progenitor cells. *Proc. Natl. Acad. Sci. USA 84:6889*.

Rennick, D., J. Jackson, G. Yang, J. Wideman, F. Lee, and S. Hudak. 1989. Interleukin-6 interacts with interleukin-4 and other hematopoietic growth factors to selectively enhance the growth of megakaryocytic, erythroid, myeloid, and multipotential progenitor cells. *Blood* 73:1828.

Rettenmier, C.W., M.F. Roussel, R.A. Ashmun, P. Ralph, K. Price, and C.J. Sherr. 1987. Synthesis of membrane-bound colony-stimulating factor 1 (CSF-1) and down modulation of CSF-1 receptors in NIH 3T3 cells transformed by cotransfection of the human CSF-1 and *c-fms* (CSF-1 receptor) genes. *Mol. Cell. Biol.* 7:2378.

Rettenmier, C.W., and M.F. Roussel. 1988. Differential processing of colonystimulating factor 1 precursors encoded by two human cDNAs. *Mol. Cell. Biol.* 8:5026.

Riedl, E., H. Strobl, O. Majdic, and W. Knapp. 1997. TGF β 1 promotes in vitro generation of dendritic cells by protecting progenitor cells from apoptosis. *J. Immunol.* 158:1591.

Ritter, L.M., M. Bryans, O. Abdo, V. Sharma, and N.M. Wilkie. 1995. *MIP-1* α nuclear protein (MNP), a novel transcription factor expressed in hematopoietic cells that is crucial for transcription of the human *MIP-1* α gene. *Mol. Cell. Biol.* 15:3110.

Rodewald, H-R., M. Dessing, A.M. Dvorak, and S.J. Galli. 1996. Identification of a committed precursor for the mast cell lineage. *Science*. 271:818.

Roberts, A.B., M.A. Anzano, L.M. Wakefield, N.S. Roche, D.F. Stern, and M.B. Sporn. 1985. Type β transforming growth factor: A bifunctional regulator of cellular growth. *Proc. Natl. Acad. Sci. USA* 82:119.

Roberts, R., J. Gallagher, E. Spooncer, T. D. Allen, F. Bloomfield, and T.M. Dexter. 1988. Heparan sulphate bound growth factors: a mechanism for stromal cell mediated haemopoiesis. *Nature 332:376*.

Rollins, B.J., E.D. Morrison, and C.D. Stiles. 1988. Cloning and expression of *JE*, a gene inducible by platelet-derived growth factor and whose product has cytokine-like properties. *Proc. Natl. Acad. Sci. USA* 85:3738.

Rollins, B.J., P. Stier, T.E. Ernst, and G.G. Wong. 1989. The human homologue of the *JE* gene encodes a monocyte secretory protein. *Mol. Cell. Biol.* 9:4687.

Rollins, B.J. 1997. Chemokines. Blood 90:909.

Romano, M., M. Sironi, C. Toniatti, N. Polentarutti, P. Fruscella, P. Ghezzi, R. Faggioni, W. Luini, V. van Hinsbergh, S. Sozzani, F. Bussolino, V. Poli, G. Ciliberto, and A. Mantovani. 1997. Role of IL-6 and its soluble receptor in induction of chemokines and leukocyte recruitment. *Immunity* 6:315.

Rosenfeld, C.S, C. Evans, and R.K Shadduck. 1990. Human macrophage colonystimulating factor induces macrophage colonies after L-phenylalanine methylester treatment of human marrow. *Blood* 76:1783.

Rot, A., M. Kreiger, T. Brunner, S.C. Bischoff, T.J. Schall, and C. A. Dahinden. 1992. RANTES and macrophage inflammatory protein 1α induce the migration and activation of normal human eosinophil granulocytes. 1992. *J. Exp. Med.* 176:1489.

Roth, S.J., M.W. Carr, and T.A. Springer. 1995. C-C chemokines, but not the C-X-C chemokines interleukin-8 and interferon- γ inducible protein-10, stimulate transendothelial chemotaxis of T lymphocytes. *Eur. J. Immunol.* 25:3482.

Rumsaeng, V., H. Vliagoftis, C.K. Oh, and D.D. Metcalf. 1997. Lymphotactin gene expression in mast cells following $Fc \varepsilon$ receptor I aggregation. J. Immunol. 158:1353.

Rutledge, B.J., H. Rayburn, R. Rosenberg, R.J. North, R.P. Gladue, C.L. Corless, and B.J. Rollins. 1995. High level monocyte chemoattractant protein-1 expression transgenic mice increases their susceptibility to intracellular pathogens. *J. Immunol.* 155: 4838.

Salazar-Mather, T.P., J.S. Orange, and C.A. Biron. 1998. Early murine cytomegalovirus (MCMV) infection induces liver natural killer (NK) cell inflammation and protection through macrophage inflammatory protein 1α (MIP- 1α)-dependent pathways. *J. Exp. Med.* 187:1.

Sallusto, F., C.R. Mackay, and A. Lanzavechhia. 1997. Selective expression of the eotaxin receptor CCR3 by human T helper 2 cells. *Science* 277:2005.

Samson, M., O. Labbe, C. Mollereau, G. Vassart, and M. Parmentier. 1996a. Molecular cloning and functional expression of a new human CC-chemokine receptor gene. *Biochem.* 35:3362.

Samson, M., F. Libert, B.J. Doranz, J. Rucker, C. Liesnard, C-M. Farber, S. Saragosti, C. Lapoumeroulie, J. Cognaux, C. Forceille, G. Muyldermans, C. Verhofstede, G. Burtonboy, M. Georges, T. Imai, S. Rana, Y. Yi, R.J. Smyth, R.G. Collman, R.W. Doms, G. Vassart, and M. Parmentier. 1996b. Resistance to HIV-1 infection in Caucasian individuals bearing mutant alleles of the CCR-5 chemokine receptor gene. *Nature* 382:722.

Samson, M., G. LaRosa, F. Libert, P. Paindavoine, M. Detheux, G. Vassart, and M. Parmentier. 1997. The second extracellular loop of CCR5 is the major determinant of ligand specificity. *J. Biol. Chem.* 272:24934.

Sanderson, C.J., D.J. Warren, and M. Strath. 1985. Identification of a lymphokine that stimulates eosinophil differentiation in vitro. Its relationship to interleukin 3, and functional properties of eosinophils produced in cultures. *J. Exp. Med.* 162:60.

Sanderson, C.J. 1992. Interleukin-5, eosinophils, and disease. Blood 79:3101.

Sarafi, M.N., E.A. Garcia-Zepeda, J.A. MacLean, I. F. Charo, and A.D. Luster. 1997. Murine monocyte chemoattractant protein (MCP)-5: A novel CC chemokine that is a structural and functional homologue of human MCP-1. J. Exp. Med. 185:99.

Sarris, A.H., H.E. Broxmeyer, U. Wirthmueller, N. Karasavvas, S. Cooper, L. Lu, J. Drueger, and J.C. Ravetch. 1993. Human interferon-inducible protein 10: Expression and purification of recombinant protein demonstrate inhibition of early human hematopoietic progenitors. *J. Exp. Med.* 178:1127.

Sarris, A.H., M. Talpaz, A.B. Deisseroth, and Z. Estrov. 1996. Human recombinant interferon-inducible protein-10 inhibits the proliferation of normal and acute myelogenous leukemia. *Leukemia* 10:757.

Schaafsma, M.R., W.E. Fibbe, J. Van Damme, N. Duinkerken, P. Ralph, K. Kaushansky, B.W. Altrock, R. Willemze, and J.H.F. Falkenburg. 1989. Interleukin-6 is not involved in the interleukin-1-induced production of colony-stimulating factors by human bone marrow stromal cells and fibroblasts. *Blood* 74:2619.

Schall, T.J. 1991. Biology of the RANTES/SIS cytokine family. Cytokine 3:165.

Schall, T.J., J. Jongstra, B.J. Dryer, J. Jorgensen, C. Clayberger, M.M. Davis, and A.M. Krensky. 1988. A human T-cell specific molecule is a member of a new gene family. *J. Immunol.* 141:1018.

Schall, T.J., K. Bacon, K.J. Toy, and D.V. Goeddel. 1990. Selective attraction of monocytes and T lymphocytes of the memory phenotype by cytokine RANTES. *Nature* 347:669.

Schall, T.J., R.E. O'Hehir, D.V. Goeddel, and J.R. Lamb. 1992. Uncoupling of cytokine mRNA expression and protein secretion during the induction phase of T cell anergy. *J. Immunol.* 148:381.

Schall, T.J., K. Bacon, R.D.R. Camp, J.W. Kaspari, and D.V. Goeddel. 1993. Human macrophage inflammatory protein α (MIP-1 α) and MIP-1 β chemokines attract distinct populations of lymphocytes. *J. Exp. Med.* 177:1821.

Schroeder, J-M., U. Mrowietz, E. Morita, and E. Christophers. 1987. Purification and partial biochemical characterization of a human monocyte-derived, neutrophilactivating peptide that lacks interleukin 1 activity. J. Immunol. 139:3474.

Schwartz, G.N., F. Liao, R.E. Gress, and J.M. Farber. 1997. Suppressive effects of recombinant human monokine induced by IFN- γ (rHuMig) chemokine on the number of committed and primitive hemopoietic progenitors in liquid cultures of CD34⁺ human bone marrow cells. *J. Immunol.* 159:895.

Shabo, Y., J. Lotem, M. Rubinstein, M. Revel, S.C. Clark, S.F. Wolf, R. Kamen, and L. Sachs. 1988. The myeloid blood cell differentiation-inducing protein MGI-2A is interleukin-6. *Blood* 72:2070.

Shah, N.M., M.A. Marchionni, I. Isaacs, P. Stroorbant, and D.J. Anderson. 1994. Glial growth factor restricts mammalian neural crest stem cells to a glial fate. *Cell* 77:349.

Shah, N.M., A.K. Groves, and D.J. Anderson. 1996. Alternative neural crest cell fates are instructively promoted by TGF β superfamily members. *Cell* 85:331.

Sherry, B., P. Tekamp-Olson, C. Gallegos, D. Bauer, G. Davatelis, S.D. Wolpe, K. Masiarz, D. Coit, and A. Cerami. 1988. Resolution of the two components of macrophage inflammatory protein 1, and cloning and characterization of one of those components, macrophage inflammatory protein 1β. J. Exp. Med. 168:2251.

Shibuya, A, K. Nagayoshi, K. Nakamura, and H. Nakauchi. 1995. Lymphokine requirement for the generation of natural killer cells from CD34⁺ hematopoietic progenitor cells. *Blood* 85:3538.

Shieh, J-H., R.H.F. Peterson, and M.S. Moore. 1991. Modulation of granulocyte colony-stimulating factor receptors on murine peritoneal exudate macrophages by tumour necrosis factor- α . J. Immunol. 146:2648.

Shipley, G.D., M.R. Pittelkow, J.J. Wille Jr., R.E. Scott, and H.L. Moses. 1986. Reversible inhibition of normal human prokeratinocyte proliferation by type β transforming growth factor-growth inhibitor in serum-free medium. *Cancer Res.* 46:2068.

Shirafuji, N., S. Matsuda, H. Ogura, K. Tani, H. Kodo, K. Ozawa, S. Nagata, S. Asano, and F. Takaku. 1990 Granulocyte colony-stimulating factor stimulates human mature neutrophilic granulocytes to produce interferon- α . *Blood.* 75:17.

Shull, M.M., I. Ormsby, A.B. Kier, S. Pawlowski, R.J. Diebold, M. Yin, R. Allen, C. Sidman, G. Proetzel, D. Calvin, N. Annunziata, and T. Doetschman. 1992. Targeted disruption of the mouse transforming growth factor- β 1 gene results in multifocal inflammatory disease. *Nature 359:693*.

Sica, A., A. Saccani, A. Borsatti, C.A. Power, T.N.C. Wells, W. Luini, N. Polentarutti, S. Sozzani, and A. Mantovani. 1997. Bacterial lipopolysaccharide rapidly inhibits expression of C-C chemokine receptors in human monocytes. *J. Exp. Med.* 185:969.

Sieff, C.A., S.G. Emerson, R.E. Donahue, D.G. Nathan, E.A. Wang, G.G. Wong, and S.C. Clark. 1985. Human recombinant granulocyte colony-stimulating factor: A multilineage hematopoietin. *Science 230:1171*.

Sieff, C.A., C.M. Niemeyer, S.J. Mentzer, and D.V. Faller. 1988. Interleukin-1, tumor necrosis factor, and the production of colony-stimulating factors by cultured mesenchymal cells. *Blood* 72:1316.

Siminovich, L., E.A. McCulloch, and J.E. Till. 1964. The distribution of colony-forming cells among spleen colonies. J. Cell. Comp. Physiol. 62:327.

Simon, M.I., M.P. Strathmann, N. Gautam. 1991. Diversity of G proteins in signal transduction. *Science 252:802*.

Smith, J.B., and H.R. Herschman. 1995. Glucocorticoid-attenuated response genes encode intercellular mediators, including a new C-X-C chemokine. J. Biol. Chem. 270:16756.

Smith, C.A., and D.M. Rennick. 1986. Characterization of a murine lymphokine distinct from interleukin 2 and interleukin 3 (IL-3) possessing a T-cell growth factor activity and a mast-cell growth factor activity that synergizes with IL-3. *Proc. Natl. Acad. Sci. USA 83:1857*.

Smith, D.R., P.J. Polverini, S.L. Kunkel, M.B. Orringer, R.I. Whyte, M.D. Burdick, C.A. Wilke, and R.M. Strieter. 1994a. Inhibition of interleukin 8 attenuates angiogenesis in human bronchogenic carcinoma. *J. Exp. Med.* 179:1409.

Smith, R.E., R.M. Strieter, S.H. Phan, N.W. Lukacs, G.B. Huffnagle, C.A. Wilke, M.D. Burdick, P. Lincoln, H. Evanoff, and S.L. Kunkel. 1994b. Production and function of murine macrophage inflammatory protein- 1α in bleomycin-induced lung injury. *J. Immunol.* 153:4704.

Smith, M.W., Dean, M., M. Carrington, C. Winkler, G.A. Huttley, D.A. Lomb, J.J. Goedert, T.R. O'Brien, L.P. Jacobson, R. Kaslow, S. Buchbinder, E. Vittinghoff, D. Vlahov, K. Hoots, M.W. Hilgartner, Hemophilia Growth and Development Study (HGDS), Multicenter AIDS Cohort Study (MACS), Multicenter Hemophilia Cohort Study (MHCS), San Francisco City Cohort (SFCC), ALIVE Study, and S.J. O'Brien. 1997. Contrasting genetic influences of CCR2 and CCR5 variants on HIV-1 infection and disease progression. *Science 277:959*.

Sonoda, Y., Y-C Yang, G.G. Wong, S.C. Clark and M. Ogawa. 1988. Analysis in serum-free culture of the targets of recombinant human hemopoietic growth factors: Interleukin 3 and granulocyte/macrophage-colony-stimulating factor are specific for early developmental stages. *Proc. Natl. Acad. Sci USA*. 85:4360.

Sonoda, Y., T. Okuda, S. Yokota, T. Maekawa, Y. Shizumi, H. Nishigaki, S. Misawa, H. Fujii, and T. Abe. 1990. Actions of human interleukin-4 / B-cell stimulatory factor-1 on proliferation and differentiation of enriched hematopoietic progenitor cells in culture. *Blood* 75:1615.

Souza, L.M. T.C. Boone, J. Gabrilove, P.H. Lai, K.M. Zsebo, D.C. Murdock, V.R. Chazin, J. Bruszewski, H. Lu, K.K. Chen, J. Barendt, E. Platzer, M.A.S. Moore, R. Mertelsmann, and K. Welte. 1986. Recombinant human granulocyte colony-stimulating factor: Effects on normal and leukemic myeloid cells. *Science.* 232:61.

Spivak-Kroizman, T., M.A. Lemmon, I. Dikic, J.E. Ladbury, D. Pinchasi, J. Huang, M. Jaye, G. Crumley, J. Schlessinger, and I. Lax. 1994. Heparin-induced oligmerization of FGF molecules is responsible for FGF receptor dimerization, activation, and cell proliferation. *Cell* 79:1015.

Sprenger, H., R.G. Meyer, A. Kaufmann, D. Bussfeld, E. Rischkowsky, and D. Gemsa. 1996. Selective induction of monocyte and not neutrophil-attracting chemokines after influenza A virus infection. J. Exp. Med. 184:1191.

Stafford, S., H. Li, P.A. Forsythe, M. Ryan, R. Bravo, and R. Alam. 1997. Monocyte chemotactic protein-3 (MCP-3)/fibroblast-induced cytokine (FIC) in eosinophilic inflammation of the airways and the inhibitory effects of an anti-MCP-3/FIC antibody. *J. Immunol.* 158:4953.

Standiford, T.J., S.L. Kunkel, J.M. Liebler, M.D. Burdick, A.R. Gilbert, and R.M. Strieter. 1993a. Gene expression of macrophage inflammatory protein-1 α from human blood monocytes and alveolar macrophages is inhibited by interleukin-4. *Am. J. Respir. Cell Mol. Biol.* 9:192.

Standiford, T.J., M.W. Rolfe, S.L. Kunkel, J.P. Lynch III, M.D. Burdick, A.R. Gilbert, M.B. Orringer, R.I. Whyte, and R.M. Strieter. 1993b. Macrophage inflammatory protein-1 α expression in interstitial lung disease. *J. Immunol.* 151:2852.

Standiford, T.J., S.L. Kunkel, N.W. Lukacs, M.J. Greenberger, J.M. Danforth, R.G. Kunkel, and R.M. Strieter: 1995. Macrophage inflammatory protein-1 α mediates lung leukocyte recruitment, lung capillary leak, and early mortality in murine endotoxemia. *J. Immunol.* 155:1515.

Stanley, R.E., and P.M. Heard. 1977. Factors regulating macrophage production and growth. Purification and some properties of the colony stimulating factor from medium conditioned by L cells. J. Biol. Chem. 252:4305.

Stanley, E., D. Metcalf, P. Sobieszczuk, N.M. Gough and A.R. Dunn. 1985. The structure and expression of the murine gene encoding granulocyte-macrophage colony-stimulating factor: evidence for utilisation of alternative promoters. *EMBO J.* 4:2569.

Stanley, E.R., A. Bartocci, D. Patinkin, M. Rosendaal, and T.R. Bradley. 1986. Regulation of very primitive, multipotent, hemopoietic cells by hemopoietin-1. *Cell* 45:667.

Stanley, E.R., G.J. Lieshke, D. Grail, D. Metcalf, G. Hodgson, J.A.M. Gall, D.W. Mahler, J. Cebon, V. Sinickas, and A.R. Dunn. 1994. Granulocyte/macrophage colony-stimulating factor-deficient mice show no major pertubation of hematopoiesis but develop a characteristic pulmonary pathology. *Proc. Natl. Acad. Sci. USA* 91:5592.

Stein, J., G.V. Borzilla, and C.W. Rettenmier. 1990. Direct stimulation of cells expressing receptors for macrophage colony-stimulating factor (CSF-1) by a plasma membrane-bound precursor of human CSF-1. *Blood* 76:1308.

Stephenson, J.R., A.A. Axelrad, D.L. McLeod, and M.M. Shreeve. 1971. Induction of colonies of hemoglobin-synthesizing cells by erythropoietin *in vitro*. *Proc. Natl. Acad. Sci. USA 68:1542*.

Stoeckle, M.Y., and K.A. Barker. 1990. Two burgeoning families of platelet factor 4-related proteins: Mediators of the inflammatory response. *The New Biologist 2:*313.

Strife, A., C. Lambek, D. Wisniewski, S. Gulati, J. Gasson, D. Golde, K. Welte, J. Gabrilove, and B. Clarkson. 1987. Activities of four purified growth factors on highly enriched human hematopoietic progenitor cells. *Blood* 69:1508.

Strobl, H., E. Riedl, C. Schneinecker, C. Bello-Fernandez, W.F. Pickl, K. Rappersberger, O. Majdic, and W. Knapp. 1996. TGF β 1 promotes the development of dendritic cells from CD34⁺ progenitors. *J. Immunol.* 157:1499.

Stuber-Roos, R., M. Loetscher, D.F. Legler, I. Clark-Lewis, M. Baggiolini, and B. Moser. 1997. Identification of CCR8, the receptor for the human CC chemokine I-309. J. Biol. Chem. 272:17251.

Su, S., N. Mukaida, J. Wang, Y. Zhang, A. Takami, S. Nakao, and K. Matsushima. 1997. Inhibition of immature erythroid progenitor cell proliferation by macrophage inflammatory protein- 1α by interacting mainly with a C-C chemokine receptor, CCR1. *Blood 90:605*.

Suda, T., J. Suda, M. Ogawa, and J.N. Ihle. 1985. Permissive role of interleukin 3 (IL-3) in proliferation and differentiation of multipotential haemopoietic progenitors in culture. J. Cell Physiol. 124:182.
Suda, T., S. Okada, J. Suda, Y. Miura, M. Ito, T. Sudo, S. Hayashi, S. Nishikawa, and H. Nakauchi. 1989. A stimulatory effect of recombinant murine interleukin-7 (IL-7) on B-cell colony formation and an inhibitory effect of IL-1 α . *Blood* 74:1936.

Suematsu, S., T. Matsuda, K. Aozasa, S. Akira, N. Nakano, S. Ohno, J. Miyazaki, K.I. Yamamura, T. Hirano, and T. Kishimoto. 1989. IgG1 plasmacytosis in interleukin 6 transgenic mice. *Proc. Natl. Acad. Sci. USA* 86:7547.

Sutherland, H.J., C.J. Eaves, A.C. Eaves, W. Dragowska, and P.M. Landsdorp. 1989. Characterization and partial purification of human marrow cells capable of initiating long-term hematopoiesis in vitro. *Blood* 74:1563.

Szabolcs, P., M.A.S. Moore, J.W. Young. 1995. Expansion of immunostimulatory dendritic cells among the myeloid progeny of human $CD34^+$ bone marrow precursors cultured with *c-kit* ligand, granulocyte-macrophage colony-stimulating factor, and TNF- α . J. Immunol. 154:5851.

Taetle, R., C. Payne, B.D. Santos, M. Russell, and P. Segarini. 1993. Effects of transforming growth factor $\beta 1$ on growth and apoptosis of human acute myelogenous leukemia cells. *Cancer Res.* 53:3386.

Taipale, J., K. Miyazono, C-H. Heldin, and J. Keski-Oja. 1994. Latent transforming growth factor- β 1 associates to fibroblast extracellular matrix via latent TGF- β binding protein. *J.Cell. Biol.* 124:171.

Takagi, M., T. Hara, M. Ichihara, K. Takatsu, and A. Miyajima. 1995. Multi-colony stimulating activity of interleukin 5 (IL-5) on hematopoietic progenitors from transgenic mice that express IL-5 receptor α subunit constitutively. *J. Exp. Med.* 181:889.

Takahashi, G.W., D.F. Andrews III, M.B. Lilly, J.W. Singer, and M.R. Alderson. 1993. Effect of Granulocyte-macrophage colony-stimulating factor and interleukin-3 on interleukin-8 production by human neutrophils and monocytes. *Blood* 81:357.

Takaki, S., A. Tominaga, Y. Hitoshi, S. Mita, E. Sonoda, N. Yamaguchi, and K. Takatsu. 1990. Molecular cloning and expression of the murine interleukin-5 receptor *EMBO J.* 9:4367.

Takamatsu, S., I. Nakashima, and K. Nakano. 1996. Modulation of endotoxin-induced histamine synthesis by cytokines in mouse bone marrow-derived macrophages. J. Immunol. 156:778.

Tanabe, S., M. Heesen, M.A. Berman, M.B. Fischer, I. Yoshizawa, Y. Luo, and M.E. Dorf. 1997. Murine astrocytes express a functional chemokine receptor. *J. Neurosci.* 17:6522.

Tanaka, Y., D.H. Adams, S. Hubscher, H. Hirano, U. Siebenlist, and S. Shaw. 1993. T-cell adhesion induced by proteoglycan-immobilized cytokine MIP-1β. *Nature* 361:79.

Tangirala, R.K., K. Murao, and O. Quehenberger. 1997. Regulation of expression of the human monocyte chemotactic protein-1 receptor (hCCR2) by cytokines. J. Biol. Chem. 272:8050.

Tashiro, K., H. Tada, R. Heilker, M. Shirozu, T. Nakano, and T. Honjo. 1993. Signal sequence trap: A cloning strategy for secreted proteins and type I membrae proteins. *Science 261:600*.

Taub, D.D., K. Conlon, A.R. Lloyd, J.J. Oppenheim, and D.J. Kelvin. 1993. Preferential migration of activated CD4⁺ and CD8⁺ T cells in response to MIP-1 α and MIP-1 β . *Science 260:355*.

Taub, D.D., T.J. Sayers, C.R.D. Carter, and J.R. Ortaldo. 1995a. α and β chemokines induce NK cell migration and enhance NK-mediated cytolysis. *J. Immunol.* 155:3877.

Taub, D., J. Daystych, N. Inamura, J. Upton, D. Kelvin, D. Metcalfe, and J. Oppenheim. 1995b. Bone marrow-derived murine mast cells migrate but do not degranulate, in response to chemokines. *J. Immunol.* 154:2393.

Taub, D.D., D.L. Longo, and W.J. Murphy. 1996. Human interferon-inducible protein-10 induces mononuclear cell infiltration in mice and promotes the migration of human T lymphocytes into the peripheral tissues of human peripheral blood lymphocytes-SCID mice. *Blood* 87:1423.

Tawid, H., W. Labastide, C. Barker, and J. Rees. 1989. Cooperative effects of human recombinant granulocyte-macrophage colony stimulating factor and human recombinant erythropoietin in inducing erythroid differentiation of the human erythroleukaemia cell line K 562 clonogenic cells. *Leukemia Res. 13:127*.

Temeles, D.S., H.E. McGrath, E.L.W. Kittler, R.K. Shadduck, V.K. Kister, R.B. Crittenden, B.L. Turner, and P.J. Quesenberry. 1993. Cytokine expression from bone marrow derived macrophages. *Exp. Haematol.* 21:388.

Tessier, P.A., P.H. Naccache, I. Clark-Lewis, R.P. Gladue, K.S. Neote, and S.R. McColl. 1997. Chemokine networks in vivo. *J. Immunol.* 159:3595.

Thomas, R.S., M.J. Tymms, L.H. McKinlay, M.F. Shannon, A. Seth, and I. Kola. 1997. ETS1, NF-κB and AP1 synergistically transactivate the human GM-CSF promoter. *Oncogene 14:2845*.

Thorens, B., J-J. Mermod, and P. Vassalli. 1987. Phagocytosis and inflammatory stimuli induce GM-CSF mRNA in macrophages through posttranscriptional regulation. *Cell* 48:671.

Thornberry, N.A. H.G. Bull, J.R. Calaycay, K.T. Chapman, A.D. Howard, M.J. Kostura, D.K. Miller, S.M. Molineaux, J.R. Weidner, J. Aunins, K.O. Elliston, J.M. Ayala, F.J. Casano, J. Chin, G.J-F. Ding, L.A. Egger, E.P. Gaffney, G. Limjuco, O.C. Palyha, S.M. Raju, A.M. Rolando, J.P. Salley, T-T. Yamin, T.D. Lee, J.E. Shively, M. MacCross, R.A. Mumford, J.A. Schmidt, and M.J. Tocci. 1992. A novel heterodimeric cysteine protease is required for interleukin-1 β processing in monocytes. *Nature 356:768*.

Thompson, N.L., K.C. Flanders, J.M. Smith, L.R. Ellingsworth, A.B. Roberts, and M.B. Sporn. 1989. Expression of transforming growth factor- β 1 in specific cells and tissues of adult and neonatal mice. *J. Cell. Biol.* 108:661.

Tiffany, H.L., L.L. Lautens, J-L. Gao, J. Pease, M. Locati, C. Combadiere, W. Modi, T.I. Bonner, and P.M. Murphy. 1997. Identification of CCR8: A monocyte and thymus receptor for the CC chemokine I-309. *J. Exp. Med.* 186:165.

Till, J.E., and E.A. McCulloch. 1961. A direct measurement of the radiation sensitivity of normal mouse bone marrow cells. *Radiat. Res.* 14:213.

Till, J.E., E.A. McCulloch, and L. Siminovich. 1964. A stochastic model of stem cell proliferation, based on the growth of spleen colony-forming cells. *Proc. Natl. Acad. Sci. USA 51:29*.

Tomida, M., Y. Yamamoto-Yamaguchi, and M. Hozumi. 1984. Purification of a factor inducing differentiation of mouse myeloid leukemic M1 cells from conditioned medium of mouse fibroblast L929 cells. *J. Biol. Chem.* 259:10978.

Tominaga, A., S. Takaki, N. Koyama, S. Katoh, R. Matsumoto, M. Migita, Y. Hitoshi, Y. Hoyosa, S. Yamauchi, Y. Kanai, J-I. Miyazaki, G. Usuku, K-I. Yamamura, K. Takatsu. 1991. Transgenic mice expressing a B cell growth and differentiation factor gene (interleukin 5) develop eosinophilia and autoantibody production. J. Exp. Med. 173:429.

Tomonaga, M., D.W. Golde, and J.C. Gasson. 1986. Biosynthetic (recombinant) human granulocyte-macrophage colony-stimulating factor: Effect on normal bone marrow and leukemic cell lines. *Blood* 67:31.

Tsuchiya, S., M. Yamabe, Y. Yamaguchi, Y. Kobayashi, T. Konno, and K. Tada. 1980. Establishment and characterization of a human acute monocytic leukemia cell line (THP-1). *Int. J. Cancer. 26:171*.

Tsunawaki, S., M. Sporn, A. Ding, and C. Nathan. 1988. Deactivation of macrophages by transforming growth factor- β . *Nature 334:260*.

Tucker, R.F., G.D. Shipley, H.L. Moses, and R.W. Holley. 1984. Growth inhibitor from BSC-1 cells is closely related to platelet type β transforming growth factor. *Science* 226:705.

Tushinski, R.J., I.B. McAlister, D.E. Williams, and A.E. Namen. 1991. The effects of interleukin 7 (IL-7) on human bone marrow in vitro. *Exp. Hematol.* 19:749.

Ueda, A., Y. Ishigatsubo, T. Okubo, and T. Yoshimura. 1997. Transcriptional regulation of the human monocyte chemoattractant protein-1 gene. J. Biol. Chem. 272:31092.

Uguccioni, M., P. Loetscher, U. Forssmann, B. Dewald, H. Li, S. H. Lima, Y. Li, B. Kreider, G. Garotta, M. Thelen, and M. Baggiolini. 1996. Monocyte chemotactic protein 4 (MCP-4), a novel structural and functional analogue of MCP-3 and Eotaxin. *J. Exp. Med.* 183:2379.

Uguccioni, M., C.R. Mackay, B. Ochensberger, P. Loetscher, S. Rhis, G.J. LaRosa, P. Rao, P.D. Ponath, M. Baggiolini and C. Dahinden. 1997. High expression of the chemokine receptor CCR3 in human blood basophils. *J. Clin. Invest.* 100:1137.

Ullrich, A., and J. Schlessinger. 1990. Signal transduction by receptors with tyrosine kinase activity. *Cell* 61:203.

Van Damme, J., P. Proost, J.P. Lenaerts, and G. Opdenakker. 1992. Structural and functional identification of two human, tumor-derived monocyte chemotactic proteins (MCP-2 and MCP-3) belonging to the chemokine family. *J. Exp. Med.* 176:59.

Van der Sluijs, J.P., J.P. de Jong, N.H.C. Brons, and R.E. Ploemacher. 1990. Marrow repopulating cells, but not CFU-S, establish long-term in vitro hemopoiesis on a marrow-derived stromal layer. *Exp. Hematol.* 18:893.

VanZant, G. 1984. Studies of hematopoietic stem cells spared by 5-fluorouracil. J. Exp. Med. 159:679.

Vaux, D.L., P.A. Lalor, S. Cory, and G.R. Johnson. 1990. In vivo expression of interleukin 5 induces an eosinophilia and expanded Ly-1B lineage populations. *Int. Immunol.* 2:965.

Vellenga, E., A. Rambaldi, T.J. Ernst, D. Ostapovicz, J.D. Griffin. 1988. Independent regulation of M-CSF and G-CSF gene expression in human monocytes. *Blood* 71:1529.

Verfaillie, C.M. 1992. Direct contact between human primitive hematopoietic progenitors and bone marrow stroma is not required for long-term in vitro hematopoiesis. *Blood* 79:2821.

Verfaillie, C.M. 1993. Soluble factor(s) produced by human bone marrow stroma increase cytokine-induced proliferation and maturation of primitive hematopoietic progenitors while preventing their terminal differentiation. *Blood* 82:2045.

Verfaillie, C.M., P.M. Catanzarro, and W. Li. 1994. Macrophage inflammatory protein 1α , interleukin 3 and diffusible marrow stromal factors maintain human hematopoietic stem cells for at least eight weeks in vitro. J. Exp. Med. 179:643.

Vicari, A.P., D.J. Figuerosa, J.A. Hedrick, J.S. Foster, K.P. Singh, S. Menon, N.G. Copeland, D.J. Gilbert, N.A. Jenkins, K.B. Bacon, and A. Zlotnik. 1997. TECK: A novel CC chemokine specifically expressed by thymic dendritic cells and potentially involved in T cell development. *Immunity* 7:291.

Wahl, S.M., D.A. Hunt, L.M. Wakefield, N. McCartney-Francis, L.M. Wahl, A.B. Roberts, and M.B. Sporn. 1987. Transforming growth factor β induces monocyte chemotaxis and growth factor production. *Proc. Natl. Acad. Sci. USA* 84:5788.

Wahl, S.M., D.A. Hunt, H.L. Wong, S. Dougherty, N. McCartney-Francis, L.M. Wahl, L. Ellingsworth, J.A. Schmidt, G. Hall, A.B. Roberts, and M.B. Sporn. 1988. Transforming growth factor- β is a potent immunosuppressive agent that inhibits IL-1 dependent lymphocyte proliferation. *J. Immunol.* 140:3026.

Wakefield, L.M., D.M. Smith, K.C. Flanders, M.B. Sporn. 1988. Latent transforming growth factor- β from human platelets. A high molecular weight complex containing precursor sequences. *J. Biol. Chem.* 263:7646.

Wakefield, L.M., D.M. Smith, S. Broz, M Jackson, A.D. Levinson, and M.B. Sporn. 1989. Recombinant TGF-beta 1 is synthesised as a two-component latent complex that shares some structural features with the native platelet latent TGF-beta 1 complex. *Growth Factors 1:203*.

Walker, F., N.A. Nicola, D. Metcalf, and A.W. Burgess. 1985. Hierachical down-modulation of hemopoietic growth factor receptors. *Cell* 43:269.

Walz, A., P. Peveri, H. Aschauer, and M. Baggiolini. 1987. Purification and amino acid sequencing of NAF, a novel neutrophil-activating factor produced by monocytes. *Biochem. Biophys. Res. Commun.* 149:755.

Walz, A., R. Burgener, B. Car, M. Baggiolini, S.L. Kunkel, and R.M. Strieter. 1991. Structure and neutrophil-activating properties of a novel inflammatory peptide (ENA-78) with homology to interleukin 8. *J. Exp. Med.* 174:1355.

Wang, X., T-L. Yue, E.H. Ohlstein, C-P. Sung, and G. Z. Feuerstein. 1996. Interferoninducible protein-10 involves vascular smooth muscle cell migration, proliferation, and inflammatory response. *J. Biol. Chem.* 271:24286.

Warren, M., and P. Ralph. 1986. Macrophage growth factor CSF-1 stimulates human monocyte production of interferon, tumor necrosis factor, and colony-stimulating activity. *J. Immunol.* 137:2281.

Weber, M., M. Uguccioni, M. Baggiolini, I. Clark-Lewis, and C.A. Dahinden. 1996. Deletion of the NH₂-terminal residue converts monocyte chemotactic protein 1 from an activator of basophil mediator release to an eosinophil chemoattractant. J. Exp. Med. 183:681.

Weissman, D., R.L. Rabin, J. Arthos, A. Rubbert, M. Dybul, R. Swofford, S. Venkatesan, J.M. Farber, and A.S. Fauci. 1997. Macrophage-tropic HIV and SIV envelope proteins induce a signal through the CCR5 chemokine receptor. *Nature* 389:981.

Wesselingh, S.L., N.M. Gough, J.J. Finlay-Jones, and P.J. McDonald. 1990. Detection of cytokine mRNA in astrocyte cultures using the polymerase chain reaction. *Lymphokine Res.* 9:177.

Wetzler, M., M. Talpaz, D.G. Lowe, G. Baiocchi, J.U. Gutterman, and R. Kurzrock. 1990. Constitutive expression of leukemia inhibitory factor RNA by human bone marrow stromal cells and modulation by IL-1, TNF α , and TGF β . *Exp. Hematol* 19:347.

Widmer, U., Z. Yang, S. van Deventer, K.R. Manogue, B. Sherry, and A. Cerami. 1991. Genomic structure of murine macrophage inflammatory protein- 1α and conservation of potential regulatory sequences with a human homolog, LD78. *J. Immunol.* 146:4031.

Widmer, U., K.R. Manogue, A. Cerami, and B. Sherry. 1993. Genomic cloning and promoter analysis of macrophage inflammatory protein (MIP)-2, MIP-1 α , and MIP-1 β , members of the chemokine superfamily of proinflammatory cytokines. *J. Immunol.* 150:4996.

Wiktor-Jedrzejczak, W., A. Bartocci, A.W. Ferrante Jr, A. Ahmed-Ansari, K.W. Sell, J.W. Pollard, and E.R. Stanley. 1990. Total absence of colony-stimulating factor 1 in the macrophage-deficient osteopetrotic (*op/op*) mouse. *Proc. Natl. Acad. Sci. USA* 87:4828.

Wiktor-Jedrzejczak, W., E. Urbanowska, S.L. Auckerman, J.W. Pollard, E.R. Stanley, P. Ralph, A.A. Ansari, K.W. Sell, and M. Szperl. 1991. Correction by CSF-1 of defects in the osteopetrotic op/op mouse suggests local, developmental, and humoral requirements for this growth factor. *Exp. Hematol.* 19:1049.

Wilcox, J.N., and R. Derynck. 1988. Developmental expression of transforming growth factors alpha and beta in mouse fetus. *Mol. Cell. Biol.* 8:3415.

Williams, D.E., and P.J. Morrissey. 1989. Alterations in megakaryocye and platelet compartments following in vivo IL-1 β administration to normal mice. *J. Immunol.* 142:4361.

Williams, R.L., D.J. Hilton, S. Pease, T.A. Willson, C.L. Stewart, D.P. Gearing, E.F. Wagner, D. Metcalf, N.A. Nicola, and N.M. Gough. 1988. Myeloid leukemia inhibitory factor maintains the developmental potential of embryonic stem cells. *Nature* 336:684.

Williams, G.T., C.A. Smith, E. Spooncer, and T.M. Dexter. 1990. Haemopoietic colony stimulating factors promote cell survival by suppressing apoptosis. *Nature* 343:76.

Wilson, K.P., J-A.F. Black, J.A. Thomson, E.E. Kim, J.P. Griffith, M.A. Murcko, S.P. Chambers, R.A. Aldape, S.A. Raybuck, and D.J. Livingston. 1994. Structure and mechanism of interleukin-1 β converting enzyme. *Nature* 370:270.

Wing, E.J., A. Waheed, R. Shadduck, L.S. Nagle, and K. Stephenson. 1982. Effect of colony-stimulating factor on murine macrophages. Induction of antitumor activity. *J. Clin. Invest.* 69:270.

Winkler, C., W. Modi, M.W. Smith, G.W. Nelson, X. Wu, M. Carrington, M. Dean, T. Honjo, K. Tashiro, D. Yabe, S. Buchbinder, E. Vittinghoff, J. Goedert, T.R. O'Brien, L.P. Jacobson, R. Detels, S. Donfield, A. Willoughby, E. Gomperts, D. Vlahov, J. Phair, ALIVE Study, Hemophilia Growth and Development Study (HGDS), Multicenter AIDS Cohort Study (MACS), Multicenter Hemophilia Cohort Study (MHCS), San Francisco City Cohort (SFCC), and S.J. O'Brien. 1998. Genetic restriction of AIDS pathogenesis by an SDF-1 chemokine gene variant. *Science* 279:389.

Wodnar-Filipowicz, A., C.H. Heusser, and C. Moroni. 1989. Production of the haemopoietic growth factors GM-CSF and interleukin-3 by mast cells in response to IgE receptor-mediated activation. *Nature 339:150*.

Wolpe, S.D., G. Davatelis, B. Sherry, B. Beutler, D.G. Hesse, H.T. Nguyen, L.L. Moldawer, C.F. Nathan, S.F. Lowry, and A. Cerami. 1988. Macrophages secrete a novel heparin-binding protein with inflammatory and neutrophil chemokinetic properties. *J. Exp. Med.* 167:570.

Wolf, N.S., and G.V. Priestly. 1986. Kinetics of early and late spleen colony development. *Exp. Hematol.* 14:676.

Wong, G.G., P.A. Temple, A.C. Leary, J.S. Witek-Giannotti, Y-C. Yang, A.B. Ciarletta, M. Chung, P. Murtha, R. Kriz, R.J. Kaufman, C.R. Ferenz, B.S. Sibley, K.J. Turner, R.M. Hewick, S.C. Clark, N. Yanai, H. Yokota, M. Yamada, M. Saito, K. Motoyoshi, and F. Takaku. 1987. Human CSF-1: Molecular cloning and expression of 4 kb cDNA encoding the human urinary protein. *Science 235:1504*.

Wong, L-M., S.J. Myers, C-L. Tsou, J. Gosling, H. Arai, and I.F. Charo. 1997. Organization and differential expression of the human monocyte chemoattractant protein 1 receptor gene. *J. Biol. Chem.* 272:1038.

Woodcock, J.M., B.J. McClure, F.C. Stomski, M.J. Elliot, C.J. Bagley, and A.F. Lopez. 1997. The human granulocyte-macrophage colony-stimulating factor (GM-CSF) receptor exists as a preformed receptor complex that can be activated by GM-CSF, interleukin-3, or interleukin-5. *Blood* 90:3005.

Wrenshall, L.E., F.B. Cerra, A. Carlson, F.H. Bach, and J.L. Platt. 1991. Regulation of murine splenocyte responses by heparan sulfate. *J. Immunol.* 147:455.

Wu, D., G.J. LaRosa, and M.I. Simon. 1993. G Protein-coupled signal transduction pathways for interleukin-8. *Science 261:101*.

Wu, L., N.P. Gerard, R. Wyatt, H. Choe, C. Parolin, N. Ruffing, A. Borsetti, A.A. Cardoso, E. Desjardin, W. Newman, C. Gerard, and J. Sodroski. 1996. CD4-induced interaction of primary HIV-1 gp120 glycoproteins with the chemokine receptor CCR-5. *Nature 384:179*.

Yang, L., and Y-C. Yang. 1995. Heparin inhibits the expression of interleukin-11 and granulocyte-macrophage colony-stimulating factor in primate bone marrow stromal fibroblasts through mRNA stabilisation. *Blood* 86:2526.

Yoshida, H., S-I. Hayashi, T. Kunisada, M. Ogawa, S. Nishikawa, H. Okamura, T. Sudo, L.D. Shultz, and S-I. Nishikawa. 1990. The murine mutation osteopetrosis is in the coding region of the macrophage colony stimulating factor gene. *Nature*. 345:442.

Yoshida, T., T. Imai, M. Kakizaki, M. Nishimura, and O. Yoshie. 1995. Molecular cloning of a novel C or γ type chemokine, SCM-1. *FEBS Lett.* 360:155.

Yoshida, R., T. Imai, K. Hieshima, J. Kusuda, M. Baba, M. Kitaura, M. Nishimura, M. Kakizaki, H. Nomiyama, and O. Yoshie. 1997. Molecular cloning of a novel human CC chemokine EBI1-ligand chemokine that is a specific functional ligand for EBI1, CCR7. J. Biol. Chem. 272:13803.

Yoshimura, T., K. Matsushima, S. Tanaka, E.A. Robinson, E. Appella, J.J. Oppenheim, and E.J. Leonard. 1987. Purification of a human monocyte-derived neutrophil chemotactic factor that has peptide sequence similarity to other host-defence cytokines. *Proc. Natl. Acad. Sci. USA* 84:9233.

Yoshimura, T.K., E.A. Robinson, S. Tanaka, E. Appella, and E.J. Leonard. 1989. Purification and amino acid analysis of two human monocyte chemoattractants produced by phytohemagglutinin-stimulated human blood mononuclear leukocytes. *J. Immunol.* 142:1956.

Youn, B-S., I-K. Jang, H.E. Broxmeyer, S. Cooper, N.A. Jenkins, D.J. Gilbert, N.G. Copeland, T.A. Elick, M.J. Fraser, Jr., and B.S. Kwon. 1995. A novel chemokine, macrophage inflammatory protein-related protein-2, inhibits colony formation of bone marrow myeloid progenitors. *J. Immunol.* 155:2661.

Yuo, A, S. Kitagawa, A. Ohsaka, M. Ohta, K. Miyazono, T. Okabe, A. Urabe, M. Saito, and F. Takaku. 1989. Recombinant human granulocyte colony-stimulating factor as an activator of human granulocytes: Potentiation of responses triggered by receptor-mediated agonists and stimulation of C3bi receptor expression and adherence. *Blood* 74:2144.

Zhou, Y-Q., E.R. Stanley, S.C. Clark, J.A. Hatzfeld, J-P. Levesque, C. Frederici, S.M. Watt, and A. Hatzfeld. 1988. Interleukin-3 and interleukin-1 α allow earlier bone marrow progenitors to respond to human colony-stimulating factor 1. *Blood* 72:1870.

Zsebo, K.M., D.A. Williams, E.N. Geissler, V.C. Broudy, F.H. Martin, H.L. Atkins, R-Y Hsu, N.C. Birkett, K.H. Okino, D.C. Murdock, F.W. Jacobsen, K.E. Langley, K.A. Smith, T. Takeishi, B.M. Cattanach, S.J. Galli, and S.V. Suggs. 1990. Stem cell factor is encoded at the *Sl* locus of the mouse and is the ligand for the *c-kit* tyrosine kinase receptor. *Cell* 63:213.

Zucali, J.R., C.A. Dinarello, D.J. Oblon, M.A. Gross, L. Anderson and R.S. Weiner. 1986. Interleukin 1 stimulates fibroblasts to produce granulocyte-macrophage colony-stimulating activity and prostaglandin E_2 . J. Clin. Invest. 77:1857.

